

# RWE Renewables UK Dogger Bank South (West) Limited RWE Renewables UK Dogger Bank South (East) Limited

Dogger Bank South Offshore Wind Farms

Book of Reference Volume 4

November 2024 January 2025

Application Reference: 4.2 APFP Regulation: 5(2)(d) Revision: 0304

Unrestricted



Company:	RWE Renewables UK Dogger Bank South (West) Limited and RWE Renewables UK Dogger Bank South (East) Limited	Asset:	Development
Project:	Dogger Bank South Offshore Wind Farms	Sub Project / Package:	Consents
Document Title or Description:	Book of Reference		
Document Number:	005028761- <del>02<u>04</u></del>	Contract or Referenc e Number:	N/A

COPYRIGHT © RWE Renewables UK Dogger Bank South (West) Limited and RWE Renewables UK Dogger Bank South (East) Limited, 2024. All rights reserved.

This document is supplied on and subject to the terms and conditions of the Contractual Agreement relating to this work, under which this document has been supplied, in particular:

#### LIABILITY

In preparation of this document RWE Renewables UK Dogger Bank South (West) Limited and RWE Renewables UK Dogger Bank South (East) Limited has made reasonable efforts to ensure that the content is accurate, up to date and complete for the purpose for which it was contracted. RWE Renewables UK Dogger Bank South (West) Limited and RWE Renewables UK Dogger Bank South (East) Limited makes no warranty as to the accuracy or completeness of material supplied by the client or their agent.

Other than any liability on RWE Renewables UK Dogger Bank South (West) Limited and RWE Renewables UK Dogger Bank South (East) Limited detailed in the contracts between the parties for this work RWE Renewables UK Dogger Bank South (West) Limited and RWE Renewables UK Dogger Bank South (East) Limited shall have no liability for any loss, damage, injury, claim, expense, cost or other consequence arising as a result of use or reliance upon any information contained in or omitted from this document.

Any persons intending to use this document should satisfy themselves as to its applicability for their intended purpose.

The user of this document has the obligation to employ safe working practices for any activities referred to and to adopt specific practices appropriate to local conditions.

Rev No.	Date	Status/Reason for Issue	Author	Checked by	Approved by
01	February 2024	Draft to PINS	DM	RWE	RWE
02	June 2024	Final for DCO Application	DM	RWE	RWE
03	November 2024	Submission at previous Draft Deadline 1	DM	RWE	RWE
04	January 2025	<u>Change request</u>	DM	RWE	RWE

#### Unrestricted



Revision Change Log			
Rev No.	Page	Section	Description
01	N/A	N/A	Submitted for DCO Application
02	All	1	The Guide to the Application has been updated to account for the addition of the Pre-Examination Procedural Deadline documents and the updates of documents previously submitted as part of the DCO Submission
03	All	N/A	Updates to plot descriptions, addresses and new or archived interests.
<u>04</u>	<u>All</u>	<u>N/A</u>	New plots, removed plots and updates to plot interests due to <u>change request.</u>

# Contents

1	Boo	ok of Reference	1
1	1	Introduction	1
1	2	Rights which may be acquired	4
1	3	Structure of this book of reference	34
1	.4	Book of reference notes	35
1	5	Schedule of Changes to Plot Interests	36
		Schedule of Changes to Plots	
		5	

### Tables

Table 1-1 Land plan colours relating to acquisition	3
Table 1-2 Rights sought for the Projects	5



# **1** Book of Reference

# **1.1 Introduction**

- 1. This Book of Reference is submitted by RWE Renewables UK Dogger Bank South (West) Limited (incorporated under company number 13656525) and RWE Renewables UK Dogger Bank South (East) Limited (incorporated under company number 13656240) who have their registered office at Windmill Hill Business Park, Whitehill Way, Swindon, Wiltshire, United Kingdom, SN5 6PB ("the Applicants") in relation to the development consent order ("DCO") under the Planning Act 2008 ("the 2008 Act") for the construction and operation of offshore energy generating stations and electrical connections comprising of the Dogger Bank South East ("DBS East") and Dogger Bank South West ("DBS West") offshore wind farms (together referred to as "Dogger Bank South Offshore Wind Farms" and "the Projects") as required by Regulation 5(2)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 ("the Regulations").
- 2. The land described in this Book of Reference, which is to be subject to powers of compulsory acquisition and/or temporary possession, is required to deliver the works comprising the Projects which are described in detail in Part 1 of Schedule 1 of Volume 3, Draft DCO (application ref: 005028758-0204) and shown on Volume 2, Works Plans (application ref: 004525725-0203). This Book of Reference is part of the application documents for the authorised project it should be read in conjunction with Volume 2, Land Plans (application ref: 005028748-0102), Volume 4, Statement of Reasons (application ref: 005028760-0204) and Volume 3, Draft DCO (application ref: 005028768-0204).
- 3. Every parcel of land that is affected is identified on a plot-by-plot basis and a unique number has been ascribed to each plot. The plots are shown on the Land Plans which accompany the Draft DCO. For each plot it identifies whether the Applicants are seeking the power to acquire the freehold, the power to create and/or acquire permanent rights, or the power to create and/or acquire temporary rights of possession and use. It also lists the persons with land outside the order limits who may be entitled to make a relevant claim for compensation due to the effects of construction or when the authorised project is operational.
- 4. The draft DCO seeks powers to compulsorily acquire land and new rights (both temporary and permanent) for the purposes of the construction, operation, maintenance and decommissioning of the Projects. Some of the plots identified in Part 1 of the Book of Reference will be subject to acquisition of permanent rights (including restrictive covenants) pursuant to Article 22 (Compulsory acquisition of rights) and Schedule 7 (Land in which only new rights etc. may be required) of the draft DCO. These plots are shown coloured blue on the land plans.

Unrestricted 005028761-<mark>02</mark>04 Page 1



- 5. Plots that are subject to powers of temporary possession only, such as for the purpose of access or areas only required during construction are listed in Schedule 9 of the Order and shown coloured yellow on the land plans.
- 6. The colours shown on the land plans indicate the type of acquisition sought as set out in the **Table 1-1** below:





Table 1-1 Land plan colours relating to acquisition

Colour of the plot on land plans	Description of acquisition sought in Book of Reference	Acquisition sought	Principal Relevant DCO Article(s)
Pink	"Acquisition of freehold"	Compulsory acquisition of all interests and rights in land (including as required, subsoil, surface land)	
Blue	"Acquisition of new rights and imposition of restrictions" and "Acquisition of new rights"	Creation and compulsory acquisition of new rights (including where necessary, a right to impose restrictive covenants)	
Yellow	"Temporary possession"	Temporary possession and use of land, primarily during construction	



7. The Applicants have taken the cautious approach of seeking powers of compulsory acquisition (or rights of use) in respect of all plots of land required for the Projects even where it already holds an interest or presumes it holds an interest in the land. This approach has been taken to ensure that the Projects have the right to acquire the interests they need in the whole of the Order limits even where an unidentified owner later asserts an interest in land which the Applicants believe they own or have rights.

# 1.2 Rights which may be acquired

- 8. For plots in which rights are to be acquired or restrictive covenants are to be imposed, the rights sought for the Projects have been categorised as shown in the **Table 1-2** below. This means that where the entry in this Book of Reference specifies that 'all cable rights' are to be acquired, all of the rights shown in the cable rights line of table will be sought. Where only some of a category of rights is required, the specific rights sought will be listed for the relevant plot.
- 9. In some cases more than one category of rights will be sought in a single plot. For example cable rights to construct, install, operate, maintain and decommission the cable and restrictive covenants to protect the cable once it has been installed will be required on the cable route plots.



#### Table 1-2 Rights sought for the Projects

(1) Number of plot shown on land plans	(2) Purpose for which rights may be acquired
Cable rights and restrictive	covenants
02-001, 02-002, 02-003, 02-004, 02-005, 02-006,	1. Cable rights
02-007, 02-008, 02-009, 02-010, 02-011, 02-012,	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised project to—
02-013, 02-014, 02-015, 02-017, 02-021, 02-022, 02-024, 02-026, 02-030, 02-031, 02-034, 02-038, 03-001, 03-002, 03-010, 03-011, 03-012, 03-013, 04-008, 04-009, 04-010, 04-014, 04-018, 04-021, 04-022, 04-023, 04-024,	<ul> <li>(a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables (including the removal of materials including spoil) in, under, over and/or on the Land, together with such telemetry and fibre-optic lines, ducting, jointing bays, allow the installation and use of electricity cables in the additional ducts, and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (all collectively referred to as the "cables"), and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling beneath sea defences, watercourses, roads and railways;</li> <li>(b) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew,</li> </ul>



(1) Number of plot shown on land plans	(2) Purpose for which rights may be acquired
05-005, 06-001, 06-002, 06-007, 06-017, 06-018, 06-025, 07-001, 07-002, 07-003, 07-004, 07-005, 08-008, 08-009, 08-013, 08-015, 08-016, 08-017, 08-025, 08-026, 09-001, 09-002, 09-003, 09-008, 09-009, 09-010, 09-013, 10-005, 10-006, 10-007, 10-008, 10-009, 10-010, 10-011, 10-012, 10-013, 10-014, 10-019, 11-008, 11-018, 12-004, 12-011, 12-012, 12-013, 12-014, 13-001, 13-002, 13-003, 13-004, 13-005, 13-009, 00-000, 00-000, 00-000, 00-000, 00-000, 00-000, 00-000, 00-000, 00-000, 00-000, 00-000, 00-000, 00-000, 00-000, 00-000, 00-000, 00-000, 00-000, 00-000, 10-0	<ul> <li>materials including spoil) in, under, over and/or on the Land, together with such telemetry and fibre-optic lines, ducting, jointing bays, allow the installation and use of electricity cables in the additional ducts, and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (all collectively referred to as the "cables"), and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling beneath sea defences, watercourses, roads and railways;</li> <li>(c) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the additional ducts for electricity cables (including the removal of materials including spoil) in, under, over and/or on the Land, allow the installation and use of electricity cables in the additional ducts, together with such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (all collectively referred to as the "cables"), and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling beneath sea defences, watercourses, roads and railways;</li> <li>(d) enter, be on, and break open and break up the surface of the Land and remain with or without plant, vehicles, machinery, apparatus and equipment which is ancillary to the purposes of transmitting electricity along the cables;</li> </ul>

Unrestricted 005028761-<del>02</del>04



(1) Number of plot shown on	(2) Purpose for which rights may be acquired
land plans	
13-014, 13-015, 13-016, 13-017, 14-005, 14-006, 14-007, 14-008, 14-013, 14-014, 15-006, 15-007, 15-008, 16-001, 16-002, 16-003, 16-004, 16-008, 16-009, 17-002, 17-008, 17-009, 17-010, 17-011, 18-001, 18-002, 18-007, 18-009, 18-010a, 18-012, 18-014a, 18-027, 18-038, 18-039, 18-043, 18-050, 18-052, 18-054, 19-003, 19-004, 19-005, 19-007, 20-002	<ul> <li>(e) benefit from continuous vertical and lateral support for the authorised project;</li> <li>(f) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, inspecting, removing and replacing the ducting, electrical infrastructure and cables;</li> <li>(g) construct and install and thereafter use the Land for all necessary purposes for the commissioning, construction, repair, testing and maintenance of the ducting, electrical infrastructure and cables in, on or under the Land;</li> <li>(h) place and use plant, machinery, structures and temporary structures within the Land for the purposes of the installation, construction, maintenance, repairing, renewing, upgrading, inspecting, removal and replacing of the ducting, electrical infrastructure and cables and to erect temporary signage and provide measures for the benefit of public and personnel safety;</li> <li>(i) erect fencing, gates, walls, barriers or other means of enclosure, and create secure working areas and compounds including trenchless installation technique compounds and working areas;</li> <li>(j) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water</li> </ul>



(1)	(2)	
Number of plot shown on land plans	Purpose for which rights may be acquired	
	<ul> <li>courses and drains during any period during which construction, maintenance, repair or renewal is being carried out;</li> <li>(k) erect temporary bridges and supporting or protective structures for the purposes of access to adjoining land;</li> </ul>	
	<ul> <li>(I) effect access and egress to the highway;</li> <li>(m) make such investigations in or on the Land as required;</li> <li>(n) alter, fell, lop or cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be standing on the Land or other land which would if not felled, lopped, cut or removed</li> </ul>	
	<ul> <li>would obstruct or interfere with the operation of the cables and ancillary equipment including ducting;</li> <li>(o) take and use, remove and discharge water from the Land, and to install, retain, use, maintain,</li> </ul>	
	inspect, alter, remove, refurbish, reconstruct, replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and manage waterflows in any drains, watercourses and culverts, install, use, inspect, maintain, adjust, alter, renew, repair, test or cleanse drainage schemes on the Land or reinstate any existing drainage scheme on the Land;	
	<ul> <li>(p) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus including but not limited to electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications cables and any ancillary equipment and</li> </ul>	

Unrestricted 005028761-<del>02</del>04



(1)	(2)	
Number of plot shown on land plans	Purpose for which rights may be acquired	
	<ul> <li>apparatus public and private drains, watercourses, sewers, ponds or culverts, service media (including the pipes, cables or conduits or apparatus of statutory undertakers);</li> <li>(q) erect and remove temporary fencing, gates, walls, barriers or other means of enclosure, remove fences and structures within the Land during any period during which construction, maintenance, repair or renewal is being carried out (subject to erection of any temporary stock-proof fencing as is reasonably required and the re-instatement or suitable replacement of the fences or structures following the exercise of the rights);</li> <li>(r) store and stockpile materials (including excavated material);</li> <li>(s) create boreholes and trial excavation pits for the purposes of intrusively surveying the land and monitoring the use of any trenchless installation technique, to keep in place and monitor the same through construction, maintenance repair, replacement or decommissioning and to reinstate the Land;</li> </ul>	
	<ul> <li>(t) excavate materials below ground level, including soils, and to store and re-use or dispose of the same, and in so excavating to undertake any works, including works of protection or removal of archaeological remains as may be required by any written scheme of investigation approved under this Order;</li> </ul>	
	<ul> <li>(u) lay out temporary paths and bridleways for public use as temporary diversions for public rights of way which are interfered with during any period in which construction, maintenance, repair, renewal or decommissioning is being carried out;</li> </ul>	



(1)	(2)	
Number of plot shown on land plans	Purpose for which rights may be acquired	
	<ul> <li>(v) install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out mitigation, maintenance, remediation works for environmental or ecological mitigation or enhancement works, including temporary works for noise alleviation measures and the installation of temporary barriers for the protection of fauna;</li> <li>(w) carry out such works (together with associated fencing) required by a planning permission and/or consent now or to be granted over the Land and/or in accordance with any necessary licences relating to protected species and/or wildlife; and</li> <li>(x) (in an emergency only when the cables are temporarily unusable) to lay down, install, use, maintain and inspect underground cables, telephone signalling and fibre-optic cables and ancillary equipment, associated works and other conducting media together with conduits or pipes for containing the same in and under the Land.</li> </ul>	
	2. Restrictive covenants	
	A restrictive covenant over the Land for the benefit of the remainder of the Order land to:	
	<ul> <li>(y) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erection of any buildings or construction erection or works of any kind (including the foundations or footings thereto);</li> </ul>	



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	<ul> <li>(z) prevent anything to be done by way of hard surfacing of the Land with concrete of any kind or with any other material or surface whatsoever without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed if the proposed surfacing would not cause damage to relevant part of the authorised project nor make it materially more difficult or expensive to maintain the authorised project);</li> <li>(aa) prevent anything to be done by way of excavation of any kind in the Land nor any activities which would alter, increase or decrease ground cover or soil levels in any manner whatsoever without the consent in writing of the undertaker save as are reasonably required for agricultural activities (being ploughing to no deeper than 0.6m for the purposes of arable farming);</li> </ul>
	<ul> <li>(bb) prevent the planting or growing within the Land of any trees, shrubs or underwood without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed trees, shrubs or underwood would not cause damage to the relevant part of the authorised project nor make it materially more difficult or expensive to access the relevant part of the authorised project);</li> <li>(cc) prevent anything being done which may interfere with free flow and passage of electricity or telecommunications through the cables or support for the authorised project;</li> </ul>



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	<ul> <li>(dd) prevent carrying out operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with the exercise of the rights or damage the authorised project; and</li> <li>(ee) prevent any activity which would in the reasonable opinion of the undertaker result in the disturbance of ecological mitigation areas or areas of habitat creation including any ploughing or grazing without the prior written consent of the undertaker</li> </ul>
Cable rights, transition join	bay rights and restrictive covenants
02-016, 02-018, 02-019, 02-020, 02-023, 02-025	1. Cable rights
	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised project to—
	(a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables (including the removal of materials including spoil) in, under, over and/or on the Land, together with such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (all

Unrestricted 005028761-<del>02</del>04



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	<ul> <li>collectively referred to as the "cables"), and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling beneath sea defences, watercourses, roads and railways;</li> <li>(b) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the additional ducts for electricity cables (including the removal of materials including spoil) in, under, over and/or on the Land, allow the installation and use of electricity cables in the additional ducts, together with such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (all collectively referred to as the "cables"), and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling beneath sea defences, watercourses, roads and railways;</li> <li>enter, be on, and break open and break up the surface of the Land and remain with or without plant, vehicles, machinery, apparatus and equipment which is ancillary to the purposes of transmitting electricity along the cables;</li> <li>(c) benefit from continuous vertical and lateral support for the authorised project;</li> <li>(d) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using,</li> </ul>



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	<ul> <li>maintaining, repairing, renewing, inspecting, removing and replacing the ducting, electrical infrastructure and cables;</li> <li>(e) construct and install and thereafter use the Land for all necessary purposes for the commissioning, construction, repair, testing and maintenance of the ducting, electrical infrastructure and cables in, on or under the Land;</li> <li>(f) place and use plant, machinery, structures and temporary structures within the Land for the purposes of the installation, construction, maintenance, repairing, renewing, upgrading, inspecting, removal and replacing of the ducting, electrical infrastructure and cables and to erect temporary signage and provide measures for the benefit of public and personnel safety;</li> <li>(g) erect fencing, gates, walls, barriers or other means of enclosure, and create secure working areas;</li> <li>(h) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair or renewal is being carried out;</li> </ul>



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	<ul> <li>erect temporary bridges and supporting or protective structures for the purposes of access to adjoining land;</li> <li>(i) effect access and egress to the highway;</li> <li>(j) make such investigations in or on the Land as required;</li> <li>(k) alter, fell, lop or cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be standing on the Land or other land which would if not felled, lopped, cut or removed would obstruct or interfere with the operation of the cables and ancillary equipment including ducting;</li> <li>(l) take and use, remove and discharge water from the Land, and to install, retain, use, maintain, inspect, alter, remove, refurbish, reconstruct, replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and manage waterflows in any drains, watercourses and culverts, install, use, inspect, maintain, adjust, alter, renew, repair, test or cleanse drainage schemes on the Land or reinstate any existing drainage scheme on the Land;</li> <li>(m) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus including but not limited to electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications cables and any ancillary equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, service media (including the pipes, cables or conduits or apparatus of statutory undertakers);</li> </ul>



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	<ul> <li>(n) erect and remove temporary fencing, gates, walls, barriers or other means of enclosure, remove fences and structures within the Land during any period during which construction, maintenance, repair or renewal is being carried out (subject to erection of any temporary stock-proof fencing as is reasonably required and the re-instatement or suitable replacement of the fences or structures following the exercise of the rights);</li> <li>(o) store and stockpile materials (including excavated material);</li> <li>(p) create boreholes and trial excavation pits for the purposes of intrusively surveying the Land and monitoring the use of any trenchless installation technique, to keep in place and monitor the same through construction, maintenance repair, replacement or decommissioning and to reinstate the Land;</li> </ul>
	<ul> <li>(q) excavate materials below ground level, including soils, and to store and re-use or dispose of the same, and in so excavating to undertake any works, including works of protection or removal of archaeological remains as may be required by any written scheme of investigation approved under this Order;</li> </ul>
	<ul> <li>(r) lay out temporary paths and bridleways for public use as temporary diversions for public rights of way which are interfered with during any period in which construction, maintenance, repair, renewal or decommissioning is being carried out;</li> </ul>
	(s) install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out mitigation, maintenance, remediation works for environmental or ecological mitigation or



(1) Number of plot shown on	(2) Purpose for which rights may be acquired
land plans	Purpose for which rights hidy be acquired
	<ul> <li>enhancement works, including temporary works for noise alleviation measures and the installation of temporary barriers for the protection of fauna;</li> <li>(t) carry out such works (together with associated fencing) required by a planning permission and/or consent now or to be granted over the Land and/or in accordance with any necessary licences relating to protected species and/or wildlife; and</li> <li>(u) (in an emergency only when the cables are temporarily unusable) to lay down, install, use, maintain and inspect underground cables, telephone signalling and fibre-optic cables and ancillary equipment, associated works and other conducting media together with conduits or pipes for containing the same in and under the Land.</li> <li>2. Transition joint bay rights</li> <li>Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised project to lay down, install, adjust, alter, construct, create, use, maintain, repair, renew, upgrade, inspect, remove and replace transition joint bays.</li> <li>3. Restrictive covenants</li> </ul>

Unrestricted 005028761-<del>02</del>04



(1) Number of plot shown on	(2) Purpose for which rights may be acquired
land plans	
	A restrictive covenant over the Land for the benefit of the remainder of the Order land to:
	<ul> <li>(v) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erection of any buildings or construction erection or works of any kind (including the foundations or footings thereto);</li> </ul>
	<ul> <li>(w) prevent anything to be done by way of hard surfacing of the Land with concrete of any kind or with any other material or surface whatsoever without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed if the proposed surfacing would not cause damage to relevant part of the authorised project nor make it materially more difficult or expensive to maintain the authorised project);</li> </ul>
	<ul> <li>(x) prevent anything to be done by way of excavation of any kind in the Land nor any activities which would alter, increase or decrease ground cover or soil levels in any manner whatsoever without the consent in writing of the undertaker save as are reasonably required for agricultural activities (being ploughing to no deeper than 0.6m for the purposes of arable farming);</li> </ul>
	<ul> <li>(y) prevent the planting or growing within the Land of any trees, shrubs or underwood without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed trees, shrubs or underwood would not cause damage to</li> </ul>



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	<ul> <li>the relevant part of the authorised project nor make it materially more difficult or expensive to access the relevant part of the authorised project);</li> <li>(z) prevent anything being done which may interfere with free flow and passage of electricity or telecommunications through the cables or support for the authorised project;</li> <li>(aa) prevent carrying out operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with the exercise of the rights or damage the authorised project; and</li> <li>(bb) prevent any activity which would in the reasonable opinion of the undertaker result in the disturbance of ecological mitigation areas or areas of habitat creation including any ploughing or grazing without the prior written consent of the undertaker.</li> </ul>
Cable rights and restrictive	covenants under existing infrastructure
02-027,02-033,03-006, 04-004,04-013,06-004,	1. Cable rights
06-012,06-021,08-005, 08-014,08-021,09-006,	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised project to—
10-002, 11-004, 11-010, 11-011, 11-012, 12-005,	(a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables (including the removal of

Unrestricted 005028761-<del>02</del>04



(1) Number of plot shown on land plans	(2) Purpose for which rights may be acquired
14-011, 15-004, 16-007, 17-001, 17-005, 18-003, 18-004, <del>18-026</del> , 18-027, 18-029, 18-032, <del>18-033,</del> <del>18-034, 18-036, 18-037,</del> 18-040	<ul> <li>materials including spoil) in, under, over and/or on the Land, allow the installation and use of electricity cables in the additional ducts, together with such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (all collectively referred to as the "cables"), and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling beneath sea defences, watercourses, roads and railways;</li> <li>(b) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the additional ducts for electricity cables (including the removal of materials including spoil) in, under, over and/or on the Land, allow the installation and use of electricity cables in the additional ducts, together with such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (all collectively referred to as the "cables"), and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling beneath sea defences, watercourses, roads and railways;</li> </ul>



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	<ul> <li>(c) enter, be on, and break open and break up the surface of the Land and remain with or without plant, vehicles, machinery, apparatus and equipment which is ancillary to the purposes of transmitting electricity along the cables;</li> <li>(d) benefit from continuous vertical and lateral support for the authorised project;</li> <li>(e) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, inspecting, removing and replacing the ducting, electrical infrastructure and cables;</li> <li>(f) construct and install and thereafter use the Land for all necessary purposes for the commissioning, construction, repair, testing and maintenance of the ducting, electrical infrastructure and cables in, on or under the Land;</li> <li>(g) place and use plant, machinery, structures and temporary structures within the Land for the purposes of the installation, construction, maintenance, repairing, renewing, upgrading, inspecting, removal and replacing of the ducting, electrical infrastructure and cables and to erect temporary signage and provide measures for the benefit of public and personnel safety;</li> <li>(h) erect fencing, gates, walls, barriers or other means of enclosure, and create secure working areas and compounds including trenchless installation technique compounds and working areas;</li> </ul>



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	<ul> <li>(i) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair or renewal is being carried out;</li> <li>(j) Erect temporary bridges and supporting or protective structures for the purposes of access to adjoining land;</li> <li>(k) effect access and egress to the highway;</li> <li>(l) make such investigations in or on the Land as required;</li> <li>(m) alter, fell, lop or cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be standing on the Land or other land which would if not felled, lopped, cut or removed would obstruct or interfere with the operation of the cables and ancillary equipment including ducting;</li> <li>(n) take and use, remove and discharge water from the Land, and to install, retain, use, maintain, inspect, alter, remove, refurbish, reconstruct, replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and manage waterflows in any drains, watercourses and culverts, install, use, inspect, maintain, adjust, alter, renew, repair, test or cleanse drainage schemes on the Land or reinstate any existing drainage scheme on the Land;</li> </ul>



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	<ul> <li>(o) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus including but not limited to electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications cables and any ancillary equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, service media (including the pipes, cables or conduits or apparatus of statutory undertakers);</li> <li>(p) erect and remove temporary fencing, gates, walls, barriers or other means of enclosure, remove fences and structures within the Land during any period during which construction, maintenance, repair or renewal is being carried out (subject to erection of any temporary stock-proof fencing as is reasonably required and the re-instatement or suitable replacement of the fences or structures following the exercise of the rights);</li> <li>(q) store and stockpile materials (including excavated material);</li> <li>(r) create boreholes and trial excavation pits for the purposes of intrusively surveying the land and monitoring the use of any trenchless installation technique, to keep in place and monitor the same through construction, maintenance repair, replacement or decommissioning and to reinstate the Land;</li> <li>(s) excavate materials below ground level, including soils, and to store and re-use or dispose of the same, and in so excavating to undertake any works, including works of protection or removal of archaeological remains as may be required by any written scheme of investigation approved under this Order;</li> </ul>



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	<ul> <li>(t) lay out temporary paths and bridleways for public use as temporary diversions for public rights of way which are interfered with during any period in which construction, maintenance, repair, renewal or decommissioning is being carried out; install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out mitigation, maintenance, remediation works for environmental or ecological mitigation or enhancement works, including temporary works for noise alleviation measures and the installation of temporary barriers for the protection of fauna;</li> <li>(u) carry out such works (together with associated fencing) required by a planning permission and/or consent now or to be granted over the Land and/or in accordance with any necessary licences relating to protected species and/or wildlife; and</li> <li>(v) (in an emergency only when the cables are temporarily unusable) to lay down, install, use, maintain and inspect underground cables, telephone signalling and fibre-optic cables and ancillary equipment, associated works and other conducting media together with conduits or pipes for containing the same in and under the Land.</li> <li>2. Restrictive covenants</li> <li>A restrictive covenant over the Land for the benefit of the remainder of the Order land to:</li> </ul>

Unrestricted 005028761-<del>02</del>04



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	<ul> <li>(w) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erection of any buildings or construction, erection or works of any kind (including the foundations or footings thereto) other than those related to works for the benefit of existing highway or railway infrastructure;</li> <li>(x) prevent anything to be done by way of excavation of any kind in the Land nor any activities which would alter, increase or decrease ground cover or soil levels by greater than 1 metre whatsoever without the consent in writing of the undertaker, save where such works are reasonably required for the exercise of statutory functions or rights in relation any public highway or railway on the Land and will not damage, undermine or interfere with the cables;</li> <li>(y) prevent the planting or growing within the Land of any trees, shrubs or underwood without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed trees, shrubs or underwood would not cause damage to the relevant part of the authorised project;</li> <li>(z) prevent anything being done which may interfere with free flow and passage of electricity or telecommunications through the cables or support for the authorised project;</li> <li>(aa) prevent carrying out operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with the exercise of the rights or damage the authorised project; and</li> </ul>



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	(bb) prevent any activity which would in the reasonable opinion of the undertaker result in the disturbance of ecological mitigation areas or areas of habitat creation including any ploughing or grazing without the prior written consent of the undertaker.
Mitigation work areas access rights, mitigation rights and restrictive covenants	
<u>18-010a, 18-014a,</u> <u>18-015,</u> 18-016, 18-017, 18-019, 18-020, <u>18-021,</u> <u>18-022a</u> 18-023, <u>18-025a,</u> 18-024	<ol> <li>Permanent access rights         Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised project to—         (a) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of the authorised project, the inspection, testing, maintenance, renewal, upgrading, replacement and removal of the ducting, electrical infrastructure and cables and connection into any adjacent cables and associated works, to take plant and equipment on to adjoining land and make such investigations in or on the Land which is     </li> </ol>
	<ul> <li>ancillary for the purposes of exercise of the rights;</li> <li>(b) enter, be on, and break open and break up the surface of the Land and remain with or without plant, vehicles, machinery, apparatus and equipment which is ancillary to the purposes of transmitting electricity along the cables;</li> </ul>



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	<ul> <li>(c) construct, use, maintain and improve a permanent means of access including visibility splays, and retain, maintain, straighten, widen, repair, alter, upgrade and use existing access routes for the purposes of accessing the Land, adjoining land and highway;</li> <li>(d) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning is being carried out;</li> <li>(e) erect temporary bridges and supporting or protective structures for the purposes of access to adjoining land;</li> <li>(f) place and use plant, machinery, structures and temporary structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel safety;</li> <li>(g) alter, fell, lop, cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land;</li> <li>(h) repair, improve, renew, remove, relocate and plant trees, woodland, shrubs, hedgerows, seeding and other ecological measures together with the right to maintain, inspect and replant such trees, shrubs and landscaping;</li> </ul>



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	<ul> <li>(i) erect and remove temporary fencing, gates, walls, barriers or other means of enclosure; and</li> <li>(j) lay out temporary paths and bridleways for public use as temporary diversions for public rights of way which are interfered with during any period in which construction, maintenance, repair or renewal decommissioning is being carried out.</li> </ul>
	2. Mitigation works rights
	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised project to—
	(k) install, execute, implement, retain, repair, improve, renew, remove, relocate and plant trees, woodland, shrubs, hedgerows, seeding and other ecological measures together with the right to maintain, inspect and replant such trees, shrubs and landscaping;
	<ul> <li>(I) install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out mitigation, maintenance, remediation works for environmental or ecological mitigation or enhancement works, including temporary works and the installation of temporary barriers for the protection of fauna</li> </ul>
	<ul> <li>(m) erect maintain, repair, improve and remove permanent fencing, gates, barriers or other means of enclosure; and</li> </ul>



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	<ul> <li>(n) carry out such works (together with associated fencing) required by a planning permission and/or consent now or to be granted over the Land in accordance with any necessary licences relating to protected species and/or wildlife</li> </ul>
	3. Restrictive covenants
	A restrictive covenant over the Land for the benefit of the remainder of the Order land to prevent any activity which would in the reasonable opinion of the undertaker result in the disturbance of ecological mitigation areas or areas of habitat creation including any ploughing or grazing, during the period within which the undertaker is bound by any consent to maintain that ecological mitigation areas or areas of habitat creation, without the prior written consent of the undertaker.
National Grid substation we	orks area rights
20-004	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised project to—
	<ul> <li>(a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables (including the removal of materials including spoil) in, under, over and/or on the Land, allow the installation and use of</li> </ul>



(1) Number of plot shown on land plans	(2) Purpose for which rights may be acquired
	<ul> <li>electricity cables in the additional ducts, together with such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (all collectively referred to as the "cables") and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling beneath sea defences, watercourses, roads and railways;</li> <li>(b) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the additional ducts for electricity cables (including the removal of materials including spoil) in, under, over and/or on the Land, allow the installation and use of electricity cables in the additional ducts, together with such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (all collectively referred to as the "cables") and in doing so to use or resort to trenchless installation techniques including (but not limited to) directional drilling beneath sea defences, watercourses, roads and railways;</li> </ul>



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	<ul> <li>(c) enter, be on, and break open and break up the surface of the Land and remain with or without plant, vehicles, machinery, apparatus and equipment which is ancillary to the purposes of transmitting electricity along the cables;</li> <li>(d) benefit from continuous vertical and lateral support for the authorised project;</li> <li>(e) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, inspecting the authorised project and for removing and replacing the ducting, electrical infrastructure and cables;</li> <li>(f) use, maintain, renew improve and alter existing accesses, roads, streets, tracks or ways over the Land, providing that such use is not exclusive and exercise of this right must not prevent or unreasonably inhibit use by other parties;</li> <li>(g) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning is being carried out;</li> </ul>



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	<ul> <li>(h) place and use plant, machinery, structures and temporary structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel safety;</li> </ul>
	<ul> <li>alter, fell, lop, cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land; and</li> </ul>
	<ul> <li>take and use, remove and discharge water from the Land and to install, retain, use, maintain, inspect, alter, remove, refurbish, reconstruct, replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and manage waterflows in any drains, watercourses and culverts, to lay down, install, adjust, alter, construct, create, use, maintain, repair, renew, upgrade, inspect, remove and replace a drainage scheme on the Land; and</li> </ul>
	(k) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus including but not limited to electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications cables and any ancillary equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, service media (including the pipes, cables or conduits or apparatus of statutory undertakers).



(1)	(2)
Number of plot shown on land plans	Purpose for which rights may be acquired
	A restrictive covenant over the Land for the benefit of the remainder of the Order land—
	<ul> <li>(I) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erection of any buildings or construction, erection or works of any kind (including the foundations or footings thereto) without the prior written consent of the undertaker;</li> <li>(m) prevent the planting or growing within the Land of any trees, shrubs or underwood without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed trees, shrubs or underwood would not cause damage to the relevant part of the authorised project nor make it materially more difficult or expensive to access the relevant part of the authorised project);</li> <li>(n) prevent anything being done which may interfere with free flow and passage of electricity or telecommunications through the cables or support for the authorised project; and</li> <li>(o) to prevent carrying out operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with the exercise of the rights or damage the authorised project.</li> </ul>



## **1.3 Structure of this book of reference**

- 10. This Book of Reference is comprised of five Parts, in accordance with Regulation 7(1) of the Regulations, as follows:
  - Part 1 contains the names and addresses of those who own, lease, occupy or have another interest in the land that will be affected by the authorised development and the rights contained in the draft DCO. A person is within Category 1 if the Applicants after making diligent inquiry, know that they are an owner, lessee, tenant, or occupier of the land. A person is within Category 2 if they are interested in the land or have the power to sell and convey or release the land. Parties recorded in Part 3 of this Book of Reference have also been included in Part 1;
  - Part 2 lists persons who may be entitled to make a relevant claim, also called category 3 persons. A person is within Category 3 if the Applicants believes that, if the Order were to be made and fully implemented, they would or might be entitled to make a relevant claim as defined in section 57(6) of the 2008 Act. A relevant claim is a claim under section 10 of the Compulsory Acquisition Act 1965, a claim under part 1 of the Land Compensation Act 1973, or a claim under section 152(3) of the 2008 Act. This part therefore contains the names and addresses of those who have an interest in the Order land and may be entitled to claim compensation for loss resulting from the implementation of the Order and use of the Projects. This part also contains the names and addresses of those without an interest in the draft DCO land, who may be entitled to claim compensation for loss resulting from the implementation of the Order and use of the authorised development;
  - Part 3 contains the names and addresses of those entitled to enjoy easements or other private rights which it is proposed may be extinguished, suspended or interfered with in connection with the authorised development, pursuant to the Order;
  - Part 4 identifies plots in which there is a Crown interest that will be affected by the authorised development and the rights contained in the Order; and
  - Part 5 identifies plots which constitute "special category land" for the purposes of section 132 of the 2008 Act that will be affected by the authorised development and the rights contained in the draft DCO. This is land –
    - the acquisition of which is subject to special parliamentary procedure;
    - which is special category land
    - which is replacement land

Unrestricted 005028761-<del>02</del>04

### Page 34



The Applicants believes that if the Order land listed within Part 5, when burdened with the Order right, will be no less advantageous than it was before to the following persons-

- the persons in whom it is vested,
- other persons, if any, entitled to rights of common or other rights, and
- the public.

The Applicants are therefore of the opinion that while it has identified open space within the Order Land, there is no requirement for the provision of replacement or exchange land.

### **1.4 Book of reference notes**

- 11. Part 1 of this Book of Reference provides the area in square metres of all land included in the DCO. Areas have not been repeated in Parts 2 to 5 of this Book of Reference.
- 12. The term 'approximately' is used before all plot area measurements as these measurements are given in square metres and each measurement is rounded up to the nearest whole square metre.
- 13. The tracked version of the clean Book of Reference highlights the updates to land interest information since the previous submission of the Book of reference (APP-031):
  - a. Those entries struck through and highlighted in red have been removed
  - b. Those entries highlighted in green have been added; and
  - c. Those entries highlighted in orange have been updated, name, address or qualifier.

Page 35



# **1.5 Schedule of Changes to Plot Interests**

Plot Number	Date of Change	Contact Name	Nature of Interest	Change Made	Reasons for Change	Revision
18-010a, 18- 014a, 18-015a, 18-021a, 18- 022a, 18-025a, 18-027a	10/12/2024	Keyland Developments Limited	Category 2	Interest added	Plot created due to change request application, new interests added.	04
18-010a, 18- 014a, 18-015a, 18-021a, 18- 022a, 18-025a, 18-027a	10/12/2024	Albanwise Limited	Category 1	Interest added	Plot created due to change request application, new interests added.	04
18-010a, 18- 014a, 18-015a, 18-021a, 18- 022a, 18-025a, 18-027a	10/12/2024	National Westminster Bank PLC	Category 2	Interest added	Plot created due to change request application, new interests added.	04



Plot Number	Date of Change	Contact Name	Nature of Interest	Change Made	Reasons for Change	Revision
18-010a, 18- 014a, 18-015a, 18-021a, 18- 022a, 18-025a, 18-027a	10/12/2024	Orsted Hornsea Project Four Limited	Category 2	Interest added	Plot created due to change request application, new interests added.	04
18-010a, 18- 014a, 18-015a, 18-021a, 18- 025a, 18-027a	10/12/2024	Ineos Manufacturing (Hull) Limited	Category 2	Interest added	Plot created due to change request application, new interests added.	04
18-010a, 18- 015a, 18-025a	10/12/2024	Yorkshire Water Services Limited	Category 2	Interest added	Plot created due to change request application, new interests added.	04
18-010a, 18- 027a	10/12/2024	Oliver White	Category 1	Interest added	Plot created due to change request application, new interests added.	04
18-010a, 18- 027a	10/12/2024	Pamela White	Category 1	Interest added	Plot created due to change request application, new interests added.	04



Plot Number	Date of Change	Contact Name	Nature of Interest	Change Made	Reasons for Change	Revision
18-010a, 18- 027a	10/12/2024	The Executor Of The Estate Of The Late Andrew White	Category 1	Interest added	Plot created due to change request application, new interests added.	04
18-014a, 18- 015a, 18-021a, 18-022a, 18- 025a	10/12/2024	Unknown       Category 2       Interest added       Plot created due to change request application, new interests added.		04		
18-014a, 18- 015a, 18-021a, 18-022a, 18- 025a	10/12/2024	Unknown	Category 2	Interest added	Plot created due to change request application, new interests added.	04
18-014a, 18- 022a	10/12/2024	Robert Charles Elvidge	Category 1	Interest added	Plot created due to change request application, new interests added.	04
18-015	10/12/2024	Ineos Manufacturing (Hull) Limited	Category 2	Removed Interest	Plot reduced in area size due to change request, interest removed.	04
18-015	10/12/2024	KCOM Group Limited	Category 2	Removed Interest	Plot reduced in area size due to change request, interest removed.	04



Plot Number	Date of Change	Contact Name	Nature of Interest	Change Made	Reasons for Change	Revision
18-015a	10/12/2024	KCOM Group Limited	Category 2	Interest added	Plot created due to change request application, new interests added.	04
18-015a, 18- 021a, 18-025a	10/12/2024	Andrew Woodmansey	Category 1	Interest added	Plot created due to change request application, new interests added.	04
18-015a, 18- 022a	10/12/2024	National Gas Transmission PLC	Category 2	Interest added	Plot created due to change request application, new interests added.	04
18-021a	10/12/2024	Barclays Bank PLC	Category 2	Interest added	Plot created due to change request application, new interests added.	04
18-021a	10/12/2024	Unknown	Category 2	Interest added	Plot created due to change request application, new interests added.	04
18-021a	10/12/2024	Ineos Manufacturing (Hull) Limited	Category 1	Interest added	Plot created due to change request application, new interests added.	04
18-026	10/12/2024	Northern Powergrid (Yorkshire) PLC	Category 2	Removed Interest	Plot archived due to change request application, interest removed.	04



Plot Number	Date of Change	Contact Name	Nature of Interest	Change Made	Reasons for Change	Revision
18-026, 18-033, 18-034, 18-036, 18-037	10/12/2024	Albanwise Limited Category 1 Removed Interest Plot archived due to change request application, interest removed.		04		
18-027	10/12/2024	National Grid Electricity Transmission PLC	Category 2	Removed Interest	Plot reduced in area size due to change request, interest removed.	04
18-027	10/12/2024	Northern Powergrid (Yorkshire) PLC	Category 2	Removed Interest	Plot reduced in area size due to change request, apparatus interest removed.	04
18-027	10/12/2024	Yorkshire Water Services Limited	Category 2	Removed Interest	Plot reduced in area size due to change request, interest removed.	04
18-027a	10/12/2024	National Grid Electricity Transmission PLC	Category 2	Interest added	Plot created due to change request application, new interests added.	04
18-027a	10/12/2024	Northern Powergrid (Yorkshire) PLC	Category 2	Interest added	Plot created due to change request application, new interests added.	04
18-033, 18-034, 18-036, 18-037	10/12/2024	Keyland Developments Limited	Category 2	Removed Interest	Plot archived due to change request application, interest removed.	04



Plot Number	Date of Change	Contact Name	Nature of Interest	Change Made	Reasons for Change	Revision
18-033, 18-034, 18-036, 18-037	10/12/2024	2024 Ineos Manufacturing (Hull) Limited		Removed Interest	Plot archived due to change request application, interest removed.	04
18-033, 18-034, 18-036, 18-037	10/12/2024	National Westminster Bank PLC	Category 2	Removed Interest	Plot archived due to change request application, interest removed.	04
18-033, 18-034, 18-036, 18-037	10/12/2024	Orsted Hornsea Project Four Limited	Category 2	Removed Interest	Plot archived due to change request application, interest removed.	04
18-033, 18-034, 18-036, 18-037	10/12/2024	Unknown	Category 2	Removed Interest	Plot archived due to change request application, interest removed.	04
18-033, 18-034, 18-036, 18-037	10/12/2024	Unknown	Category 2	Removed Interest	Plot archived due to change request application, interest removed.	04
18-033, 18-034, 18-037	10/12/2024	Matthew Yeo	Category 1	Removed Interest	Plot archived due to change request application, interest removed.	04
18-033, 18-034, 18-037	10/12/2024	Moira Yeo	Category 1	Removed Interest	Plot archived due to change request application, interest removed.	04



Plot Number	Date of Change	Contact Name	ntact Name Nature of Interest Change Made Reasons for Change		Reasons for Change	Revision
18-033, 18-034, 18-037	10/12/2024	Stuart Yeo	Category 1	Removed Interest	Plot archived due to change request application, interest removed.	04
18-036	10/12/2024	Andrew Woodmansey	Category 1	Interest removed	Plot archived due to change request application, interest removed.	04

Unrestricted 005028761-<del>02</del>04 Page 42



# **1.6 Schedule of Changes to Plots**

Plot Number	Date of Change	Change Made	Reasons for Change	Revision
18-010	09/12/2024	Extent of area size updated.	Change application caused the area of this plot to change.	04
18-010a	09/12/2024	New Plot added to Book of Reference and introduction pages	Change application created new plot.	04
18-014	09/12/2024	Extent of area size and land description updated.	Change application caused the description and area of this plot to change.	04
18-014a	09/12/2024	New Plot added to Book of Reference and introduction pages	Change application created new plot.	04
18-015	09/12/2024	Extent of area size, land description and acquisition type updated.	Change application caused the acquisition type, land description and area of this plot to change.	04
18-015a	09/12/2024	New Plot added to Book of Reference and introduction pages	Change application created new plot.	04
18-021	09/12/2024	Extent of area size and acquisition type updated.	Change application caused the acquisition type and area of this plot to change.	04



Plot Number	Date of Change	Change Made	Reasons for Change	Revision
18-021a	09/12/2024	New Plot added to Book of Reference and introduction pages	Change application created new plot.	04
18-022	09/12/2024	Extent of area size updated.	Change application caused the area of this plot to change.	04
18-022a	09/12/2024	New Plot added to Book of Reference and introduction pages	Change application created new plot.	04
18-025	09/12/2024	Extent of area size updated.	Change application caused the area of this plot to change.	04
18-025a	09/12/2024	New Plot added to Book of Reference and introduction pages	Change application created new plot.	04
18-026	09/12/2024	Plot removed	Plot was not required as part of the change application request.	04
18-027	09/12/2024	Extent of area size and land description updated.	Change application caused the description and area of this plot to change.	04
18-027a	09/12/2024	New Plot added to Book of Reference and introduction pages	Change application created new plot.	04



Plot Number	Date of Change	Change Made	Reasons for Change	Revision
18-029	09/12/2024	Extent of area size and land description updated.	Change application caused the description and area of this plot to change.	04
18-032	09/12/2024	Extent of area size and land description updated.	Change application caused the description and area of this plot to change.	04
18-033	09/12/2024	Plot removed	Plot was not required as part of the change application request.	04
18-034	09/12/2024	Plot removed	Plot was not required as part of the change application request.	04
18-036	09/12/2024	Plot removed	Plot was not required as part of the change application request.	04
18-037	09/12/2024	Plot removed	Plot was not required as part of the change application request.	04



	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	rind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	
01-001	Temporary Possession	2797 square metres of public road and verges (North Turnpike)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Robert Andrew Watson Corner Farm Eastgate Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TS (in respect of subsoil beneath public highway) The Executor of the Estate of the Late Jean Catherine Frank East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT (in respect of subsoil beneath public highway) Unsme	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Unknown



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	nd Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-001 cont'd			Wilfred Michael Houghton East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT (in respect of subsoil beneath public highway)			
01-002	Temporary Possession	road and verges (North Turnpike)	Unknown (in respect of mines and minerals) York Diocesan Board Of Finance Limited Diocese Of York Amy Johnson Way York YO30 4XT	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of an Option Agreement dated 16 April 2019) Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of an Option Agreement dated 16 April 2019)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
01-003	Temporary Possession		Unknown (in respect of mines and minerals) York Diocesan Board Of Finance Limited Diocese Of York Amy Johnson Way York YO30 4XT	NONE	Diocese Of York	Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of an Option Agreement dated 16 April 2019) Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of an Option Agreement dated 16 April 2019)			
01-004	Temporary Possession	track (east of North Turnpike)	The Executor of the Estate of the Late Jean Catherine Frank East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public footpath (Ulrome FP6))	Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of an Option Agreement dated 29 April 2019)			



	Land w	/hich is proposed to be subject to	BOOK OF R (i) powers of compulsory acquisition, (ii)	•	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	County of Eas Qualifying persons under Regulation 7(1	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans	·		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-004 cont'd			Wilfred Michael Houghton East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT		The Executor of the Estate of the Late Jean Catherine Frank East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT Wilfred Michael Houghton East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT	Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of an Option Agreement dated 29 April 2019) Unknown (in respect of rights, covenants and restrictions as contained in a Transfer dated 3 February 1977)
01-005	Temporary Possession	1037 square metres of agricultural land and access track (east of North Turnpike)	The Executor of the Estate of the Late Jean Catherine Frank East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT Wilfred Michael Houghton East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT	NONE	The Executor of the Estate of the Late Jean Catherine Frank East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT Wilfred Michael Houghton East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT	Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect to an Option Agreement dated 29 April 2019) Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect to an Option Agreement dated 29 April 2019)



	Land w	hich is proposed to be subject to	BOOK OF R	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans 01-006	Extent of acquisition or use Description of land Regulations 2009					Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 Doggerbank Offshore Wind Farm Project
			Jean Catherine Frank East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT Wilfred Michael Houghton East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT		Jean Catherine Frank East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT Wilfred Michael Houghton East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT	1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 29 April 2019) Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 29 April 2019) Unknown (in respect of rights of access to maintain and repair service media dated 3 February 1977)
01-007	Temporary Possession		The Executor of the Estate of the Late Jean Catherine Frank East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT	NONE	The Executor of the Estate of the Late Jean Catherine Frank East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT	Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 29 April 2019)



	Land w	hich is proposed to be subject to	BOOK OF F : (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-007 cont'd			Wilfred Michael Houghton East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT		Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT	Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 29 April 2019) Unknown (in respect of rights of access to maintain and repair service media dated 3 February 1977)
01-008	Temporary Possession	landward of the Mean High Water at Ulrome (east of North Turnpike)	Glendon Estates Gresham House 5-7 St. Pauls Street LEEDS West Yorkshire LS1 2JG	Clifford Noel Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD Graham Clifford Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD	Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD	Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 28 May 2019)





				ind Farm Development Consent Order EFERENCE - PART 1		
	Land w	hich is proposed to be subject to		right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-008 cont'd				Martin Geoffrey Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD Neil Anthony Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD	Martin Geoffrey Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD Neil Anthony Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD	
01-009	Temporary Possession	landward of the Mean High Water at Ulrome (east of North Turnpike)	Glendon Estates Gresham House 5-7 St. Pauls Street LEEDS West Yorkshire LS1 2JG	NONE	5-7 St. Pauls Street LEEDS West Yorkshire LS1 2JG	Clifford Noel Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD (in respect of rights contained in a Lease dated 5 August 2005)



	Land wh	nich is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to o st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))		
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
01-009 cont'd						Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 28 May 2019) Graham Clifford Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD (in respect of rights contained in a Lease dated 5 August 2005) Martin Geoffrey Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD (in respect of rights contained in a Lease dated 5 August 2005) Martin Geoffrey Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD (in respect of rights contained in a Lease dated 5 August 2005)		





	Land w	which is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-009 cont'd						Neil Anthony Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD (in respect of rights contained in a Lease dated 5 August 2005) Unknown (in respect of rights of access)
01-010	Temporary Possession	seaward of the Mean High	The Executor of the Estate of the Late Jean Catherine Frank East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT	NONE	The Executor of the Estate of the Late Jean Catherine Frank East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT	Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 29 April 2019)
			The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH		The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 29 April 2019)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)		carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	County of East Riding of Yorkshire         Category 1         Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure Regulations 2009         Extent of acquisition or use       Description of land         Owners       Lessees or Tenants       Occupiers				Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and
01-010 cont'd			or Reputed Owners Wilfred Michael Houghton East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT	or Reputed Lessees or Tenants	or Reputed Occupiers Wilfred Michael Houghton East Farm Ulrome DRIFFIELD East Riding of Yorkshire YO25 8TT	Procedures) Regulations 2009 Unknown (in respect of rights of access)
01-011	Temporary Possession	Water lying to the north east of Ulrome (excluding all interests of the Crown)	Gresham House 5-7 St. Pauls Street	NONE	Glendon Estates Gresham House 5-7 St. Pauls Street LEEDS West Yorkshire LS1 2JG The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Clifford Noel Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD (in respect of rights contained in a Lease dated 5 August 2005) Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 28 May 2019)



	Land wh	nich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to c st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-011 cont'd						Graham Clifford Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD (in respect of rights contained in a Leas dated 5 August 2005) Martin Geoffrey Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD (in respect of rights contained in a Leas dated 5 August 2005) Neil Anthony Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD (in respect of rights contained in a Leas dated 5 August 2005) Neil Anthony Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD (in respect of rights contained in a Leas dated 5 August 2005) Unknown (in respect of rights of access)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)		carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners	st Riding of Yorkshire Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009 Lessees or Tenants or Reputed Lessees or Tenants	Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-012	Temporary Possession	foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests	1 St. James's Market	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	or Reputed Occupiers East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	NONE
01-013	Temporary Possession	foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests	1 St. James's Market	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)



	Land w	hich is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-013 cont'd						Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
01-014	Temporary Possession	foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests	1 St. James's Market	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)



	Land w	hich is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
Lanu Fians			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
01-014 cont'd						Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)	
01-015	Temporary Possession	foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests	1 St. James's Market	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	BEVERLEY East Riding of Yorkshire HU17 9BA	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)	



	Land w	hich is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-015 cont'd						Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
02-001	Acquisition of Rights	foreshore and sea landward and seaward of the Mean Low	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	BEVERLEY East Riding of Yorkshire HU17 9BA	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1)	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-001 cont'd						Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
02-002	Acquisition of Rights	seaward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	NONE
02-003	Acquisition of Rights	30458 square metres of foreshore seaward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	1 St. James's Market	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)





	Land w	which is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-003 cont'd						(Applications: Prescribed Forms and
02-004	Acquisition of Rights	122 square metres of foreshore seaward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	Michael Braddock 165 Sissons Road LEEDS West Yorkshire LS10 4LP	NONE	West Yorkshire LS10 4LP	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)



	Land w	hich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-004 cont'd			The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH		The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) Rodrica Consuelo Straker Burton Constable Hall Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
02-005	Acquisition of Rights	Water lying to the east of	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner) Unknown	NONE	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner) Unknown	Unknown



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)		carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	County of East Riding of Yorkshire  Category 1  Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures  Regulations 2009  Owners Lessees or Tenants Occupiers				Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and
02-006	Acquisition of Rights	landward of the Mean High Water lying to the east of Skipsea (north of Hornsea Road)	or Reputed Owners East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	or Reputed Lessees or Tenants NONE	or Reputed Occupiers East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	Procedures) Regulations 2009 NONE
02-007	Acquisition of Rights	seaward of the Mean High Water lying to the east of Skipsea (excluding all interests	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-007 cont'd						Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
02-008	Acquisition of Rights	Water lying to the east of Skipsea (excluding all interests of the Crown)	165 Sissons Road LEEDS West Yorkshire LS10 4LP	NONE	165 Sissons Road LEEDS West Yorkshire LS10 4LP	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
			The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH		Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to a t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-008 cont'd						Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
02-009	Acquisition of Rights	7 square metres of shore landward of the Mean High Water at Skipsea (north of Hornsea Road)	Michael Braddock 165 Sissons Road LEEDS West Yorkshire LS10 4LP		West Yorkshire LS10 4LP	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)



	Land w	hich is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-009 cont'd						Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
02-010	Acquisition of Rights	foreshore seaward of the Mean High Water lying to the east of Skipsea (excluding all interests	1 St. James's Market	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)



	Land w	/hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-010 cont'd						Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
02-011	Acquisition of Rights	Water lying to the east of	Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE	NONE	Office F12 Beverley Enterprise Centre	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
			The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH		Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-011 cont'd						Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
02-012	Acquisition of Rights	Water at Skipsea (north of Cliff Road, excluding all interests of the Crown)	Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE	NONE	Beck View Road BEVERLEY East Riding of Yorkshire HU17 OJT	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
			The John Raleigh Chichester Constable Will Trust Burton Constable Foundation Burton Constable Hall Burton Constable HULL HU11 4LN (in respect of mines and minerals)		SW1Y 4AH	James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)



	Land w	hich is proposed to be subject to	BOOK OF R	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land. and/or (iii) rights to	carry out protective works (Regulation 7(1	)(a))
				st Riding of Yorkshire Category 1		
Number on Land Plans	Extent of acquisition or use	Description of land		ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-012 cont'd			The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH			Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
02-013	Acquisition of Rights	599 square metres of shore landward of the Mean High Water lying to the east of Skipsea (north of Hornsea Road)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of mines and minerals) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner) Unknown	NONE	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner) Unknown	Unknown
02-014	Acquisition of Rights	8 square metres of shore landward of the Mean High Water lying to the east of Skipsea (north of Hornsea Road)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)	NONE	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)	Unknown



	Land w	hich is proposed to be subject to	BOOK O c: (i) powers of compulsory acquisition,	e Wind Farm Development Consent Order F REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights t East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 7(1)(a) of the Infrastructure Planning (Appl Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-014 cont'd			Unknown		Unknown	
02-015	Acquisition of Rights	2144 square metres of shore landward of the Mean High Water lying to the east of Skipsea (north of Hornsea Road)	Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE	NONE	Carr Hill Farm Limited Office F12 Beverley Enterprise Centre Beck View Road BEVERLEY East Riding of Yorkshire HU17 OJT	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) Rodrica Consuelo Straker Burton Constable Hall Burton Constable Hall



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	<pre>/ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire</pre>	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-015 cont'd						Unknown (in respect of right of way contained in a Conveyance dated 4 June 1963)
02-016	Acquisition of Rights		Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of mines and minerals) Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE	NONE	Office F12 Beverley Enterprise Centre Beck View Road BEVERLEY East Riding of Yorkshire HU17 OJT	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	Vind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-016 cont'd						Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) Unknown (in respect of rights of way contained in a Conveyance dated 4 June 1963)
02-017	Acquisition of Rights	6728 square metres of agricultural land and drains (east of Cleeton Lane and north of Hornsea Road, B1242)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of mines and minerals) Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE	NONE	Office F12 Beverley Enterprise Centre Beck View Road BEVERLEY East Riding of Yorkshire HU17 OJT	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Fiding			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-017 cont'd						Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) Unknown (in respect of rights of way contained in a Conveyance dated 4 June 1963)
02-018	Acquisition of Rights	(east of Cleeton Lane)	Elliott Eggs Limited Manor Farm Catfoss Road Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX	NONE	Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester Constable Will Trust) (in respect of mines and minerals)





	Land w	which is proposed to be subject t	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-018 cont'd						Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Elliot Eggs Limited) Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester Constable Will Trust) (in respect of mines and minerals)
02-019	Acquisition of Rights	319 square metres of agricultural land and drains (east of Cleeton Lane)	Elliott Eggs Limited Manor Farm Catfoss Road Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX (as reputed owner) John Henry Warkup Church Farm Beeford Road Skipsea DRIFFIELD East Riding of Yorkshire YO25 8TG (as reputed owner)	NONE	Elliott Eggs Limited Manor Farm Catfoss Road Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX (as reputed owner) John Henry Warkup Church Farm Beeford Road Skipsea DRIFFIELD East Riding of Yorkshire YO25 8TG (as reputed owner)	Unknown



	Land v	which is proposed to be subject	BOOK O to: (i) powers of compulsory acquisition,	e Wind Farm Development Consent Order F REFERENCE - PART 1 . (ii) right to use the land, and/or (iii) rights t East Riding of Yorkshire	o carry out protective works (Regulation 7(2	.)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-019 cont'd			Unknown		Unknown	
02-020	Acquisition of Rights	85758 square metres of agricultural land and drains (north of Hornsea Road, B1242)	Elliott Eggs Limited Manor Farm Catfoss Road Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX	NONE	Elliott Eggs Limited Manor Farm Catfoss Road Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Elliott Eggs Limited)





	Land w	rhich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	.)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-020 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
02-021	Acquisition of Rights	1461 square metres of agricultural land and drains (north of Hornsea Road, B1242)	Elliott Eggs Limited Manor Farm Catfoss Road Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX	NONE	Elliott Eggs Limited Manor Farm Catfoss Road Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)



	Land wh	ich is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ca ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 ?(1)(a) of the Infrastructure Planning (Applicati Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-021 cont'd						James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Elliott Eggs Limited) Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)



	Land w	which is proposed to be subject	BOOK C to: (i) powers of compulsory acquisition	e Wind Farm Development Consent Order )F REFERENCE - PART 1 , (ii) right to use the land, and/or (iii) rights t <sup>:</sup> East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Owners	Category 1 7(1)(a) of the Infrastructure Planning (Appl Regulations 2009 Lessees or Tenants	Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-022	Acquisition of Rights	16 square metres of agricultural land (north of Hornsea Road, B1242)	or Reputed Owners Elliott Eggs Limited Manor Farm Catfoss Road Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX	NONE	or Reputed Occupiers Elliott Eggs Limited Manor Farm Catfoss Road Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester Constable Will Trust) (in respect of mines and minerals) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Elliot Eggs Limited)



	Land w	hich is proposed to be subject to	BOOK OF RI : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-022 cont'd						Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
02-023	Acquisition of Rights	B1242)	Elliott Eggs Limited Manor Farm Catfoss Road Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX	NONE	Elliott Eggs Limited Manor Farm Catfoss Road Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester Constable Will Trust) (in respect of mines and minerals)





	Land w	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-023 cont'd						Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Elliot Eggs Limited) Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
02-024	Acquisition of Rights	25135 square metres of agricultural land (north of Hornsea Road, B1242)	Elliott Eggs Limited Manor Farm Catfoss Road Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX	NONE	Elliott Eggs Limited Manor Farm Catfoss Road Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester Constable Will Trust) (in respect of mines and minerals)



	Land wh	ich is proposed to be subject t	BOOK OF F o: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to ca Ist Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-024 cont'd						James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester Constable Will Trust) (in respect of mines and minerals) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Elliot Eggs Limited) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Rodrica Consuelo Straker Burton Constable Hall Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)



	Land w	hich is proposed to be subject t	BOOK C to: (i) powers of compulsory acquisition	e Wind Farm Development Consent Order DF REFERENCE - PART 1 , (ii) right to use the land, and/or (iii) rights t f East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 n 7(1)(a) of the Infrastructure Planning (Appl Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-025	Acquisition of Rights	3873 square metres of agricultural land (north of Hornsea Road, B1242)	Elliott Eggs Limited Manor Farm Catfoss Road Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX	NONE	Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester Constable Will Trust) (in respect of mines and minerals) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Elliot Eggs Limited) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)



	Land w	hich is proposed to be subject to	BOOK OF RI : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-025 cont'd						Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
02-026	Acquisition of Rights	and verges (Hornsea Road, B1242)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Elliott Eggs Limited Manor Farm Catfoss Road Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX (in respect of subsoil beneath public highway) Unknown	NONE	BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown



	Land w	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1 Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
20110 1 10110			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
02-027	Acquisition of Rights	1607 square metres of public road and verges (Hornsea Road, B1242)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Elliott Eggs Limited Manor Farm Catfoss Road Bewholme DRIFFIELD East Riding of Yorkshire YO25 8DX (in respect of subsoil beneath public highway) Unknown	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown Unknown Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
02-028	Temporary Possession	20 square metres of public road and verges (Hornsea Road, B1242)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Unknown	



	Land w	hich is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, (i	Wind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to ast Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-028 cont'd			Unknown			Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
02-029	Temporary Possession	1 square metres of verges (south of Hornsea Road, B1242)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as reputed owner) Unknown	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as reputed owner) Unknown	Unknown
02-030	Acquisition of Rights	632 square metres of verges (south of Hornsea Road, B1242)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as reputed owner) Unknown	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as reputed owner) Unknown	Unknown



	Land w	hich is proposed to be subject to	BOOK C c: (i) powers of compulsory acquisition	e Wind Farm Development Consent Order )F REFERENCE - PART 1 , (ii) right to use the land, and/or (iii) rights t <sup>;</sup> East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-031	Acquisition of Rights	10884 square metres of agricultural land and verges (east of Hornsea Road, B1242)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	Geoffrey Henry Maltas Clements Farm Hornsea Road Skipsea DRIFFIELD YO25 8ST	Geoffrey Henry Maltas Clements Farm Hornsea Road Skipsea DRIFFIELD YO25 8ST	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
02-031 cont'd						Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)	
02-032	Temporary Possession	(east of Hornsea Road, B1242)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	Geoffrey Henry Maltas Clements Farm Hornsea Road Skipsea DRIFFIELD YO25 8ST	Geoffrey Henry Maltas Clements Farm Hornsea Road Skipsea DRIFFIELD YO25 8ST	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)	





	Land w	/hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009		Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-032 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Rodrica Consuelo Straker Burton Constable Hall Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
02-033	Acquisition of Rights	2484 square metres of public road, verges and hedgerow (Hornsea Road, B1242)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority and in respect of subsoil beneath public highway) Unknown	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to a t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-033 cont'd						Unknown Yorkshire Water Services Limited
						Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
02-034	Acquisition of Rights	(west of Hornsea Road, B1242)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	East Riding of Yorkshire	DRIFFIELD East Riding of Yorkshire YO25 8EE	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)





	Land w	which is proposed to be subject to	BOOK O : (i) powers of compulsory acquisition,	e Wind Farm Development Consent Order F REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights East Riding of Yorkshire	to carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	lications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-034 cont'd						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus)
						Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
02-035	Temporary Possession	agricultural land and verges (west of Hornsea Road, B1242)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE	Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)





	Land w	which is proposed to be subject to	BOOK OF RE b: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-035 cont'd						James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) Rodrica Consuelo Straker Burton Constable Hall Burton Constable Hall Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
02-036	Temporary Possession	881 square metres of public road, access track and verges (Hornsea Road, B1242)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority and in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-036 cont'd			Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE (in respect of subsoil beneath public highway) Unknown			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground sewerage and water apparatus)
02-037	Temporary Possession	570 square metres of agricultural land and access track (west of Hornsea Road, B1242)	Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE	NONE	Carr Hill Farm Limited Office F12 Beverley Enterprise Centre Beck View Road BEVERLEY East Riding of Yorkshire HU17 OJT	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of rights reserved by a Transfer dated 15 February 2012)





	Land wh	ich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o st Riding of Yorkshire	carry out protective works (Regulation 7(1)	)(a))		
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
02-037 cont'd						Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) Rodrica Consuelo Straker Burton Constable Hall Burton Constable Hall Gas a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)		



	Land w	hich is proposed to be subject to	BOOK OF c (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to cast Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land		cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-038	Acquisition of Rights	72661 square metres of agricultural land and public footpath (Skipsea Footpath No.6) (west of Hornsea Road, B1242)	Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE	NONE	Carr Hill Farm Limited Office F12 Beverley Enterprise Centre Beck View Road BEVERLEY East Riding of Yorkshire HU17 OJT East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public footpath (Skipsea Footpath No.6))	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of rights reserved by a Transfer dated 15 February 2012) Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)



	Land w	hich is proposed to be subject to	BOOK OF RE b: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-038 cont'd						Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
03-001	Acquisition of Rights	(east of Bewholme Lane)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as reputed owner) Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE (as reputed owner) Unknown	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as reputed owner) Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE (as reputed owner) Unknown	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights) Unknown



	Land w	which is proposed to be subject	BOOK ( to: (i) powers of compulsory acquisition	re Wind Farm Development Consent Order DF REFERENCE - PART 1 n, (ii) right to use the land, and/or (iii) rights t f East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulatio	Category 1 n 7(1)(a) of the Infrastructure Planning (Appl Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-002	Acquisition of Rights	26356 square metres of agricultural land (east of Bewholme Lane)	Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE	NONE	Carr Hill Farm Limited Office F12 Beverley Enterprise Centre Beck View Road BEVERLEY East Riding of Yorkshire HU17 OJT	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of rights of access) Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Karl Jonathan Warkup) James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	· · ·	Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009 Lessees or Tenants	ations: Prescribed Forms and Procedures) Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and
03-002 cont'd			or Reputed Owners	or Reputed Lessees or Tenants		Procedures) Regulations 2009 Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
03-003	Temporary Possession	(east of Bewholme Lane)	Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE	NONE	Beck View Road BEVERLEY East Riding of Yorkshire HU17 OJT	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Karl Jonathan Warkup)



	Land w	hich is proposed to be subject to	BOOK OF R	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	County of Eas Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-003 cont'd						James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Rodrica Consuelo Straker Burton Constable Hall Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
03-004	Temporary Possession	417 square metres of public road and verges (Bewholme Lane)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)



	Land w	hich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	/ind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning
Lund Huns			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-004 cont'd			Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE (in respect of subsoil beneath public highway) Unknown			Unknown
03-005	Temporary Possession	52 square metres of agricultural land (west of Bewholme Lane)	Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE	NONE	Carr Hill Farm Limited Office F12 Beverley Enterprise Centre Beck View Road BEVERLEY East Riding of Yorkshire HU17 OJT	Unknown (in respect of restrictive covenants as may have been imposed before 11 May 1791) Unknown (in respect of rights of way, water support, drainage, light and other easements granted by a Conveyance
03-006	Acquisition of Rights	730 square metres of public road and verges (Bewholme Lane)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	dated 20 April 1988) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-006 cont'd			Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE (in respect of subsoil beneath public highway) Unknown			Unknown
03-007	Temporary Possession	Bewholme Lane)	Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE	NONE	BEVERLEY East Riding of Yorkshire HU17 OJT	Edward Anthony Bromet The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Karl Jonathan Warkup)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-007 cont'd						James Andrew John Straker The Estate Office South East Wing Burton Constable Hall HULL HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Rodrica Consuelo Straker Burton Constable Hall Burton Constable HULL East Riding of Yorkshire HU11 4LN (as a trustee of the John Raleigh Chichester-Constable Will Trust) (in respect of mines and minerals)
03-008	Temporary Possession	590 square metres of public road and verges (Bewholme Lane)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)



	Land w	hich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009		7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-008 cont'd			Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE (in respect of subsoil beneath public highway) Unknown			Unknown
03-009	Temporary Possession	127 square metres of agricultural land (west of Bewholme Lane)	Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE	NONE	Carr Hill Farm Limited Office F12 Beverley Enterprise Centre Beck View Road BEVERLEY East Riding of Yorkshire HU17 0JT	Unknown (in respect of restrictive covenants as may have been imposed before 11 May 1971) Unknown (in respect of rights of way, water support, drainage, light and other easements granted by a Conveyance dated 20 April 1988)
03-010	Acquisition of Rights	9561 square metres of agricultural land (west of Bewholme Lane)	Karl Jonathan Warkup Carr Hill Farm Bewholme DRIFFIELD East Riding of Yorkshire YO25 8EE	NONE	Carr Hill Farm Limited Office F12 Beverley Enterprise Centre Beck View Road BEVERLEY East Riding of Yorkshire HU17 OJT	Unknown (in respect of restrictive covenants as may have been imposed before 11 May 1971)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	nd Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1)	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Lanu Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-010 cont'd						Unknown (in respect of rights of way, water support, drainage, light and other easements granted by a Conveyance dated 20 April 1988)
03-011	Acquisition of Rights		Carr Hill Farm Limited Office F12 Beverley Enterprise Centre Beck View Road BEVERLEY East Riding of Yorkshire HU17 OJT	NONE	Carr Hill Farm Limited Office F12 Beverley Enterprise Centre Beck View Road BEVERLEY East Riding of Yorkshire HU17 OJT	NONE
03-012	Acquisition of Rights	agricultural land and drain (Low Bonwick Farm, south of Dunnington Lane)	Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG	NONE	YO25 8EG	Carr Hill Farm Limited Office F12 Beverley Enterprise Centre Beck View Road BEVERLEY East Riding of Yorkshire HU17 0JT (in respect of rights of access to maintain, inspect and repair service media contained in a Transfer dated 30 November 2017) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Michael Geoffrey Kirkwood)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009 Lessees or Tenants	ations: Prescribed Forms and Procedures) Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and
03-013	Acquisition of Rights	agricultural land (Manor Farm, south of Dunnington Lane)	or Reputed Owners Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG	or Reputed Lessees or Tenants	Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG	Procedures) Regulations 2009 Manor Farm Energy Limited Medina House 2 Station Avenue BRIDLINGTON East Riding of Yorkshire YO16 4LZ (in respect of rights of access and maintaining, inspecting and repairing conducting media contained in a Lease dated 5 September 2014)
04-001	Temporary Possession	splay and hedgerow (Manor	Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG	NONE	Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG	Manor Farm Energy Limited Medina House 2 Station Avenue BRIDLINGTON East Riding of Yorkshire YO16 4LZ (in respect of rights of access and maintaining, inspecting and repairing conducting media contained in a Lease dated 5 September 2014)
04-002	Temporary Possession	road and verges (Skipsea Lane)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE	County Hall Cross Street BEVERLEY East Riding of Yorkshire	Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-002 cont'd			Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG (in respect of subsoil beneath public highway) Unknown			Unknown Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
04-003	Temporary Possession	agricultural land (Manor Farm, west of Skipsea Lane)	Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG	NONE	YO25 8EG	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in Deeds dated 11 May 1972, 18 February 1981 and 26 March 1991)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1								
	Land w	hich is proposed to be subject to		right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))			
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners	ations: Prescribed Forms and Procedures) Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and				
			or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	or Reputed Occupiers	Procedures) Regulations 2009			
04-004	Acquisition of Rights	road and verges (Skipsea Lane)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown			
			Unknown			Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)			



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)		carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	· ·	t Riding of Yorkshire Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009 Lessees or Tenants or Reputed Lessees or Tenants	ations: Prescribed Forms and Procedures) Occupiers or Reputed Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-005	Temporary Possession	(east of Skipsea Lane)	Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG	NONE		Manor Farm Energy Limited Medina House 2 Station Avenue BRIDLINGTON East Riding of Yorkshire YO16 4LZ (in respect of rights of access and maintaining, inspecting and repairing media contained in a Lease dated 5 September 2014)
04-006	Temporary Possession	road and verges (Skipsea Lane)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG (in respect of subsoil beneath public highway)	NONE	BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown



	Land w	hich is proposed to be subject to	BOOK OF R	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	County of Eas	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans		Description of fund	Owners or Reputed Owners	Regulations 2009 Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-006 cont'd			Unknown			Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
04-007	Temporary Possession	agricultural land (Manor Farm, west of Skipsea Lane)	Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG	NONE	Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in Deeds dated 11 May 1972, 18 February 1981 and 26 March 1991)
04-008	Acquisition of Rights	agricultural land (Manor Farm, west of Skipsea Lane)	Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG	NONE	Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in Deeds dated 11 May 1972, 18 February 1981 and 26 March 1991)



	Land w	which is proposed to be subject t	BOOK OF R co: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order XEFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-008 cont'd						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)
04-009	Acquisition of Rights	490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)	James Anthony Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP (as reputed owner) Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG (as reputed owner) Sharon Julie Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP (as reputed owner) Unknown	NONE	James Anthony Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP (as reputed owner) Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG (as reputed owner) Sharon Julie Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP (as reputed owner) Unknown	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights) Unknown



	Land w	hich is proposed to be subject t	BOOK OF BOOK OF i	Vind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to ast Riding of Yorkshire	carry out protective works (Regulation 7(1	)(ə))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-010	Acquisition of Rights	42868 square metres of agricultural land and drain (east of Dunnington Lane)	James Anthony Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP Sharon Julie Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP	NONE	James Anthony Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP Sharon Julie Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of rights of access) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in a Deed of Grant dated 25 June 1971) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of underground gas apparatus)



	Land w	hich is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-011	Temporary Possession	2103 square metres of agricultural land (east of Dunnington Lane)	James Anthony Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP Sharon Julie Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP	NONE	DRIFFIELD East Riding of Yorkshire	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in a Deed of Grant dated 25 June 1971)
04-012	Temporary Possession	79 square metres of public road and verge (Dunnington Lane)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Unknown



	Land wh	ich is proposed to be subject	BOOK OF RE to: (i) powers of compulsory acquisition, (ii) I	nd Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to c : Riding of Yorkshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-012 cont'd			East Yorkshire Concrete Products Limited Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG (in respect of subsoil beneath public highway) James Anthony Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP (in respect of subsoil beneath public highway) Sharon Julie Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP (in respect of subsoil beneath public highway) Sharon Julie Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP (in respect of subsoil beneath public highway) Unknown			



	Land w	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)		o carry out protective works (Regulation 7(1)	(a))	
Number on Land Plans	Extent of acquisition or use	Description of land		County of East Riding of Yorkshire Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
04-013	Acquisition of Rights	858 square metres of public road and verge (Dunnington Lane)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) East Yorkshire Concrete Products Limited Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG (in respect of subsoil beneath public highway) James Anthony Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP (in respect of subsoil beneath public highway) Sharon Julie Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP (in respect of subsoil beneath public highway) Sharon Julie Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Unknown	



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	nd Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-013 cont'd			Unknown			
04-014	Acquisition of Rights	125 square metres of verge (west of Dunnington Lane)	East Yorkshire Concrete Products Limited Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG (as reputed owner) Unknown	NONE	East Yorkshire Concrete Products Limited Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG (as reputed owner) Unknown	Unknown
04-015	Temporary Possession	2432 square metres of public road and verge (Dunnington Lane)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) East Yorkshire Concrete Products Limited Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG (in respect of subsoil beneath public highway)	NONE	Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Unknown



	Land wh	nich is proposed to be subject t	BOOK OF RE to: (i) powers of compulsory acquisition, (ii) i	nd Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o t Riding of Yorkshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-015 cont'd			James Anthony Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP (in respect of subsoil beneath public highway) Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG (in respect of subsoil beneath public highway) Sharon Julie Dean Grove Farm Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EP (in respect of subsoil beneath public highway) Unknown			



	Land w	hich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-016	Temporary Possession	305 square metres of verge (west of Dunnington Lane)	East Yorkshire Concrete Products Limited Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG (as reputed owner) Unknown	NONE	East Yorkshire Concrete Products Limited Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG (as reputed owner) Unknown	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Unknown
04-017	Temporary Possession	3499 square metres of agricultural land (west of Dunnington Lane)	East Yorkshire Concrete Products Limited Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG	NONE	East Yorkshire Concrete Products Limited Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)
04-018	Acquisition of Rights	34002 square metres of agricultural land, electicity poles, electricity cables and verge (west of Dunnington Lane)	East Yorkshire Concrete Products Limited Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG	NONE	East Yorkshire Concrete Products Limited Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG	NONE



	Land w	hich is proposed to be subject to	BOOK OF R	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to	carry out protective works (Regulation 7(1	)(a))
			County of Eas	st Riding of Yorkshire		
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-019	Temporary Possession	agricultural land and access track (south of Dunnington Lane)	East Yorkshire Concrete Products Limited Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG	NONE	East Yorkshire Concrete Products Limited Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG	NONE
04-020	Temporary Possession	road and verges (Dunnington Lane and Beverley Road, A165)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) East Yorkshire Concrete Products Limited Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Unknown



	Land wh	ich is proposed to be subject	BOOK OF RE to: (i) powers of compulsory acquisition, (ii) I	nd Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to c t Riding of Yorkshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 )(a) of the Infrastructure Planning (Applicat Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-020 cont'd			Investacc Pension Trustees Limited Solway House Business Park Kingstown Carlisle CA6 4BY (in respect of subsoil beneath public highway) Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG (in respect of subsoil beneath public highway) Unknown Violet Elizabeth Jefferson Southfield Farm Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD (in respect of subsoil beneath public highway)			Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)



	Land w	hich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-021	Acquisition of Rights	21402 square metres of agricultural land (west of Dunnington Lane)	Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG	NONE	Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Michael Geoffrey Kirkwood) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)
04-022	Acquisition of Rights	15941 square metres of agricultural land (west of Dunnington Lane)	Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG	NONE	Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG	NONE
04-023	Acquisition of Rights	18898 square metres of agricultural land, access track and drain (Pinderhill Farm, south of Dunnington Lane)	Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG	NONE	Michael Geoffrey Kirkwood Manor Farm Dunnington DRIFFIELD East Riding of Yorkshire YO25 8EG	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Michael Geoffrey Kirkwood)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii	/ind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
	Acquisition of Rights	Dunnington Lane)	East Yorkshire Concrete Products Limited Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG	NONE	East Yorkshire Concrete Products Limited Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG	Gregory Peter Atkin Moor Grange Beverley Road Beeford Driffield East Riding of Yorkshire YO25 8AE (in respect of rights of drainage and access to the retained land contained in a Transfer dated 14 September 2010) Lucy Sheila Elizabeth Atkin Moor Grange Cottage Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AE (in respect of rights of drainage and access to the retained land contained in a Transfer dated 14 September 2010)
05-001	Acquisition of Rights	agricultural land and hedgerow (Moor Grange Farm, east of Moor Grange Access Road,	Eastview Properties Limited 2nd Floor St Mary's Court 20 Hill Street Douglas Isle of Man IM1 1EU	NONE	Robin Rivis Beachcomber Lodge 56 South Marine Drive BRIDLINGTON YO15 3JN	Alexander Douglas Robinson Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG (in respect of rights to maintain a ditch contained in a Conveyance dated 1 August 1984)



	Land whi	ich is proposed to be subject t	BOOK OF co: (i) powers of compulsory acquisition, (i	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to can ast Riding of Yorkshire	rry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
05-001 cont'd						Gregory Peter Atkin Moor Grange Beverley Road Beeford Driffield East Riding of Yorkshire YO25 8AE (in respect of rights of way contained in a Transfer dated 10 May 1988 and in respect of rights of drainage and access to the retained land contained in a Transfer dated 14 September 2010) Lucy Sheila Elizabeth Atkin Moor Grange Cottage Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AE (in respect of rights of way contained in a Transfer dated 10 May 1988 and in respect of rights of drainage and access to the retained land contained in a	



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and		
			or Reputed Owners	or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Procedures) Regulations 2009
05-001 cont'd						R & CM Rivis Limited Beachcomber Lodge 56 South Marine Drive BRIDLINGTON YO15 3JN (in respect of rights of way and paying a proportion of costs to repair, maintain, clean, dredging and renewing drains contained in a Transfer dated 14 September 2010)
05-002	Temporary Possession	Moor Grange Access Road, west of Billings Lane)			BRIDLINGTON YO15 3JN	Alexander Douglas Robinson Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG (in respect of rights to maintain a ditch contained in a Conveyance dated 1 August 1984)



	Land whi	ich is proposed to be subject t	BOOK OF BOOK OF I	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to cau ast Riding of Yorkshire	rry out protective works (Regulation 7(1	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-002 cont'd						Gregory Peter Atkin Moor Grange Beverley Road Beeford Driffield East Riding of Yorkshire YO25 8AE (in respect of rights of way contained in a Transfer dated 10 May 1988 and in respect of rights of drainage and access to the retained land contained in a Transfer dated 14 September 2010) Lucy Sheila Elizabeth Atkin Moor Grange Cottage Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AE (in respect of rights of way contained in a Transfer dated 10 May 1988 and in respect of rights of drainage and access to the retained land contained in a Transfer dated 10 May 1988 and in



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-002 cont'd 05-003	Acquisition of Rights	6575 square metres of agricultural land and drain	Copeland R G & Sons Billings Hill Farm	NONE		R & CM Rivis Limited Beachcomber Lodge 56 South Marine Drive BRIDLINGTON YO15 3JN (in respect of rights of way and paying a proportion of costs to repair, maintain, clean, dredging and renewing drains contained in a Transfer dated 14 September 2010) Unknown
05-004	Acquisition of Rights	(west of Billings Lane)	Dunnington Driffield YO25 8EQ (as reputed owner) Unknown Christopher Andrew Arnott	NONE	Dunnington Driffield YO25 8EQ (as reputed owner) Unknown Christopher Andrew Arnott	Clydesdale Bank PLC
05-004	Acquisition of Rights	agricultural land and hedgerow (east of New Road) (excluding all interests of the Crown)	Manor House	INOINE	Manor House Nunkeeling DRIFFIELD	Clydesdale Bank PLC 177 Bothwell Street Glasgow G2 7ER (as mortgagee for Christopher Andrew Arnott and David John Arnott)



	Land wh	ich is proposed to be subject	BOOK OF to: (i) powers of compulsory acquisition, (i	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights t ast Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Appl Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-004 cont'd			David John Arnott Manor House Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH		David John Arnott Manor House Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH	Julia Alexa Clover Todd Nunkeeling Priory Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH (in respect of rights to maintain the dwellinghouse, drains and septic tank stated in the Conveyance dated 14 January 1966) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of wayleave consents dated 28 September 1967, 13 May 1969 and 28 November 1983)



	Land w	hich is proposed to be subject to	BOOK OF RE c (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-004 cont'd						Nunkeeling Renewables Limited Hall Garth Manor Farm Nunkeeling Driffield East Yorkshire YO25 8EH (in respect of rights contained in a Lease dated 1 May 2012 and 3 December 2012) Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of restrictive covenants and rights listed in a Deed dated 30 May 1949) Unknown (in respect of rights granted and terms listed in a Conveyance dated 18 March 1987)
05-005	Acquisition of Rights	29550 square metres of agricultural land, access track and hedgerow (Moor House Farm, Moor House Access Road)	Moor House Farming Limited Moor House Farm New Road Brandesburton Driffield North Humberside YO25 8EN	NONE	Moor House Farming Limited Moor House Farm New Road Brandesburton Driffield North Humberside YO25 8EN	Unknown (in respect of rights of way)



	Land w	hich is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Lanu Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-005 cont'd			Unknown (in respect of mines and minerals)			Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
06-001	Acquisition of Rights	,	Laurazena Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB	NONE	DRIFFIELD East Riding of Yorkshire	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown (in respect of rights reserved by a Conveyance dated 28 October 1965)
06-002	Acquisition of Rights	agricultural land (west of Main Road)	Moor House Farming Limited Moor House Farm New Road Brandesburton Driffield North Humberside YO25 8EN	NONE	Moor House Farming Limited Moor House Farm New Road Brandesburton Driffield North Humberside YO25 8EN	NONE



	Land w	hich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7( Owners	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and	
			or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Procedures) Regulations 2009
06-003	Temporary Possession	519 square metres of public road and verge (Main Road)	Charles Thompson Taylor 62 Barley Gate Leven BEVERLEY HU17 5NU (in respect of subsoil beneath public highway) East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Harry Thompson Taylor 2 Altoft Close Brandesburton DRIFFIELD YO25 8QH (in respect of subsoil beneath public highway) Moor House Farming Limited Moor House Farm New Road Brandesburton Driffield North Humberside YO25 8EN (in respect of subsoil beneath public highway)	NONE		Unknown Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of abandoned underground water apparatus)



	Land w	hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-003 cont'd			Unknown			
			Wendy Taylor Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of subsoil beneath public highway)			
06-004	Acquisition of Rights		Charles Thompson Taylor 62 Barley Gate Leven BEVERLEY HU17 5NU (in respect of subsoil beneath public highway) East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE		Unknown Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)



	Land w	hich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Farm Development Consent Order IEFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1)	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-004 cont'd			Harry Thompson Taylor 2 Altoft Close Brandesburton DRIFFIELD YO25 8QH (in respect of subsoil beneath public highway) Moor House Farming Limited Moor House Farm New Road Brandesburton Driffield North Humberside YO25 8EN (in respect of subsoil beneath public highway) Unknown Wendy Taylor Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of subsoil beneath public highway)			
06-005	Temporary Possession	30 square metres of verge (east of Main Road)	Charles Thompson Taylor 62 Barley Gate Leven BEVERLEY HU17 5NU (as reputed owner)	NONE	Charles Thompson Taylor 62 Barley Gate Leven BEVERLEY HU17 5NU (as reputed owner)	Unknown



	Land wh	nich is proposed to be subject	BOOK OF RE to: (i) powers of compulsory acquisition, (ii)	nd Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-005 cont'd			Harry Thompson Taylor 2 Altoft Close Brandesburton DRIFFIELD YO25 8QH (as reputed owner)		Brandesburton DRIFFIELD YO25 8QH (as reputed owner)	Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
			Laurazena Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (as reputed owner) Unknown		Laurazena Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (as reputed owner) Unknown	
			Wendy Taylor Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (as reputed owner)		Wendy Taylor Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (as reputed owner)	



	Land w	hich is proposed to be subject	BOOK ( to: (i) powers of compulsory acquisition	re Wind Farm Development Consent Order DF REFERENCE - PART 1 n, (ii) right to use the land, and/or (iii) rights t f East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulatio	Category 1 n 7(1)(a) of the Infrastructure Planning (App Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-006	Temporary Possession	136 square metres of agricultural land and verge (east of Main Road)	Charles Thompson Taylor 62 Barley Gate Leven BEVERLEY HU17 5NU Harry Thompson Taylor 2 Altoft Close Brandesburton DRIFFIELD YO25 8QH Wendy Taylor Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB	NONE	Wendy Taylor Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB	Foster Farms Limited Willow Croft Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 SPR (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1 February 1980 and 17 July 1973) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Wendy Taylor, Charles Thompson Taylor and Harry Thompson Taylor) Laurazena Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of easements as contained in a Transfer dated 4 May 2023)



	Land w	hich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-006 cont'd						Roy Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of easements as contained in a Transfer dated 4 May 2023)
06-007	Acquisition of Rights	55074 square metres of agricultural land (north of Catfoss Road, east of Main Road)	Charles Thompson Taylor 62 Barley Gate Leven BEVERLEY HU17 5NU Harry Thompson Taylor 2 Altoft Close Brandesburton DRIFFIELD YO25 8QH	NONE		Foster Farms Limited Willow Croft Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1 February 1980 and 17 July 1973) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Wendy Taylor, Charles Thompson Taylor and Harry Thompson Taylor)



	Land v	which is proposed to be subject t	BOOK O o: (i) powers of compulsory acquisition,	e Wind Farm Development Consent Order F REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights to East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1 7(1)(a) of the Infrastructure Planning (Appli Regulations 2009		7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-007 cont'd			Wendy Taylor Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB			Laurazena Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of easements as contained in a Transfer dated 4 May 2023) Roy Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of easements as contained in a Transfer dated 4 May 2023)
06-008	Temporary Possession	2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)	Charles Thompson Taylor 62 Barley Gate Leven BEVERLEY HU17 5NU	NONE	Wendy Taylor Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB	Foster Farms Limited Willow Croft Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1 February 1980 and 17 July 1973)



	Land wi	nich is proposed to be subject t		/ind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to a	carry out protective works (Regulation 7(1	)(a))			
	County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
06-008 cont'd			Harry Thompson Taylor 2 Altoft Close Brandesburton DRIFFIELD YO25 8QH Wendy Taylor Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB			HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Wendy Taylor, Charles Thompson Taylor and Harry Thompson Taylor) Laurazena Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of easements as contained in a Transfer dated 4 May 2023) Roy Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of easements as contained in a Transfer dated 4 May 2023)			



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order IEFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
06-009	Temporary Possession	16 square metres of hedgerow and drain (north of Catfoss Road)	Christopher Andrew Arnott Manor House Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH David John Arnott Manor House Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH	NONE	Christopher Andrew Arnott Manor House Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH David John Arnott Manor House Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deeds dated 27 June 1972 and 10 April 1975) Clydesdale Bank PLC 177 Bothwell Street Glasgow G2 7ER (as mortgagee for David John Arnott and Christopher Andrew Arnott) Nunkeeling Renewables Limited Hall Garth Manor Farm Nunkeeling Driffield East Yorkshire YO25 8EH (in respect of the rights reserved by the Transfer dated 9 October 2013)	
06-010	Temporary Possession	1004 square metres of public road and verge (Catfoss Road)	Charles Thompson Taylor 62 Barley Gate Leven BEVERLEY HU17 5NU (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	



	Land wh	ich is proposed to be subject t		nd Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to c	arry out protective works (Regulation 7(1	)(a))
				Riding of Yorkshire Category 1		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	(a) of the Infrastructure Planning (Applicat Regulations 2009	ions: Prescribed Forms and Procedures)	Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-010 cont'd			Christopher Andrew Arnott Manor House Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH (in respect of subsoil beneath public highway) David John Arnott Manor House Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH (in respect of subsoil beneath public highway) East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Harry Thompson Taylor 2 Altoft Close Brandesburton DRIFFIELD YO25 8QH (in respect of subsoil beneath public highway)			Unknown Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)



	Land w	hich is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-010 cont'd			Moor House Farming Limited Moor House Farm New Road Brandesburton Driffield North Humberside YO25 8EN (in respect of subsoil beneath public highway) Unknown Wendy Taylor Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of subsoil beneath public highway)			
06-011	Temporary Possession	4 square metres of verge (south of Catfoss Road)	Moor House Farming Limited Moor House Farm New Road Brandesburton Driffield North Humberside YO25 8EN (as reputed owner) Unknown	NONE	Moor House Farming Limited Moor House Farm New Road Brandesburton Driffield North Humberside YO25 8EN (as reputed owner) Unknown	Unknown



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii		o carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	County of East Riding of Yorkshire Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
06-012	Acquisition of Rights		Charles Thompson Taylor 62 Barley Gate Leven BEVERLEY HU17 5NU (in respect of subsoil beneath public highway) East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Harry Thompson Taylor 2 Altoft Close Brandesburton DRIFFIELD YO25 8QH (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
			Moor House Farming Limited Moor House Farm New Road Brandesburton Driffield North Humberside YO25 8EN (in respect of subsoil beneath public highway)				



	Land w	hich is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-012 cont'd			Unknown			
			Wendy Taylor Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of subsoil beneath public highway)			
06-013	Temporary Possession	43 square metres of verge (north of Catfoss Road)	Charles Thompson Taylor 62 Barley Gate Leven BEVERLEY HU17 SNU	NONE	Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB	Foster Farms Limited Willow Croft Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1 February 1980 and 17 July 1973)
			Harry Thompson Taylor 2 Altoft Close Brandesburton DRIFFIELD YO25 8QH			HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Wendy Taylor, Charles Thompson Taylor and Harry Thompson Taylor)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order IEFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-013 cont'd			Wendy Taylor Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB			Laurazena Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of easements as contained in a Transfer dated 4 May 2023) Roy Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of easements as contained in a Transfer dated 4 May 2023)
06-014	Temporary Possession	road and verge (Catfoss Road	ABI (UK) Limited Swinemoor Lane BEVERLEY East Riding of Yorkshire HU17 OLJ (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)



	Land wh	ich is proposed to be subject t	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) r	nd Farm Development Consent Order FERENCE - PART 1 ight to use the land, and/or (iii) rights to ca Riding of Yorkshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-014 cont'd			Andrew Berzins Woodnook Fold Nunkeeling DRIFFIELD YO25 8EH (in respect of subsoil beneath public highway) Charles Thompson Taylor 62 Barley Gate Leven BEVERLEY HU17 5NU (in respect of subsoil beneath public highway) Christopher Scott White The Old Barn Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH (in respect of subsoil beneath public highway) David White 7 Maple Walk Brandesburton DRIFFIELD YO25 8SH (in respect of subsoil beneath public highway)			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)



	Land wh	nich is proposed to be subject	BOOK OF RE to: (i) powers of compulsory acquisition, (ii) r	nd Farm Development Consent Order FERENCE - PART 1 ight to use the land, and/or (iii) rights to ca Riding of Yorkshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Applicat Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-014 cont'd			East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Gemma Smithson Pasture Field House Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH (in respect of subsoil beneath public highway) Harry Thompson Taylor 2 Altoft Close Brandesburton DRIFFIELD YO25 8QH (in respect of subsoil beneath public highway) Henry Gordon Waterhouse Pasture Field House Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH (in respect of subsoil beneath public highway) Henry Gordon Waterhouse Pasture Field House Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH (in respect of subsoil beneath public highway)			



	Land wh	ich is proposed to be subject	BOOK OF RE to: (i) powers of compulsory acquisition, (ii) r	nd Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to c Riding of Yorkshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land		Category 1 (a) of the Infrastructure Planning (Applicat Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-014 cont'd			Joanne White The Old Barn Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH (in respect of subsoil beneath public highway) Moor House Farming Limited Moor House Farm New Road Brandesburton Driffield North Humberside YO25 8EN (in respect of subsoil beneath public highway) Suzanne Berzins Woodnook Fold Nunkeeling DRIFFIELD YO25 8EH (in respect of subsoil beneath public highway) Unknown			



	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	/ind Farm Development Consent Order IEFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-014 cont'd			Wendy Taylor Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of subsoil beneath public highway)			
06-015	Temporary Possession	6834 square metres of agricultural land and drain (south of Catfoss Road)	Moor House Farming Limited Moor House Farm New Road Brandesburton Driffield North Humberside YO25 8EN	NONE	Moor House Farming Limited Moor House Farm New Road Brandesburton Driffield North Humberside YO25 8EN	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Moor House Farming Limited)
06-016	Temporary Possession	8429 square metres of agricultural land and access track (south of Catfoss Road)	Moor House Farming Limited Moor House Farm New Road Brandesburton Driffield North Humberside YO25 8EN	NONE	Moor House Farming Limited Moor House Farm New Road Brandesburton Driffield North Humberside YO25 8EN	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect of the right to lay, maintain, inspect and repair and in respect to easement stated in the Deed dated 27 June 1972) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Moor House Farming Limited)



	Land w	which is proposed to be subject to	BOOK O o: (i) powers of compulsory acquisition,	Wind Farm Development Consent Order FREFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights t East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1 7(1)(a) of the Infrastructure Planning (Appl Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-017	Acquisition of Rights	31268 square metres of agricultural land and drain (south of Catfoss Road)	Moor House Farming Limited Moor House Farm New Road Brandesburton Driffield North Humberside YO25 8EN	NONE	Moor House Farming Limited Moor House Farm New Road Brandesburton Driffield North Humberside YO25 8EN	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the right to lay, maintain, inspect and repair and in respect to easement stated in the Deed dated 27 June 1972) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Moor House Farming Limited)
06-018	Acquisition of Rights	30325 square metres of agricultural land and drain (north of Harsell Lane, excluding all interests of the Crown)	Mark Wilson Mewburn Dacre Lodge Catwick Lane Brandesburton DRIFFIELD YO25 8SB	NONE	Mark Wilson Mewburn Dacre Lodge Catwick Lane Brandesburton DRIFFIELD YO25 8SB	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights contained in Deeds dated 19 April 1972 and 6 February 1973) Handelsbanken PLC 3 Thomas More Square LONDON E1W 1WY (as mortgagee for Mark Wilson Mewburn)



	Land whi	ich is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to ca st Riding of Yorkshire	arry out protective works (Regulation 7(1	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Applicati Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms ar Procedures) Regulations 2009
06-018 cont'd						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Peter James Boynton Hind House Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HU11 5QN (as The Executor of the Estate of the Late Patricia Anne Boynton) (in respect of rights reserved by a Conveyance dated 24 April 1981) Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of covenants and grants contained in a Deed dated 11 Noven 1948) Unknown (in respect to rights reserved by a Conveyance dated 1 October 1965)



			BOOK OF	Wind Farm Development Consent Order REFERENCE - PART 1		
	Land w	hich is proposed to be subject to		ii) right to use the land, and/or (iii) rights to East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-019	Temporary Possession	136 square metres of verge (north of Harsell Lane, excluding all interests of the Crown)	Mark Wilson Mewburn Dacre Lodge Catwick Lane Brandesburton DRIFFIELD YO25 8SB	NONE	Mark Wilson Mewburn Dacre Lodge Catwick Lane Brandesburton DRIFFIELD YO25 8SB	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights contained in Deeds dated 19 April 1972 and 6 February 1973) Colin Stuart Hazell Astral House Catfoss HULL East Riding of Yorkshire HU11 5SP (in respect of rights of access as contained in a Transfer dated 16 December 2010) Handelsbanken PLC 3 Thomas More Square LONDON E1W 1WY (as mortgagee for Mark Wilson Mewburn)



	Land wh	nich is proposed to be subject t	BOOK OF to: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ca ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))	
Number on	Extent of acquisition or use	t of acquisition or use Description of land		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
06-019 cont'd						Peter James Boynton Hind House Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HU11 5QN (as The Executor of the Estate of the Late Patricia Anne Boynton) (in respect of rights reserved by a Conveyance dated 24 April 1981) Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of covenants and grants contained in a Deed dated 11 November 1948) Unknown (in respect to rights reserved by a Conveyance dated 1 October 1965)	



			BOOK OF R	Vind Farm Development Consent Order REFERENCE - PART 1		
	Land w	hich is proposed to be subject to		) right to use the land, and/or (iii) rights to st Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-020	Temporary Possession	374 square metres of public road and verge (Harsell Lane, excluding all interests of the Crown)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Mark Wilson Mewburn Dacre Lodge Catwick Lane Brandesburton DRIFFIELD YO25 8SB	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights contained in Deeds dated 19 April 1972 and 6 February 1973) Colin Stuart Hazell Astral House Catfoss HULL East Riding of Yorkshire HU11 5SP (in respect of rights listed in a Transfer dated 16 December 2010) Handelsbanken PLC 3 Thomas More Square LONDON E1W 1WY (as mortgagee for Mark Wilson Mewburn)



	Land wł	nich is proposed to be subject t	BOOK OF to: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ca ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))	
Number on	Extent of acquisition or use	tent of acquisition or use Description of land		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
06-020 cont'd						Peter James Boynton Hind House Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HU11 5QN (as The Executor of the Estate of the Late Patricia Anne Boynton) (in respect of rights reserved by a Conveyance dated 24 April 1981) Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of covenants and grants contained in a Deed dated 11 November 1948) Unknown (in respect to rights reserved by a Conveyance dated 1 October 1965)	



	Land w	hich is proposed to be subject to	BOOK OF R	/ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to	carry out protective works (Regulation 7(1	)(a))	
			County of Ea	st Riding of Yorkshire			
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
06-021	Acquisition of Rights	road (Harsell Lane, exlucing all interests of the Crown)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Mark Wilson Mewburn Dacre Lodge Catwick Lane Brandesburton DRIFFIELD YO25 8SB	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights contained in Deeds dated 19 April 1972 and 6 February 1973) Colin Stuart Hazell Astral House Catfoss HULL East Riding of Yorkshire HU11 5SP (in respect of rights listed in a Transfer dated 16 December 2010) Handelsbanken PLC 3 Thomas More Square LONDON E1W 1WY (as mortgagee for Mark Wilson Mewburn)	



	Land wh	nich is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ca ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))	
Number on	Extent of acquisition or use	ent of acquisition or use Description of land		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
06-021 cont'd						Peter James Boynton Hind House Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HU11 5QN (as The Executor of the Estate of the Late Patricia Anne Boynton) (in respect of rights reserved by a Conveyance dated 24 April 1981) Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of covenants and grants contained in a Deed dated 11 November 1948) Unknown (in respect to rights reserved by a Conveyance dated 1 October 1965)	



	Landu	which is proposed to be subject to	BOOK OF	Wind Farm Development Consent Order REFERENCE - PART 1	o carry out protective works (Regulation 7(1	
		mentis proposed to be subject to		ast Riding of Yorkshire		)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-022	Temporary Possession	116 square metres of verge (north of Harsell Lane, excluding all interests of the Crown)	Mark Wilson Mewburn Dacre Lodge Catwick Lane Brandesburton DRIFFIELD YO25 8SB	NONE	Mark Wilson Mewburn Dacre Lodge Catwick Lane Brandesburton DRIFFIELD YO25 8SB	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights contained in Deeds dated 19 April 1972 and 6 February 1973) Colin Stuart Hazell Astral House Catfoss HULL East Riding of Yorkshire HU11 5SP (in respect of rights of access as contained in a Transfer dated 16 December 2010) Handelsbanken PLC 3 Thomas More Square LONDON E1W 1WY (as mortgagee for Mark Wilson Mewburn)



	Land w	nich is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to c ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-022 cont'd						Peter James Boynton Hind House Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HU11 5QN (as The Executor of the Estate of the Late Patricia Anne Boynton) (in respect of rights reserved by a Conveyance dated 24 April 1981) Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of covenants and grants contained in a Deed dated 11 November 1948) Unknown (in respect to rights reserved by a Conveyance dated 1 October 1965)



	Land w	hich is proposed to be subject to	BOOK C : (i) powers of compulsory acquisition	e Wind Farm Development Consent Order )F REFERENCE - PART 1 , (ii) right to use the land, and/or (iii) rights t <sup>E</sup> East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulatior	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-023	Temporary Possession	road (Harsell Lane, exlucing all interests of the Crown)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Mark Wilson Mewburn Dacre Lodge Catwick Lane Brandesburton DRIFFIELD YO25 8SB	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Category 2 S) Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and



	Land wh	ich is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (i	Wind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to c ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-023 cont'd						Peter James Boynton Hind House Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HU11 5QN (as The Executor of the Estate of the Late Patricia Anne Boynton) (in respect of rights of way contained in a Conveyance dated 24 April 1981) Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights to maintain hedges and trees contained in a Deed dated 11 November 1948) Unknown (in respect to rights of way contained in a Conveyance dated 1 October 1965)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
06-024	Temporary Possession		Mark Wilson Mewburn Dacre Lodge Catwick Lane Brandesburton DRIFFIELD YO25 8SB	NONE	Mark Wilson Mewburn Dacre Lodge Catwick Lane Brandesburton DRIFFIELD YO25 8SB	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, maintain and renew a pipe contained in a Deed dated 19 April 1972 and in respect of rights to lay, maintain and renew a pipe contained in a Deed dated 6 February 1973) Colin Stuart Hazell Astral House Catfoss HULL East Riding of Yorkshire HU11 5SP (in respect of rights to maintain, repair and renew a fence contained in a Transfer dated 16 December 2010) Handelsbanken PLC 3 Thomas More Square LONDON E1W 1WY (as mortgagee for Mark Wilson Mewburn)			



	Land wh	ich is proposed to be subject t	BOOK OF to: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ca ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-024 cont'd						Peter James Boynton Hind House Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HU11 5QN (as The Executor of the Estate of the Late Patricia Anne Boynton) (in respect of rights of way contained in a Conveyance dated 24 April 1981) Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights to maintain hedges and trees contained in a Deed dated 11 November 1948) Unknown (in respect to rights of way contained in a Conveyance dated 1 October 1965)



	Land w	hich is proposed to be subject to	BOOK O c: (i) powers of compulsory acquisition,	e Wind Farm Development Consent Order F REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights t East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Fians			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
06-025	Acquisition of Rights	64907 square metres of agricultural land, drain and hedgerow (south of Harsell Lane, east of Catwick Road, excluding all interests of the Crown)	Mark Wilson Newburn Dacre Lodge Catwick Lane Brandesburton DRIFFIELD YO25 8SB	NONE	Mark Wilson Mewburn Dacre Lodge Catwick Lane Brandesburton DRIFFIELD YO25 8SB	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, maintain and renew a pipe contained in a Deed dated 19 April 1972 and in respect of rights to lay, maintain and renew a pipe contained in a Deed dated 6 February 1973) Colin Stuart Hazell Astral House Catfoss HULL East Riding of Yorkshire HU11 5SP (in respect of rights to maintain, repair and renew a fence contained in a Transfer dated 16 December 2010) Handelsbanken PLC 3 Thomas More Square LONDON E1W 1WY (as mortgagee for Mark Wilson Mewburn)	



	Land wh	ich is proposed to be subject t	BOOK OF to: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ca ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	nt of acquisition or use Description of land	Qualifying persons under Regulation 7	Category 1 7(1)(a) of the Infrastructure Planning (Applicat Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-025 cont'd						Peter James Boynton Hind House Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HU11 5QN (as The Executor of the Estate of the Late Patricia Anne Boynton) (in respect of rights of way contained in a Conveyance dated 24 April 1981) Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights to maintain hedges and trees contained in a Deed dated 11 November 1948) Unknown (in respect to rights of way contained in a Conveyance dated 1 October 1965)



	Land w	hich is proposed to be subject to	BOOK OF RE c (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-001	Acquisition of Rights	20924 square metres of agricultural land (east of Catwick Road)	The Executor of the Estate of the Late Walter Trevor Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR (as The Executor of the Estate of the Late Walter Trevor Kirkwood)	NONE	(as The Executor of the Estate of the Late Walter Trevor Kirkwood)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect to the rights to lay and maintain pipes and in respect to a right to easements as contained in a Deed dated 19 April 1972)
07-002	Acquisition of Rights	6926 square metres of agricultural land (east of Catwick Road)	Andrew Stanley Tomlinson Moor House Farm Brandesburton Grange Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EL	NONE	Andrew Stanley Tomlinson Moor House Farm Brandesburton Grange Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EL	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect to the right to easement and in respect to the right to lay, maintain, inspect and repair a gas pipeline contained in a Deed dated 19 April 1972)



	Land w	hich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 I right to use the land, and/or (iii) rights to st Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-002 cont'd			James Leonard Tomlinson Woodlands Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH		James Leonard Tomlinson Woodlands Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH	
07-003	Acquisition of Rights	4081 square metres of agricultural land, copse and drain (east of Catwick Road)	Andrew Stanley Tomlinson Moor House Farm Brandesburton Grange Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EL	NONE	Andrew Stanley Tomlinson Moor House Farm Brandesburton Grange Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EL	David Hobson Manor Lodge Catfoss Lane Sigglesthorne HULL HU11 5QN (in respect of rights of wayleave easements, rights of way and drainage contained in a transfer dated 1 August 1990)
			James Leonard Tomlinson Woodlands Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH		James Leonard Tomlinson Woodlands Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH	
07-004	Acquisition of Rights	21226 square metres of agricultural land, drain and public footpath (Seaton Footpath No.10) (east of Catwick Road)	Andrew Stanley Tomlinson Moor House Farm Brandesburton Grange Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EL	NONE	Andrew Stanley Tomlinson Moor House Farm Brandesburton Grange Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8EL	Andrew Stewart Buchan 346 Hathersage Road HULL East Riding of Yorkshire HU8 0EY (in respect of rights of light or air contained in a Conveyance dated 6 November 1990)



	Land w	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-004 cont'd			James Leonard Tomlinson Woodlands Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH		East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public footpath (Seaton Footpath No.10)) James Leonard Tomlinson Woodlands Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH	Nicola Jayne Smith 346 Hathersage Road HULL East Riding of Yorkshire HU8 0EY (in respect of rights of light and air contained in a Conveyance dated 6 November 1990)
07-005	Acquisition of Rights	copse and watercourse (Catfoss Drain) (Catfoss Hall	Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire HU11 5BL (as trustee to the R A Bethell Trust)	NONE	Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire HU11 5BL (as trustee to the R A Bethell Trust)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)



	Land w	hich is proposed to be subject to	BOOK OF RI : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009 Lessees or Tenants	ations: Prescribed Forms and Procedures) Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 Timothy Duncan Rose Catfoss Hall Catfoss Hall Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HU11 5QN (in respect of rights of way contained in a Conveyances dated 26 January 1999 and 29 October 1970)
07-005 cont'd			or Reputed Owners	or Reputed Lessees or Tenants	or Reputed Occupiers	
08-001	Temporary Possession	splay (east of Catwick Road)	Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire HU11 5BL (as trustee to the R A Bethell Trust)	NONE	Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire HU11 5BL (as trustee to the R A Bethell Trust)	Timothy Duncan Rose Catfoss Hall Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HU11 5QN (in respect of rights of way contained in a Conveyances dated 26 January 1999 and 29 October 1970)
08-002	Temporary Possession	road, access splays, drains and verges (Catwick Road)	Allan Charles Walker Westlands Hornsea Road Sigglesthorne HULL East Riding of Yorkshire HU11 5QL (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)



	Land wh	nich is proposed to be subject	BOOK OF RE to: (i) powers of compulsory acquisition, (ii)	nd Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to c t Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Applicat Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-002 cont'd			Dorothy Elizabeth Walker Westlands Hornsea Road Sigglesthorne HULL East Riding of Yorkshire HU11 5QL (in respect of subsoil beneath public highway) East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire HULL East Riding of Yorkshire HULL East Riding of Yorkshire HULL East Riding of Yorkshire HULL East Riding of Yorkshire HU11 5BL (in respect of subsoil beneath public highway)			Unknown Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)



	Land w	hich is proposed to be subject to	BOOK OF RE c (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1)	)(a))
Number on	Extent of acquisition or use	Description of land		Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-002 cont'd			Robert Charles Orlando Hellyer Paxmere House Peasemore NEWBURY Berkshire RG20 7JH (in respect of subsoil beneath public highway) Unknown William Anthony Bethell Arnold Manor Black Tup Lane Arnold HULL East Riding of Yorkshire HU11 5JA (in respect of subsoil beneath public highway)			
08-003	Temporary Possession	230 square metres of bridge, verge and copse (Washdike Bridge, east of Catwick Road)	Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire HU11 5BL (as reputed owner) Unknown	NONE	Rise HULL East Riding of Yorkshire HU11 5BL (as reputed owner)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
08-004	Temporary Possession	515 square metres of public road and verges (West Road, A1035)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Unknown Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground sewerage and water apparatus)	
08-005	Acquisition of Rights	1912 square metres of public road and verges (West Road, A1035)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus)	



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-005 cont'd			Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire HU11 5BL (in respect of subsoil beneath public highway) Unknown			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground sewerage and water apparatus)
08-006	Temporary Possession	966 square metres of public road and verges (West Road, A1035)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-006 cont'd			Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire HU11 5BL (in respect of subsoil beneath public highway) Unknown			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground sewerage and water apparatus)
08-007	Temporary Possession	agricultural land (south of West Road, A1035, east of Catwick Heads)	Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire HU11 5BL	NONE	Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire HU11 5BL	Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-007 cont'd						Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
08-008	Acquisition of Rights	(south of West Road, A1035, east of Catwick Heads)	Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire HU11 5BL	NONE		Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus)
08-009	Acquisition of Rights	14359 square metres of agricultural land, access splay, pond and drain (south of West Road, A1035 and east of Catwick Heads)	Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire HU11 5BL	F D Bird & Sons Limited The Old Hall Lakeview Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR	Riston Road	Unknown (in respect of rights of easements and covenants contained in a Conveyance dated 19 January 1987)



	Land w	hich is proposed to be subject to	BOOK OF RE DOCK OF RE	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-009 cont'd			Robert Charles Orlando Hellyer Paxmere House Peasemore NEWBURY Berkshire RG20 7JH William Anthony Bethell Arnold Manor Black Tup Lane Arnold HULL East Riding of Yorkshire HU11 5JA			
08-010	Temporary Possession	218 square metres of verge (south of West Road, A1035, east of Catwick Heads)	Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire HU11 5BL Robert Charles Orlando Hellyer Paxmere House Peasemore NEWBURY Berkshire RG20 7JH	F D Bird & Sons Limited The Old Hall Lakeview Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR	F D Bird & Sons Limited The Old Hall Lakeview Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR	Unknown (in respect of rights of easements and covenants contained in a Conveyance dated 19 January 1987)



	Land w	/hich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	/ind Farm Development Consent Order IEFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	T	Category 1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-010 cont'd			William Anthony Bethell Arnold Manor Black Tup Lane Arnold HULL East Riding of Yorkshire HU11 5JA			
08-011	Temporary Possession	583 square metres of public road, access splay and verge (Catwick Heads)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire HU11 5BL (in respect of subsoil beneath public highway) Robert Charles Orlando Hellyer Paxmere House Peasemore NEWBURY Berkshire RG20 7JH (in respect of subsoil beneath public highway)	NONE	County Hall Cross Street BEVERLEY East Riding of Yorkshire	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown



	Land w	hich is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-011 cont'd			Unknown			
			Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR (in respect of subsoil beneath public highway) William Anthony Bethell Arnold Manor Black Tup Lane Arnold HULL East Riding of Yorkshire HU11 5JA (in respect of subsoil beneath public highway)			
08-012	Temporary Possession	4675 square metres of agricultural land, access splay and verge (west of Catwick Heads)	Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR	NONE	Criftins Farm	Unknown (in respect of rights of light, air, footpaths and carriageways contained in a Conveyance dated 6 April 1965)



Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
t of acquisition or use	Description of land		Category 1	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
tion of Rights	1637 square metres of agricultural land (east of Catwick Heads)	Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire HU11 5BL (as reputed owner) James Heppell Mewburn Sandsfield Brandesburton DRIFFIELD East Riding of Yorkshire YO25 8SB (as reputed owner) Robert Charles Orlando Hellyer Paxmere House Peasemore NEWBURY Berkshire RG20 7JH (as reputed owner) Unknown William Anthony Bethell Arnold Manor Black Tup Lane Arnold HULL East Riding of Yorkshire	NONE	Hugh Adrian BethellRise ParkRiseHULLEast Riding of YorkshireHU11 5BL(as reputed owner)James Heppell MewburnSandsfieldBrandesburtonDRIFFIELDEast Riding of YorkshireYO25 8SB(as reputed owner)Robert Charles Orlando HellyerPaxmere HousePeasemoreNEWBURYBerkshireRG20 7JH(as reputed owner)UnknownWilliam Anthony BethellArnold ManorBlack Tup LaneArnoldHULLEast Riding of Yorkshire	Unknown			
	of acquisition or use	of acquisition or use Description of land ion of Rights 1637 square metres of agricultural land (east of	BOOK O Land which is proposed to be subject to: (i) powers of compulsory acquisition, County of County of County of Description of land Description of land Counters ion of Rights 1637 square metres of agricultural land (east of Catwick Heads) 1637 square metres of 1637 square metres of 1638 squa	BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to use the land, and/or (iii) right to use the land, and	BOCK OF REFERENCE - PART1           Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to set he land, and/or (iii) right to carry out protective works (Regulation 7(1) County of East Riding of Yorkshire           of acquisition or use         Description of land         Category 1           Description of land         Qualifying persons under Regulation 7(1)(1)(1) of the Inforstructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009         Occupiers           100 of Rights         2637 square metres of agricultural land (esst of Catwick Heads)         High Adrian Bethell         NONE         High Adrian Bethell         Rese Park           Rise         High Adrian Bethell         NONE         Rise Park         Rise Park           Rise         High Adrian Bethell         NONE         High Adrian Bethell         Rise Park           Rise         Hull         East Riding of Yorkshire         Hull         East Riding of Yorkshire         Hull           Lass Right of Statis         Rise Park         Rise         Rise         Hull         East Riding of Yorkshire           Hull         East Riding of Yorkshire         Hull Statis         Garreputed owner)         James Heppell Mewburn         Sandsfield         Sandsfield         Sandsfield         Sandsfield         Sandsfield         Sandsfield         Sandsfield         Sandsfield         Sandsf			



	Land w	hich is proposed to be subject to	BOOK OF F c (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to ast Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-014	Acquisition of Rights	1919 square metres of public road and verge (Catwick Heads)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire HU11 5BL (in respect of subsoil beneath public highway) Robert Charles Orlando Hellyer Paxmere House Peasemore NEWBURY Berkshire RG20 7JH (in respect of subsoil beneath public highway) Unknown	NONE	Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown



	Land w	hich is proposed to be subject to	BOOK OF RI : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-014 cont'd			Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR (in respect of subsoil beneath public highway) William Anthony Bethell Arnold Manor Black Tup Lane Arnold HULL East Riding of Yorkshire HU11 5JA (in respect of subsoil beneath public highway)			
08-015	Acquisition of Rights	agricultural land, access track, drain and public footpath (Catwick Footpath No.8) (west of Catwick Heads)	Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public footpath (Catwick Footpath No.8))	Unknown (in respect of rights of light, air, footpaths and carriageways contained in a Conveyance dated 6 April 1965)



	Land w	hich is proposed to be subject to	BOOK OF RE c (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Lanu Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-015 cont'd					Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR	
08-016	Acquisition of Rights	, , , , , , , , , , , , , , , , , , ,	Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR	NONE	Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR	Unknown (in respect of right of way as contained in a Conveyance dated 10 May 1989)
08-017	Acquisition of Rights	Farm, north of Rise Lane)	George Richard Williams Flat 10 Tennyson Mansions Queen's Club Gardens LONDON W14 9TJ (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable)	P C Foster & Son Cobble Hall Farm Rise Lane Catwick BEVERLEY East Riding of Yorkshire HU17 5PN	P C Foster & Son Cobble Hall Farm Rise Lane Catwick BEVERLEY East Riding of Yorkshire HU17 5PN	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)



	Land w	hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-017 cont'd			Nigel Richard Wild Foxdale Mill Lane Oasby GRANTHAM Lincolnshire NG32 3NA (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable) Rupert Edward Odo Russell Highleaze House Oare MARLBOROUGH Wiltshire SN8 4JE (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable)			
08-018	Temporary Possession	10 square metres of agricultural land (Cobble Hall Farm, north of Rise Lane)	George Richard Williams Flat 10 Tennyson Mansions Queen's Club Gardens LONDON W14 9TJ (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable)	P C Foster & Son Cobble Hall Farm Rise Lane Catwick BEVERLEY East Riding of Yorkshire HU17 5PN	P C Foster & Son Cobble Hall Farm Rise Lane Catwick BEVERLEY East Riding of Yorkshire HU17 5PN	NONE



	Land w	hich is proposed to be subject to	BOOK OF RE DOCK OF RE	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1)	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-018 cont'd			Nigel Richard Wild Foxdale Mill Lane Oasby GRANTHAM Lincolnshire NG32 3NA (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable) Rupert Edward Odo Russell Highleaze House Oare MARLBOROUGH Wiltshire SN8 4JE (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable)			
08-019	Temporary Possession	147 square metres of public road and verges (Rise Lane)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	P C Foster & Son Cobble Hall Farm Rise Lane Catwick BEVERLEY East Riding of Yorkshire HU17 5PN	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE



	Land wh	ich is proposed to be subject t	BOOK OF REF co: (i) powers of compulsory acquisition, (ii) ri	nd Farm Development Consent Order ERENCE - PART 1 ight to use the land, and/or (iii) rights to c Riding of Yorkshire	arry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(	Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-019 cont'd			George Richard Williams Flat 10 Tennyson Mansions Queen's Club Gardens LONDON W14 9TJ (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable) Nigel Richard Wild Foxdale Mill Lane Oasby GRANTHAM Lincolnshire NG32 3NA (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable) Rupert Edward Odo Russell Highleaze House Oare MARLBOROUGH Wiltshire SN8 4JE (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable)			



	Land w	hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land		Category 1	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-020	Temporary Possession	61 square metres of agricultural land (Cobble Hall Farm, south of Rise Lane)		P C Foster & Son Cobble Hall Farm Rise Lane Catwick BEVERLEY East Riding of Yorkshire HU17 5PN		NONE



	Land w	hich is proposed to be subject to	BOOK OF RI DOCK OF RI	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	(a))	
Number or Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
08-021	Acquisition of Rights	212 square metres of public road and verges (Rise Lane)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) George Richard Williams Flat 10 Tennyson Mansions Queen's Club Gardens LONDON W14 9TJ (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable) Nigel Richard Wild Foxdale Mill Lane Oasby GRANTHAM Lincolnshire NG32 3NA (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable)	P C Foster & Son Cobble Hall Farm Rise Lane Catwick BEVERLEY East Riding of Yorkshire HU17 5PN	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE	



	Land w	which is proposed to be subject to	BOOK OF RE b: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-021 cont'd			Rupert Edward Odo Russell Highleaze House Oare MARLBOROUGH Wiltshire SN8 4JE (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable)			
08-022	Temporary Possession	35 square metres of agricultural land (Cobble Hall Farm, north of Rise Lane)	George Richard Williams Flat 10 Tennyson Mansions Queen's Club Gardens LONDON W14 9TJ (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable) Nigel Richard Wild Foxdale Mill Lane Oasby GRANTHAM Lincolnshire NG32 3NA (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable)	P C Foster & Son Cobble Hall Farm Rise Lane Catwick BEVERLEY East Riding of Yorkshire HU17 5PN	P C Foster & Son Cobble Hall Farm Rise Lane Catwick BEVERLEY East Riding of Yorkshire HU17 5PN	NONE



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009		Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
08-022 cont'd			Rupert Edward Odo Russell Highleaze House Oare MARLBOROUGH Wiltshire SN8 4JE (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable)						
08-023	Temporary Possession	159 square metres of public road and verges (Rise Lane)	County Hall Cross Street BEVERLEY East Riding of Yorkshire	P C Foster & Son Cobble Hall Farm Rise Lane Catwick BEVERLEY East Riding of Yorkshire HU17 5PN	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE			



	Land w	hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-023 cont'd			Nigel Richard Wild Foxdale Mill Lane Oasby GRANTHAM Lincolnshire NG32 3NA (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable) Rupert Edward Odo Russell Highleaze House Oare MARLBOROUGH Wiltshire SN8 4JE (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable)			
08-024	Temporary Possession	32 square metres of agricultural land (Cobble Hall Farm, south of Rise Lane)	George Richard Williams Flat 10 Tennyson Mansions Queen's Club Gardens LONDON W14 9TJ (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable)	P C Foster & Son Cobble Hall Farm Rise Lane Catwick BEVERLEY East Riding of Yorkshire HU17 5PN	P C Foster & Son Cobble Hall Farm Rise Lane Catwick BEVERLEY East Riding of Yorkshire HU17 5PN	NONE



	Land w	hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-024 cont'd			Nigel Richard Wild Foxdale Mill Lane Oasby GRANTHAM Lincolnshire NG32 3NA (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable) Rupert Edward Odo Russell Highleaze House Oare MARLBOROUGH Wiltshire SN8 4JE (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable)			
08-025	Acquisition of Rights	14445 square metres of agricultural land (Cobble Hall Farm, south of Rise Lane)	George Richard Williams Flat 10 Tennyson Mansions Queen's Club Gardens LONDON W14 9TJ (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable)	P C Foster & Son Cobble Hall Farm Rise Lane Catwick BEVERLEY East Riding of Yorkshire HU17 5PN	P C Foster & Son Cobble Hall Farm Rise Lane Catwick BEVERLEY East Riding of Yorkshire HU17 5PN	NONE



	Land w	hich is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-025 cont'd			Nigel Richard Wild Foxdale Mill Lane Oasby GRANTHAM Lincolnshire NG32 3NA (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable) Rupert Edward Odo Russell Highleaze House Oare MARLBOROUGH Wiltshire SN8 4JE (as The Executor of the Estate of the Late Lady Ernestine Caroline Antoinette Valerie Stickland-Constable)			
08-026	Acquisition of Rights	· ·	Christine Ann Ellerington Old Hall Farm Long Lane BEVERLEY East Riding of Yorkshire HU17 ORN	NONE	Old Hall Farm Long Lane BEVERLEY East Riding of Yorkshire HU17 ORN	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as Mortgagee to James Anthony Ellerington and Christine Ann Ellerington)



	Land wh	ich is proposed to be subject		nd Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights t	o carry out protective works (Regulation 7(1	)(a))
				t Riding of Yorkshire		
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-026 cont'd			James Anthony Ellerington Field House Farm Rise Lane Catwick BEVERLEY HU17 5PN John Christopher Bird The Old Hall Lakeview Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of mines and minerals) Mark Desmond Bird Old Hall Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of mines and minerals) Paul Adrian Bird Old Hall Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of mines and minerals) Paul Adrian Bird Old Hall Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR		James Anthony Ellerington Field House Farm Rise Lane Catwick BEVERLEY HU17 5PN	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown (in respect of rights contained in a Transfer dated 2 September 2009)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	/ind Farm Development Consent Order IEFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Hans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-001	Acquisition of Rights		Christine Ann Ellerington Old Hall Farm Long Lane BEVERLEY East Riding of Yorkshire HU17 ORN James Anthony Ellerington Field House Farm Rise Lane Catwick BEVERLEY HU17 5PN	NONE	Christine Ann Ellerington Old Hall Farm Long Lane BEVERLEY East Riding of Yorkshire HU17 ORN James Anthony Ellerington Field House Farm Rise Lane Catwick BEVERLEY HU17 5PN	Amy Clark-Blakey Glenholme Leven Road Catwick BEVERLEY HU17 5PT (in respect of rights contained in a Transfer dated 2 March 2008) Gemma Louise Elizabeth Clark-Blakey Glenholme Leven Road Catwick BEVERLEY HU17 5PT (in respect of rights contained in a Transfer dated 2 March 2008)
			John Christopher Bird The Old Hall Lakeview Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of mines and minerals)			John Christopher Bird The Old Hall Lakeview Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of rights contained in a Transfer dated 30 July 2009 )



	Land wh	ich is proposed to be subject	BOOK OF R to: (i) powers of compulsory acquisition, (ii)	/ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to c st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-001 cont'd			Mark Desmond Bird Old Hall Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of mines and minerals) Paul Adrian Bird Old Hall Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of mines and minerals)			Mark Desmond Bird Old Hall Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of rights contained in a Transfer dated 30 July 2009 ) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Paul Adrian Bird, Mark Desmond Bird and John Christopher Bird) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of rights contained in a Deed of Grant of Easement dated 5 September 1967. Also in respect of rights and restrictive covenants contained in a Deed of Grant of Easement dated 3 October 1966)	



	Land v	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii	/ind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Euria Fians			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-001 cont'd						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights contained in a Lease dated 14 October 2021) Paul Adrian Bird Old Hall Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of rights contained in a Transfer dated 30 July 2009 )
09-002	Acquisition of Rights	5341 square metres of agricultural land (east of Riston Road)	Christine Ann Ellerington Old Hall Farm Long Lane BEVERLEY East Riding of Yorkshire HU17 ORN James Anthony Ellerington Field House Farm Rise Lane Catwick BEVERLEY HU17 5PN	NONE	Christine Ann Ellerington Old Hall Farm Long Lane BEVERLEY East Riding of Yorkshire HU17 ORN James Anthony Ellerington Field House Farm Rise Lane Catwick BEVERLEY HU17 5PN	Unknown



	Land w	/hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-002 cont'd			Unknown		Unknown	
09-003	Acquisition of Rights		The Executor of the Estate of the Late Walter Trevor Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR Unknown (in respect of mines and minerals) Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR	NONE	Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR	NONE
09-004	Temporary Possession	92 square metres of agricultural land (east of Riston Road)	The Executor of the Estate of the Late Walter Trevor Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR	NONE	Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR	NONE



	Land w	hich is proposed to be subject to	BOOK OF RE b: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009		Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-004 cont'd			Unknown (in respect of mines and minerals) Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR			
09-005	Temporary Possession	89 square metres of public road and verges (Riston Road)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) The Executor of the Estate of the Late Walter Trevor Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR (in respect of subsoil beneath public highway) Unknown	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Unknown



	Land w	/hich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-005 cont'd			Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR (in respect of subsoil beneath public highway)			
09-006	Acquisition of Rights		East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) The Executor of the Estate of the Late Walter Trevor Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR (in respect of subsoil beneath public highway) Unknown	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Unknown



	Land w	which is proposed to be subject to	BOOK OF F o: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order IEFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-006 cont'd			Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR (in respect of subsoil beneath public highway)			
09-007	Temporary Possession	117 square metres of agricultural land (west of Riston Road)	Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR	NONE	Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR	Holderness Hunt (Holdings) Limited Citadel House 58 High Street HULL East Riding of Yorkshire HU1 1QE (in respect of rights of access) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Walter Stuart Leonard Kirkwood) Unknown (in respect of right of access contained in a Conveyance dated 6 April 1918)



	Land w	hich is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to Ist Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-008	Acquisition of Rights	15705 square metres of agricultural land and drain (west of Riston Road)	Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR	NONE	Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR	Holderness Hunt (Holdings) Limited Citadel House 58 High Street HULL East Riding of Yorkshire HU1 1QE (in respect of rights of access) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Walter Stuart Leonard Kirkwood) Unknown (in respect of right of access contained in a Conveyance dated 6 April 1918)
09-009	Acquisition of Rights	19717 square metres of agricultural land and drain (Willow Croft Farm, west of Riston Road)	Yarrows Aggregates Limited Little Catwick Quarry Leven Bypass Leven BEVERLEY East Riding of Yorkshire HU17 5AA	Ashley Nigel Foster Cruckley Farm Foston-on-the-Wolds DRIFFIELD East Riding of Yorkshire YO25 8BS	Ashley Nigel Foster Cruckley Farm Foston-on-the-Wolds DRIFFIELD East Riding of Yorkshire YO25 8BS	John Christopher Bird The Old Hall Lakeview Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of rights granted by a Conveyance dated 27 November 1925)



	Land wh	nich is proposed to be subject to	BOOK OF RI D: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to a st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-009 cont'd						Mark Desmond Bird Old Hall Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of rights granted by a Conveyance dated 27 November 1925) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Yarrows Aggregates Limited) Paul Adrian Bird Old Hall Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of rights granted by a Conveyance dated 27 November 1925)



			BOOK OF RE	ind Farm Development Consent Order FERENCE - PART 1		
	Land w	nich is proposed to be subject to		right to use the land, and/or (III) rights to t Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners or Reputed Owners	Category 1 )(a) of the Infrastructure Planning (Appli Regulations 2009 Lessees or Tenants or Reputed Lessees or Tenants	cations: Prescribed Forms and Procedures) Occupiers or Reputed Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-010	Acquisition of Rights	and drain (Prospect Farm, east of Whitecross Road, A165 and west of Catwick Lane)	Hugh Adrian Bethell Rise Park	NONE	Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire	JBM Solar Projects 33 Limited Windmill Hill Business Park Whitehill Way SWINDON SN5 6PB (as Beneficiary in respect of rights contained in a document dated 24 May 2023) Unknown (in respect of rights listed in a conveyance dated 19 January 1987)



			BOOK OF RE	ind Farm Development Consent Order FERENCE - PART 1		
	Land w	hich is proposed to be subject to		right to use the land, and/or (iii) rights to t Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners or Reputed Owners	Category 1 )(a) of the Infrastructure Planning (Appli Regulations 2009 Lessees or Tenants or Reputed Lessees or Tenants	cations: Prescribed Forms and Procedures) Occupiers or Reputed Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-011	Temporary Possession	agricultural land and hedgerow (Prospect Farm, east of Whitecross Road, A165 and west of Catwick Lane)	Hugh Adrian Bethell	NONE	Hugh Adrian Bethell Rise Park Rise HULL East Riding of Yorkshire	JBM Solar Projects 33 Limited Windmill Hill Business Park Whitehill Way SWINDON SN5 6PB (as Beneficiary in respect of rights



	Land w	/hich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
09-012	Temporary Possession	6000 square metres of agricultural land and drain (Riston Grange Farm, east of Whitecross Road, A165)	or Reputed Owners Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	or Reputed Lessees or Tenants	or Reputed Occupiers Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	Procedures) Regulations 2009 Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of a gas main and rights listed in a Deed dated 3 October 1966) JBM Solar Projects 33 Limited Windmill Hill Business Park Whitehill Way SWINDON SN5 6PB (as Beneficiary in respect of rights contained in a document dated 30 May 2023) Unknown (in respect of right of entry and access to services as contained in a Conveyance dated 4 October 1977)	
09-013	Acquisition of Rights	39182 square metres of agricultural land and drain (Riston Grange Farm, east of Whitecross Road, A165)	Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	NONE	Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of a gas main and rights listed in a Deed dated 3 October 1966)	



	Land w	which is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009 Lessees or Tenants	ations: Prescribed Forms and Procedures) Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and
			or Reputed Owners	or Reputed Lessees or Tenants	or Reputed Occupiers	Procedures) Regulations 2009
09-013 cont'd						JBM Solar Projects 33 Limited Windmill Hill Business Park Whitehill Way SWINDON SN5 6PB (as Beneficiary in respect of rights contained in a document dated 30 May 2023) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Unknown (in respect of right of entry and access to services as contained in a Conveyance dated 4 October 1977)
10-001	Temporary Possession	agricultural land (Riston Grange Farm, east of Whitecross Road, A165)		NONE	Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of a gas main and rights listed in a Deed dated 3 October 1966)





	Land w	hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-001 cont'd						JBM Solar Projects 33 Limited Windmill Hill Business Park Whitehill Way SWINDON SN5 6PB (as Beneficiary in respect of rights contained in a document dated 30 May 2023) Unknown (in respect of right of entry and access to services as contained in a Conveyance dated 4 October 1977)
10-002	Acquisition of Rights	2119 square metres of public road and verges (Whitecross Road, A165)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of subsoil beneath public highway) Unknown	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown



	Land w	hich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-002 cont'd						Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
10-003	Temporary Possession	3692 square metres of public road and verges (Whitecross Road, A165)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of subsoil beneath public highway) Unknown	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown



	Land w	hich is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Lanu Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-003 cont'd						Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
10-004	Temporary Possession	agricultural land (Riston Grange Farm, west of Whitecross Road, A165)		NONE		Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of rights of access) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of a gas main and rights listed in a Deed dated 3 October 1966)



	Land w	which is proposed to be subject to	BOOK OF c (i) powers of compulsory acquisition, (i	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ast Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-004 cont'd						JBM Solar Projects 33 Limited Windmill Hill Business Park Whitehill Way SWINDON SN5 6PB (as Beneficiary in respect of rights contained in a document dated 30 May 2023) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown (in respect of right of entry and access to services as contained in a Conveyance dated 4 October 1977)
10-005	Acquisition of Rights	44238 square metres of agricultural land and drain (Arnold and Riston Drain) (Riston Grange Farm, west of Whitecross Road, A165)	Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	NONE	Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights and rights of access)



	Land wh	ich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii	/ind Farm Development Consent Order IEFERENCE - PART 1 ) right to use the land, and/or (iii) rights to c st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-005 cont'd						Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of a gas main and rights liste in a Deed dated 3 October 1966)
						JBM Solar Projects 33 Limited Windmill Hill Business Park Whitehill Way SWINDON SN5 6PB (as Beneficiary in respect of rights contained in a document dated 30 May 2023)
						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)
						Unknown (in respect of right of entry and access f services as contained in a Conveyance dated 4 October 1977)



	Land w	which is proposed to be subject to	BOOK ( c: (i) powers of compulsory acquisition	re Wind Farm Development Consent Order DF REFERENCE - PART 1 n, (ii) right to use the land, and/or (iii) rights t f East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
10-006	Acquisition of Rights	27667 square metres of agricultural land and drain (Arnold and Riston Drain) (Riston Grange Farm, west of Whitecross Road, A165)	Richard Guy Caley North Grange Meaux BEVERLEY East Riding of Yorkshire HU17 9SS	NONE	Richard Guy Caley North Grange Meaux BEVERLEY East Riding of Yorkshire HU17 9SS	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of rights of access) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Richard Guy Caley) JBM Solar Projects 33 Limited Windmill Hill Business Park Whitehill Way SWINDON SN5 6PB (as Beneficiary in respect of rights contained in a document dated 31 May 2023)	
10-007	Acquisition of Rights	18189 square metres of agricultural land and public footpath (Riston Footpath No.2) (north of Carr Lane)	Richard Guy Caley North Grange Meaux BEVERLEY East Riding of Yorkshire HU17 9SS	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public footpath (Riston Footpath No.2))	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Richard Guy Caley)	



	Land w	which is proposed to be subject t	BOOK OF F o: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to ist Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-007 cont'd						JBM Solar Projects 33 Limited Windmill Hill Business Park Whitehill Way SWINDON SN5 6PB (as Beneficiary in respect of rights contained in a document dated 31 May 2023) Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR (as The Executor of the Estate of the Late Walter Trevor Kirkwood) (in respect of a right of way as contained in a Conveyance dated 19 July 1967)
10-008	Acquisition of Rights	2 square metres of verge (north of Carr Lane)	Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	T H Caley and Sons Limited Church Farm Meaux Road Routh BEVERLEY North Humberside HU17 9SR	T H Caley and Sons Limited Church Farm Meaux Road Routh BEVERLEY North Humberside HU17 9SR	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights listed in a Deed of Grant dated 5 September 1967)



	Land whi	ch is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order :FERENCE - PART 1 right to use the land, and/or (iii) rights to ca t Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-008 cont'd						Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of the rights of access to maintain drainage as contained in a Conveyance dated 4 February 1957) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Thomas Stephen Caley) JBM Solar Projects 33 Limited Windmill Hill Business Park Whitehill Way SWINDON SN5 6PB (as Beneficiary in respect of rights contained in a document dated 30 Ma 2023) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights contained in a Lea dated 14 July 2015)



	Land w	hich is proposed to be subject to	BOOK OF R	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to	carry out protective works (Regulation 7(1	)(a))
				t Riding of Yorkshire Category 1	(	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	)(a) of the Infrastructure Planning (Applica) Regulations 2009	ations: Prescribed Forms and Procedures)	Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-008 cont'd						T H Caley and Sons Limited Church Farm Meaux Road Routh BEVERLEY North Humberside HU17 9SR (in respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)
10-009	Acquisition of Rights	2273 square metres of agricultural land, verges and drain (north of Carr Lane)	Unknown	T H Caley and Sons Limited Church Farm Meaux Road Routh BEVERLEY North Humberside HU17 9SR	T H Caley and Sons Limited Church Farm Meaux Road Routh BEVERLEY North Humberside HU17 9SR Unknown	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights to maintain) Richard Guy Caley North Grange Meaux BEVERLEY East Riding of Yorkshire HU17 9SS (in respect of rights of drainage) Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of rights of drainage)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	<pre>/ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire</pre>	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-009 cont'd						Unknown
10-010	Acquisition of Rights	16062 square metres of agricultural land (Church Farm, south of Main Road, A1035)	Caroline Mary Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	NONE		Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights of access to enter and conduct works to maintain drainage as contained in a Conveyance dated 28 March 1956) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus)
10-011	Acquisition of Rights		Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE



	Land w	hich is proposed to be subject to	BOOK OF I : (i) powers of compulsory acquisition, (i	Wind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to ast Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7 Owners	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009		
10-012	Acquisition of Rights	(Routh and Meaux East Drain, Routh) (south of Main Road, A1035)	or Reputed Owners Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	or Reputed Lessees or Tenants	or Reputed Occupiers Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	Procedures) Regulations 2009 NONE
10-013	Acquisition of Rights	and drain (Routh and Meaux East Drain, Routh) (south of Main Road, A1035)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE
10-014		agricultural land, access tracks and drains (Church Farm, south of Main Road, A1035 and east of Meaux Lane)		NONE	HU17 9SR	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights of access to enter and conduct works to maintain drainage as contained in a Conveyance dated 28 March 1956)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Lanu Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-015	Temporary Possession		Caroline Mary Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	NONE	Routh BEVERLEY North Humberside HU17 9SR	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights of access to enter and conduct works to maintain drainage as contained in a Conveyance dated 28 March 1956)
10-016	Temporary Possession	and drains (south of Main Road, A1035)	Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	T H Caley and Sons Limited Church Farm Meaux Road Routh BEVERLEY North Humberside HU17 9SR	Meaux Road Routh BEVERLEY North Humberside HU17 9SR	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights listed in a Deed of Grant dated 5 September 1967) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of the rights of access to maintain drainage as contained in a Conveyance dated 4 February 1957)



	Land whi	ich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to ca st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Applicati Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms an Procedures) Regulations 2009
10-016 cont'd						HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Thomas Stephen Caley) JBM Solar Projects 33 Limited Windmill Hill Business Park Whitehill Way SWINDON SN5 6PB (as Beneficiary in respect of rights contained in a document dated 30 M 2023) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead and underground electricity apparatus) Northern Powergrid (Yorkshire) PLC
						Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights contained in a Le dated 14 July 2015)



	Land w	hich is proposed to be subject to	BOOK OF RI : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		ations: Prescribed Forms and Procedures) Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and	
			or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	or Reputed Occupiers	Procedures) Regulations 2009
10-016 cont'd						T H Caley and Sons Limited Church Farm Meaux Road Routh BEVERLEY North Humberside HU17 9SR (in respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)
10-017	Temporary Possession		East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	NONE		KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead and underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)



	Land w	hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-017 cont'd						Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
10-018	Temporary Possession	789 square metres of public road and access splay (Main Road, A1035)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority and in respect of subsoil beneath public highway) John Christopher Bird The Old Hall Lakeview Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead and underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on Land Plans	Extent of acquisition or use	Description of land		ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
10-018 cont'd			Paul Adrian Bird Old Hall Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of subsoil beneath public highway) Unknown			Unknown			
10-019	Acquisition of Rights	35853 square metres of agricultural land and drains (Manor House Farm, east of Meaux Lane)	Howard Noel Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR Ian Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	NONE	Howard Noel Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR Ian Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	Benjamin David James Mackrill West Barn Furnace Farm Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986) HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR (in respect of rights contained in a Lease of a wind turbine site and solar site dated 13 March 2014)			



	Land wh	ich is proposed to be subject t	BOOK OF REI co: (i) powers of compulsory acquisition, (ii) r	nd Farm Development Consent Order FERENCE - PART 1 ight to use the land, and/or (iii) rights to Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-019 cont'd			The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR		The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR	National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for The Executor of the Estate of the Late Harold Sinkler, Howard Noel Sinkler and Ian Harold Sinkler) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights contained in a Lease dated 29 March 2012) Robin John Mackrill 38 Blandford Road LONDON W4 1DX (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	/ind Farm Development Consent Order EFERENCE - PART 1 I right to use the land, and/or (iii) rights to st Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Lund Fluits			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-019 cont'd						Unknown (in respect of rights contained in a lease dated 29 March 2012)
11-001	Temporary Possession	119 square metres of agricultural land (Manor House Farm, east of Meaux Lane)	Howard Noel Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	NONE	Howard Noel Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	Benjamin David James Mackrill West Barn Furnace Farm Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)
			Ian Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR		Ian Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR (in respect of rights contained in a Lease of a wind turbine site and solar site dated 13 March 2014)
			The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR		The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR	National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for The Executor of the Estate of the Late Harold Sinkler, Howard Noel Sinkler and Ian Harold Sinkler)



	Land w	hich is proposed to be subject to	BOOK OF RI : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-001 cont'd						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights contained in a Lease dated 29 March 2012) Robin John Mackrill 38 Blandford Road LONDON W4 1DX (in respect of rights to drainage as
						contained in a Conveyance dated 12 June 1986) Unknown (in respect of rights contained in a lease dated 29 March 2012)
11-002	Temporary Possession		East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)



	Land wh	nich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) r	nd Farm Development Consent Order FERENCE - PART 1 ight to use the land, and/or (iii) rights to c Riding of Yorkshire	arry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-002 cont'd			Howard Noel Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of subsoil beneath public highway) Ian Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of subsoil beneath public highway) The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY Hu17 9SR (in respect of subsoil beneath public highway) The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR (in respect of subsoil beneath public highway) Unknown			Unknown Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-003	. ,		Howard Noel Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	NONE	Howard Noel Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	Benjamin David James Mackrill West Barn Furnace Farm Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)
			Ian Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR		Ian Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR (in respect of a Lease of a wind turbine site and solar site dated 13 March 2014)
			The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR		The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR	National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for The Executor of the Estate of the Late Harold Sinkler, Howard Noel Sinkler and Ian Harold Sinkler)





	Land w	hich is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-003 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Robin John Mackrill 38 Blandford Road LONDON W4 1DX (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)
11-004	Acquisition of Rights	775 square metres of public road and verges (Meaux Lane)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Howard Noel Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown



	Land w	hich is proposed to be subject to	BOOK OF RI : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and		
11-004 cont'd			or Reputed Owners Ian Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of subsoil beneath public highway) The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR (in respect of subsoil beneath public highway) Unknown	or Reputed Lessees or Tenants		Procedures) Regulations 2009 Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
11-005	Temporary Possession	136 square metres of agricultural land (Manor House Farm, east of Meaux Lane)	Howard Noel Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	NONE	Howard Noel Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	Benjamin David James Mackrill West Barn Furnace Farm Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)



	Land wh	ich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) r	nd Farm Development Consent Order FERENCE - PART 1 ight to use the land, and/or (iii) rights to Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-005 cont'd			Ian Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR		Ian Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR	HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR (in respect of rights contained in a Lease of a wind turbine site and solar site dated 13 March 2014) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for The Executor of the Estate of the Late Harold Sinkler, Howard Noel Sinkler and Ian Harold Sinkler) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights contained in a Lease dated 29 March 2012) Robin John Mackrill 38 Blandford Road LONDON W4 1DX (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)



				/ind Farm Development Consent Order EFERENCE - PART 1		
	Land w	hich is proposed to be subject to		•	carry out protective works (Regulation 7(1	)(a))
			County of Ea	st Riding of Yorkshire		
				Category 1		Category 2
Number on			Qualifying persons under Regulation 7(2)		ations: Prescribed Forms and Procedures)	Qualifying persons under Regulation
Land Plans	Extent of acquisition or use	Description of land		Regulations 2009		7(1)(a) of the Infrastructure Planning
			Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and
			or Reputed Owners	or Reputed Lessees or Tenants	or Reputed Occupiers	Procedures) Regulations 2009
11-006	Temporary Possession	321 square metres of public	East Riding Of Yorkshire Council	NONE	East Riding Of Yorkshire Council	Openreach Limited
		road and verges (Meaux Lane)	County Hall		County Hall	6 Gracechurch Street
			Cross Street		Cross Street	LONDON
			BEVERLEY		BEVERLEY	EC3V 0AT
			East Riding of Yorkshire		East Riding of Yorkshire	(in respect of underground
			HU17 9BA		HU17 9BA	telecommunication apparatus)
			(as highway authority)		(as highway authority)	
			Howard Noel Sinkler			Unknown
			Manor House			
			Meaux Road			
			Routh			
			BEVERLEY			
			East Riding of Yorkshire			
			HU17 9SR			
			(in respect of subsoil beneath public			
			highway)			
			Ian Harold Sinkler			Yorkshire Water Services Limited
			Manor House			Western House
			Meaux Road			Western Way
			Routh			Halifax Road
			BEVERLEY			Bradford
			East Riding of Yorkshire			West Yorkshire
			HU17 9SR			BD6 2SZ
			(in respect of subsoil beneath public			(in respect of underground water
			highway)			apparatus)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-006 cont'd			The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR (in respect of subsoil beneath public highway) Unknown			
11-007	Temporary Possession	160 square metres of agricultural land (Manor House Farm, west of Meaux Lane)	Howard Noel Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	NONE	Howard Noel Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	Benjamin David James Mackrill West Barn Furnace Farm Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)
			lan Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR		lan Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR (in respect of a Lease of a wind turbine site and solar site dated 13 March 2014)



				ind Farm Development Consent Order EFERENCE - PART 1		
	Land w	hich is proposed to be subject to		right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners or Reputed Owners	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009 Lessees or Tenants or Reputed Lessees or Tenants	ations: Prescribed Forms and Procedures) Occupiers or Reputed Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-007 cont'd			The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR		The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR	National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for The Executor of the Estate of the Late Harold Sinkler, Howard Noel Sinkler and Ian Harold Sinkler) Robin John Mackrill 38 Blandford Road LONDON W4 1DX (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)
11-008	Acquisition of Rights	and drain (Manor House Farm, east of A1035, Routh and west of Meaux Lane)		NONE	Howard Noel Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	Benjamin David James Mackrill West Barn Furnace Farm Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)



	Land wh	nich is proposed to be subject t	BOOK OF RE co: (i) powers of compulsory acquisition, (ii) r	nd Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to : Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land		ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-008 cont'd			lan Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR		Ian Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR (in respect of a Lease of a wind turbine site and solar site dated 13 March 2014)
			The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR		Manor House Meaux Road Routh BEVERLEY HU17 9SR	National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for The Executor of the Estate of the Late Harold Sinkler, Howard Noel Sinkler and Ian Harold Sinkler)
						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)
						Robin John Mackrill 38 Blandford Road LONDON W4 1DX (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)



	Land w	hich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-008 cont'd						Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
11-009	Temporary Possession	24147 square metres of agricultural land (east of A1035, Routh)	Howard Noel Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR Ian Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	NONE	Howard Noel Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR Ian Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR	Benjamin David James Mackrill West Barn Furnace Farm Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986) HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR (in respect of a Lease of a wind turbine site and solar site dated 13 March 2014)



	Land wl	nich is proposed to be subject t	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)		carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land					
11-009 cont'd			The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR	or Reputed Lessees or Tenants	or Reputed Occupiers The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR	Procedures) Regulations 2009 National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for The Executor of the Estate of the Late Harold Sinkler, Howard Noel Sinkler and Ian Harold Sinkler) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Robin John Mackrill 38 Blandford Road LONDON W4 1DX (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)	



	Land w	hich is proposed to be subject to	BOOK OF : (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ast Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-010	Acquisition of Rights	1415 square metres of public road, verges, footway and access splay (A1035, Routh)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Howard Noel Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of rights to maintain hedges as contained in a Conveyance dated 6 June 1974) Ian Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of rights to maintain hedges as contained in a Conveyance dated 6 June 1974) KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus)



	Land wh	nich is proposed to be subject t	BOOK OF co: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to c ast Riding of Yorkshire	arry out protective works (Regulation 7(1	L)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-010 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR (in respect of rights to maintain hedges as contained in a Conveyance dated 6 June 1974) Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)



	Londu	hiak ia maaaaad ka ka ambiaat ta	BOOK OF R	ind Farm Development Consent Order EFERENCE - PART 1	a name automatica status (Desculation 7/4	
	Land w	nich is proposed to be subject to		right to use the land, and/or (III) rights to t Riding of Yorkshire	o carry out protective works (Regulation 7(1	J(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners or Reputed Owners	Category 1 L)(a) of the Infrastructure Planning (Appli Regulations 2009 Lessees or Tenants or Reputed Lessees or Tenants	cations: Prescribed Forms and Procedures) Occupiers or Reputed Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-011	Acquisition of Rights	road, verges, footway and access splay (A1035, Routh)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority and in respect of subsoil beneath public highway) Unknown	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown
11-012	Acquisition of Rights	road, verges, footway and access splay (A1035, Routh)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE



	Land w	hich is proposed to be subject to	BOOK OI c (i) powers of compulsory acquisition,	Wind Farm Development Consent Order F REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights t East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-013	Temporary Possession	520 square metres of agricultural land and access track (Hall Farm and Field House Farm, south of Tickton Bypass, A1035)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	NONE	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH CB2 1PH	Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of rights contained in a Deed of Grant dated 31 July 2019) Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights and easements contained in a Lease dated 21 February 2020. Also in respect of rights granted as contained in a Deed dated 12 October 2023.) Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of rights contained in a Deed of Grant dated 31 July 2019)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-013 cont'd						Doggerbank Offshore Wind Farm Proje 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights of access to conduct works as contained in a Deed of Grant dates 6 April 1984. Also in respect of rights contained in a Lease dated 21 February 2020) East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of rights contained in an Option Agreement dated 24 October 2017) Hall Farm Wind Farm Limited C/O Res Limited Beaufort Court Egg Farm Lane Kings Langley Hertfordshire WD4 8LR (as Beneficiary in respect of rights contained in a document dated 23 February 2009)			



	Land wh	ich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to c st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-013 cont'd						Hall Farm Wind Farm Limited C/O Res Limited Beaufort Court Egg Farm Lane Kings Langley Hertfordshire WD4 8LR (in respect of rights contained in a Leas dated 24 November 2011) National Westminster Bank PLC 250 Bishopsgate LONDON EC2IM 4AA (as mortgagee for Albanwise Limited) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights contained in a Leas
						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Fians			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-014	Temporary Possession	77 square metres of public road, verge and access splay (A1035, Routh)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)			
11-015	Temporary Possession	607 square metres of public road, verges and access splay (A1035, Routh)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus)			



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire							
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
11-015 cont'd			East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority and in respect of subsoil beneath public highway)			Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus)		
			Howard Noel Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of subsoil beneath public highway)			Unknown		
			Ian Harold Sinkler Manor House Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of subsoil beneath public highway)			Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)		



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-015 cont'd			The Executor of the Estate of the Late Harold Sinkler Manor House Meaux Road Routh BEVERLEY HU17 9SR (in respect of subsoil beneath public highway) Unknown				
11-016	Temporary Possession	road, verges, footway and access splay (A1035, Routh)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	NONE		Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	



	Land w	hich is proposed to be subject to	BOOK OF : (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights t ast Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-017	Temporary Possession	<b>U</b>	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road	NONE	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH CB2 1PH	Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and



	Land wh	ich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	/ind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to ca st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans		·	Owners or Reputed Owners	Regulations 2009 Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-017 cont'd						Doggerbank Offshore Wind Farm Proje 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights of access to conduct works as contained in a Deed of Grant dates 6 April 1984. Also in respect of rights contained in a Lease dated 21 February 2020) East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of rights contained in an Option Agreement dated 24 October 2017) Hall Farm Wind Farm Limited C/O Res Limited Beaufort Court Egg Farm Lane Kings Langley Hertfordshire WD4 8LR (as Beneficiary in respect of rights contained in a document dated 23 February 2009)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2)	cions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-017 cont'd						Hall Farm Wind Farm Limited C/O Res Limited Beaufort Court Egg Farm Lane Kings Langley Hertfordshire WD4 8LR (in respect of rights contained in a Lease dated 24 November 2011) KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)			



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-017 cont'd						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights contained in a Lease dated 14 September 2018)			
11-018	Acquisition of Rights	Farm and Field House Farm, east of Eske Lane and west of	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	NONE	c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of rights contained in a document dated 14 August 2019) Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights and easements contained in a Lease dated 21 February 2020. Also in respect of rights granted as contained in a Deed dated 12 October 2023.)			



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-018 cont'd						Doggerbank Offshore Wind Farm Proje 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of rights contained in a Deed of Grant dated 31 July 2019) Doggerbank Offshore Wind Farm Proje 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights of access to condu works as contained in a Deed of Grant dates 6 April 1984. Also in respect of rights contained in a Lease dated 21 February 2020) East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of rights contained in an Option Agreement dated 24 October 2017)			



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
11-018 cont'd						Hall Farm Wind Farm Limited C/O Res Limited Beaufort Court Egg Farm Lane Kings Langley Hertfordshire WD4 8LR (as Beneficiary in respect of rights contained in a document dated 23 February 2009) Hall Farm Wind Farm Limited C/O Res Limited Beaufort Court Egg Farm Lane Kings Langley Hertfordshire WD4 8LR (in respect of rights contained in a Lease dated 24 November 2011) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)			



	Land wh	nich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	/ind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to a st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-018 cont'd						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)
						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights contained in a Lease dated 14 September 2018)
						Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of rights to maintain a watermain as contained in a Deed dated 6 April 1984)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-018 cont'd						Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
12-001	Temporary Possession	agricultural land and drain (Hall Farm and Field House Farm, east of Eske Lane)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	NONE	c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of rights contained in a document dated 14 August 2019) Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights and easements contained in a Lease dated 21 February 2020. Also in respect of rights granted as contained in a Deed dated 12 October 2023.)



	Land whi	ich is proposed to be subject t	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to c st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applicat Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
12-001 cont'd						Doggerbank Offshore Wind Farm Pro 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of rights contained in a Deed of Grant dated 3 July 2019) Doggerbank Offshore Wind Farm Pro 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights of access to cond works as contained in a Deed of Gran dates 6 April 1984. Also in respect of rights contained in a Lease dated 21 February 2020) East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of rights contained in an Option Agreement dated 24 October 2017)



	Land wh	ich is proposed to be subject t	BOOK OF BOOK OF i	Vind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to c ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-001 cont'd						Hall Farm Wind Farm Limited C/O Res Limited Beaufort Court Egg Farm Lane Kings Langley Hertfordshire WD4 8LR (as Beneficiary in respect of rights contained in a document dated 23 February 2009) Hall Farm Wind Farm Limited C/O Res Limited Beaufort Court Egg Farm Lane Kings Langley Hertfordshire WD4 8LR (in respect of rights contained in a Lease dated 24 November 2011) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)



	Land w	rhich is proposed to be subject to	BOOK OF RE b: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
12-001 cont'd						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights contained in a Lease dated 14 September 2018)
12-002	Temporary Possession	127 square metres of agricultural land and drain (east of Eske Lane)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (as reputed owner) Unknown	NONE	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (as reputed owner) Unknown	Unknown



	Land w	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to ast Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
12-003	Temporary Possession	959 square metres of public road and verge (Eske Lane)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (in respect of subsoil beneath public highway) East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) T H Caley and Sons Limited Church Farm Meaux Road Routh BEVERLEY North Humberside HU17 9SR (in respect of subsoil beneath public highway) Unknown	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Unknown	



	Land w	hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
12-004	Acquisition of Rights	85 square metres of agricultural land and drain (east of Eske Lane)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (as reputed owner) Unknown	NONE	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (as reputed owner) Unknown	Unknown
12-005	Acquisition of Rights	564 square metres of public road and verge (Eske Lane)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) T H Caley and Sons Limited Church Farm Meaux Road Routh BEVERLEY North Humberside HU17 9SR (in respect of subsoil beneath public highway) Unknown	NONE	County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
12-006	Temporary Possession	5985 square metres of public road and verges (Eske Lane and Tickton Bypass, A1035)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (in respect of subsoil beneath public highway) Dianne Marion Burnett	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Northern Powergrid (Yorkshire) PLC
			2 High Eske Farm Cottages Eske Lane Tickton BEVERLEY East Riding of Yorkshire HU17 9SG (in respect of subsoil beneath public highway)			Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground and overhead electricity apparatus)
			East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority and in respect of subsoil beneath public highway)			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)



	Land wh	ich is proposed to be subject	BOOK OF RE to: (i) powers of compulsory acquisition, (ii)	nd Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to c t Riding of Yorkshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applicat Regulations 2009	cions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
12-006 cont'd			Geraldine Burnett 1 Eske Lane Tickton BEVERLEY East Riding of Yorkshire HU17 9SG (in respect of subsoil beneath public highway) Michael Burnett 2 High Eske Farm Cottages Eske Lane Tickton BEVERLEY East Riding of Yorkshire HU17 9SG (in respect of subsoil beneath public highway) T H Caley and Sons Limited Church Farm Meaux Road Routh BEVERLEY North Humberside HU17 9SR (in respect of subsoil beneath public highway) Unknown			Unknown



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii	/ind Farm Development Consent Order IEFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7( Owners	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009 Lessees or Tenants	ations: Prescribed Forms and Procedures) Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and
			or Reputed Owners	or Reputed Lessees or Tenants	or Reputed Occupiers	Procedures) Regulations 2009
12-007	Temporary Possession	agricultural land and copse (north of Tickton Bypass, A1035 and west of Eske Lane)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	NONE	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	John Scott Tickton Farm 56 Main Street Tickton BEVERLEY East Riding of Yorkshire HU17 9RZ (in respect of rights to inspect, clean, maintain, repair and renew the buildings, walls and fences contained in a Transfer dated 19 December 2017)
12-008	Temporary Possession	road and verges (Eske Lane and Tickton Bypass, A1035)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus)



	Land w	hich is proposed to be subject to	BOOK OF : (i) powers of compulsory acquisition,	Wind Farm Development Consent Order REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights t ast Riding of Yorkshire	o carry out protective works (Regulation 7(1	.)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
12-009		Bypass, A1035)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead and underground electricity apparatus) Unknown (in respect of rights contained in a Conveyance dated 19 September 1972)	
12-010	Temporary Possession	2920 square metres of agricultural land (west of Eske Lane)	T H Caley and Sons Limited Church Farm Meaux Road Routh BEVERLEY North Humberside HU17 9SR	NONE	T H Caley and Sons Limited Church Farm Meaux Road Routh BEVERLEY North Humberside HU17 9SR	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	



	Land w	/hich is proposed to be subject to	BOOK OF : (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights t ast Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Appl Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
12-011	Acquisition of Rights	96989 square metres of agricultural land, hedgerows and drains (west of Eske Lane)	T H Caley and Sons Limited Church Farm Meaux Road Routh BEVERLEY North Humberside HU17 9SR	NONE	T H Caley and Sons Limited Church Farm Meaux Road Routh BEVERLEY North Humberside HU17 9SR	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights and rights of access) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)
12-012	Acquisition of Rights	2659 square metres of grassland, river bank and public footpath (Tickton Footpath No.1) (east of the River Hull, Beverley)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public footpath (Tickton Footpath No.1))	NONE



	Land w	hich is proposed to be subject to	BOOK OF R (i) powers of compulsory acquisition, (ii)	•	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	County of Eas Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
12-012 cont'd					Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	
12-013	Acquisition of Rights	101 square metres of river bank (east of the River Hull, Beverley)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as reputed owner) Unknown	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as reputed owner) Unknown	Unknown
12-014	Acquisition of Rights	2356 square metres of foreshore and bed (River Hull, Beverley) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE
13-001	Acquisition of Rights	770 square metres of grassland, river bank, public bridleway (Leconfield Bridleway No.27) and public footpath (Leconfield Footpath No.33) (west of the River Hull, Beverley)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as reputed owner)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public footpath (Leconfield Footpath No.33) and public bridleway (Leconfield Bridleway No.27))	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus)



	Land w	hich is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1)	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
13-001 cont'd			Unknown		Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as reputed owner) Unknown	Unknown	
13-002	Acquisition of Rights	grassland, river bank, public bridleway (Leconfield Bridleway No.27) and public footpath (Leconfield Footpath	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public footpath (Leconfield Footpath No.33) and public bridleway (Leconfield Bridleway No.27)) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus) Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of rights contained in a Transfer scheme dated 1 September 1989)	



	Land w	hich is proposed to be subject to	BOOK OF F : (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order XEFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
13-003	Acquisition of Rights	828 square metres of drain (Beverley and Barmston Drain) (west of the River Hull, Beverley)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE
13-004	Acquisition of Rights	71530 square metres of agricultural land and copse (Carr Farm, east of Carr Road and north of Hull Bridge Road, A1035)	Angela Elizabeth Sellers Grange Farm Arram BEVERLEY East Riding of Yorkshire HU17 7NR	NONE	Angela Elizabeth Sellers Grange Farm Arram BEVERLEY East Riding of Yorkshire HU17 7NR	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of rights of access) Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access and the free passage of water and electricity contained in a Conveyance dated 25 March 1974) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)



	Land v	which is proposed to be subject to	BOOK Of c: (i) powers of compulsory acquisition,	Wind Farm Development Consent Order FREFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights to East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
13-004 cont'd						Phillip Brumfield Dunkenhill Farm Highgate Cherry Burton BEVERLEY HU17 7RT (as The Executor of the Estate of The Late David Roy Duncan Brumfield) (in respect of rights contained in a Transfer dated 26 February 1993) The Executor Of The Estate Of The Late David Roy Duncan Brumfield The Beeches Seven Corners Lane BEVERLEY East Riding of Yorkshire HU17 7AJ (in respect of rights contained in a Transfer dated 26 February 1993)
13-005	Acquisition of Rights	243 square metres of drain (west of the River Hull, Beverley)	Angela Elizabeth Sellers Grange Farm Arram BEVERLEY East Riding of Yorkshire HU17 7NR (as reputed owner)	NONE	Angela Elizabeth Sellers Grange Farm Arram BEVERLEY East Riding of Yorkshire HU17 7NR (as reputed owner)	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1)	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
13-005 cont'd			Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (as reputed owner) Unknown		Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (as reputed owner) Unknown	Unknown
13-006	Temporary Possession	agricultural land and dike (South Bullock Dike, Carr Farm, east of Carr Road and north of Hull Bridge Road, A1035)	Angela Elizabeth Sellers Grange Farm Arram BEVERLEY East Riding of Yorkshire HU17 7NR	NONE	Grange Farm Arram BEVERLEY East Riding of Yorkshire HU17 7NR	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights and rights of access) Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access and the free passage of water and electricity contained in a Conveyance dated 25 March 1974)



	Land w	which is proposed to be subject to	BOOK OF F : (i) powers of compulsory acquisition, (i	Vind Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to Ist Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
13-006 cont'd						Phillip Brumfield Dunkenhill Farm Highgate Cherry Burton BEVERLEY HU17 7RT (as The Executor of the Estate of The Late David Roy Duncan Brumfield) (in respect of rights contained in a Transfer dated 26 February 1993) The Executor Of The Estate Of The Late David Roy Duncan Brumfield The Beeches Seven Corners Lane BEVERLEY East Riding of Yorkshire HU17 7AJ (in respect of rights contained in a Transfer dated 26 February 1993)
13-007	Temporary Possession	4825 square metres of agricultural land and dike (South Bullock Dike, east of Carr Road)	Tamara Clare Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	NONE	Tamara Clare Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	nd Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
13-007 cont'd						Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Tamara Clare Watson Hall) Unknown (in respect of rights to service media reserved by a Transfer dated 22 June 1989)
13-008	Temporary Possession	agricultural land, culvert and drain (Molescroft Grange Farm, east of Carr Road)	Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	NONE	Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights and rights of access) Michael Jonathan Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (in respect of restrictions contained in a Transfer dated 12 November 2012)



	Land w	hich is proposed to be subject to	BOOK OF RI : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure: Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-008 cont'd						S R J Developments (Yorkshire) Limited Belgrave House 15 Belgrave Crescent SCARBOROUGH North Yorkshire YO11 1UB (in respect of rights of access and others listed in a Transfer dated 14 July 2000) Yorkshire Water Services Limited Western House Western House	
						Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of rights and easements relating to water mains, passage of water, entry and covenants listed in a Deed dated 26 September 1995)	
13-009	Acquisition of Rights		Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	NONE	Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of rights of access)	



	Land wi	nich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	/ind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to c st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
13-009 cont'd						Michael Jonathan Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (in respect of restrictions contained in a Transfer dated 12 November 2012) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) S R J Developments (Yorkshire) Limited Belgrave House 15 Belgrave Crescent SCARBOROUGH North Yorkshire YO11 1UB (in respect of rights of access and others listed in a Transfer dated 14 July 2000)





	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	nd Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o t Riding of Yorkshire	carry out protective works (Regulation 7(1)	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners	tions: Prescribed Forms and Procedures) Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and	
13-009 cont'd			or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	or Reputed Occupiers	Procedures) Regulations 2009 Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of rights and easements relating to water mains, passage of water, entry and covenants listed in a Deed dated 26 September 1995)
13-010	Temporary Possession	47466 square metres of agricultural land, hedgerows, drains and access tracks (Molescroft Grange Farm, east of Grange Way, A1035)	Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	NONE	Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights and rights o access) Michael Jonathan Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (in respect of restrictions contained in a Transfer dated 12 November 2012)



	Land wh	ich is proposed to be subject to	BOOK OF RI b: (i) powers of compulsory acquisition, (ii)	'ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to ca st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
13-010 cont'd						Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a Deed of Release dated 27 February 2019) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) S R J Developments (Yorkshire) Limited Belgrave House 15 Belgrave Crescent SCARBOROUGH North Yorkshire YO11 1UB (in respect of rights of access and other listed in a Transfer dated 14 July 2000)





	Land w	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to a t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
13-010 cont'd						Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of rights and easements relating to water mains, passage of water, entry and covenants listed in a Deed dated 26 September 1995)
13-011	Temporary Possession	2687 square metres of agricultural land, drains and access track (east of Grange Way, A1035)	Tamara Clare Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	NONE	Tamara Clare Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights and rights of access) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Tamara Clare Watson Hall) Unknown (in respect of rights to service media reserved by a Transfer dated 22 June 1989)



	Land w	hich is proposed to be subject t	BOOK O o: (i) powers of compulsory acquisition,	e Wind Farm Development Consent Order F REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights t East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
Land Hans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
13-012	Temporary Possession	752 square metres of public road and verge (Hull Bridge Road, A1035)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus) Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 Owners Lessees or Tenants Occupiers			
			or Reputed Owners	or Reputed Lessees or Tenants	or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
13-013	Temporary Possession	road (Hull Bridge Road, A1035)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority and in respect of subsoil beneath public highway)	NONE	County Hall Cross Street BEVERLEY East Riding of Yorkshire	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	
			Unknown			Unknown	
13-014	Acquisition of Rights	,	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (as reputed owner) Unknown	NONE	Waterloo General Office LONDON	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) Unknown	
13-015	Acquisition of Rights	(north of Carr Road)	Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	NONE	Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rights to access a level crossing contained in a Deed of Release dated 12 September 1995)	



	Land wh	ich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	/ind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to c st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
13-015 cont'd						Phillip Brumfield Dunkenhill Farm Highgate Cherry Burton BEVERLEY HU17 7RT (as The Executor of the Estate of the Late Barbara Brumfield and The Late David Roy Duncan Brumfield) (in respect of rights contained in a Transfer dated 22 June 1989) S R J Developments (Yorkshire) Limited Belgrave House 15 Belgrave Crescent SCARBOROUGH North Yorkshire YO11 1UB (in respect of rights, agreements and covenants listed in a Transfer dated 14 July 2000) The Executor Of The Estate Of The Late David Roy Duncan Brumfield The Beeches Seven Corners Lane BEVERLEY East Riding of Yorkshire HU17 7AJ (in respect of rights contained in a Transfer dated 22 June 1989)



	Land w	hich is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
13-016	Acquisition of Rights	and public footpath (Molescroft Footpath No.5) (Low Parks Farm, north of Carr Road)	Phillip Brumfield Dunkenhill Farm Highgate Cherry Burton BEVERLEY HU17 7RT (as The Executor of the Estate of the Late Barbara Brumfield and The Late David Roy Duncan Brumfield) The Executor of the Estate of the Late Barbara Brumfield Dunkenhill Farm	NONE	HU17 9BA (in respect of public footpath (Molescroft Footpath No.5)) Phillip Brumfield Dunkenhill Farm Highgate	Catherine Suzanne Mace The Chestnuts Carr Road BEVERLEY East Riding of Yorkshire HU17 7JZ (in respect of rights of access as contained in an Indenture dated 20 September 1920) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street
			Highgate Cherry Burton BEVERLEY HU17 7RT		Late Barbara Brumfield and The Late David Roy Duncan Brumfield)	Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)
			The Executor Of The Estate Of The Late David Roy Duncan Brumfield The Beeches Seven Corners Lane BEVERLEY East Riding of Yorkshire HU17 7AJ		The Executor of the Estate of the Late Barbara Brumfield Dunkenhill Farm Highgate Cherry Burton BEVERLEY HU17 7RT	Thomas Anthony Richard Mace The Chestnuts Carr Road BEVERLEY East Riding of Yorkshire HU17 7JZ (in respect of rights of access as contained in an Indenture dated 20 September 1920)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	nd Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1)	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
13-016 cont'd					The Executor Of The Estate Of The Late David Roy Duncan Brumfield The Beeches Seven Corners Lane BEVERLEY East Riding of Yorkshire HU17 7AJ	
13-017	Acquisition of Rights	agricultural land and drains (north of Carr Road)	Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	NONE	Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	Michael Jonathan Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (as mortgagee for Molescroft Farms Ltd) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a Deed of Release dated 27 February 2019)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
13-017 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) S R J Developments (Yorkshire) Limited Belgrave House 15 Belgrave Crescent SCARBOROUGH North Yorkshire YO11 1UB (in respect of rights of access and others listed in a Transfer dated 14 July 2000)
14-001	Temporary Possession	agricultural land (north of Carr Road, Molescroft and north of Ings Road)		NONE	East Riding of Yorkshire	Michael Jonathan Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (as mortgagee for Molescroft Farms Ltd)



	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	/ind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
14-001 cont'd						Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a Deed of Release dated 27 February 2019)
						S R J Developments (Yorkshire) Limited Belgrave House 15 Belgrave Crescent SCARBOROUGH North Yorkshire YO11 1UB (in respect of rights of access and others listed in a Transfer dated 14 July 2000)
14-002	Temporary Possession	13 square metres of public road and verge (Carr Road)	Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rights to access a level crossing contained in a Deed of Release dated 12 September 1995)



	Land wh	ich is proposed to be subject t	BOOK OF co: (i) powers of compulsory acquisition, (i	Wind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to ca ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applicat Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
14-002 cont'd						Phillip Brumfield Dunkenhill Farm Highgate Cherry Burton BEVERLEY HU17 7RT (as The Executor of the Estate of the Late Barbara Brumfield and The Late David Roy Duncan Brumfield) (in respect of rights contained in a Transfer dated 22 June 1989) The Executor Of The Estate Of The Late David Roy Duncan Brumfield The Beeches Seven Corners Lane BEVERLEY East Riding of Yorkshire HU17 7AJ (in respect of rights contained in a Transfer dated 22 June 1989) Unknown (in respect of rights contained in a Conveyance dated 11 September 1975)



				ind Farm Development Consent Order EFERENCE - PART 1						
	Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))									
			County of Eas	t Riding of Yorkshire						
				Category 1		Category 2				
Number on			Qualifying persons under Regulation 7(1		ations: Prescribed Forms and Procedures)	Qualifying persons under Regulation				
Land Plans	Extent of acquisition or use	Description of land		Regulations 2009		7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Procedures) Regulations 2009				
14-003	Temporary Possession	2309 square metres of public	East Riding Of Yorkshire Council	NONE	East Riding Of Yorkshire Council	KCOM Group Limited				
		road and verges (Ings Road)	County Hall		County Hall	37 Carr Lane				
			Cross Street		Cross Street	Hull				
			BEVERLEY		BEVERLEY	East Yorkshire				
			East Riding of Yorkshire		East Riding of Yorkshire	HU1 3RE				
			HU17 9BA		HU17 9BA	(in respect of underground				
			(as highway authority and in respect of		(as highway authority)	telecommunication apparatus)				
			subsoil beneath public highway)							
			Ian Peter Robson			Northern Powergrid (Yorkshire) PLC				
			4 Chapel Street			Lloyds Court				
			Nunnington			78 Grey Street				
			YORK			Newcastle Upon Tyne				
			YO62 5UP			NE1 6AF				
			(in respect of subsoil beneath public			(in respect of underground electricity				
			highway)			apparatus)				
			John Michael Mudryk			Unknown				
			31 Westgate							
			North Cave							
			BROUGH							
			HU15 2NG							
			(in respect of subsoil beneath public							
			highway)							



	Land w	hich is proposed to be subject to	BOOK OF RE c (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
14-003 cont'd			Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (in respect of subsoil beneath public highway) Richard Hendrik Los 34 Sellers Drive Leconfield BEVERLEY HU17 7NA (in respect of subsoil beneath public highway) Unknown			
14-004	Temporary Possession	roads and verges (Ings Road and Grange Way, A1035)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus)



	Land wh	ich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to c st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 ()(a) of the Infrastructure Planning (Applica) Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
14-004 cont'd						Richmond Properties (UK) Limited Field Head Driffield Road Molescroft Beverley East Yorkshire HU17 7LU (in respect of rights of access to development sites as contained in a Transfer dated 31 December 1993) Shirethorn Limited
						Unit 8B Marina Court Castle Street HULL East Riding of Yorkshire HU1 1TJ (in respect of rights of access to development sites as contained in a Transfer dated 31 December 1993)
						Tamara Clare Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (in respect of rights contained in a Transfer dated 22 June 1989)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
14-005	Acquisition of Rights	agricultural land, access track and public footpath (Molescroft Footpath No.3) (Low Parks Farm, north of Ings Road)	Phillip Brumfield Dunkenhill Farm Highgate Cherry Burton BEVERLEY HU17 7RT (as The Executor of the Estate of the Late Barbara Brumfield and The Late David Roy Duncan Brumfield) The Executor of the Estate of the Late Barbara Brumfield Dunkenhill Farm Highgate Cherry Burton BEVERLEY HU17 7RT	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public footpath (Molescroft Footpath No.3)) Phillip Brumfield Dunkenhill Farm Highgate Cherry Burton BEVERLEY HU17 7RT (as The Executor of the Estate of the Late Barbara Brumfield and The Late David Roy Duncan Brumfield)	NONE
			The Executor Of The Estate Of The Late David Roy Duncan Brumfield The Beeches Seven Corners Lane BEVERLEY East Riding of Yorkshire HU17 7AJ		Ine Executor of the Estate of the Late Barbara Brumfield Dunkenhill Farm Highgate Cherry Burton BEVERLEY HU17 7RT	



	Land w	hich is proposed to be subject to	BOOK OF RI b: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
14-005 cont'd					The Executor Of The Estate Of The Late David Roy Duncan Brumfield The Beeches Seven Corners Lane BEVERLEY East Riding of Yorkshire HU17 7AJ	
14-006	Acquisition of Rights	32110 square metres of agricultural land and drains (Ings Drain) (east of Driffield Road, A164)	Ian Peter Robson 4 Chapel Street Nunnington YORK YO62 5UP John Michael Mudryk 31 Westgate North Cave BROUGH HU15 2NG Richard Hendrik Los 34 Sellers Drive Leconfield BEVERLEY HU17 7NA Unknown (in respect of mines and minerals)	NONE	Bishop Burton College York Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QG	NONE



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1)	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
14-007	Acquisition of Rights	railway track, verges (Former Market Weighton to Beverley railway line) and public footpath (Molescroft Footpath	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public footpath (Molescroft Footpath No.6))	NONE
14-008	Acquisition of Rights	26576 square metres of agricultural land (east of Driffield Road, A164)	Tamara Clare Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	NONE	BEVERLEY East Riding of Yorkshire HU17 8SU	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Tamara Clare Watson Hall) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus)



	Land w	hich is proposed to be subject to	BOOK OF I	Vind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	County of Ea Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
20110110110			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
14-008 cont'd						Unknown (in respect of rights to service media reserved by a Transfer dated 22 June 1989)
14-009	Temporary Possession	A164)	Tamara Clare Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	NONE	Tamara Clare Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Tamara Clare Watson Hall) Unknown (in respect of rights to service media reserved by a Transfer dated 22 June
14-010	Temporary Possession	Road, A164)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	1989) KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus)



	Land wh	ich is proposed to be subject	BOOK OF RE to: (i) powers of compulsory acquisition, (ii)	nd Farm Development Consent Order :FERENCE - PART 1 right to use the land, and/or (iii) rights to o t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
14-010 cont'd			Tamara Clare Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (in respect of subsoil beneath public highway) Unknown			Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)



				/ind Farm Development Consent Order EFERENCE - PART 1		
	Land w	hich is proposed to be subject to		right to use the land, and/or (iii) rights to st Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	l · · · ·	Category 1	cations: Prescribed Forms and Procedures) Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and
			or Reputed Owners	or Reputed Lessees or Tenants	or Reputed Occupiers	Procedures) Regulations 2009
14-011	Acquisition of Rights	road and verges (Driffield Road, A164)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Tamara Clare Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (in respect of subsoil beneath public highway) Unknown	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	/ind Farm Development Consent Order IEFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Lanu Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
14-011 cont'd						Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
14-012	Temporary Possession	land (west of Driffield Road, A164)	Tamara Clare Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	NONE	BEVERLEY	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Tamara Clare Watson Hall) Unknown (in respect of rights to service media reserved by a Transfer dated 22 June
14-013	Acquisition of Rights	Driffield Road, A164)	Tamara Clare Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	NONE	Tamara Clare Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	1989) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Tamara Clare Watson Hall)



	Land w	hich is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
14-013 cont'd						Unknown (in respect of rights to service media reserved by a Transfer dated 22 June 1989)
14-014	Acquisition of Rights	agricultural land and hedgerow (Constitution Hill Farm, north of Malton Road, A1035)	Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	NONE	Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the option agreement dated 24 November 2021) Unknown (in respect of rights relating to a sewer listed in a Conveyance dated 9 November 1920)
15-001	Temporary Possession	agricultural land and hedgerow (Constitution Hill Farm, north of Malton Road, A1035)	Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	NONE	Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the option agreement dated 24 November 2021) Unknown (in respect of rights relating to a sewer listed in a Conveyance dated 9 November 1920)



	Land w	hich is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights t ast Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
15-002	Temporary Possession	973 square metres of public road, access splay and verges (Malton Road, A1035)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (in respect of subsoil beneath public highway) Unknown	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus) Unknown Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	<pre>/ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire</pre>	carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			or Reputed Owners	or Reputed Lessees or Tenants	or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
15-003	Temporary Possession	Hill Farm, south of Malton Road, A1035)	Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	NONE	Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Molescroft Farms Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the option agreement dated 24 November 2021)	
15-004	Acquisition of Rights	1845 square metres of public road and verges (Malton Road, A1035)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus)	



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to a t Riding of Yorkshire	carry out protective works (Regulation 7(1)	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners	tions: Prescribed Forms and Procedures) Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and	
			or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	or Reputed Occupiers	Procedures) Regulations 2009
15-004 cont'd			Unknown			Unknown
						Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
15-005	Temporary Possession	Grange Farm, south of Malton Road, A1035)	Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	NONE	BEVERLEY East Riding of Yorkshire HU17 8SU	Michael Jonathan Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (in respect of restrictions contained in a Transfer dated 12 November 2012) S R J Developments (Yorkshire) Limited Belgrave House 15 Belgrave Crescent SCARBOROUGH North Yorkshire YO11 1UB (in respect of rights of access and others listed in a Transfer dated 14 July 2000)



	Land w	hich is proposed to be subject to	BOOK O : (i) powers of compulsory acquisition,	e Wind Farm Development Consent Order F REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights t East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Lanu Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
15-006	Acquisition of Rights	Grange Farm, south of Malton Road, A1035)	Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	NONE	Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU	Michael Jonathan Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (in respect of restrictions contained in a Transfer dated 12 November 2012) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a Deed of Release dated 27 February 2019) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)



	Land w	hich is proposed to be subject to	BOOK OF R	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	County of Eas Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Regulations 2009 Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
15-006 cont'd						S R J Developments (Yorkshire) Limited Belgrave House 15 Belgrave Crescent SCARBOROUGH North Yorkshire YO11 1UB (in respect of rights of access and others listed in a Transfer dated 14 July 2000)
15-007	Acquisition of Rights	166 square metres of drain (south of Malton Road, A1035)	Andrew Digby Cooke Mount Pleasant Farm Beverley Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 & QY (as reputed owner) Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 & SU (as reputed owner) Nicholas John Cooke 10 Bryan Mere Bishop Burton BEVERLEY HU17 & QW (as reputed owner)	NONE	Andrew Digby Cooke Mount Pleasant Farm Beverley Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QY (as reputed owner) Molescroft Farms Limited Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (as reputed owner) Nicholas John Cooke 10 Bryan Mere Bishop Burton BEVERLEY HU17 8QW (as reputed owner)	Unknown



	Land w	hich is proposed to be subject to	BOOK OF I : (i) powers of compulsory acquisition, (i	Wind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights t ast Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Appl Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
15-007 cont'd			Unknown		Unknown	
15-008	Acquisition of Rights			NONE	Andrew Digby Cooke Mount Pleasant Farm Beverley Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QY Nicholas John Cooke 10 Bryan Mere Bishop Burton BEVERLEY HU17 8QW	National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Andrew Digby Cooke and Nicholas John Cooke) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (in respect of rights reserved by a Deed dated 4 March 1988)
16-001	Acquisition of Rights	and hedgerows (Mount Pleasant Farm, north of York Road)	Andrew Digby Cooke Mount Pleasant Farm Beverley Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QY	NONE	Andrew Digby Cooke Mount Pleasant Farm Beverley Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QY	Lindum Developments Limited Lindum Business Park Station Road North Hykeham Lincoln LN6 3QX (in respect of rights of access and covenants listed in a Transfer dated 2 March 2021)





	l and u		BOOK OF R	/ind Farm Development Consent Order EFERENCE - PART 1	) carry out protective works (Regulation 7(1	
	Land w	vnich is proposed to be subject to		st Riding of Yorkshire	carry out protective works (Regulation 7(1	.)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
16-001 cont'd			Nicholas John Cooke 10 Bryan Mere Bishop Burton BEVERLEY HU17 8QW		Nicholas John Cooke 10 Bryan Mere Bishop Burton BEVERLEY HU17 8QW	National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Andrew Digby Cooke and Nicholas John Cooke) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Unknown (in respect of rights of way, light and drainage listed in a Conveyance dated 11 September 1953)
16-002	Acquisition of Rights	1418 square metres of public road, verges, scrubland and access splay (York Road)	Andrew Digby Cooke Mount Pleasant Farm Beverley Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QY	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus)



	Land wh	ich is proposed to be subject	BOOK OF to: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ca ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applicat Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
16-002 cont'd			Nicholas John Cooke 10 Bryan Mere Bishop Burton BEVERLEY HU17 8QW			Lindum Developments Limited Lindum Business Park Station Road North Hykeham Lincoln LN6 3QX (in respect of rights of access and covenants listed in a Transfer dated 2 March 2021) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Andrew Digby Cooke and Nicholas John Cooke) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)



Land w	hich is proposed to be subject to	BOOK OF F : (i) powers of compulsory acquisition, (ii	REFERENCE - PART 1 ) right to use the land, and/or (iii) rights t	o carry out protective works (Regulation 7(1	)(a))
Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Appli Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
					Unknown (in respect of rights of way, light and drainage listed in a Conveyance dated 11 September 1953)
Acquisition of Rights	(Mount Pleasant Farm, north of York Road)	Mount Pleasant Farm Beverley Road Bishop Burton BEVERLEY East Riding of Yorkshire	NONE	Andrew Digby Cooke Mount Pleasant Farm Beverley Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QY	Lindum Developments Limited Lindum Business Park Station Road North Hykeham Lincoln LNG 3QX (in respect of rights of access and covenants listed in a Transfer dated 2 March 2021)
		10 Bryan Mere Bishop Burton BEVERLEY		Nicholas John Cooke 10 Bryan Mere Bishop Burton BEVERLEY HU17 8QW	National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Andrew Digby Cooke and Nicholas John Cooke) Unknown (in respect of rights of way, light and drainage listed in a Conveyance dated 11 September 1953)
	Extent of acquisition or use	Extent of acquisition or use Description of land Acquisition of Rights 624 square metres of scrubland (Mount Pleasant Farm, north of York Road)	Extent of acquisition or use Description of land Acquisition of Rights Cauty of East Riding of Yorkshire Acquisition of Rights Cauty of East Riding of Yorkshire County of East Riding of Yorkshire HU17 8QY BOOK OF F County of East County of Eas	County of East Riding of Yorkshire         Extent of acquisition or use       Description of land       Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Apple Regulations 2009         Owners       Lessees or Tenants         Owners       Lessees or Tenants         Owners       Lessees or Tenants         Or Reputed Owners       NONE         Acquisition of Rights       624 square metres of scrubland         Andrew Digby Cooke       NONE         York Road)       Beverley Road         Bishop Burton       BEVERLEY         East Riding of Yorkshire       HU17 8QY         Nicholas John Cooke       Nicholas John Cooke         10 Bryan Mere       Bishop Burton       BEVERLEY       East Riding of Yorkshire	BOOK OF REFERENCE - PART 1         Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1) County of East Riding of Yorkshire         Extent of acquisition or use       Description of land       Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009         Owners       Lessees or Tenants       Occupiers         Or Reputed Owners       or Reputed Lessees or Tenants       Occupiers         Acquisition of Rights       624 square metres of scrubland       Andrew Digby Cooke       NONE       Andrew Digby Cooke         Mount Pleasant Farm, north of Mount Pleasant Farm       Beverley Road       Bishop Burton       Beverley Road         Bishop Burton       BEVERLEY       East Riding of Yorkshire       Hu17 8QY       Hu17 8QY         Nicholas John Cooke       10 Bryan Mere       Bishop Burton       Bishop Burton         BEVERLEY       Bishop Burton       Bishop Burton       Bishop Burton         BEVERLEY       Nicholas John Cooke       10 Bryan Mere       Bishop Burton         BEVERLEY       Bishop Burton       Bishop Burton       Bishop Burton         BEVERLEY       Bishop Burton       Bishop Burton       Bishop Biston



	Land w	hich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii		o carry out protective works (Regulation 7(1)	(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	County of East Riding of Yorkshire Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
16-004	Acquisition of Rights	91 square metres of public road, verge and access splay (York Road, Beverley) (excluding all interests of the Crown)	Andrew Digby Cooke Mount Pleasant Farm Beverley Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QY (in respect of subsoil beneath public highway) East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Nicholas John Cooke 10 Bryan Mere Bishop Burton BEVERLEY HU17 8QW (in respect of subsoil beneath public highway) Secretary Of State For Transport Great Minster House 33 Horseferry Road Marsham LONDON SW1P 4DR (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Unknown	



	Land w	which is proposed to be subject t	BOOK OF F co: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to sst Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 (1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
16-004 cont'd			Unknown			
16-005	Temporary Possession	27 square metres of public road and verges (York Road, Beverley) (excluding all interests of the Crown)	Andrew Digby Cooke Mount Pleasant Farm Beverley Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QY (in respect of subsoil beneath public highway) East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Nicholas John Cooke 10 Bryan Mere Bishop Burton BEVERLEY HU17 8QW (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Unknown



	Land w	hich is proposed to be subject to	BOOK OF R b: (i) powers of compulsory acquisition, (ii)		carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	County of East Riding of Yorkshire Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
16-005 cont'd			Secretary Of State For Transport Great Minster House 33 Horseferry Road Marsham LONDON SW1P 4DR (in respect of subsoil beneath public highway) Unknown				
16-006	Temporary Possession	road and verges (York Road, Beverley) (excluding all interests of the Crown)	Secretary Of State For Transport Great Minster House 33 Horseferry Road Marsham LONDON SW1P 4DR	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Unknown (in respect of restrictive covenants imposed before 11 February 2009)	
16-007	Acquisition of Rights	road and verges (York Road, Beverley) (excluding all	Secretary Of State For Transport Great Minster House 33 Horseferry Road Marsham LONDON SW1P 4DR	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Unknown (in respect of restrictive covenants imposed before 11 February 2009)	



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
16-008	Acquisition of Rights	agricultural land and hedgerow (Mount Pleasant Farm, south of York Road)	Andrew Digby Cooke Mount Pleasant Farm Beverley Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QY Nicholas John Cooke 10 Bryan Mere Bishop Burton BEVERLEY HU17 8QW	NONE	Andrew Digby Cooke Mount Pleasant Farm Beverley Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QY Nicholas John Cooke 10 Bryan Mere Bishop Burton BEVERLEY HU17 8QW	National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Andrew Digby Cooke and Nicholas John Cooke) Unknown (in respect of rights of way, light and drainage listed in a Conveyance dated 11 September 1953)
16-009	Acquisition of Rights	agricultural land and hedgerow (west of Newbald Road, Walkington and east of A1079, Beverley)	Beverley Road	NONE	Andrew Digby Cooke Mount Pleasant Farm Beverley Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QY Nicholas John Cooke 10 Bryan Mere Bishop Burton BEVERLEY HU17 8QW	NONE



	Land w	<i>r</i> hich is proposed to be subject t	BOOK OF	Wind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights t	o carry out protective works (Regulation 7(1	)(a))		
Number on	County of East Riding of Yorkshire Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures)							
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Regulations 2009 Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
17-001	Acquisition of Rights	832 square metres of public road and verges (Newbald Road)	Andrew Digby Cooke Mount Pleasant Farm Beverley Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QY (in respect of subsoil beneath public highway) Ann Christine Jackson 82 Moor Lane Carnaby BRIDLINGTON YO16 4UU (in respect of subsoil beneath public highway) East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Nicholas John Cooke 10 Bryan Mere Bishop Burton BEVERLEY HU17 8QW (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Unknown		



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1)	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Lanu Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
17-001 cont'd			The Executor of The Estate of The Late David Robert Jackson Carnaby House Farm Carnaby BRIDLINGTON East Riding of Yorkshire YO16 4UJ (in respect of subsoil beneath public highway) Unknown			
17-002	Acquisition of Rights	agricultural land, hedgerows, footpath, drain and and public footpath (Walkington Footpath No.6) (Broadgate Farm, north of Broadgate, B1230 and south of Newbald Road)	BRIDLINGTON YO16 4UU	NONE	Ann Christine Jackson 82 Moor Lane Carnaby BRIDLINGTON YO16 4UU East Riding Of Yorkshire Council	NONE
			David Robert Jackson Carnaby House Farm Carnaby BRIDLINGTON East Riding of Yorkshire YO16 4UJ		County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public footpath (Walkington Footpath No.6))	



	Land w	hich is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
17-002 cont'd					The Executor of The Estate of The Late David Robert Jackson Carnaby House Farm Carnaby BRIDLINGTON East Riding of Yorkshire YO16 4UJ	
17-003	Temporary Possession	Broadgate, B1230)	Ann Christine Jackson 82 Moor Lane Carnaby BRIDLINGTON YO16 4UU The Executor of The Estate of The Late David Robert Jackson Carnaby House Farm Carnaby BRIDLINGTON East Riding of Yorkshire YO16 4UJ	NONE	Ann Christine Jackson 82 Moor Lane Carnaby BRIDLINGTON YO16 4UU The Executor of The Estate of The Late David Robert Jackson Carnaby House Farm Carnaby BRIDLINGTON East Riding of Yorkshire YO16 4UJ	NONE
17-004	Temporary Possession	road and verge (Broadgate, B1230)	Ann Christine Jackson 82 Moor Lane Carnaby BRIDLINGTON YO16 4UU (in respect of subsoil beneath public highway)	NONE	Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to a t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
17-004 cont'd			East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority and in respect of subsoil beneath public highway) The Executor of The Estate of The Late David Robert Jackson Carnaby House Farm Carnaby BRIDLINGTON East Riding of Yorkshire YO16 4UJ (in respect of subsoil beneath public highway) Unknown			Unknown Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground sewerage and water apparatus)
17-005	Acquisition of Rights	· · · · · · · · · · · · · · · ·	Ann Christine Jackson 82 Moor Lane Carnaby BRIDLINGTON YO16 4UU (in respect of subsoil beneath public highway)	NONE	5	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus)



	Land wh	ich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) ri	nd Farm Development Consent Order ERENCE - PART 1 ight to use the land, and/or (iii) rights to c Riding of Yorkshire	arry out protective works (Regulation 7(1)	i(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		Owners Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
17-005 cont'd			Bryant Homes Northern Limited Gate House Turnpike Road HIGH WYCOMBE Buckinghamshire HP12 3NR (in respect of subsoil beneath public highway) East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority and in respect of subsoil beneath public highway) The Executor of The Estate of The Late David Robert Jackson Carnaby House Farm Carnaby BRIDLINGTON East Riding of Yorkshire YO16 4UJ (in respect of subsoil beneath public			Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus) Unknown Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground sewerage
			highway) Unknown			and water apparatus)



	Land w	hich is proposed to be subject to	BOOK OF RI	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to	o carry out protective works (Regulation 7(1	)(a))
		,		st Riding of Yorkshire		
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners or Reputed Owners	Category 1 L)(a) of the Infrastructure Planning (Appli Regulations 2009 Lessees or Tenants or Reputed Lessees or Tenants	cations: Prescribed Forms and Procedures) Occupiers or Reputed Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-006		541 square metres of public road and verge (Broadgate, B1230)	Ann Christine Jackson 82 Moor Lane Carnaby BRIDLINGTON YO16 4UU (in respect of subsoil beneath public highway) East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority and in respect of subsoil beneath public highway) The Executor of The Estate of The Late David Robert Jackson Carnaby House Farm Carnaby BRIDLINGTON East Riding of Yorkshire YO16 4UJ (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus) Unknown



	Land w	hich is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	· · ·	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
17-006 cont'd			Unknown			Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground sewerage and water apparatus)
17-007	Temporary Possession	(south of Broadgate, B1230)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	NONE	НU17 9ВА	Bryant Homes Northern Limited Gate House Turnpike Road HIGH WYCOMBE Buckinghamshire HP12 3NR (in respect of the rights of access to sewage reserved by the Conveyance dated 28 September 1967) KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
17-007 cont'd						The Lord Mayor Aldermen and Citizens of the City and County of Kingston Upon Hull Kingston upon Hull City Council The Guildhall Alfred Gelder Street HULL East Riding of Yorkshire HU1 2AA (in respect of the right of access granted by a Deed of Grant dated 21 November 1972) Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
17-008	Acquisition of Rights	512 square metres of car park and hardstanding (south of Broadgate, B1230)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	Bryant Homes Northern Limited Gate House Turnpike Road HIGH WYCOMBE Buckinghamshire HP12 3NR (in respect of reserved rights of way and others including restrictive covenants as contained in a Transfer dated 7 May 2010)



			BOOK OF	Nind Farm Development Consent Order REFERENCE - PART 1	7/0	V- W
	Land w	nich is proposed to be subject to		i) right to use the land, and/or (iii) rights to ast Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7 Owners or Reputed Owners	Category 1 (1)(a) of the Infrastructure Planning (Appli Regulations 2009 Lessees or Tenants or Reputed Lessees or Tenants	cations: Prescribed Forms and Procedures) Occupiers or Reputed Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-009	Acquisition of Rights		East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	Bryant Homes Northern Limited Gate House Turnpike Road HIGH WYCOMBE Buckinghamshire HP12 3NR (in respect of the rights of access to sewage reserved by the Conveyance dated 28 September 1967) KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus)



	Land wh	ich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	'ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to c st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
17-009 cont'd						The Lord Mayor Aldermen and Citizen: of the City and County of Kingston Upo Hull Kingston upon Hull City Council The Guildhall Alfred Gelder Street HULL East Riding of Yorkshire HU1 2AA (in respect of the right of access granted by a Deed of Grant dated 21 November 1972) Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of rights of access) Yorkshire Water Services Limited Western House Western House Western House West Yorkshire BD6 2SZ (in respect of rights of access) Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)



			88	Wind Farm Development Consent Order REFERENCE - PART 1		
	Land w	hich is proposed to be subject to		ii) right to use the land, and/or (iii) rights t East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 7(1)(a) of the Infrastructure Planning (Appli Regulations 2009 Lessees or Tenants	cations: Prescribed Forms and Procedures) Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and
			or Reputed Owners	or Reputed Lessees or Tenants	or Reputed Occupiers	Procedures) Regulations 2009
17-010	Acquisition of Rights	grassland, copse and building (Broadgates Hospital, south of Broadgate, B1230) (excluding all interests of the Crown)	Bryant Homes Northern Limited Gate House Turnpike Road HIGH WYCOMBE Buckinghamshire HP12 3NR	NONE	Stephen Holtby Broadgate Farm Beverley Road Walkington BEVERLEY East Riding of Yorkshire HU17 8RP	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)
						The Lord Mayor Aldermen and Citizens of the City and County of Kingston Upon Hull Kingston upon Hull City Council The Guildhall Alfred Gelder Street HULL East Riding of Yorkshire HU1 2AA (in respect of rights to construct and maintain a 14 inch water pipe listed in a Deed dated 09 June 1972)
						Unknown (in respect of rights relating to supply of water listed in a Conveyance dated 14 April 1983)



	Land w	hich is proposed to be subject to	BOOK OF RE c (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to a t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
17-010 cont'd						Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground sewerage and water apparatus)
17-011	Acquisition of Rights	agricultural land (Vinegar Hill Farm, north of A1079, Beverley)	Riplingham Estates Limited Tranby Croft Tranby Lane Anlaby HULL East Riding of Yorkshire HU10 7EE	Mark Faulkingham West End Farm Yokefleet Goole East Riding Of Yorkshire DN14 7XY	DN14 7XY	Beazer Homes Doncaster Limited Persimmon House Fulford YORK North Yorkshire YO19 4FE (in respect of the rights of access reserved by the Conveyance dated 24 January 1962) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of the Agreement relating to the placing, maintaining, repairing, renewing and inspecting of overhead power lines dated 2 July 1931 and 9 July 1931)



	Land wh	ich is proposed to be subject	BOOK OF to: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to o ast Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
17-011 cont'd						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights relating to the placing, maintaining, repairing, renewing, inspecting, retaining and removal of poles and overhead and underground lines as contained in Deeds dated 2 August 1957 and 8 August 1961 and 15 March 1962)



	Land w	hich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
17-011 cont'd						Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of restrictive covenants and the rights granted by a Deed dated 19 March 1968. Also in respect of the rights of access to maintain a water main granted by a Deed dated 19 April 1995) Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
18-001	Acquisition of Rights	32638 square metres of agricultural land (north of A1079, Beverley)	Bishop Burton College York Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QG	NONE	Bishop Burton College York Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QG	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)





	Land w	which is proposed to be subject	BOOK ( to: (i) powers of compulsory acquisition	re Wind Farm Development Consent Order OF REFERENCE - PART 1 n, (ii) right to use the land, and/or (iii) rights to of East Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulatio	Category 1 on 7(1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-002	Acquisition of Rights	379 square metres of agricultural land (north of A1079, Beverley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)
				Pamela White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Pamela White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)
				The Executor Of The Estate Of The Late Andrew White Manor Farm Newland GOOLE DN14 7XD	The Executor Of The Estate Of The Late Andrew White Manor Farm Newland GOOLE DN14 7XD	National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)



	Land w	hich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1	ations: Prescribed Forms and Procedures) Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and
18-002 cont'd			or Reputed Owners	or Reputed Lessees or Tenants		Procedures) Regulations 2009 Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)
18-003	Acquisition of Rights	1483 square metres of public road and verges (A1079, Beverley) (excluding all interests of the Crown)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (in respect of subsoil beneath public highway) East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Secretary Of State For Transport Great Minster House 33 Horseferry Road Marsham LONDON SW1P 4DR (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Unknown



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii	/ind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Hans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-003 cont'd			Unknown			
18-004	Acquisition of Rights	1535 square metres of public road and verges (A1079, Beverley) (excluding all interests of the Crown)	Secretary Of State For Transport Great Minster House 33 Horseferry Road Marsham LONDON SW1P 4DR	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE
18-005	Temporary Possession	2307 square metres of public road (A1079, Beverley) (excluding all interests of the Crown)	Secretary Of State For Transport Great Minster House 33 Horseferry Road Marsham LONDON SW1P 4DR	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE
18-006	Freehold Acquisition	1836 square metres of public road and verge (A1079, Beverley) (excluding all interests of the Crown)	Secretary Of State For Transport Great Minster House 33 Horseferry Road Marsham LONDON SW1P 4DR	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE
18-007	Acquisition of Rights	51 square metres of agricultural land (west of A1079, Beverley)	Bishop Burton College York Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QG	NONE	Bishop Burton College York Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QG	NONE



	Land w	hich is proposed to be subject t	BOOK OI co: (i) powers of compulsory acquisition,	Wind Farm Development Consent Order F REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights t East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-008	Freehold Acquisition	26374 square metres of agricultural land (west of A1079, Beverley)	Bishop Burton College York Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QG	NONE	Bishop Burton College York Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8QG	NONE
18-009	Acquisition of Rights	6202 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ Pamela White	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ Pamela White	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Grid Electricity Transmission
				Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	PLC 1-3 Strand LONDON WC2N 5EH (in respect of overhead electricity apparatus)





	Land wi	nich is proposed to be subject	BOOK C to: (i) powers of compulsory acquisition	e Wind Farm Development Consent Order DF REFERENCE - PART 1 , (ii) right to use the land, and/or (iii) rights to f East Riding of Yorkshire	carry out protective works (Regulation 7(1	.)(a))
Number on	Extent of acquisition or use	Description of land		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-009 cont'd				The Executor Of The Estate Of The Late Andrew White Manor Farm Newland GOOLE DN14 7XD	The Executor Of The Estate Of The Late Andrew White Manor Farm Newland GOOLE DN14 7XD	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)



	Land w	hich is proposed to be subject to	BOOK C : (i) powers of compulsory acquisition	re Wind Farm Development Consent Order DF REFERENCE - PART 1 n, (ii) right to use the land, and/or (iii) rights to f East Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
18-010	Freehold Acquisition	scrubland, hedgerows, drains, pylon and public footpath (Walkington Footpath No.4)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public footpath (Walkington Footpath No.4))	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
				Pamela White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)	
				The Executor Of The Estate Of The Late Andrew White Manor Farm Newland GOOLE DN14 7XD	Pamela White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of overhead electricity apparatus)	



	Land wh	ich is proposed to be subject	BOOK OF to: (i) powers of compulsory acquisition, (i	Wind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to ast Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-010 cont'd					The Executor Of The Estate Of The Late Andrew White Manor Farm Newland GOOLE DN14 7XD	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)





	Land v	vhich is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to ist Riding of Yorkshire	carry out protective works (Regulation 7(1	.)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-010 cont'd						Procedures) Regulations 2009 Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021) Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water and sewerage apparatus)
18-010a	Acquisition of Rights	48523 square metres of agricultural land (The Risby Estate, south of A1079, Beverley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)



	Land wh	nich is proposed to be subject t	BOOK OF b o: (i) powers of compulsory acquisition, (i	Vind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to ast Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
18-010a cont'd				Pamela White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Pamela White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)	



	Land w	which is proposed to be subject t	BOOK O o: (i) powers of compulsory acquisition,	e Wind Farm Development Consent Order DF REFERENCE - PART 1 , (ii) right to use the land, and/or (iii) rights t f East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 n 7(1)(a) of the Infrastructure Planning (Appl Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-010a cont'd						Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water and sewerage apparatus)
18-011	Temporary Possession	27093 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)
				Pamela White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Pamela White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)





	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	/ind Farm Development Consent Order EFERENCE - PART 1 I right to use the land, and/or (iii) rights to	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	County of East Riding of Yorkshire Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-011 cont'd				The Executor Of The Estate Of The Late Andrew White Manor Farm Newland GOOLE DN14 7XD	The Executor Of The Estate Of The Late Andrew White Manor Farm Newland GOOLE DN14 7XD	National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)
18-012	Acquisition of Rights	10430 square metres of agricultural land, hedgerow, drain and public footpath (Walkington Footpath No.4) (The Risby Estate, west of A1079, Beverley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public footpath (Walkington Footpath No.4))	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)



	Land whi	ich is proposed to be subject	BOOK O to: (i) powers of compulsory acquisition,	e Wind Farm Development Consent Order IF REFERENCE - PART 1 , (ii) right to use the land, and/or (iii) rights to East Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-012 cont'd				Pamela White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ The Executor Of The Estate Of The Late	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ Pamela White	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Grid Electricity Transmission
				Andrew White Manor Farm Newland GOOLE DN14 7XD	Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 Jul 1969)
					The Executor Of The Estate Of The Late Andrew White Manor Farm Newland GOOLE DN14 7XD	National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)



	Land wh	ich is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ca ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applicat Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-012 cont'd						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)
						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)
						Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water and sewerage apparatus)



	Land w	/hich is proposed to be subject to	BOOK ( c (i) powers of compulsory acquisition	re Wind Farm Development Consent Order DF REFERENCE - PART 1 n, (ii) right to use the land, and/or (iii) rights to f East Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009		
20110110110			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-013	Temporary Possession	footpath (Walkington Footpath	Botanic House	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public footpath (Walkington Footpath No.9))	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)
				Pamela White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)
				The Executor Of The Estate Of The Late Andrew White Manor Farm Newland GOOLE DN14 7XD	Pamela White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)



	Land w	hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-013 cont'd					The Executor Of The Estate Of The Late Andrew White Manor Farm Newland GOOLE DN14 7XD	Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)
18-014	Freehold Acquisition	131236 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Robert Charles Elvidge West End Farm Bentley BEVERLEY HU17 8PP	HU17 8PP	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Keyland Developments Limited Western House Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 ?(1)(a) of the Infrastructure Planning (Applicat Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-014 cont'd						National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)			
						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)			
						Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)			
						Unknown (in respect of rights listed in a Deed dated 17 August 1993)			



			BOOK	re Wind Farm Development Consent Order DF REFERENCE - PART 1		
	Land w	hich is proposed to be subject to		n, (ii) right to use the land, and/or (iii) rights t f East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and		
18-014a	Acquisition of Rights	8619 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	or Reputed Owners Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	or Reputed Lessees or Tenants Robert Charles Elvidge West End Farm Bentley BEVERLEY HU17 8PP	or Reputed Occupiers Robert Charles Elvidge West End Farm Bentley BEVERLEY HU17 8PP	Procedures) Regulations 2009 Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)
						Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)
						National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	· · · ·	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-014a cont'd						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)
18-015	Acquisition of Rights		Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43-7FG (in respect of rights in regards to a- pipeline and valve as contained in a- Deed and Lease dated 22 May 2002)



	Land wh	ich is proposed to be subject t	BOOK OF co: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to c ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
18-015 cont'd						Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus) KCOM Group Limited 37 Carr Lane Hull East Yorkshire HUL 3RE (in respect of underground- telecommunication apparatus) Keyland Developments Limited Western House Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)	



	Land whi	ich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii	/ind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to ca st Riding of Yorkshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Applicati Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-015 cont'd						National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.) National Gas Transmission PLC National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of underground gas apparatus) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 Septembe 2021)





	Land w	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-015 cont'd						Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993) Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
18-015a	Freehold Acquisition	25825 square metres of agricultural land, scrubland and drain (The Risby Estate, west of A164, Bentley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
18-015a cont'd						Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus) KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)			



	Land wh	ich is proposed to be subject t	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to ca st Riding of Yorkshire	arry out protective works (Regulation 7(1	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	cions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-015a cont'd						National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.) National Gas Transmission PLC National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of underground gas apparatus) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)



	Land w	/hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	/ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-015a cont'd						Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993) Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
18-016	Acquisition of Rights	agricultural land and drain (The	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Robert Charles Elvidge West End Farm Bentley BEVERLEY HU17 8PP	Robert Charles Elvidge West End Farm Bentley BEVERLEY HU17 8PP	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-016 cont'd						Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)			





	Land w	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-016 cont'd						Unknown (as successor of Arthur Thomas Filmer Wilson Filmer) (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (as successor of Charles Henry Woodmansey) (in respect of rights listed in a Deed dated 17 August 1993)
18-017	Acquisition of Rights	agricultural land (The Risby	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-017 cont'd						Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of rights & easements listed in a Grant of Easement dated 3 April 2006.) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of underground gas apparatus)			



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
18-017 cont'd						National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)			
						Unknown (as successor of Arthur Thomas Filmer Wilson Filmer) (in respect of a right of way listed in a conveyance dated 12 August 1953. Als in respect of rights to lay a pipe to carr and supply water and entry listed in a Deed dated 1 December 1954) Unknown (as successor of Charles Henry Woodmansey) (in respect of rights listed in a Deed dated 17 August 1993)			



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))									
	County of East Riding of Yorkshire									
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applic: Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Fians			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
18-017 cont'd						Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)				
18-018	Freehold Acquisition	4 square metres of agricultural land (The Risby Estate, west of A164, Bentley)		Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG Robert Charles Elvidge West End Farm Bentley BEVERLEY HU17 8PP	Robert Charles Elvidge West End Farm Bentley BEVERLEY HU17 8PP	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Ineos Manufacturing (Hull) Limited) Keyland Developments Limited Western House Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)				



	Land w	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1)	)(a))
Number on	Extent of acquisition or use	Description of land		ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-018 cont'd						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of provisions contained in a Conveyance dated 3 September 1976) Unknown (in respect of rights listed in a Deed dated 17 August 1993)
18-019	Acquisition of Rights	417 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG	Robert Charles Elvidge West End Farm Bentley BEVERLEY HU17 8PP	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Ineos Manufacturing (Hull) Limited)



	Land wh	ich is proposed to be subject t	BOOK C o: (i) powers of compulsory acquisition	e Wind Farm Development Consent Order IF REFERENCE - PART 1 , (ii) right to use the land, and/or (iii) rights to ca East Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 7(1)(a) of the Infrastructure Planning (Applicati Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-019 cont'd				Robert Charles Elvidge West End Farm Bentley BEVERLEY HU17 8PP		Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)
						Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Als in respect of rights to lay a pipe to carr and supply water and entry listed in a Deed dated 1 December 1954)



			BOOK OF RI	ind Farm Development Consent Order EFERENCE - PART 1					
	Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009					
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-019 cont'd						Unknown (in respect of provisions contained in a Conveyance dated 3 September 1976) Unknown (in respect of rights listed in a Deed dated 17 August 1993)			
18-020	Acquisition of Rights	agricultural land (The Risby Estate, west of A164, Bentley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG	Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Ineos Manufacturing (Hull) Limited) Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)			



	Land whi	ich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii	/ind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to ca st Riding of Yorkshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Applicati Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-020 cont'd						National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of rights & easements liste in a Grant of Easement dated 3 April 2006.) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of underground gas apparatus) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 Septembe 2021)





	Land w	hich is proposed to be subject to	BOOK OF I : (i) powers of compulsory acquisition, (i	Vind Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	County of Ea Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-020 cont'd						Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of provisions contained in a Conveyance dated 3 September 1976) Unknown (in respect of rights listed in a Deed dated 17 August 1993)
18-021	Acquisition of Rights		Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Ineos Manufacturing (Hull) Limited) Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus)



	Land whi	ich is proposed to be subject t	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	'ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to ca st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Applicat Regulations 2009	cions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-021 cont'd						Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)
						Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Al in respect of rights to lay a pipe to car and supply water and entry listed in a Deed dated 1 December 1954)





	Land w	vhich is proposed to be subject to	BOOK OF : (i) powers of compulsory acquisition,	Wind Farm Development Consent Order REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights to East Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 7(1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-021 cont'd						Unknown (in respect of provisions contained in a Conveyance dated 3 September 1976)
						Unknown (in respect of rights listed in a Deed dated 17 August 1993)
						Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford
						West Yorkshire BD6 2SZ (in respect of underground water apparatus)
18- <b>021</b> a	Freehold Acquisition	87 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Ineos Manufacturing (Hull) Limited)
				Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG		Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus)



	Land wh	ich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to c t Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applicat Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-021a cont'd						Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021) Unknown (in respect of a right of way listed in a
						Conveyance dated 12 August 1953. in respect of rights to lay a pipe to c and supply water and entry listed in Deed dated 1 December 1954)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-021a cont'd						Unknown (in respect of provisions contained in a Conveyance dated 3 September 1976) Unknown (in respect of rights listed in a Deed dated 17 August 1993)			
18-022	Freehold Acquisition	37869 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Robert Charles Elvidge West End Farm Bentley BEVERLEY HU17 8PP	Bentley BEVERLEY HU17 8PP	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus)			



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-022 cont'd						Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of underground gas apparatus)			



	Land wh	ich is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ca ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 ?(1)(a) of the Infrastructure Planning (Applicat Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-022 cont'd						National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)
						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)
						Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)
						Unknown (in respect of rights listed in a Deed dated 17 August 1993)



				e Wind Farm Development Consent Order F REFERENCE - PART 1		
	Land w	hich is proposed to be subject t		(ii) right to use the land, and/or (iii) rights t East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation Owners or Reputed Owners	Category 1 7(1)(a) of the Infrastructure Planning (Appl Regulations 2009 Lessees or Tenants or Reputed Lessees or Tenants	ications: Prescribed Forms and Procedures) Occupiers or Reputed Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-022a	Acquisition of Rights	408 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Robert Charles Elvidge West End Farm Bentley BEVERLEY HU17 8PP	Robert Charles Elvidge West End Farm Bentley BEVERLEY HU17 8PP	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)



	Land w	which is proposed to be subject to	BOOK OF F : (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-022a cont'd						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Als in respect of rights to lay a pipe to car and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)
18-023	Acquisition of Rights	agricultural land and drain (The Risby Estate, west of A164, Bentley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Robert Charles Elvidge West End Farm Bentley BEVERLEY HU17 8PP	Robert Charles Elvidge West End Farm Bentley BEVERLEY HU17 8PP	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)



	Land wi	hich is proposed to be subject to	BOOK OF R b: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to c st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-023 cont'd						Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus) Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.)



	Land whi	ich is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to c t Riding of Yorkshire	arry out protective works (Regulation 7(1)	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Plannin
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-023 cont'd						National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of underground gas apparatus) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 Septembe 2021) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. A in respect of rights to lay a pipe to ca and supply water and entry listed in a Deed dated 1 December 1954)



	Land w	hich is proposed to be subject to	BOOK OF c (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights t East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-023 cont'd						Unknown (in respect of rights listed in a Deed dated 17 August 1993)
18-024	Acquisition of Rights	13813 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP		Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus)



	Land wh	ich is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to c ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-024 cont'd						Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of rights & easements listed in a Grant of Easement dated 3 April 2006.) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of underground gas apparatus)



	Land whi	ich is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ca ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applicat Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-024 cont'd						National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)
						Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed



	Land w	hich is proposed to be subject to	BOOK OF F : (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-025	Freehold Acquisition	26351 square metres of agricultural land and tree (The Risby Estate, west of A164, Bentley)	Albanwise Limited	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Incos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Incos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus) KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus)



	Land wh	ich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to ca st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 L)(a) of the Infrastructure Planning (Applicat Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-025 cont'd						Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September
						2021) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Al in respect of rights to lay a pipe to car and supply water and entry listed in a Deed dated 1 December 1954)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-025 cont'd						Unknown (in respect of rights listed in a Deed dated 17 August 1993) Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
18-025a	Acquisition of Rights	5052 square metres of agricultural land and tree (The Risby Estate, west of A164, Bentley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP		Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus)



	Land wh	ich is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to c t Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-025a cont'd						Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Westminster Bank PLC 250 Bishopsgate LONDON EC2IM 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 Septembe 2021) Unknown (in respect of a right of way listed in a
						Conveyance dated 12 August 1953. Al in respect of rights to lay a pipe to car and supply water and entry listed in a Deed dated 1 December 1954)





	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	'ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning	
Land Fians			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-025a cont'd						Unknown (in respect of rights listed in a Deed dated 17 August 1993) Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
<del>18 026</del>	Acquisition of Rights	<del>2 square metres of drain</del> <del>(south of A1079, Beverley)</del>	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	NONE	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity- apparatus)
18-027	Acquisition of Rights	23332 square metres of agricultural land, scrubland and copse (The Risby Estate, west of A164, Bentley and south of A1079, Beverley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)



	Land wh	nich is proposed to be subject t	BOOK C co: (i) powers of compulsory acquisition	e Wind Farm Development Consent Order DF REFERENCE - PART 1 , (ii) right to use the land, and/or (iii) rights to f East Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulatior	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-027 cont'd				Pamela White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ The Executor Of The Estate Of The Late Andrew White Manor Farm Newland GOOLE DN14 7XD	Pamela White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N SEH (in respect of overhead electricity apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N SEH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)



	Land whi	ch is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to ca st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-027 cont'd						National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) <del>Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity- apparatus)</del>	
						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 Septembe 2021)	
						Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water- apparatus)	



	Land w	hich is proposed to be subject to	BOOK OF : (i) powers of compulsory acquisition,	Wind Farm Development Consent Order REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights to East Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
18-027a	Temporary Possession	scrubland and drain (The Risby	100 Hills Road	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
				Pamela White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Pamela White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)	
				The Executor Of The Estate Of The Late Andrew White Manor Farm Newland GOOLE DN14 7XD	The Executor Of The Estate Of The Late Andrew White Manor Farm Newland GOOLE DN14 7XD	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)	



	Land whi	ich is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ca ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 7(1)(a) of the Infrastructure Planning (Applicat Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-027a cont'd						National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)



	Land w	hich is proposed to be subject to	BOOK OF : (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ast Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-028	Freehold Acquisition	3970 square metres of woodland (Bentley Moor Wood, west of A164, Bentley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)



	Land w	which is proposed to be subject to	BOOK OF : c: (i) powers of compulsory acquisition, (i	Vind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to ast Riding of Yorkshire	carry out protective works (Regulation 7(1	.)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-028 cont'd						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)
18-029	Acquisition of Rights	270 square metres of drain (The Risby Estate, west of A164, Bentley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Oliver White Butt Farm Victoria Road BEVERLEY East Riding of Yorkshire HU17 8PJ	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)



	Land wh	nich is proposed to be subject t	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to a st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))		
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
18-029 cont'd						Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)		



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to a t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-029 cont'd						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)
18-030	Freehold Acquisition		Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	NONE	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)
						Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)



	Land wh	ich is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ca ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 ?(1)(a) of the Infrastructure Planning (Applicat Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-030 cont'd						National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)
						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)
						Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)
						Unknown (in respect of rights listed in a Deed dated 17 August 1993)



	Land w	hich is proposed to be subject to	BOOK OF F c (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to ist Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-031	woodland and drain (Bent	woodland and drain (Bentley Moor Wood, west of A164, Bentley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE	Matthew Yeo Highfield Farm Walshford WETHERBY West Yorkshire LS22 5HT		Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)
				Moira Yeo 48 Blackburn Avenue BRIDLINGTON YO15 2ES	Moira Yeo 48 Blackburn Avenue BRIDLINGTON YO15 2ES	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)
				Stuart Yeo 48 Blackburn Avenue BRIDLINGTON YO15 2ES	Stuart Yeo 48 Blackburn Avenue BRIDLINGTON YO15 2ES	National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)





	Land w	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, i	Wind Farm Development Consent Order REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights to East Riding of Yorkshire	o carry out protective works (Regulation 7(1	.)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-031 cont'd						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)
18-032	Acquisition of Rights	17850 square metres of agricultural land, hedgerows and drain (The Risby Estate, west of A164, Bentley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Matthew Yeo Highfield Farm Walshford WETHERBY West Yorkshire LS22 5HT	Matthew Yeo Highfield Farm Walshford WETHERBY West Yorkshire LS22 5HT	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)



	Land whi	ich is proposed to be subject t	BOOK Of co: (i) powers of compulsory acquisition,	e Wind Farm Development Consent Order F REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights t East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	7(1)(a) of the Infrastructure Planning		
Lund Huns			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-032 cont'd				Moira Yeo 48 Blackburn Avenue BRIDLINGTON YO15 2ES	Moira Yeo 48 Blackburn Avenue BRIDLINGTON YO15 2ES	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)
				Stuart Yeo 48 Blackburn Avenue BRIDLINGTON YO15 2ES	Stuart Yeo 48 Blackburn Avenue BRIDLINGTON YO15 2ES	National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)
						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)
						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)



	Land w	hich is proposed to be subject to	BOOK OF RI : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Fidins			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-032 cont'd						Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)
<del>18-033</del>	Acquisition of Rights	<del>12 square metres of agricultural land (The Risby Estate, west of A164, Bentley)</del>	Al <del>banwise Limited</del> c <del>/o Mills &amp; Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH</del>	<del>Matthew Yeo</del> <del>Highfield Farm</del> <del>Walshford</del> <del>WETHERBY</del> <del>West Yorkshire</del> <del>LS22 5HT</del>	<del>Matthew Yeo</del> H <del>ighfield Farm</del> <del>Walshford</del> <del>WETHERBY</del> <del>West Yorkshire</del> <del>LS22 5HT</del>	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43-7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)
				<del>Moira Yeo</del> 4 <del>8 Blackburn Avenue</del> <del>BRIDLINGTON</del> <del>¥015-2ES</del>	<del>Moira Yeo</del> 4 <del>8 Blackburn Avenue</del> <del>BRIDLINGTON</del> <del>YO15 2ES</del>	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August- 2021)



	Land wh	ich is proposed to be subject t	BOOK ( o: (i) powers of compulsory acquisition	re Wind Farm Development Consent Order DF REFERENCE - PART 1 n, (ii) right to use the land, and/or (iii) rights t f East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulatio	Category 1 n 7(1)(a) of the Infrastructure Planning (Appl Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
<del>18-033-</del> cont'd				Stuart Yeo 48 Blackburn Avenue BRIDLINGTON ¥015 2ES	Stuart Yeo 48 Blackburn Avenue BRIDLINGTON YO15 2ES	National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September- 2021)
						Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed- dated 17 August 1993)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii	/ind Farm Development Consent Order IEFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009 Lessees or Tenants	ations: Prescribed Forms and Procedures) Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and
<del>18-034</del>	Acquisition of Rights	14-square metres of- agricultural land and- hedgerow (The Risby Estate, west of A164, Bentley)	or Reputed Owners Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBERIDGE Cambridgeshire CB2 1PH	or Reputed Lessees or Tenants Matthew Yeo Highfield Farm Walshford WETHERBY West Yorkshire LS22 5HT	or Reputed Occupiers Matthew Yeo Highfield Farm Walshford WETHERBY West Yorkshire LS22 5HT	Procedures) Regulations 2009 Incos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43-7FG (in respect of rights in regards to a
				<del>Moira Yeo</del> 4 <del>8 Blackburn Avenue</del> <del>BRIDLINGTON</del>	<del>Moira Yeo</del> 4 <del>8 Blackburn Avenue</del> <del>BRIDLINGTON</del>	pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Keyland Developments Limited Western House Western Way
				<del>Y015 2ES</del>	<del>¥015-2ES</del>	Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)
				<del>Stuart Yeo</del> 4 <del>8 Blackburn Avenue</del> <del>BRIDLINGTON</del> <del>YO15 2ES</del>	<del>Stuart Yeo</del> 4 <del>8 Blackburn Avenue</del> <del>BRIDLINGTON</del> <del>YO15-2ES</del>	<del>National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)</del>



Land w	which is proposed to be subject to	BOOK : (i) powers of compulsory acquisitio	OF REFERENCE - PART 1 n, (ii) right to use the land, and/or (iii) rights t	o carry out protective works (Regulation 7(1	)(a))
Extent of acquisition or use	Description of land			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
					Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the- Option Agreement dated 1 September 2021) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Als in respect of rights to lay a pipe to carr and supply water and entry listed in a Deed dated 1 December 1954) Unknown
Freehold Acquisition	scrubland (The Risby Estate,	c/o Mills & Reeve LLP	Matthew Yeo Highfield Farm Walshford WETHERBY West Yorkshire LS22 5HT	Matthew Yeo Highfield Farm Walshford WETHERBY West Yorkshire LS22 5HT	(in respect of rights listed in a Deed- dated 17 August 1993) Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a
	Extent of acquisition or use	Extent of acquisition or use       Description of land         reehold Acquisition       222 square metres of drain and scrubland (The Risby Estate,	Extent of acquisition or use Description of land Extent of acquisition or use Description of land Cualifying persons under Regulation Owners or Reputed Owners or Reputed Owners Prechold Acquisition 222 square metres of drain and Scrubland (The Risby Estate, west of A164, Bentley) Solution Hills Read Communication Solution Hills Read Communication Commit Regulation Communication Communication Description of land Communication Description of land Communication	County of East Riding of Yorkshire       Extent of acquisition or use     Description of land       Description of land     Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Appl Regulations 2009       Owners     Lessees or Tenants       or Reputed Owners     Lessees or Tenants       or Reputed Owners     or Reputed Lessees or Tenants       reehold Acquisition     222 square metres of drain and scrubland (The Risby Estate, west of A164, Bentley)     Albanwise Limited     Matthew Yeo       C/O Mills Read CAMBRIDGE     C/O Mills Read West Yorkshire     C/O Mills Read West Yorkshire     Matthew Yeo	BOOK OF REFERENCE - NART 1         Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1) County of Exa Ruling of Yorkshire         Extent of acquisition or use       Description of land       Category 1         Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures))       Regulations 2009         Owners       Lessees or Tenants       Occupiers         or Reputed Owners       or Reputed Lessees or Tenants       Or Reputed Occupiers         reehold Acquisition       Z22 square metres of drain and scrubing (Mathew Yeo       Mathew Yeo       Mathew Yeo         reehold Acquisition       Z22 square metres of drain and scrubing (Mathew Yeo       Mathew Yeo       Mathew Yeo         west of A164, Bentley)       Kollis & Reeve LLP       Highfield Farm       Highfield Farm       Highfield Farm         0Hills Road       WETHERBY       WETHERBY       WETHERBY       WETHERBY       WETHERBY



	Land whi	ich is proposed to be subject to	BOOK O c: (i) powers of compulsory acquisition,	e Wind Farm Development Consent Order F REFERENCE - PART 1 , (ii) right to use the land, and/or (iii) rights to East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 7(1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-035 cont'd			or Reputed Owners	Moira Yeo 48 Blackburn Avenue BRIDLINGTON YO15 2ES Stuart Yeo 48 Blackburn Avenue BRIDLINGTON YO15 2ES	Moira Yeo 48 Blackburn Avenue BRIDLINGTON YO15 2ES Stuart Yeo 48 Blackburn Avenue BRIDLINGTON YO15 2ES	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited
						5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Als in respect of rights to lay a pipe to carr and supply water and entry listed in a Deed dated 1 December 1954)



	Land w	/hich is proposed to be subject to	BOOK C : (i) powers of compulsory acquisition	e Wind Farm Development Consent Order DF REFERENCE - PART 1 , (ii) right to use the land, and/or (iii) rights t f East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulatior	Category 1 n 7(1)(a) of the Infrastructure Planning (Appl Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		Owners Lessees or Tenants Occupiers (Ap	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-035 cont'd						Unknown (in respect of rights listed in a Deed dated 17 August 1993)
<del>18-036</del>	Acquisition of Rights	8 square metres of drain and scrubland (The Risby Estate, west of A164, Bentley)	Albanwise Limited e/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	BEVERLEY East Riding of Yorkshire HU17-8PP	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43-7FG (in respect of rights in regards to a- pipeline and valve as contained in a- Deed and Lease dated 22 May 2002) Keyland Developments Limited Western House Western House Western Way Buttershaw BRADFORD West Yorkshire BD6-2SZ (as Beneficiary as contained in the Option Agreement dated 12 August- 2021) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited as)



	Land w	hich is proposed to be subject to	BOOK OF RI : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to blicing of Yarkshing	carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	· · · ·	County of East Riding of Yorkshire Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
<del>18-036-</del> <del>cont'd</del>						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the- Option Agreement dated 1 September- 2021) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed- dated 17 August 1993)	
<del>18-037</del>	Acquisition of Rights	<del>62 square metres of scrubland (The Risby Estate, west of</del> A <del>164, Bentley)</del>	Albanwise Limited <del>c/o Mills &amp; Reeve LLP</del> <del>Botanic House</del> <del>100 Hills Road</del> CAMBRIDGE Cambridgeshire CB2 1PH	Matthew Yeo Highfield Farm Walshford WETHERBY West Yorkshire LS22 5HT	<del>Matthew Yeo</del> H <del>ighfield Farm</del> <del>Walshford</del> <del>WETHERBY</del> <del>West Yorkshire</del> <del>LS22 5HT</del>	Inees Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43-7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	



	Land whi	ich is proposed to be subject t	BOOK ( o: (i) powers of compulsory acquisition	re Wind Farm Development Consent Order DF REFERENCE - PART 1 n, (ii) right to use the land, and/or (iii) rights t f East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 n 7(1)(a) of the Infrastructure Planning (Appl Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
<del>18 037</del> <del>cont'd</del>				<del>Moira Yeo</del> 4 <del>8 Blackburn Avenue</del> <del>BRIDLINGTON</del> <del>YO15 2ES</del>	<del>Moira Yeo</del> 4 <del>8 Blackburn Avenue</del> <del>BRIDLINGTON</del> <del>YO15 2ES</del>	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD <del>G 2SZ</del> (as Beneficiary as contained in the Option Agreement dated 12 August- 2021)
				<del>Stuart Yeo</del> 4 <del>8 Blackburn Avenue</del> <del>BRIDLINGTON</del> <del>YO15 2ES</del>	<del>Stuart Yeo</del> 4 <del>8 Blackburn Avenue</del> <del>BRIDLINGTON</del> <del>YO15-2ES</del>	National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)
						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)
						Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Al in respect of rights to lay a pipe to car and supply water and entry listed in a Deed dated 1 December 1954)



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	/ind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
<del>18-037</del> cont'd						<del>Unknown</del> <del>(in respect of rights listed in a Deed- dated 17 August 1993)</del>
18-038	Acquisition of Rights		Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	NONE	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)
18-039	Acquisition of Rights	road and verge (A164, Bentley)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus)
18-040	Acquisition of Rights		Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus)



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-040 cont'd			East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority and in respect of subsoil beneath public highway) Unknown			Unknown Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
18-041	Temporary Possession	49 square metres of public road and verge (A164, Bentley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (in respect of subsoil beneath public highway)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Unknown



	Land w	hich is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Lanu Pians			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-041 cont'd			East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority) Unknown			
18-042	Temporary Possession	166 square metres of agricultural land (The Risby Estate, east of A164, Bentley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Grid Electricity Transmission PLC 1-3 Strand LONDON
			East Riding of Yorkshire HU17 9BA		East Riding of Yorkshire HU17 9BA	WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)



	Land wh	ich is proposed to be subject t	BOOK OF to: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to c ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
18-042 cont'd						National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021) Unknown	
						(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)	
						Unknown (in respect of rights listed in a Deed dated 17 August 1993)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-043	Acquisition of Rights	agricultural land, access tracks, verges, hedgerows, copse, drain and public bridleway (Rowley Bridleway No.13) (The	Botanic House 100 Hills Road	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus)			
			East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA		East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)			
					East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public bridleway (Rowley Bridleway No.13))	National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.)			



	BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-043 cont'd						National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of underground gas apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 Ju 1969)			
						National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)			



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-043 cont'd						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)			
						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)			
						Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)			
						Unknown (in respect of rights listed in a Deed dated 17 August 1993)			



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on Land Plans	Extent of acquisition or use	Description of land	T	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-044	Temporary Possession	9168 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)			



	Land w	hich is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to Ist Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	l · · ·	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-044 cont'd						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)
18-045	Temporary Possession	2756 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)



Number on Land Plans	Extent of acquisition or use	Description of land		County of East Riding of Yorkshire Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
18-045 cont'd						National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 Ju 1969) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire							
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
18-045 cont'd						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)		
						Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)		
						Unknown (in respect of rights listed in a Deed dated 17 August 1993)		
18-046	Temporary Possession	27 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Ineos Manufacturing (Hull) Limited)		



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
18-046 cont'd				Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG		Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)			



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-046 cont'd						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of provisions contained in a Conveyance dated 3 September 1976)			
						Unknown (in respect of rights listed in a Deed dated 17 August 1993)			



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-047	Temporary Possession	No.13) (The Risby Estate, south of A1079, Beverley)	Botanic House 100 Hills Road	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)			
					East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public bridleway (Rowley Bridleway No.13))	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus) KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus)			



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-047 cont'd						Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)			



	Land whi	ich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-047 cont'd						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)	
						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021)	
						Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)	
						Unknown (in respect of rights listed in a Deed dated 17 August 1993)	



	Land w	hich is proposed to be subject to	BOOK OF : (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ast Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(ə))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-048	Temporary Possession		Albanwise Synergy Limited Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (in respect of a right of access to service contained in a Conveyance dated 22 May 1978. Also in respect of rights of access to maintain service media as contained in a Transfer dated 13 January 2021) Doggerbank Offshore Wind Farm Project
					County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public bridleway (Rowley Bridleway No.13))	1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 3 March 2020) Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of restrictive covenants contained in a Deed dated 3 March 2020)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 1)(a) of the Infrastructure Planning (Applicati Regulations 2009	ons: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulatior 7(1)(a) of the Infrastructure Plannin			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-048 cont'd						KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of proposed underground telecommunication apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to maintaining, repairing, renewing, inspecting or removing electric lines contained in a Deed dated 7 March 1 and 28 July 1969) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee to Albanwise Synergy Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary to Albanwise Synergy Limited dated 1 September 2021)			



	Land w	which is proposed to be subject t	BOOK O o: (i) powers of compulsory acquisition,	e Wind Farm Development Consent Order F REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights t East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 7(1)(a) of the Infrastructure Planning (Appl Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-049	Temporary Possession	2533 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus) Keyland Developments Limited Western House Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August
						2021) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)



	Land w	which is proposed to be subject to	BOOK OF F : (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-049 cont'd						National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021) Unknown (in respect of a right of way listed in a
						Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)
18-050	Acquisition of Rights	67 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Ineos Manufacturing (Hull) Limited)



	Land wh	ich is proposed to be subject to	BOOK O c: (i) powers of compulsory acquisition,	e Wind Farm Development Consent Order F REFERENCE - PART 1 . (ii) right to use the land, and/or (iii) rights to c East Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
18-050 cont'd				Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG		Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms an Procedures) Regulations 2009				
18-050 cont'd						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 Septemb 2021) The Executor Of The Estate Of The La Andrew White Manor Farm				
						Newland GOOLE DN14 7XD (in respect of rights to use service m and other rights and easements liste a Lease dated 8 November 2013)				
						Unknown (in respect of a right of way listed in Conveyance dated 12 August 1953. / in respect of rights to lay a pipe to ca and supply water and entry listed in Deed dated 1 December 1954)				
						Unknown (in respect of provisions contained ir Conveyance dated 3 September 197				



	Land w	which is proposed to be subject	BOOK to: (i) powers of compulsory acquisition	re Wind Farm Development Consent Order OF REFERENCE - PART 1 n, (ii) right to use the land, and/or (iii) rights t of East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulatio	Category 1 on 7(1)(a) of the Infrastructure Planning (Appl Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-050 cont'd						Unknown (in respect of rights listed in a Deed dated 17 August 1993)
18-051	Temporary Possession	21 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Ineos Manufacturing (Hull) Limited) Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)



	Land whi	ich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	iind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to ca st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applicati Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-051 cont'd						National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 Septembe 2021) The Executor Of The Estate Of The La Andrew White Manor Farm Newland GOOLE DN14 7XD (in respect of rights to use service me and other rights and easements listed a Lease dated 8 November 2013) Unknown
						(in respect of a right of way listed in a Conveyance dated 12 August 1953. A in respect of rights to lay a pipe to ca and supply water and entry listed in a Deed dated 1 December 1954)



	Land w	hich is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009		Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-051 cont'd						Unknown (in respect of provisions contained in a Conveyance dated 3 September 1976) Unknown (in respect of rights listed in a Deed dated 17 August 1993)
18-052		agricultural land, verge and drain (The Risby Estate, west of A1079, Beverley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	HU17 8PP	Joanne Dransfield Jillywood Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP (in respect of rights to pass along a track, lay telephone cables, electricity poles and erect poles to carry said cables; to use a water pipe, drainage and entry listed in a Conveyance dated 7 January 1994) Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)



Number on Land Plans	Extent of acquisition or use	Description of land		East Riding of Yorkshire Category 1 7(1)(a) of the Infrastructure Planning (Applications) Regulations 2009	ons: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-052 cont'd						National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 Ju 1969) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-052 cont'd						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 September 2021) Padero Solaer Ltd 2 Crossways Business Centre Bicester Road Kingswood Aylesbury Buckinghamshire HP18 0RA (as Beneficiary in respect of rights to construct) Paul Dransfield Jillywood Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP (in respect of rights to pass along a tra lay telephone cables, electricity poles and erect poles to carry said cables; to use a water pipe, drainage and entry listed in a Conveyance dated 7 January 1994)			



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order :FERENCE - PART 1 right to use the land, and/or (iii) rights to a t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
	and Plans		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-052 cont'd						Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)
18-053	Temporary Possession	Beverley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Bentley BEVERLEY	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus) Keyland Developments Limited Western House Western Way Buttershaw BRADFORD West Yorkshire BD6 2SZ (as Beneficiary as contained in the Option Agreement dated 12 August 2021)



	Land whi	ch is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii	/ind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to ca st Riding of Yorkshire	arry out protective works (Regulation 7(1)	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 1)(a) of the Infrastructure Planning (Applicati Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-053 cont'd						National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 Ju 1969)
						National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Albanwise Limited)
						Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in the Option Agreement dated 1 Septembe 2021)
						Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. A in respect of rights to lay a pipe to ca and supply water and entry listed in a Deed dated 1 December 1954)



	Land w	hich is proposed to be subject to	BOOK ( b: (i) powers of compulsory acquisition	re Wind Farm Development Consent Order OF REFERENCE - PART 1 n, (ii) right to use the land, and/or (iii) rights t of East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulatio	Category 1 n 7(1)(a) of the Infrastructure Planning (Appl Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-053 cont'd						Unknown (in respect of rights listed in a Deed dated 17 August 1993)
18-054	Acquisition of Rights	22194 square metres of agricultural land, copse, drain and public bridleway (Rowley Bridleway No.13) (Burn Park Farm, south of A1079, Beverley)	Albanwise Synergy Limited Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (in respect of a right of access to service contained in a Conveyance dated 22 May 1978. Also in respect of rights of access to maintain service media as contained in a Transfer dated 13 January 2021)
					East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public bridleway (Rowley Bridleway No.13))	BP Chemicals Limited Chertsey Road SUNBURY ON THAMES Middlesex TW16 7BP (in respect of rights to construct a pipeline contained in Agreements dated 22 May 1998 and 17 June 1998. Also in respect of rights to use a pipeline contained in a Deed dated 14 November 2000. Also in respect of rights to construct and maintain a pipeline contained in a Lease of Easements dated 16 May 2002)



	Land wi	nich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to a st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-054 cont'd						Creyke Beck Solar Limited 2 Crossways Business Centre Bicester Road Kingswood Aylesbury Buckinghamshire HP18 ORA (as Beneficiary to Albanwise Synergy Limited dated 11 August 2021)
						Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 3 March 2020)
						Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of restrictive covenants contained in a Deed dated 3 March 2020)



	Land wh	nich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to a st Riding of Yorkshire	carry out protective works (Regulation 7(1)	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-054 cont'd						Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access contained in a Lease dated 16 May 2002)
						National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of overhead electricity apparatus)
						National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right to maintaining, repairing, renewing, inspecting or removing electric lines contained in a Deed dated 7 March 1960 and 28 July 1969)



	Land wh	ich is proposed to be subject t	BOOK OF to: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights to ca East Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 7(1)(a) of the Infrastructure Planning (Applicat Regulations 2009	cions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infractructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-054 cont'd						Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and



	Land w	hich is proposed to be subject to	BOOK O : (i) powers of compulsory acquisition,	e Wind Farm Development Consent Order F REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights t East Riding of Yorkshire	o carry out protective works (Regulation 7(1	.)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation Owners	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 Owners Lessees or Tenants Occupiers			
			or Reputed Owners	or Reputed Lessees or Tenants	or Reputed Occupiers	Procedures) Regulations 2009	
18-055	Temporary Possession		Albanwise Synergy Limited Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (in respect of a right of access to service contained in a Conveyance dated 22 May 1978. Also in respect of rights of access to maintain service media as contained in a Transfer dated 13 January 2021)	
					East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public bridleway (Rowley Bridleway No.13))	BP Chemicals Limited Chertsey Road SUNBURY ON THAMES Middlesex TW16 7BP (in respect of rights to construct a pipeline contained in Agreements dated 22 May 1998 and 17 June 1998. Also in respect of rights to use a pipeline contained in a Deed dated 11 November 2000. Also in respect of rights to construct and maintain a pipeline contained in a Lease of Easements dated 16 May 2002)	



	Land wh	ich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to o st Riding of Yorkshire	arry out protective works (Regulation 7(1)	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-055 cont'd						Creyke Beck Solar Limited 2 Crossways Business Centre Bicester Road Kingswood Aylesbury Buckinghamshire HP18 ORA (as Beneficiary to Albanwise Synergy Limited dated 11 August 2021)
						Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 3 March 2020)
						Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of restrictive covenants contained in a Deed dated 3 March 2020)



	Land wh	ich is proposed to be subject t	BOOK OF co: (i) powers of compulsory acquisition, (i	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to c ast Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-055 cont'd						Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights contained in a Lease dated 16 May 2002) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right to maintaining, repairing, renewing, inspecting or removing electric lines contained in a Deed dated 7 March 1960 and 28 July 1969) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee to Albanwise Synergy Limited)



	Land w	which is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Lund Huns			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-055 cont'd						Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary to Albanwise Synergy Limited dated 1 September 2021)
19-001	Temporary Possession	1302 square metres of agricultural land and copse (Model Farm, south of A1079, Beverley)	Andrew James Martin White Model Farm Long Lane BEVERLEY East Riding of Yorkshire HU17 ORN	NONE	Andrew James Martin White Model Farm Long Lane BEVERLEY East Riding of Yorkshire HU17 ORN	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Andrew James Martin White) Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of rights contained in a document dated 23 May 2019)



	Land w	hich is proposed to be subject to	BOOK OF RI b: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners	ations: Prescribed Forms and Procedures) Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and	
			or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	or Reputed Occupiers	Procedures) Regulations 2009
19-001 cont'd						Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of rights contained in a document dated 23 May 2019)
19-002	Temporary Possession	5820 square metres of public road and verges (A1079, Beverley) (excluding all interests of the Crown)	Albanwise Synergy Limited Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE CB2 1PH (in respect of subsoil beneath public highway) Andrew James Martin White Model Farm Long Lane BEVERLEY East Riding of Yorkshire HU17 0RN (in respect of subsoil beneath public highway) East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (as highway authority)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Unknown



	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
19-002 cont'd			Unknown			
19-003	Acquisition of Rights	(south of A1079, Beverley)	Albanwise Synergy Limited Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE CB2 1PH (as reputed owner) Andrew James Martin White Model Farm Long Lane BEVERLEY East Riding of Yorkshire HU17 0RN (as reputed owner) Unknown	NONE	Albanwise Synergy Limited Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE CB2 1PH (as reputed owner) Andrew James Martin White Model Farm Long Lane BEVERLEY East Riding of Yorkshire HU17 ORN (as reputed owner) Unknown	Unknown
19-004	Acquisition of Rights	public bridleway (Woodmansey Bridleway No.30) (Model Farm, south of A1079, Beverley)	-	NONE	Model Farm Long Lane	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Andrew James Martin White)



	Land whi	ch is proposed to be subject t	BOOK OF F co: (i) powers of compulsory acquisition, (ii	Vind Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to ist Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
19-004 cont'd					East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public bridleway (Woodmansey Bridleway No.30))	Doggerbank Offshore Wind Farm Proje 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of rights contained in a document dated 23 Mar 2019 and a Lease dated 14 April 2022) Doggerbank Offshore Wind Farm Proje 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of rights contained in a document dated 23 Mar 2019) Doggerbank Offshore Wind Farm Proje 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of rights contained in a document dated 23 Mar 2019) Doggerbank Offshore Wind Farm Proje 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Dee of Grant and Transfer dated 15 April 2019. Also rights contained in an Optio Agreement dated 4 February 2020 and rights contained in a Lease dated 1 March 2022)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
19-005	Acquisition of Rights	hardstanding, scrubland,	WFAFP Limited High Warrendale Farm Warter York YO42 1XG	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public bridleway (Woodmansey Bridleway No.6)) WFAFP Limited High Warrendale Farm Warter York YO42 1XG	AFP Assets Limited 8th Floor 6 New Stree Square New Fetter Lane LONDON EC4A 3AQ (as mortgagee for WFAFP Limited) Albert John Evans Wanlass Farm Park Lane COTTINGHAM East Riding of Yorkshire HU16 5SB (in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010) Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of option agreements dated 6 June 2019, 19 September 2019, 2 March 2020 and 2 November 2021)			



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	ent of acquisition or use Description of land	Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
19-005 cont'd						Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of option agreements dated 6 June 2019, 19 September 2019, 2 March 2020 and 2 November 2021) Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access relating to laying and maintaining a pipeline contained in a Deed dated 11 Septembe 2011. Also in respect of rights and covenants listed in a Transfer dated 11 September 2002)			



	Land wh	ich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o st Riding of Yorkshire	carry out protective works (Regulation 7(1)	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-005 cont'd						Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus)	
						Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of easements and covenants relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September 2011)	
						Wind Energy One Limited 30 Finsbury Square LONDON EC2A 1AG (in respect of rights of access granted by a Lease dated 16 January 2017)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
19-006	Temporary Possession	agricultural land (Poplar Farm, south of A1079, Beverley)	WFAFP Limited	NONE	WFAFP Limited High Warrendale Farm Warter York YO42 1XG	AFP Assets Limited 8th Floor 6 New Stree Square New Fetter Lane LONDON EC4A 3AQ (as mortgagee for WFAFP Limited) Albert John Evans Wanlass Farm Park Lane COTTINGHAM East Riding of Yorkshire HU16 5SB (in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010) Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of option agreements dated 6 June 2019, 19 September 2019, 2 March 2020 and 2 November 2021)			



	Land wh	ich is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (	Wind Farm Development Consent Order REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights to ca East Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
19-006 cont'd						Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of option agreements dated 6 June 2019, 19 September 2019, 2 March 2020 and 2 November 2021) Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access relating to laying and maintaining a pipeline contained in a Deed dated 11 September 2011. Also in respect of rights and covenants listed in a Transfer dated 11 September 2002)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
19-006 cont'd						Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of easements and covenants relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September 2011)			



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on Land Plans	Extent of acquisition or use	Description of land		ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
19-007	Acquisition of Rights	13773 square metres of agricultural land, drains and access splay (Burn Park Farm, south of A1079, Beverley)	Albanwise Synergy Limited Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE CB2 1PH	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Andrew Woodmansey Manor Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (in respect of a right of access to service contained in a Conveyance dated 22 May 1978. Also in respect of rights of access to maintain service media as contained in a Transfer dated 13 January 2021) BP Chemicals Limited Chertsey Road SUNBURY ON THAMES Middlesex TW16 7BP (in respect of rights to construct a pipeline contained in Agreements dated 22 May 1998 and 17 June 1998. Also in respect of rights to use a pipeline contained in a Deed dated 11 November 2000. Also in respect of rights to construct and maintain a pipeline contained in a Lease of Easements dated 16 May 2002)			



	Land wi	nich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to a st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
19-007 cont'd						Creyke Beck Solar Limited 2 Crossways Business Centre Bicester Road Kingswood Aylesbury Buckinghamshire HP18 ORA (as Beneficiary as contained in a Unilateral Notice dated 11 August 2021)
						Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 3 March 2020)
						Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of restrictive covenants contained in a Deed dated 3 March 2020)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire							
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
19-007 cont'd			or Reputed Owners	or Reputed Lessees or Tenants		Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access contained in a Lease dated 16 May 2002) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of overhead electricity apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of overhead electricity apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH		
						(in respect of right to maintaining, repairing, renewing, inspecting or removing electric lines contained in Deeds dated 7 March 1960 and 28 July 1969)		



	Land w	hich is proposed to be subject to	BOOK OF RI : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
19-007 cont'd						National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee to Albanwise Synergy Limited) Orsted Hornsea Project Four Limited 5 Howick Place LONDON SW1P 1WG (as Beneficiary as contained in a Unilateral Notice dated 1 September 2021)
20-001	Temporary Possession	26 square metres of agricultural land (Poplar Farm, south of A1079, Beverley)	WFAFP Limited High Warrendale Farm Warter York YO42 1XG	NONE	York YO42 1XG	AFP Assets Limited 8th Floor 6 New Stree Square New Fetter Lane LONDON EC4A 3AQ (as mortgagee for WFAFP Limited) Albert John Evans Wanlass Farm Park Lane COTTINGHAM East Riding of Yorkshire HU16 5SB (in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010)



	Land wh	ich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to c st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
20-001 cont'd						Barclays Bank UK PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for INEOS Manufacturing (Hull) Limited) Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of option agreements dated 6 June 2019, 19 September 2019, 2 March 2020 and 2 November 2021) Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of option agreements dated 6 June 2019, 19 September 2019, 2 March 2020 and 2 November 2019, 2 March 2020 and 2 November 2019, 2 March 2020 and 2 November 2021)



	Land wh	nich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	'ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to c st Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
20-001 cont'd						Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access relating t laying and maintaining a pipeline contained in a Deed dated 11 Septemb 2011. Also in respect of rights and covenants listed in a Transfer dated 11 September 2002)	
						Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of easements and covenan relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September 2011) Unknown (in respect of rights listed in a Transfer dated 11 September 2002)	



	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order FFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))		
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Editor Flatio			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
20-002	Acquisition of Rights	and public bridleway (Woodmansey Bridleway No.6) (Poplar Farm, south of A1079, Beverley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public bridleway (Woodmansey Bridleway No.6))	AFP Assets Limited 8th Floor 6 New Stree Square New Fetter Lane LONDON EC4A 3AQ (as mortgagee for WFAFP Limited) Albert John Evans Wanlass Farm Park Lane COTTINGHAM		
			YO42 1XG		Hampshire SO43 7FG (being the subsoil between 0.610 metres and 9.144 metres below the surface) WFAFP Limited High Warrendale Farm Warter York YO42 1XG	East Riding of Yorkshire HU16 5SB (in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010) Barclays Bank UK PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for INEOS Manufacturing (Hull) Limited)		



	Land wh	ich is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to ca t Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applicat Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
20-002 cont'd						Doggerbank Offshore Wind Farm Proj 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of option agreements dated 6 June 2019, 19 September 2019, 2 March 2020 and 2 November 2021) Doggerbank Offshore Wind Farm Proj
						2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of option agreements dated 6 June 2019, 19 September 2019, 2 March 2020 and 2 November 2021)
						Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of easements and covenan relating to the laying and maintenanc of a pipeline listed in a Deed of Grant dated 21 September 2011)



	Land w	which is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to t Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
20-002 cont'd						Unknown (in respect of rights listed in a Transfer dated 11 September 2002) Wind Energy One Limited 30 Finsbury Square
						LONDON EC2A 1AG (in respect of rights of access granted by a Lease dated 16 January 2017)
20-003	Temporary Possession	public bridleway (Woodmansey	Warter	NONE	(in respect of public bridleway (Woodmansey Bridleway No.6)) WFAFP Limited High Warrendale Farm Warter York YO42 1XG	AFP Assets Limited 8th Floor 6 New Stree Square New Fetter Lane LONDON EC4A 3AQ (as mortgagee for WFAFP Limited) Albert John Evans Wanlass Farm Park Lane COTTINGHAM East Riding of Yorkshire HU16 5SB (in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010)



	Land wh	ich is proposed to be subject	BOOK OF to: (i) powers of compulsory acquisition,	Wind Farm Development Consent Order REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights to o Fast Riding of Yorkshire	carry out protective works (Regulation 7(1	)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
20-003 cont'd						Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of option agreements dated 6 June 2019, 19 September 2019, 2 March 2020 and 2 November 2021) Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of option agreements dated 6 June 2019, 19 September 2019, 2 March 2020 and 2 November 2021)	



	Land wh	ich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	'ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to ca st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 L)(a) of the Infrastructure Planning (Applicati Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
20-003 cont'd						Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access relating laying and maintaining a pipeline contained in a Deed dated 11 Septen 2011. Also in respect of rights and covenants listed in a Transfer dated 1 September 2002)
						Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of easements and covena relating to the laying and maintenand of a pipeline listed in a Deed of Grant dated 21 September 2011)
						Wind Energy One Limited 30 Finsbury Square LONDON EC2A 1AG (in respect of rights of access granted a Lease dated 16 January 2017)



	Land w	hich is proposed to be subject to	BOOK O c (i) powers of compulsory acquisition,	Wind Farm Development Consent Order FREFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights t East Riding of Yorkshire	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 7(1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
20-004	Acquisition of Rights	118007 square metres of agricultural land, access tracks, verges, hedgerows, electricity pylon, pond, drains, copse, scrubland, building and public bridleway (Woodmansey Bridleway No.6) (Poplar Farm, south of A1079, Beverley)	Warter York	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public bridleway (Woodmansey Bridleway No.6)) WFAFP Limited High Warrendale Farm Warter York YO42 1XG	AFP Assets Limited 8th Floor 6 New Stree Square New Fetter Lane LONDON EC4A 3AQ (as mortgagee for WFAFP Limited) Albert John Evans Wanlass Farm Park Lane COTTINGHAM East Riding of Yorkshire HU16 5SB (in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010) Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of option agreements dated 6 June 2019, 19 September 2019, 2 March 2020 and 2 November 2021)



	Land whi	ich is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to c st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applicat Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
20-004 cont'd						Doggerbank Offshore Wind Farm Proj 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of option agreements dated 6 June 2019, 19 September 2019, 2 March 2020 and 2 November 2021)
						Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access relating laying and maintaining a pipeline contained in a Deed dated 11 Septem 2011. Also in respect of rights and covenants listed in a Transfer dated 1 September 2002)
						KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus)



	Land wh	ich is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (i	Wind Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ca ast Riding of Yorkshire	arry out protective works (Regulation 7(1	.)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applicat Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
20-004 cont'd						National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of overhead electricity apparatus) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of easements and covenants relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September 2011) Wind Energy One Limited 30 Finsbury Square LONDON EC2A 1AG (in respect of rights of access granted by a Lease dated 16 January 2017)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of East Riding of Yorkshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
20-005	Temporary Possession	verges and public bridleway (Woodmansey Bridleway No.6)	WFAFP Limited High Warrendale Farm Warter	NONE	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of public bridleway (Woodmansey Bridleway No.6)) WFAFP Limited High Warrendale Farm Warter York YO42 1XG	AFP Assets Limited 8th Floor 6 New Stree Square New Fetter Lane LONDON EC4A 3AQ (as mortgagee for WFAFP Limited) Albert John Evans Wanlass Farm Park Lane COTTINGHAM East Riding of Yorkshire HU16 5SB (in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010) Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of option agreements dated 6 June 2019, 19 September 2019, 2 March 2020 and 2 November 2021)			



	Land whi	ch is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	'ind Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to ca st Riding of Yorkshire	arry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
20-005 cont'd						Doggerbank Offshore Wind Farm Proj 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (as Beneficiary in respect of option agreements dated 6 June 2019, 19 September 2019, 2 March 2020 and 2 November 2021)
						Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access relating laying and maintaining a pipeline contained in a Deed dated 11 Septemi 2011. Also in respect of rights and covenants listed in a Transfer dated 11 September 2002)
						KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus)



	Land wh	ich is proposed to be subject	BOOK OF I to: (i) powers of compulsory acquisition, (i	Wind Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to a ast Riding of Yorkshire	carry out protective works (Regulation 7(1)	)(a))
Number on	Category 1           Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures)           Extent of acquisition or use         Description of land					Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
20-005 cont'd						Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of easements and covenan relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September 2011)



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-001	2797 square metres of public road and verges (North Turnpike)	Unknown	
01-004	2586 square metres of agricultural land and access track (east of North Turnpike)	Unknown (in respect of rights, covenants and restrictions as contained in a Transfer dated 3 February 1977)	
01-006	3107 square metres of agricultural land and access track (east of North Turnpike)	Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 29 April 2019) Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 29 April 2019) Unknown (in respect of rights of access to maintain and repair service media dated 3 February 1977)	
01-007	1903 square metres of agricultural land and access track (east of North Turnpike)	Doggerbank Offshore Wind Farm Project 1 Projco LimitedNo.1 Forbury Place43 Forbury RoadReadingRG1 3JH(in respect of rights contained in a Deed of Grant dated 29 April 2019)Doggerbank Offshore Wind Farm Project 2 Projco LimitedNo.1 Forbury Place43 Forbury RoadReadingRG1 3JH(in respect of rights contained in a Deed of Grant dated 29 April 2019)Doggerbank Offshore Wind Farm Project 2 Projco LimitedNo.1 Forbury Place43 Forbury RoadReadingRG1 3JH(in respect of rights contained in a Deed of Grant dated 29 April 2019)Unknown(in respect of rights of access to maintain and repair service media dated 3 February 1977)	



Dogger Bank Sou	uth Offshore Wind Farm Development Consent Order	
BOOK OF REFERENCE - PART 2		
	County of East Riding of Yorkshire	
Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Description of Land	of the Planning Act 2008	
20 square metres of shore landward of the Mean High Water at Ulrome (east of North	Doggerbank Offshore Wind Farm Project 1 Projco Limited	
Turnpike)	No.1 Forbury Place	
	43 Forbury Road	
	Reading	
	RG1 3JH	
	(in respect of rights contained in a Deed of Grant dated 28 May 2019)	
92 square metres of shore landward of the Mean High Water at Ulrome (east of North	Clifford Noel Warkup	
Turnpike)	Marcliff	
	Beverley Road	
	Beeford	
	DRIFFIELD	
	East Riding of Yorkshire	
	Y025 8AD	
	(in respect of rights contained in a Lease dated 5 August 2005)	
	Doggerbank Offshore Wind Farm Project 1 Projco Limited	
	No.1 Forbury Place	
	43 Forbury Road	
	Reading	
	RG1 3JH	
	(in respect of rights contained in a Deed of Grant dated 28 May 2019)	
	Graham Clifford Warkup	
	Marcliff	
	Beverley Road	
	Beeford	
	DRIFFIELD	
	East Riding of Yorkshire	
	YO25 8AD	
	(in respect of rights contained in a Lease dated 5 August 2005)	
	Description of Land 20 square metres of shore landward of the Mean High Water at Ulrome (east of North Turnpike)	



	Dogger Bank So	uth Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)
Land Plans	Description of Land	of the Planning Act 2008
01-009		Martin Geoffrey Warkup
cont'd		Marcliff
		Beverley Road
		Beeford
		DRIFFIELD
		East Riding of Yorkshire
		YO25 8AD
		(in respect of rights contained in a Lease dated 5 August 2005)
		Neil Anthony Warkup
		Marcliff
		Beverley Road
		Beeford
		DRIFFIELD
		East Riding of Yorkshire
		YO25 8AD
		(in respect of rights contained in a Lease dated 5 August 2005)
		Unknown
		(in respect of rights of access)
01-010	30 square metres of foreshore seaward of the Mean High Water lying to the east of	Doggerbank Offshore Wind Farm Project 1 Projco Limited
	Ulrome (excluding all interests of the Crown)	No.1 Forbury Place
		43 Forbury Road
		Reading
		RG1 3JH
		(in respect of rights contained in a Deed of Grant dated 29 April 2019)
		Doggerbank Offshore Wind Farm Project 2 Projco Limited
		No.1 Forbury Place
		43 Forbury Road
		Reading
		RG1 3JH
		(in respect of rights contained in a Deed of Grant dated 29 April 2019)
		Unknown
		(in respect of rights of access)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
	356 square metres of foreshore seaward of the Mean High Water lying to the north east of		
	Ulrome (excluding all interests of the Crown)	Marcliff	
		Beverley Road	
		Beeford	
		DRIFFIELD	
		East Riding of Yorkshire	
		YO25 8AD	
		(in respect of rights contained in a Lease dated 5 August 2005)	
		Doggerbank Offshore Wind Farm Project 1 Projco Limited	
		No.1 Forbury Place	
		43 Forbury Road	
		Reading	
		RG1 3JH	
		(in respect of rights contained in a Deed of Grant dated 28 May 2019)	
		Graham Clifford Warkup	
		Marcliff	
		Beverley Road	
		Beeford	
		DRIFFIELD	
		East Riding of Yorkshire	
		Y025 8AD	
		(in respect of rights contained in a Lease dated 5 August 2005)	
		Martin Geoffrey Warkup	
		Marcliff	
		Beverley Road	
		Beeford	
		DRIFFIELD	
		East Riding of Yorkshire	
		Y025 8AD	
		(in respect of rights contained in a Lease dated 5 August 2005)	



	Dogger Bank Sout	th Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)
Land Plans		of the Planning Act 2008
01-011		Neil Anthony Warkup
cont'd		Marcliff
		Beverley Road
		Beeford
		DRIFFIELD
		East Riding of Yorkshire YO25 8AD
		(in respect of rights contained in a Lease dated 5 August 2005)
		(in respect of rights contained in a cease dated 5 August 2005)
		Unknown
		(in respect of rights of access)
00.005		
	3948 square metres of shore landward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	Unknown
02-013	599 square metres of shore landward of the Mean High Water lying to the east of Skipsea	Unknown
	(north of Hornsea Road)	
02-014	8 square metres of shore landward of the Mean High Water lying to the east of Skipsea	Unknown
02 014	(north of Hornsea Road)	
02-015	2144 square metres of shore landward of the Mean High Water lying to the east of Skipsea	Unknown
	(north of Hornsea Road)	(in respect of right of way contained in a Conveyance dated 4 June 1963)
02-016	185977 square metres of agricultural land and drains (east of Cleeton Lane and north of	Unknown
	Hornsea Road, B1242)	(in respect of rights of way contained in a Conveyance dated 4 June 1963)
02.047		
02-017	6728 square metres of agricultural land and drains (east of Cleeton Lane and north of	Unknown (in recreat of rights of way contained in a Conveyance dated 4 June 1962)
	Hornsea Road, B1242)	(in respect of rights of way contained in a Conveyance dated 4 June 1963)
02-019	319 square metres of agricultural land and drains (east of Cleeton Lane)	Unknown
02-026	3 square metres of public road and verges (Hornsea Road, B1242)	Unknown
02 020	o square metres of public road and verges (normsca Noad, D1242)	
02-027	1607 square metres of public road and verges (Hornsea Road, B1242)	Unknown
02-028	20 square metres of public road and verges (Hornsea Road, B1242)	Unknown



	Dogger Bank So	uth Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
02-029	1 square metres of verges (south of Hornsea Road, B1242)	Unknown
02-030	632 square metres of verges (south of Hornsea Road, B1242)	Unknown
02-033	2484 square metres of public road, verges and hedgerow (Hornsea Road, B1242)	Unknown
02-036	881 square metres of public road, access track and verges (Hornsea Road, B1242)	Unknown
02-037	570 square metres of agricultural land and access track (west of Hornsea Road, B1242)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of rights reserved by a Transfer dated 15 February 2012)
02-038	72661 square metres of agricultural land and public footpath (Skipsea Footpath No.6) (west of Hornsea Road, B1242)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of rights reserved by a Transfer dated 15 February 2012)
03-001	276 square metres of watercourse (Skipsea Drain) (east of Bewholme Lane)	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights) Unknown



	Dogger B	ank South Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
03-002	26356 square metres of agricultural land (east of Bewholme Lane)	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of rights of access)
03-004	417 square metres of public road and verges (Bewholme Lane)	Unknown
03-005	52 square metres of agricultural land (west of Bewholme Lane)	Unknown (in respect of restrictive covenants as may have been imposed before 11 May 1791) Unknown (in respect of rights of way, water support, drainage, light and other easements granted by a Conveyance dated 20 April 1988)
03-006	730 square metres of public road and verges (Bewholme Lane)	Unknown
03-008	590 square metres of public road and verges (Bewholme Lane)	Unknown
03-009	127 square metres of agricultural land (west of Bewholme Lane)	Unknown (in respect of restrictive covenants as may have been imposed before 11 May 1971) Unknown (in respect of rights of way, water support, drainage, light and other easements granted by a Conveyance dated 20 April 1988)
03-010	9561 square metres of agricultural land (west of Bewholme Lane)	Unknown (in respect of restrictive covenants as may have been imposed before 11 May 1971) Unknown (in respect of rights of way, water support, drainage, light and other easements granted by a Conveyance dated 20 April 1988)



	Dogger Bank So	outh Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
	33116 square metres of agricultural land and drain (Low Bonwick Farm, south of	Carr Hill Farm Limited
	Dunnington Lane)	Office F12
		Beverley Enterprise Centre
		Beck View Road
		BEVERLEY
		East Riding of Yorkshire
		HU17 0JT
		(in respect of rights of access to maintain, inspect and repair service media contained in a Transfer dated 30 November 2017)
03-013	26372 square metres of agricultural land (Manor Farm, south of Dunnington Lane)	Manor Farm Energy Limited
05-015		Malor Parin Energy Linited
		2 Station Avenue
		BRIDLINGTON
		East Riding of Yorkshire
		YO16 4LZ
		(in respect of rights of access and maintaining, inspecting and repairing conducting media contained in a Lease dated 5 September 2014)
04-001	76 square metres of access splay and hedgerow (Manor Farm, east of Skipsea Lane)	Manor Farm Energy Limited
		Medina House
		2 Station Avenue
		BRIDLINGTON
		East Riding of Yorkshire
		YO16 4LZ
		(in respect of rights of access and maintaining, inspecting and repairing conducting media contained in a Lease dated 5 September 2014)
04-002	523 square metres of public road and verges (Skipsea Lane)	Unknown
04-003	120 square metres of agricultural land (Manor Farm, west of Skipsea Lane)	Centrica PLC
		Millstream
		Maidenhead Road
		WINDSOR Berkshire
		SL4 5GD
		(in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in Deeds dated 11
		May 1972, 18 February 1981 and 26 March 1991)



04-006       561 square metres of public road and verges (Skipsea Lane)       Unknown         04-007       58 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millstream       Maidenhead Road         WINDSOR       Berkshire         St4 SGD       Sit A SGD         (In respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         May 1972, 18 February 1981 and 26 March 1991)         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millstream       Maidenhead Road         WINDSOR       Berkshire         St4 SGD       Sit A SGD         (In respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         Millstream       Maidenhead Road         WINDSOR       Berkshire         St4 SGD       Sit A SGD         (In respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         May 1972, 18 February 1981 and 26 March 1991)       O4-008         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board	Dogger Bank South Offshore Wind		uth Offshore Wind Farm Development Consent Order
Number on Land Plans         Description of Land         Potential claims under S10 Computisory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008           04-004         865 square metres of public road and verges (Skipsea Lane)         Unknown           04-005         78 square metres of hedgerow (cast of Skipsea Lane)         Manor Farm Energy Limited Medina House 2 Station Avenue BRD/LINGTON           04-006         561 square metres of public road and verges (Skipsea Lane)         Unknown           04-006         563 square metres of public road and verges (Skipsea Lane)         Unknown           04-007         58 square metres of public road and verges (Skipsea Lane)         Unknown           04-007         58 square metres of agricultural land (Manor Farm, west of Skipsea Lane)         Unknown           04-007         58 square metres of agricultural land (Manor Farm, west of Skipsea Lane)         Unknown           04-008         42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)         Unknown           04-008         42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)         Ministream Maidenhead Road WINDSOR           04-008         42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)         Ministream Maidenhead Road WINDSOR           04-008         42976 square metres of agricultural land (Manor Farm, west of Skipsea Lane)         Ministream Maidenhead Road WINDSOR	B		BOOK OF REFERENCE - PART 2
Land Plans       Of the Planning Act 2008         04-004       865 square metres of public road and verges (Skipsea Lane)       Unknown         04-005       78 square metres of hedgerow (east of Skipsea Lane)       Manor Farm Energy Limited Medina House 2 Station Avenue BRDLINGTOM East Riding of Yorkkine YO16 412 (in respect of rights of access and maintaining, inspecting and repairing media contained in a Lease dated 5 September 201 (in respect of rights of access and maintaining, inspecting and repairing media contained in a Lease dated 5 September 201 (in respect of rights of access and maintaining, inspecting and repairing media contained in a Lease dated 5 September 201 (in respect of rights to faccess and maintaining, inspecting and repairing media contained in a Lease dated 5 September 201 (in respect of rights to faccess and maintaining, inspect, alter, renew, replace and remove a main or pipe contained in Maidenbead Road WINOSOR         04-007       88 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC Milistream Maidenbead Road WINOSOR         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC Milistream Maidenbead Road WINOSOR         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC Milistream Maidenbead Road WINOSOR         04-008       42976 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC Milistream Maidenbead Road WINOSOR         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drai			County of East Riding of Yorkshire
Land Mans       of the Manning Act 2008         04-004       865 square metres of public road and verges (Skipsea Lane)       Unknown         04-005       78 square metres of hedgerow (east of Skipsea Lane)       Manor Farm Energy Umited Medina House         2 Station Avenue B RDUNGTON       2 Station Avenue B RDUNGTON         04-005       561 square metres of public road and verges (Skipsea Lane)       Unknown         04-006       561 square metres of public road and verges (Skipsea Lane)       Unknown         04-007       58 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Unknown         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC Milistream Maidenhead Road WINDSOR         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC Milistream Maidenhead Road WINDSOR         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC Milistream Maidenhead Road WINDSOR         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC Milistream Maidenhead Road WINDSOR         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC Milistream Maidenhead Road WINDSOR         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley &	nber on	Description of Land	
04-005       78 square metres of hedgerow (east of Skipsea Lane)       Medina House 2 Station Avenue BRIDLINGTON East Riding of Yorkshire Y016 412 (in respect of rights of access and maintaining, inspecting and repairing media contained in a Lease dated 5 September 201 (in respect of rights of access and maintaining, inspecting and repairing media contained in a Lease dated 5 September 201 (in respect of rights of access and maintaining, inspecting and repairing media contained in a Lease dated 5 September 201 (in respect of rights of access and maintaining, inspecting and repairing media contained in a Lease dated 5 September 201 (in respect of rights of access and maintaining, inspecting and repairing media contained in a Lease dated 5 September 201 (in respect of rights of access and maintaining, inspect, alter, renew, replace and remove a main or pipe contained in I Mailerheam Mailerheam Mailerheam St4 SGD (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I Militeream Mailerheam	d Plans	Description of Land	of the Planning Act 2008
Medina House       2 Station Avenue         BRIDELINGTON       East Ringing of Yorkshire         V016 4L2       (in respect of rights of access and maintaining, inspecting and repairing media contained in a Lease dated 5 September 201         04-006       561 square metres of public road and verges (Skipsea Lane)       Unknown         04-007       58 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millistream       Maidenhead Road       WiNDSOR         Berkshire       SL4 sego (Fights of access, and maintain), inspect, alter, renew, replace and remove a main or pipe contained in I         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millistream       Maidenhead Road       WiNDSOR         8erkshire       SL4 sego (fights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         Maidenhead Road       WiNDSOR         8erkshire       SL4 sego (fights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         Maidenhead Road       WiNDSOR         8erkshire       SL4 SeG0 (fights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         Maidenhead Road       WiNDSOR         8erkshire       SL4 SeG0 (fights to lay, construct, use, maintai	4-004 86	65 square metres of public road and verges (Skipsea Lane)	Unknown
Medina House       2 Station Avenue         BRIDELINGTON       East Ringing of Yorkshire         V016 4L2       (in respect of rights of access and maintaining, inspecting and repairing media contained in a Lease dated 5 September 201         04-006       561 square metres of public road and verges (Skipsea Lane)       Unknown         04-007       58 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millistream       Maidenhead Road       WiNDSOR         Berkshire       SL4 sego (Fights of access, and maintain), inspect, alter, renew, replace and remove a main or pipe contained in I         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millistream       Maidenhead Road       WiNDSOR         8erkshire       SL4 sego (fights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         Maidenhead Road       WiNDSOR         8erkshire       SL4 sego (fights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         Maidenhead Road       WiNDSOR         8erkshire       SL4 SeG0 (fights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         Maidenhead Road       WiNDSOR         8erkshire       SL4 SeG0 (fights to lay, construct, use, maintai			
2 Station Avenue       BRIDLINGTON         Bast Riding of Yorkshire       YO16 4LZ         (In respect of rights of access and maintaining, inspecting and repairing media contained in a Lease dated 5 September 201         04-006       561 square metres of public road and verges (5kipsea Lane)       Unknown         04-007       58 square metres of agricultural land (Manor Farm, west of 5kipsea Lane)       Centrica PLC         Millstream       Mailanebad Road         WINDSOR       Berkshire         SL4 5G0       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in May 1972, 18 February 1981 and 26 March 1991)         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millstream       Maidenhead Road         WINDSOR       Berkshire         SL4 5G0       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in May 1972, 18 February 1981 and 26 March 1991)         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Milistream       Maidenhead Road       WINSOR         Maidenhead Road       WINSOR       Berkshire         SL4 5G0       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contain	4-005 78	8 square metres of hedgerow (east of Skipsea Lane)	Manor Farm Energy Limited
BRIDLINGTON       Exist Riding of Yorkshire         O4-006       S61 square metres of public road and verges (Skipsea Lane)       Unknown         O4-007       S8 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millstream       Millstream         Mildenhead Road       WINDSOR         Berkshire       S14 square metres of agricultural land (Manor Farm, west of Skipsea Lane)         O4-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)         O4-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)         O4-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)         O4-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)         O4-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)         O4-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)         O4-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)         OH-009       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)         OB-009       Centrica PLC         Millstream       Millstream         Millstream       Millstream         Millstream       Millstream         Millstream			Medina House
East Riding of Yorkshire       YO16 4LZ         (In respect of rights of access and maintaining, inspecting and repairing media contained in a Lease dated 5 September 201         04-006       S61 square metres of public road and verges (Skipsea Lane)       Unknown         04-007       S8 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millstream       Maidenhead Road       WillSDSOR         Berkshire       SL4 SGD       In Krespect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I May 1972, 18 February 1981 and 26 March 1991)         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millstream       Maidenhead Road       Millstream         Maidenhead Road       WillSpSOR         Berkshire       SL4 SGD       Centrica PLC         Millstream       Maidenhead Road       WillSpSOR         Berkshire       SL4 SGD       Gentrica PLC         Millstream       Maidenhead Road       WillSpSOR         Berkshire       SL4 SGD       Gentrica PLC         Millstream       Maidenhead Road       WillSpSOR         Berkshire       SL4 SGD       Gentrica PLC         Millstream       Maidenhead Road       Gentrica PLC <td< td=""><td></td><td></td><td>2 Station Avenue</td></td<>			2 Station Avenue
04-006       S61 square metres of public road and verges (Skipsea Lane)       Unknown         04-007       S88 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Milistream       Milistream         Milisteream       Milistream         Milisteream       Milisteream         Milisteream       Milisteream <t< td=""><td></td><td></td><td>BRIDLINGTON</td></t<>			BRIDLINGTON
04-006       S61 square metres of public road and verges (Skipsea Lane)       Unknown         04-007       S8 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millstream       Maidenhead Road       WINDSOR         Berkshire       SL4 SGD       in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millstream       Maidenhead Road       WiNDSOR         Berkshire       SL4 SGD       Centrica PLC         Millstream       Maidenhead Road       WiNDSOR         Berkshire       SL4 SGD       Centrica PLC         Millstream       Maidenhead Road       WiNDSOR         Berkshire       SL4 SGD       Centrica PLC         Millstream       Maidenhead Road       WiNDSOR         Berkshire       SL4 SGD       Centrica PLC         Millstream       Maidenhead Road       WiNDSOR         Berkshire       SL4 SGD       Centrica PLC         Millstream       Maidenhead Road       WiNDSOR         Berkshire       SL4 SGD       Centrica PLC         Millstream       Maidenhead Road       WiNDSOR			East Riding of Yorkshire
04-006       561 square metres of public road and verges (Skipsea Lane)       Unknown         04-007       58 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millstream       Maidenhead Road         WINDSOR       Berkshire         St4 SGD       Sit A SGD         (In respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         May 1972, 18 February 1981 and 26 March 1991)         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millstream       Maidenhead Road         WINDSOR       Berkshire         St4 SGD       Sit A SGD         (In respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         Millstream       Maidenhead Road         WINDSOR       Berkshire         St4 SGD       Sit A SGD         (In respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         May 1972, 18 February 1981 and 26 March 1991)       O4-008         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board			YO16 4LZ
04-007       58 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Milistream       Maidenhead Road         WINDSOR       Berkshire         SL4 SGD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Willstream       Maidenhead Road         Willstream       Maidenhead Road         Willstream       Maidenhead Road         May 1972, 18 February 1981 and 26 March 1991)       Centrica PLC         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Willstream       Maidenhead Road       WiNDSOR         Berkshire       SL4 SGD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         May 1972, 18 February 1981 and 26 March 1991)       Nay 1972, 18 February 1981 and 26 March 1991)         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board			(in respect of rights of access and maintaining, inspecting and repairing media contained in a Lease dated 5 September 2014)
04-007       58 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Milistream       Maidenhead Road         WINDSOR       Berkshire         SL4 SGD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Milistream       Maidenhead Road         WINDSOR       Berkshire         SL4 SGD       Centrica PLC         Milistream       Maidenhead Road         WINDSOR       Berkshire         SL5 GD       Centrica PLC         Milistream       Maidenhead Road         WiNDSOR       Berkshire         SL4 SGD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         Maidenhead Road       WINDSOR         Berkshire       SL4 SGD         (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         May 1972, 18 February 1981 and 26 March 1991)         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board			
04-007       58 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Milistream       Maidenhead Road         WINDSOR       Berkshire         SL4 SGD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Willstream       Maidenhead Road         Willstream       Maidenhead Road         Willstream       Maidenhead Road         May 1972, 18 February 1981 and 26 March 1991)       Centrica PLC         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Willstream       Maidenhead Road       WiNDSOR         Berkshire       SL4 SGD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         May 1972, 18 February 1981 and 26 March 1991)       Nay 1972, 18 February 1981 and 26 March 1991)         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board			
Millstream       Maidenhead Road         WINDSOR       Berkshire         SL4 SGD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millstream       Maidenhead Road         WINDSOR       Berkshire         SL4 SGD       Centrica PLC         Millstream       Maidenhead Road         WINDSOR       Berkshire         SL4 SGD       Centrica PLC         Millstream       Maidenhead Road         WINDSOR       Berkshire         SL4 SGD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         Maidenhead Road       WINDSOR         Berkshire       SL4 SGD         (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         May 1972, 18 February 1981 and 26 March 1991)         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board	4-006 56	61 square metres of public road and verges (Skipsea Lane)	Unknown
Millstream       Maidenhead Road         WINDSOR       Berkshire         SL4 SGD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millstream       Maidenhead Road         WINDSOR       Berkshire         SL4 SGD       Centrica PLC         Millstream       Maidenhead Road         WINDSOR       Berkshire         SL4 SGD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         Millstream       Maidenhead Road         WINDSOR       Berkshire         SL4 SGD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         May 1972, 18 February 1981 and 26 March 1991)       Nay 1972, 18 February 1981 and 26 March 1991)         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board			
04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         04-008       4200 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         04-008       Maidenhead Road       Maidenhead Road         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board	4-007 58	8 square metres of agricultural land (Manor Farm, west of Skipsea Lane)	Centrica PLC
WINDSOR         Berkshire         SL4 5GD         (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millstream       Maidenhead Road         WINDSOR       Berkshire         SL4 5GD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         Millstream       Maidenhead Road         WINDSOR       Berkshire         SL4 5GD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         May 1972, 18 February 1981 and 26 March 1991)       SL4 5GD         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board			Millstream
Berkshire         SL4 5GD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millstream       Maidenhead Road         WINDSOR       Berkshire         SL4 5GD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board			Maidenhead Road
SL4 5GD (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I May 1972, 18 February 1981 and 26 March 1991)         04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I May 1972, 18 February 1981 and 26 March 1991)         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board			WINDSOR
04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millstream       Maidenhead Road         WINDSOR       Berkshire         SL4 5GD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board			Berkshire
04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in l May 1972, 18 February 1981 and 26 March 1991)         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board			SL4 5GD
04-008       42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)       Centrica PLC         Millstream       Maidenhead Road         WINDSOR       Berkshire         SL4 5GD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board			(in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in Deeds dated 11
04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in l May 1972, 18 February 1981 and 26 March 1991)         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board			May 1972, 18 February 1981 and 26 March 1991)
04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in l May 1972, 18 February 1981 and 26 March 1991)         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board			
04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Maidenhead Road         WINDSOR       Berkshire         SL4 5GD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         Maidenhead Road       May 1972, 18 February 1981 and 26 March 1991)	4-008 42	2776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)	
WINDSOR         Berkshire         SL4 5GD         (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         May 1972, 18 February 1981 and 26 March 1991)         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board			
04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Berkshire         Berkshire       SL4 5GD         SL4 5GD       (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I         May 1972, 18 February 1981 and 26 March 1991)       Beverley & North Holderness Internal Drainage Board			
04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       SL4 5GD (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in I May 1972, 18 February 1981 and 26 March 1991)         04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board			
04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board			
04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       May 1972, 18 February 1981 and 26 March 1991)         Beverley & North Holderness Internal Drainage Board			
04-009       490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)       Beverley & North Holderness Internal Drainage Board			
			May 1972, 18 February 1981 and 26 March 1991)
	1-009 49	90 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)	Beverley & North Holderness Internal Drainage Board
Derwent House			Derwent House
Crockey Hill			
York			
Y019 4SR			YO19 4SR
(in respect of riparian rights)			



	Dogger Bank S	South Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)
Land Plans	Description of Land	of the Planning Act 2008
04-009		Unknown
cont'd		
04-010	42868 square metres of agricultural land and drain (east of Dunnington Lane)	Beverley & North Holderness Internal Drainage Board
04-010	42000 square metres of agricultural and and drain (east of Dunnington Lane)	Derwent House
		Crockey Hill
		York
		YO19 4SR
		(in respect of rights of access)
		Centrica PLC
		Millstream
		Maidenhead Road
		WINDSOR
		Berkshire
		SL4 5GD
		(in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in a Deed of Grant
		dated 25 June 1971)
04-011	2103 square metres of agricultural land (east of Dunnington Lane)	Centrica PLC
		Millstream
		Maidenhead Road
		WINDSOR
		Berkshire
		SL4 5GD
		(in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in a Deed of Grant
		dated 25 June 1971)
04-012	70 servers matters of multiplicated and verses (Dunnington Long)	Unknown
04-012	79 square metres of public road and verge (Dunnington Lane)	Unknown
04-013	858 square metres of public road and verge (Dunnington Lane)	Unknown
04-014	125 square metres of verge (west of Dunnington Lane)	Unknown
04-014	122 square metres of verge (west of Dunnington Lane)	
04-015	2432 square metres of public road and verge (Dunnington Lane)	Unknown



	Dogger Bank Sou	ith Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
04-016	305 square metres of verge (west of Dunnington Lane)	Unknown
	18022 square metres of public road and verges (Dunnington Lane and Beverley Road, A165)	Unknown
	27263 square metres of agricultural land and drain (Moor Grange Farm, south of Dunnington Lane)	Gregory Peter Atkin Moor Grange Beverley Road Beeford Driffield East Riding of Yorkshire YO25 8AE (in respect of rights of drainage and access to the retained land contained in a Transfer dated 14 September 2010) Lucy Sheila Elizabeth Atkin Moor Grange Cottage Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AE (in respect of rights of drainage and access to the retained land contained in a Transfer dated 14 September 2010)
	45117 square metres of agricultural land and hedgerow (Moor Grange Farm, east of Moor Grange Access Road, west of Billings Lane)	Alexander Douglas Robinson Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG (in respect of rights to maintain a ditch contained in a Conveyance dated 1 August 1984)



	Dogger Bank Sou	th Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
05-001		Gregory Peter Atkin
cont'd		Moor Grange
		Beverley Road
		Beeford
		Driffield
		East Riding of Yorkshire
		Y025 8AE
		(in respect of rights of way contained in a Transfer dated 10 May 1988 and in respect of rights of drainage and access to the retained land
		contained in a Transfer dated 14 September 2010)
		Lucy Sheila Elizabeth Atkin
		Moor Grange Cottage
		Beverley Road
		Beeford
		DRIFFIELD
		East Riding of Yorkshire
		Y025 8AE
		(in respect of rights of way contained in a Transfer dated 10 May 1988 and in respect of rights of drainage and access to the retained land
		contained in a Transfer dated 14 September 2010)
		R & CM Rivis Limited
		Beachcomber Lodge
		56 South Marine Drive
		BRIDLINGTON
		Y015 3JN
		(in respect of rights of way and paying a proportion of costs to repair, maintain, clean, dredging and renewing drains contained in a
		Transfer dated 14 September 2010)
	6271 square metres of agricultural land and hedgerow (Moor Grange Farm, east of Moor	Alexander Douglas Robinson
	Grange Access Road, west of Billings Lane)	Happy Land Farm
		Dringhoe
		DRIFFIELD
		East Riding of Yorkshire
		Y025 8AG
		(in respect of rights to maintain a ditch contained in a Conveyance dated 1 August 1984)



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2	
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans	Description of Land	of the Planning Act 2008	
05-002		Gregory Peter Atkin	
cont'd		Moor Grange	
		Beverley Road	
		Beeford	
		Driffield	
		East Riding of Yorkshire	
		Y025 8AE	
		(in respect of rights of way contained in a Transfer dated 10 May 1988 and in respect of rights of drainage and access to the retained land	
		contained in a Transfer dated 14 September 2010)	
		Lucy Sheila Elizabeth Atkin	
		Moor Grange Cottage	
		Beverley Road	
		Beeford	
		DRIFFIELD	
		East Riding of Yorkshire	
		Y025 8AE	
		(in respect of rights of way contained in a Transfer dated 10 May 1988 and in respect of rights of drainage and access to the retained land	
		contained in a Transfer dated 14 September 2010)	
		R & CM Rivis Limited	
		Beachcomber Lodge	
		56 South Marine Drive	
		BRIDLINGTON	
		Y015 3JN	
		(in respect of rights of way and paying a proportion of costs to repair, maintain, clean, dredging and renewing drains contained in a	
		Transfer dated 14 September 2010)	
05-003	6575 square metres of agricultural land and drain (west of Billings Lane)	Unknown	
	69585 square metres of agricultural land and hedgerow (east of New Road) (excluding all	Julia Alexa Clover Todd	
	interests of the Crown)	Nunkeeling Priory	
		Nunkeeling	
		DRIFFIELD	
		East Riding of Yorkshire	
		Y025 8EH	
		(in respect of rights to maintain the dwellinghouse, drains and septic tank stated in the Conveyance dated 14 January 1966)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2	
	1	County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-004		Northern Powergrid (Yorkshire) PLC	
cont'd		Lloyds Court	
		78 Grey Street	
		Newcastle Upon Tyne	
		NE1 6AF	
		(in respect of wayleave consents dated 28 September 1967, 13 May 1969 and 28 November 1983)	
		Nunkeeling Renewables Limited	
		Hall Garth Manor Farm	
		Nunkeeling	
		Driffield	
		East Yorkshire	
		Y025 8EH	
		(in respect of rights contained in a Lease dated 1 May 2012 and 3 December 2012)	
		Secretary Of State For Defence	
		Main Building	
		Horse Guards Avenue	
		Whitehall	
		LONDON	
		SW1A 2HB	
		(in respect of restrictive covenants and rights listed in a Deed dated 30 May 1949)	
		Unknown	
		(in respect of rights granted and terms listed in a Conveyance dated 18 March 1987)	
06-001	10450 square metres of agricultural land and watercourse (Nunkeeling Drain, west of Main		
	Road)	(in respect of rights reserved by a Conveyance dated 28 October 1965)	
06-003	519 square metres of public road and verge (Main Road)	Unknown	
06-004	891 square metres of public road and verge (Main Road)	Unknown	
06-005	30 square metres of verge (east of Main Road)	Unknown	



	Dogger Bank So	outh Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
06-006	136 square metres of agricultural land and verge (east of Main Road)	Foster Farms Limited Willow Croft Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1 February 1980 and 17 July 1973) Laurazena Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of easements as contained in a Transfer dated 4 May 2023) Roy Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of easements as contained in a Transfer dated 4 May 2023)
06-007	55074 square metres of agricultural land (north of Catfoss Road, east of Main Road)	Foster Farms Limited Willow Croft Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1 February 1980 and 17 July 1973)



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
06-007 cont'd		Laurazena Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of easements as contained in a Transfer dated 4 May 2023) Roy Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of easements as contained in a Transfer dated 4 May 2023)	
06-008	2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)	Foster Farms Limited Willow Croft Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 SPR (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1 February 1980 and 17 July 1973) Laurazena Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of easements as contained in a Transfer dated 4 May 2023)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans		of the Planning Act 2008	
06-008		Roy Thompson	
cont'd		Haverham Farm	
		Hempholme	
		DRIFFIELD	
		East Riding of Yorkshire	
		YO25 8NB	
		(in respect of easements as contained in a Transfer dated 4 May 2023)	
06-009	16 square metres of hedgerow and drain (north of Catfoss Road)	Centrica PLC	
		Millstream	
		Maidenhead Road	
		WINDSOR	
		Berkshire	
		SL4 5GD	
		(in respect of the rights granted by the Deeds dated 27 June 1972 and 10 April 1975)	
		(in respect of the higher pranted by the beens dated 27 June 2072 and 2070) 1975)	
		Nunkeeling Renewables Limited	
		Hall Garth Manor Farm	
		Nunkeeling	
		Driffield	
		East Yorkshire	
		YO25 8EH	
		(in respect of the rights reserved by the Transfer dated 9 October 2013)	
		(in respect of the rights reserved by the mansier dated 5 october 2015)	
06-010	1004 square metres of public road and verge (Catfoss Road)	Unknown	
06-011	4 square metres of verge (south of Catfoss Road)	Unknown	
06-012	1012 square metres of public road and verge (Catfoss Road)	Unknown	



	Dogger Bank	South Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on	Description of Lond	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)
Land Plans	Description of Land	of the Planning Act 2008
06-013	43 square metres of verge (north of Catfoss Road)	Foster Farms Limited
		Willow Croft Farm
		Riston Road
		Catwick
		BEVERLEY
		East Riding of Yorkshire
		HU17 5PR
		(in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1
		February 1980 and 17 July 1973)
		Laurazena Thompson
		Haverham Farm
		Hempholme
		DRIFFIELD
		East Riding of Yorkshire
		YO25 8NB
		(in respect of easements as contained in a Transfer dated 4 May 2023)
		Roy Thompson
		Haverham Farm
		Hempholme
		DRIFFIELD
		East Riding of Yorkshire
		YO25 8NB
		(in respect of easements as contained in a Transfer dated 4 May 2023)
06-014	9777 square metres of public road and verge (Catfoss Road and Catfoss Lane)	Unknown
06-016	8429 square metres of agricultural land and access track (south of Catfoss Road)	Centrica PLC
		Millstream
		Maidenhead Road
		WINDSOR
		Berkshire
		SL4 5GD
		(in respect of the right to lay, maintain, inspect and repair and in respect to easement stated in the Deed dated 27 June 1972)



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
06-017	31268 square metres of agricultural land and drain (south of Catfoss Road)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the right to lay, maintain, inspect and repair and in respect to easement stated in the Deed dated 27 June 1972)	
06-018	30325 square metres of agricultural land and drain (north of Harsell Lane, excluding all interests of the Crown)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect of rights contained in Deeds dated 19 April 1972 and 6 February 1973) Peter James Boynton Hind House Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HULL East Riding of Yorkshire HULL East Riding of Yorkshire HULL Secret of rights reserved by a Conveyance dated 24 April 1981) Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of covenants and grants contained in a Deed dated 11 November 1948) Unknown (in respect to rights reserved by a Conveyance dated 1 October 1965)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans	Description of Land	of the Planning Act 2008	
06-019	136 square metres of verge (north of Harsell Lane, excluding all interests of the Crown)	Centrica PLC	
		Millstream	
		Maidenhead Road	
		WINDSOR	
		Berkshire	
		SL4 5GD	
		(in respect of rights contained in Deeds dated 19 April 1972 and 6 February 1973)	
		Colin Stuart Hazell	
		Astral House	
		Catfoss	
		HULL	
		East Riding of Yorkshire	
		HU11 5SP	
		(in respect of rights of access as contained in a Transfer dated 16 December 2010)	
		Peter James Boynton	
		Hind House	
		Catfoss Lane	
		Sigglesthorne	
		HULL	
		East Riding of Yorkshire	
		HU11 5QN	
		(as The Executor of the Estate of the Late Patricia Anne Boynton)	
		(in respect of rights reserved by a Conveyance dated 24 April 1981)	
		Secretary Of State For Defence	
		Main Building	
		Horse Guards Avenue	
		Whitehall	
		LONDON	
		SW1A 2HB	
		(in respect of covenants and grants contained in a Deed dated 11 November 1948)	
		Unknown	
		(in respect to rights reserved by a Conveyance dated 1 October 1965)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
	374 square metres of public road and verge (Harsell Lane, excluding all interests of the	Centrica PLC	
00-020	Crown)	Millstream	
	crowny	Maidenhead Road	
		WINDSOR	
		Berkshire	
		SL4 5GD	
		(in respect of rights contained in Deeds dated 19 April 1972 and 6 February 1973)	
		Colin Stuart Hazell	
		Astral House	
		Catfoss	
		HULL	
		East Riding of Yorkshire	
		HU11 5SP	
		(in respect of rights listed in a Transfer dated 16 December 2010)	
		Peter James Boynton	
		Hind House	
		Catfoss Lane Sigglesthorne	
		HULL	
		East Riding of Yorkshire	
		HU11 5QN	
		(as The Executor of the Estate of the Late Patricia Anne Boynton)	
		(in respect of rights reserved by a Conveyance dated 24 April 1981)	
		Secretary Of State For Defence	
		Main Building	
		Horse Guards Avenue	
1		Whitehall	
		LONDON	
		SW1A 2HB	
		(in respect of covenants and grants contained in a Deed dated 11 November 1948)	
		Unknown	
		(in respect to rights reserved by a Conveyance dated 1 October 1965)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans		of the Planning Act 2008	
06-021	472 square metres of public road (Harsell Lane, exlucing all interests of the Crown)	Centrica PLC	
		Millstream	
		Maidenhead Road	
		WINDSOR	
		Berkshire	
		SL4 5GD	
		(in respect of rights contained in Deeds dated 19 April 1972 and 6 February 1973)	
		Colin Stuart Hazell	
		Astral House	
		Catfoss	
		HULL	
		East Riding of Yorkshire	
		HU11 5SP	
		(in respect of rights listed in a Transfer dated 16 December 2010)	
		Peter James Boynton	
		Hind House	
		Catfoss Lane	
		Sigglesthorne	
		HULL	
		East Riding of Yorkshire	
		HU11 5QN	
		(as The Executor of the Estate of the Late Patricia Anne Boynton)	
		(in respect of rights reserved by a Conveyance dated 24 April 1981)	
		Secretary Of State For Defence	
		Main Building	
		Horse Guards Avenue	
		Whitehall	
		LONDON	
		SW1A 2HB	
		(in respect of covenants and grants contained in a Deed dated 11 November 1948)	
		Unknown	
I.		(in respect to rights reserved by a Conveyance dated 1 October 1965)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans	Description of Land	of the Planning Act 2008	
06-022	116 square metres of verge (north of Harsell Lane, excluding all interests of the Crown)	Centrica PLC	
		Millstream	
		Maidenhead Road	
		WINDSOR	
		Berkshire	
		SL4 5GD	
		(in respect of rights contained in Deeds dated 19 April 1972 and 6 February 1973)	
		Colin Stuart Hazell	
		Astral House	
		Catfoss	
		HULL	
		East Riding of Yorkshire	
		HU11 5SP	
		(in respect of rights of access as contained in a Transfer dated 16 December 2010)	
		Peter James Boynton	
		Hind House	
		Catfoss Lane	
		Sigglesthorne	
		HULL	
		East Riding of Yorkshire	
		HU11 5QN	
		(as The Executor of the Estate of the Late Patricia Anne Boynton)	
		(in respect of rights reserved by a Conveyance dated 24 April 1981)	
		Secretary Of State For Defence	
		Main Building	
		Horse Guards Avenue	
		Whitehall	
		LONDON	
		SW1A 2HB	
		(in respect of covenants and grants contained in a Deed dated 11 November 1948)	
		Unknown	
		(in respect to rights reserved by a Conveyance dated 1 October 1965)	



	Dogger Bank S	South Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on	Description of land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)
Land Plans	Description of Land	of the Planning Act 2008
06-023	177 square metres of public road (Harsell Lane, exlucing all interests of the Crown)	Centrica PLC
		Millstream
		Maidenhead Road
		WINDSOR
		Berkshire
		SL4 5GD
		(in respect of rights to lay, maintain and renew a pipe contained in a Deed dated 19 April 1972 and in respect of rights to lay, maintain and
		renew a pipe contained in a Deed dated 6 February 1973)
		Colin Stuart Hazell
		Astral House
		Catfoss
		HULL
		East Riding of Yorkshire
		HU11 5SP
		(in respect of rights to maintain, repair and renew a fence contained in a Transfer dated 16 December 2010)
		Peter James Boynton
		Hind House
		Catfoss Lane
		Sigglesthorne
		HULL
		East Riding of Yorkshire
		HU11 5QN
		(as The Executor of the Estate of the Late Patricia Anne Boynton)
		(in respect of rights of way contained in a Conveyance dated 24 April 1981)
		Secretary Of State For Defence
		Main Building
		Horse Guards Avenue
		Whitehall
		LONDON
		SW1A 2HB
		(in respect of rights to maintain hedges and trees contained in a Deed dated 11 November 1948)
		Unknown
		(in respect to rights of way contained in a Conveyance dated 1 October 1965)



	C	Dogger Bank South Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on	Description of Lond	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)
Land Plans	Description of Land	of the Planning Act 2008
06-024	9 square metres of agricultural land (south of Harsell Lane)	Centrica PLC
		Millstream
		Maidenhead Road
		WINDSOR
		Berkshire
		SL4 5GD
		(in respect of rights to lay, maintain and renew a pipe contained in a Deed dated 19 April 1972 and in respect of rights to lay, maintain and
		renew a pipe contained in a Deed dated 6 February 1973)
		Colin Stuart Hazell
		Astral House
		Catfoss
		HULL
		East Riding of Yorkshire
		HU11 5SP
		(in respect of rights to maintain, repair and renew a fence contained in a Transfer dated 16 December 2010)
		Peter James Boynton
		Hind House
		Catfoss Lane
		Sigglesthorne
		HULL
		East Riding of Yorkshire
		HU11 5QN
		(as The Executor of the Estate of the Late Patricia Anne Boynton)
		(in respect of rights of way contained in a Conveyance dated 24 April 1981)
		Secretary Of State For Defence
		Main Building
		Horse Guards Avenue
		Whitehall
		LONDON
		SW1A 2HB
		(in respect of rights to maintain hedges and trees contained in a Deed dated 11 November 1948)
		Unknown
		(in respect to rights of way contained in a Conveyance dated 1 October 1965)



		BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)
Land Plans	Description of Land	of the Planning Act 2008
06-025	64907 square metres of agricultural land, drain and hedgerow (south of Harsell Lane, east	Centrica PLC
	of Catwick Road, excluding all interests of the Crown)	Millstream
		Maidenhead Road
		WINDSOR
		Berkshire
		SL4 5GD
		(in respect of rights to lay, maintain and renew a pipe contained in a Deed dated 19 April 1972 and in respect of rights to lay, maintain and
		renew a pipe contained in a Deed dated 6 February 1973)
		Colin Stuart Hazell
		Astral House
		Catfoss
		HULL
		East Riding of Yorkshire
		HU11 5SP
		(in respect of rights to maintain, repair and renew a fence contained in a Transfer dated 16 December 2010)
		Peter James Boynton
		Hind House
		Catfoss Lane
		Sigglesthorne
		HULL
		East Riding of Yorkshire
		HU11 5QN
		(as The Executor of the Estate of the Late Patricia Anne Boynton)
		(in respect of rights of way contained in a Conveyance dated 24 April 1981)
		Secretary Of State For Defence
		Main Building
		Horse Guards Avenue
		Whitehall
		LONDON
		SW1A 2HB
		(in respect of rights to maintain hedges and trees contained in a Deed dated 11 November 1948)
		Unknown
		(in respect to rights of way contained in a Conveyance dated 1 October 1965)



Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
07-001	20924 square metres of agricultural land (east of Catwick Road)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect to the rights to lay and maintain pipes and in respect to a right to easements as contained in a Deed dated 19 April 1972)
07-002	6926 square metres of agricultural land (east of Catwick Road)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect to the right to easement and in respect to the right to lay, maintain, inspect and repair a gas pipeline contained in a Deed dated 19 April 1972)
07-003	4081 square metres of agricultural land, copse and drain (east of Catwick Road)	David Hobson Manor Lodge Catfoss Lane Sigglesthorne HULL HU11 5QN (in respect of rights of wayleave easements, rights of way and drainage contained in a transfer dated 1 August 1990)
	21226 square metres of agricultural land, drain and public footpath (Seaton Footpath No.10) (east of Catwick Road)	Andrew Stewart Buchan 346 Hathersage Road HULL East Riding of Yorkshire HU8 0EY (in respect of rights of light or air contained in a Conveyance dated 6 November 1990) Nicola Jayne Smith 346 Hathersage Road HULL East Riding of Yorkshire HU8 0EY (in respect of rights of light and air contained in a Conveyance dated 6 November 1990)



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-005	62273 square metres of agricultural land, access track, copse and watercourse (Catfoss Drain) (Catfoss Hall Farm Access Road, east of Catwick Road)	Timothy Duncan Rose Catfoss Hall Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HU11 5QN (in respect of rights of way contained in a Conveyances dated 26 January 1999 and 29 October 1970)	
08-001	1135 square metres of agricultural land and access splay (east of Catwick Road)	Timothy Duncan Rose Catfoss Hall Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HU11 5QN (in respect of rights of way contained in a Conveyances dated 26 January 1999 and 29 October 1970)	
08-002	6119 square metres of public road, access splays, drains and verges (Catwick Road)	Unknown	
08-003	230 square metres of bridge, verge and copse (Washdike Bridge, east of Catwick Road)	Unknown	
08-004	515 square metres of public road and verges (West Road, A1035)	Unknown	
08-005	1912 square metres of public road and verges (West Road, A1035)	Unknown	
08-006	966 square metres of public road and verges (West Road, A1035)	Unknown	
08-009	14359 square metres of agricultural land, access splay, pond and drain (south of West Road, A1035 and east of Catwick Heads)	Unknown (in respect of rights of easements and covenants contained in a Conveyance dated 19 January 1987)	
08-010	218 square metres of verge (south of West Road, A1035, east of Catwick Heads)	Unknown (in respect of rights of easements and covenants contained in a Conveyance dated 19 January 1987)	
08-011	583 square metres of public road, access splay and verge (Catwick Heads)	Unknown	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
08-012	4675 square metres of agricultural land, access splay and verge (west of Catwick Heads)	Unknown (in respect of rights of light, air, footpaths and carriageways contained in a Conveyance dated 6 April 1965)	
08-013	1637 square metres of agricultural land (east of Catwick Heads)	Unknown	
08-014	1919 square metres of public road and verge (Catwick Heads)	Unknown	
08-015	19001 square metres of agricultural land, access track, drain and public footpath (Catwick Footpath No.8) (west of Catwick Heads)	Unknown (in respect of rights of light, air, footpaths and carriageways contained in a Conveyance dated 6 April 1965)	
08-016	20212 square metres of agricultural land and pond (west of Catwick Heads)	Unknown (in respect of right of way as contained in a Conveyance dated 10 May 1989)	
	46823 square metres of agricultural land, access track and drain (Field House Farm, south of Rise Lane)	Unknown (in respect of rights contained in a Transfer dated 2 September 2009)	
09-001	15487 square metres of agricultural land (east of Riston Road)	Amy Clark-Blakey Glenholme Leven Road Catwick BEVERLEY HU17 SPT (in respect of rights contained in a Transfer dated 2 March 2008) Gemma Louise Elizabeth Clark-Blakey Glenholme Leven Road Catwick BEVERLEY HU17 SPT (in respect of rights contained in a Transfer dated 2 March 2008)	



		Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
09-001		John Christopher Bird
cont'd		The Old Hall Lakeview
		Riston Road
		Catwick
		BEVERLEY
		East Riding of Yorkshire
		HU17 5PR
		(in respect of rights contained in a Transfer dated 30 July 2009)
		Mark Desmond Bird
		Old Hall Farm
		Riston Road
		Catwick
		BEVERLEY
		East Riding of Yorkshire
		HU17 SPR
		(in respect of rights contained in a Transfer dated 30 July 2009)
		Northern Gas Networks Limited
		1100 Century Way
		Thorpe Park
		LEEDS
		West Yorkshire
		LS15 8TU
		(in respect of rights contained in a Deed of Grant of Easement dated 5 September 1967. Also in respect of rights and restrictive covenants
		contained in a Deed of Grant of Easement dated 3 October 1966)
		Northern Powergrid (Yorkshire) PLC
		Lloyds Court
		78 Grey Street
		Newcastle Upon Tyne
		NE1 6AF
		(in respect of rights contained in a Lease dated 14 October 2021)



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans		of the Planning Act 2008	
09-001		Paul Adrian Bird	
cont'd		Old Hall Farm	
		Riston Road	
		Catwick BEVERLEY	
		East Riding of Yorkshire	
		HU17 5PR	
		(in respect of rights contained in a Transfer dated 30 July 2009 )	
09-002	5341 square metres of agricultural land (east of Riston Road)	Unknown	
09-005	89 square metres of public road and verges (Riston Road)	Unknown	
09-006	1059 square metres of public road and verges (Riston Road)	Unknown	
09-007	117 square metres of agricultural land (west of Riston Road)	Holderness Hunt (Holdings) Limited	
		Citadel House	
		58 High Street	
		HULL	
		East Riding of Yorkshire	
		HU11QE	
		(in respect of rights of access)	
		Unknown	
		(in respect of right of access contained in a Conveyance dated 6 April 1918)	
09-008	15705 square metres of agricultural land and drain (west of Riston Road)	Holderness Hunt (Holdings) Limited	
		Citadel House	
		58 High Street	
		HULL	
		East Riding of Yorkshire	
		HU1 1QE	
		(in respect of rights of access)	
		Unknown	
		(in respect of right of access contained in a Conveyance dated 6 April 1918)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Lond	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans	Description of Land	of the Planning Act 2008	
09-009	19717 square metres of agricultural land and drain (Willow Croft Farm, west of Riston	John Christopher Bird	
	Road)	The Old Hall Lakeview	
		Riston Road	
		Catwick	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 5PR	
		(in respect of rights granted by a Conveyance dated 27 November 1925)	
		Mark Desmond Bird	
		Old Hall Farm	
		Riston Road	
		Catwick	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 5PR	
		(in respect of rights granted by a Conveyance dated 27 November 1925)	
		Paul Adrian Bird	
		Old Hall Farm	
		Riston Road	
		Catwick	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 5PR	
		(in respect of rights granted by a Conveyance dated 27 November 1925)	
09-010	70690 square metres of agricultural land, hedgerow and drain (Prospect Farm, east of	Unknown	
	Whitecross Road, A165 and west of Catwick Lane)	(in respect of rights listed in a conveyance dated 19 January 1987)	
09-011	9467 square metres of agricultural land and hedgerow (Prospect Farm, east of Whitecross	Unknown	
	Road, A165 and west of Catwick Lane)	(in respect of rights as contained in a Conveyance dated 19 January 1987)	
09-012	6000 square metres of agricultural land and drain (Riston Grange Farm, east of Whitecross	Centrica PLC	
	Road, A165)	Millstream	
		Maidenhead Road	
		WINDSOR	
		Berkshire	
		SL4 5GD	
		(in respect of a gas main and rights listed in a Deed dated 3 October 1966)	



	Dogger Bank Sou	uth Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
09-012 cont'd		Unknown (in respect of right of entry and access to services as contained in a Conveyance dated 4 October 1977)
	39182 square metres of agricultural land and drain (Riston Grange Farm, east of Whitecross Road, A165)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of a gas main and rights listed in a Deed dated 3 October 1966) Unknown (in respect of right of entry and access to services as contained in a Conveyance dated 4 October 1977)
	2704 square metres of agricultural land (Riston Grange Farm, east of Whitecross Road, A165)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of a gas main and rights listed in a Deed dated 3 October 1966) Unknown (in respect of right of entry and access to services as contained in a Conveyance dated 4 October 1977)
10-002	2119 square metres of public road and verges (Whitecross Road, A165)	Unknown
10-003	3692 square metres of public road and verges (Whitecross Road, A165)	Unknown
	21177 square metres of agricultural land (Riston Grange Farm, west of Whitecross Road, A165)	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of rights of access)



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-004 cont'd		Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of a gas main and rights listed in a Deed dated 3 October 1966) Unknown (in respect of right of entry and access to services as contained in a Conveyance dated 4 October 1977)	
	44238 square metres of agricultural land and drain (Arnold and Riston Drain) (Riston Grange Farm, west of Whitecross Road, A165)	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights and rights of access) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect of a gas main and rights listed in a Deed dated 3 October 1966) Unknown (in respect of right of entry and access to services as contained in a Conveyance dated 4 October 1977)	
	27667 square metres of agricultural land and drain (Arnold and Riston Drain) (Riston Grange Farm, west of Whitecross Road, A165)	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of rights of access)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire	
Number or Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-007	18189 square metres of agricultural land and public footpath (Riston Footpath No.2) (north of Carr Lane)	Walter Stuart Leonard Kirkwood Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR (as The Executor of the Estate of the Late Walter Trevor Kirkwood) (in respect of a right of way as contained in a Conveyance dated 19 July 1967)	
10-008	2 square metres of verge (north of Carr Lane)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect of rights listed in a Deed of Grant dated 5 September 1967) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 SAH (in respect of the rights of access to maintain drainage as contained in a Conveyance dated 4 February 1957) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights contained in a Lease dated 14 July 2015)	



BOCK OF REFERENCE - PAT 2 County of East Refinge of Vorkshire           Number on Land Plans         Description of Land         Potential claims under S10 Compulsion Plurchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008           10-008         If Kelay and Sans Limited Church Farm Meaux Road Routh         On the Planning Act 2008         On the Planning Act 2008           10-009         2273 square metres of agricultural land, verges and drain (north of Carr Lane)         Inverte farm (in respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)           10-009         2273 square metres of agricultural land, verges and drain (north of Carr Lane)         Findroment Agency Horton House Description           10-009         2273 square metres of agricultural land, verges and drain (north of Carr Lane)         Findroment Agency Horton House Description           10-009         2273 square metres of agricultural land, verges and drain (north of Carr Lane)         Findroment Agency Horton House Description           10-009         2273 square metres of agricultural land, verges and drain (north of Carr Lane)         Findroment Agency Horton House Description           11         2008         2273 square metres of agricultural land, verges and drain (north of Carr Lane)           11         11         11           12         11         11           13         11         11		Dogger Bank South Offshore Wind Farm Development Consent Order		
Number on Land Planst         Description of Land         Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008           10-008         TH Caley and Sons Limited Church Farm Meaux Road Routh BEVERLEY         The Caley and Sons Limited Church Farm Meaux Road Routh BEVERLEY           10-009         2273 square metres of agricultural land, verges and drain (north of Carr Lane)         Envronment Agency Horizon House Dearrey Road BRISTOL Avon BSIS SAH Lin respect of rights to maintain)         Envronment Agency Horizon House Dearrey Road BRISTOL Avon BSIS FAH Lin respect of rights to maintain)           Richard Guy Caley North Firmeset Meaux BEREIEY East Riding of Yorkshire HU3 7986 Lin respect of rights of drainage)         Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU3 7986 Lin respect of rights of drainage)		BOOK OF REFERENCE - PART 2		
Land Plans       Description of Land       of the Planning Act 2008         10-009       If Caley and Sons Limited Church Farm Mexic Road Booth BCREEY North Humberside (In respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)         10-009       2273 square metres of agricultural land, verges and drain (north of Carr Lane)       Environment Agency Horizon House Description Plant         10-009       2273 square metres of agricultural land, verges and drain (north of Carr Lane)       Environment Agency Horizon House Description Plant         10-009       2273 square metres of agricultural land, verges and drain (north of Carr Lane)       Environment Agency Horizon House Description Plant         10-009       2273 square metres of agricultural land, verges and drain (north of Carr Lane)       Environment Agency Horizon House Description Plant         10-009       2273 square metres of agricultural land, verges and drain (north of Carr Lane)       Environment Agency Horizon House Description Plant         10-009       2273 square metres of agricultural land, verges and drain (north of Carr Lane)       Environment Agency Horizon House Description Plant         10-009       2273 square metres of agricultural land, verges and drain (north of Carr Lane)       Environment Agency Horizon House Description Plant         10-009       2273 square metres of agricultural land, verges and drain (north of Carr Lane)       Environment Agency Horizon House Description Plant         10-009       2273 sqLing			County of East Riding of Yorkshire	
Land Park       Of the Marking Act 2008         10-008       TH Caley and Sons limited         Cont'd       Church Farm         Meaux Road       Routh         BVCRLFY       North Humberide         North Sector       Environment Agency         Environment Agency       Environment Agency         Horton House       Deanery Road         Bishard Guy Caley       North Stronge         Meture       Bishard Guy Caley         North Stronge       Reset of rights to maintain)         Bishard Guy Caley       North Shire         Horus Caley       Meture         Bishard Guy Caley       Meture         North Stronge       Meture         REVERTY       East Riding of Yorkshire         Huru 758       In respect of rights of drainage)	Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
cont di       Church Farm         Neuxe Road       Routh         BVFRLFY       North Humberside         HUT 95R       In respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)         10-009       2273 square metres of agricultural land, verges and drain (north of Carr Lane)       Environment Agency         Horizon House       Bainston         Non       BSISTOL         Avon       BSI. SAH         (In respect of rights of maintain)         Richard Guy Coley         North Grange         Meaux         BVERLEY         Non         BSI. SAH         (In respect of rights of maintain)         Richard Guy Coley         North Grange         Meaux         BVERLEY         Ext Riding of Yorkhire         HUT 295S         (In respect of rights of drainage)         Thomas Stephen C	Land Plans	Description of Land		
Negux Road       Routh         BEVERLEY       North Humberside         North Humberside       (In respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)         10-009       2273 square metres of agricultural land, verges and drain (north of Carr Lane)       Environment Agency         Horizon House       Deanery Road         BNISTOL       Avon         BS1 SAH       Richard Guy Caley         North Stange       Meaux         BVERLEY       East Riding of Yricklire         Huizy Post       Thomas Stephen Caley         Thomas Stephen Caley       Thomas Road         BVERLEY       East Riding of Yricklire         BVERLEY       Eas				
Bouth       BRURE         Brunch       BrUREY         North Humberside       H117 958         Brunch       Environment Agency         Horizon House       Deamery Road         BRISTOL       Avon         BS1 SAH       (in respect of rights to maintain)         Richard Guy Caley       North Humberside         Horizon House       Deamery Road         BRISTOL       Avon         BS1 SAH       (in respect of rights to maintain)         Richard Guy Caley       North Grange         Meaux       BEVERLEY         Thomas Stephen Caley       Choine spect of rights of drainage)         Thomas Stephen Caley       Chorkshire         H117 958       (in respect of rights of drainage)	cont'd			
BVERLY         North Humberside         HU17 95R         (In respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)         10-09       2273 square metres of agricultural land, verges and drain (north of Carr Lane)         Horizon House       Deanery Road         BISTORL       Avon         BS1 SAH       (in respect of rights to maintain)         Richard Guy Caley       North Krange         Meaoux       BEVERLEY         Ext Riding of Yorkshire       HU17 95S         (in respect of rights of drainage)       Thomas Stephen Caley         Church Farm       Meaaux         BEVERLY       Strokhire         HU17 95S       (in respect of rights of drainage)				
North Humerside       HU17 JSNB         10-009       2273 square metres of agricultural land, verges and drain (north of Carr Lane)       Environment Agency         Hoizon Rouse       Deanery Rouse         BellSTOL       Avon         North Farm       Rickard of rights to maintain)         Rickard of verges       Rickard of verges         North Scale       Bit STOL         Avon       Bit STOL         Avon       Bit Stol         Rickard of verges       Rickard of verges         North Grange       Meaux         BEVERLY       East Riding of Vorkshire         HU17 9SS       (in respect of rights of drainage)         Thomas Stephen Caley       Church Farm         Routh       BEVERLY         East Riding of Vorkshire       HU17 9SS         (in respect of rights of drainage)       Thomas Stephen Caley         Church Farm       Routh         BEVERLY       East Riding of Yorkshire         HU17 9SS       (in respect of rights of drainage)         Routh       BEVERLY         East Riding of Yorkshire       HU17 9SR         (in respect of rights of drainage)       (in respect of rights of drainage)				
HU17 59R       HU17 59R         10-009       2273 square metres of agricultural land, verges and drain (north of Carr Lane)       Environment Agency         Horizon House       Deanery Road         BRIST ROad       Ristron House         Deanery Road       BRISTOL         Avon       BS1 SAH         (in respect of rights to maintain)       Richard Guy Caley         North Grange       Meaux         BEVERUP       East Riding of Yorkshire         HU17 95S       (in respect of rights of drainage)         Tomas Stephen Caley       Church Farm         Meaux Road       Routh BEVERLEY         Best Reling of Yorkshire       HU17 95S         (in respect of rights of drainage)       Itomas Stephen Caley         Church Farm       Meaux Road         Routh BEVERLEY       East Riding of Yorkshire         HU17 95S       (in respect of rights of drainage)				
Image: Control of Contro				
10-009       2273 square metres of agricultural land, verges and drain (north of Carr Lane)       Environment Agency         Horizon House       Deamery Road         BRISTOL       Avon         BS1       BS1         In respect of rights to maintain)       Richard Guy Caley         North Grange       Meaux         BEVERLEY       East Riding of Yorkshire         HU170 SS       (in respect of rights of drainage)         Thomas Stephen Caley       Church Farm         Meaux Road       Routh         BEVERLEY       East Riding of Yorkshire         HU170 SR       (in respect of rights of drainage)				
Horizon House Deanery Road BRISTOL Avon BSI SAH (in respect of rights to maintain) Richard Guy Caley North Grange Meaux BEVENLEY East Riding of Yorkshire HU129 S9S (in respect of rights of drainage) Thomas Stephen Caley Church Farm Meaux Road Routh BEVENLEY East Riding of Yorkshire HU129 S9S (in respect of rights of drainage) Thomas Stephen Caley Church Farm Meaux Road Routh BEVENLEY East Riding of Yorkshire HU129 S9S (in respect of rights of drainage)			(in respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)	
Deanery Road BRISTOL BRISTOL Avon BSJ 5AH (in respect of rights to maintain) Richard Guy Caley North Grange Meaux BEVERLEY EAR Riding of Yorkshire HU17 9SS (in respect of rights of drainage) Thomas Stephen Caley Church Farm Meaux Road Routh Meaux Road Routh BEVERLEY EAR Riding of Yorkshire HU17 9SR (in respect of rights of drainage)	10-009	2273 square metres of agricultural land, verges and drain (north of Carr Lane)	Environment Agency	
BRISTOL Avon BS1 5AH (in respect of rights to maintain) Richard Guy Caley North Grange Meaux BEVERLEY East Riding of Yorkshire HU17 95S (in respect of rights of drainage) Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 95R (in respect of rights of drainage)			Horizon House	
Avon B51 5AH (in respect of rights to maintain) Richard Guy Caley North Grange Meaux BEVERLEY East Riding of Yorkshire HU17 9SS (in respect of rights of drainage) Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of rights of drainage)				
BS1 5AH (in respect of rights to maintain) Richard Guy Caley North Grange Meaux BEVERLEY East Riding of Yorkshire HU17 9SS (in respect of rights of drainage) Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of rights of drainage)				
(in respect of rights to maintain) Richard Guy Caley North Grange Meaux BVERLEY East Riding of Yorkshire HU17 9SS (in respect of rights of drainage) Thomas Stephen Caley Church Farm Meaux Road Routh BVERLEY East Riding of Yorkshire HU17 9SR (in respect of rights of drainage)				
Richard Guy Caley North Grange Meaux BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of rights of drainage)				
North Grange Meaux BEVERLEY East Riding of Yorkshire HU17 9SS (in respect of rights of drainage) Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of rights of drainage)			(in respect of rights to maintain)	
Meaux BEVERLEY East Riding of Yorkshire HU17 9SS (in respect of rights of drainage) Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of rights of drainage)			Richard Guy Caley	
BEVERLEY East Riding of Yorkshire HU17 9SS (in respect of rights of drainage) Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR HU17 9SR			North Grange	
East Riding of Yorkshire HU17 9SS (in respect of rights of drainage) Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of rights of drainage) (in respect of rights of drainage)				
HU17 9SS (in respect of rights of drainage) Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of rights of drainage)				
(in respect of rights of drainage) Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of rights of drainage)				
Thomas Stephen Caley Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of rights of drainage)				
Church Farm Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of rights of drainage)			(in respect of rights of drainage)	
Meaux Road Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of rights of drainage)				
Routh BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of rights of drainage)				
BEVERLEY East Riding of Yorkshire HU17 9SR (in respect of rights of drainage)				
East Riding of Yorkshire HU17 9SR (in respect of rights of drainage)				
HU17 9SR (in respect of rights of drainage)				
(in respect of rights of drainage)				
Unknown			(in respect of rights of drainage)	
			Unknown	



		th Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
10-010	16062 square metres of agricultural land (Church Farm, south of Main Road, A1035)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights of access to enter and conduct works to maintain drainage as contained in a Conveyance dated 28 March 1956)
	64311 square metres of agricultural land, access tracks and drains (Church Farm, south of Main Road, A1035 and east of Meaux Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights of access to enter and conduct works to maintain drainage as contained in a Conveyance dated 28 March 1956)
	15746 square metres of agricultural land and access tracks (Church Farm, south of Main Road, A1035 and east of Meaux Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights of access to enter and conduct works to maintain drainage as contained in a Conveyance dated 28 March 1956)
	18019 square metres of agricultural land, access track and drains (south of Main Road, A1035)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights listed in a Deed of Grant dated 5 September 1967)



Number of Land Plansis         Description of Land         Potential dains under 510 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008           10 015 control         Environment Agency Hisrison House Description of Land         Environment Agency Hisrison House Description and Land Plansis           10 015 control         Fill and Plansis         Environment Agency Hisrison House Description and Land Plansis           10 016 control         Fill and Plansis         Environment Agency Hisrison House Description and Land Plansis           10 016 control         Fill and Plansis         Environment Agency Hisrison House Description and Land Plansis           10 016         Respect of the rights of access to maintain drainage as contained in a Conveyance dated 4 February 1957)           Northern Powergrid (Vorkshire) PLC Lloyds Court 78 Grey Street Nexessate Upon Tyne NEI SAF (In respect of rights contained in a Lase dated 14 July 2015)           10 018         789 square metres of public road and access splay (Main Road, A1003)         Unknown           10 019         3583 square metres of agricultural land and drains (Manor House Farm, east of Meaux Land Plansis         Vers Barn Humace Farm Humace Farm		Dogger Bank South Offshore Wind Farm Development Consent Order		
Number of Land Plans         Description of Land         Potential claims under S10 Compulsory Purchase Act 1965, Pirt 1 Lind Compensation Act 1973 or 152 (3) of the Planning Act 2008           10:016         Environment Agency Horiton House Description of Lind House Description of Lind S1 SAH (In respect of the rights of access to maintain drainage as contained in a Conveyance dated 4 February 1957)           Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Type NEI SAF (In respect of rights contained in a Lease dated 14 July 2015)           10-018         789 square metres of public road and access splay (Main Road, A1035)           10-019         35853 square metres of agricultural land and drains (Manor House Farm, east of Meaux Land Plans)         Benjamin David James Mackrill West Barn Furnace Land Linde/JUNERICE           10-019         35853 square metres of agricultural land and drains (Manor House Farm, east of Meaux Land)         Benjamin David James Mackrill West Barn Furnace Land Linde/JUNERICE           10-019         35853 square metres of agricultural land and drains (Manor House Farm, east of Meaux Land)         Benjamin David James Mackrill West Barn Furnace Land Linde/JUNERICE WELLS           10-019         35853 square metres of agricultural land and drains (Manor House Farm, east of Meaux TUNBRIDGE WELLS TN 38 UE			BOOK OF REFERENCE - PART 2	
Land Plans       Other Planning Act 2008         10-015       Invition House         0.016       Dearery Road         0.017       Bit Stoll         Avoin       Bit Stoll         Avoin       Bit Stoll         Northern Powery Road       Bit Stoll         (In respect of the rights of access to maintain drainage as contained in a Conveyance dated 4 February 1957)         Northern Powerying (Yorkshire) PLC       Ludges Court         78 Grey Street       Newscatal Ludgen Tyree         Net EGAF       In respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)         10-018       785 square metres of public road and access splay (Main Road, A1035)       Unknown         10-019       2853 square metres of agricultural land and drains (Manor House Farm, east of Meaux Parm, east of Meaux Parmace Farm Farmace Farm Fa			County of East Riding of Yorkshire	
cont'd       Horizon House         horizon House       BitSTOL         Avon       BitSTOL         Northern Powergrid (Yorkshire) PLC       Northern Powergrid (Yorkshire) PLC         Undysk Court       78 Grey Street         Newcestle Upon Tynee       Nit 6AF         Nit Control       Clay and Sons Limited         Church Farm       Meaux Road         Meaux Road       Nothern Powergrid (Yorkshire) Farm, east of Meaux         Meaux Road       Respective         North Humberside       HUJ7 95R         (In respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)         10-018       R9 square metres of public road and access splay (Main Road, A1035)       Unknown         10-019       SpSt3 square metres of agricultural land and drains (Manor House Farm, east of Meaux       Berjamin David James Mackrill         Virace Lane       Furnace Lane       Furnace Lane         Virace Earm       Furnace Lane       Furnace Lane         Virace Earm       Furnace Lane       Seli Barn		Description of Land		
Image: Densey Road BitSTOL Axon BitSTOL BI	10-016		Environment Agency	
BRISTOL.         Aron         BS1 5AH         (in respect of rights of access to maintain drainage as contained in a Conveyance dated 4 February 1957)         Northern Powergrid (Vorkshire) PLC.         Updyds Court         73 Grey Street         Newcastle Upon Tyne         NEI GAF         (in respect of rights contained in a Lease dated 14 July 2015)         T H Caley and Sons Limited         Church Farm         Meaux Moad         North Humberside         HU17 95 R         (in respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)         10-018       R28 square metres of public road and access splay (Main Road, A1035)         Unknown       Benjain David James MacKrill         Variane       Benjain David James MacKrill         Variane       Benjain David James MacKrill         Variane Farm       Furnace Farm         Furnace Farm <t< td=""><td>cont'd</td><td></td><td>Horizon House</td></t<>	cont'd		Horizon House	
Aron       BS1 SAH       In respect of the rights of access to maintain drainage as contained in a Conveyance dated 4 February 1957)         Northern Powergrid (Vorkshire) PLC       Northern Powergrid (Vorkshire) PLC         Ubdys S Grey Street       Si Grey Street         S Grey Street       Northern Powergrid (Vorkshire) PLC         In Caley and Sons Limited       Northern Powergrid (Vorkshire) PLC         In Caley and Sons Limited       Th Caley and Sons Limited         North Humberside       North Humberside         In D-018       S853 square metres of public road and access splay (Main Road, A1035)       Unknown         In 0-019       S5853 square metres of agricultural land and drains (Manor House Farm, east of Meaux       Benjamin David James Mackrill         Yumae Lane       Benjamin David James Mackrill       West Barm         Yumae Erame       Iamoberluration       Iamoberluration         No Routh       Nanore Farm       Iamoberluration         No Routh       Benjamin David James Mackrill       West Barm         Yumae Lane       Benjamin David James Mackrill       West Barm         Yumae Lane       Iamoberluration       Iamoberluration         No Routh       Nanore Farm       Iamoberluration         Yumae Lane       Nanore Farm       Iamoberluration         Nanore Farm       Iamobe			Deanery Road	
Bit SAH (in respect of the rights of access to maintain drainage as contained in a Conveyance dated 4 February 1957)         Northern Powergrid (Vorkshire) PLC Uoyds Court 78 Gregs Street Need Street N			BRISTOL	
Image:			Avon	
Image: Street of public road and access splay (Main Road, A1035)       Northerm Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne Net LAF (in respect of rights contained in a Lease dated 14 July 2015)         10-018       789 square metres of public road and access splay (Main Road, A1035)       Unknown         10-019       35853 square metres of agricultural land and drains (Manor House Farm, east of Meau Lane)       Benjamin David James Mackrill West Barn Furnace Farm (unace Farm Lane)         10-019       35853 square metres of agricultural land and drains (Manor House Farm, east of Meau Lane)       Benjamin David James Mackrill West Barn Furnace Farm (unace Farm Lane)         10-019       10-018       Benjamin David James Mackrill West Barn Furnace Farm (unace Farm Lane)         10-019       10-018       Benjamin David James Mackrill West Barn Furnace Farm (unace Farm (			BS1 5AH	
Iooyds Court       Nords Court         Newszelst Upon Tyne       Newszelst Upon Tyne         NE1 GAF       (in respect of rights contained in a Lease dated 14 July 2015)         T H Caley and Sons Limited       TH Caley and Sons Limited         Church Farm       Meaux Road         North Humberside       HU17 95R         (in respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)         10-018       Z89 square metres of public road and access splay (Main Road, A1035)         10-019       J5853 square metres of agricultural land and drains (Manor House Farm, east of Meaux Lane)         BerlyEntrest TUNBRIDGE WELLS' TUNBRIDGE WELLS' TUNBRIDGE WELLS' TUNBRIDGE WELLS' TUNBRIDGE WELLS' TUNBRIDGE WELLS' TUNBRIDGE WELLS' TUNBRIDGE WELLS'			(in respect of the rights of access to maintain drainage as contained in a Conveyance dated 4 February 1957)	
10-018       78 Grey Street Nexessale Upon Tyne NEI 6AF (in respect of rights contained in a Lease dated 14 July 2015)         11-018       74 Caley and Sons Limited Church Farm Meaux Kood Routh BeVERLEY North Humberside HU17 95R (in respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)         10-018       78 square metres of public road and access splay (Main Road, A1035)       Unknown         10-019       35853 square metres of agricultural land and drains (Manor House Farm, east of Meaux Lane)       Benjamin David James Mackrill West Barm Fumace Farm Fumace Farm Fumace Farm Fumace Farm Fumace Farm Fumace Farm				
Newcastle Upon Tyne NEI 6AF (in respect of rights contained in a Lease dated 14 July 2015)TH Caley and Sons Limited Church Farm Meaux Road Routh BEVERLEY North Humberside HU127 9SR (in respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)10-018R9 square metres of public road and access splay (Main Road, A1035)Unknown10-019S8SS square metres of agricultural land and drains (Manor House Farm, east of Meaux Lane)Benjamin David James Mackrill West Barn Furmace Farm Furmace Farm 			Lloyds Court	
NE1 6AF       (in respect of rights contained in a Lease dated 14 July 2015)         T H Caley and Sons Limited       T H Caley and Sons Limited         Church Farm       Resure Neoad         Routh       BEVERLEY         J0-018       789 square metres of public road and access splay (Main Road, A1035)         U-019       J35853 square metres of agricultural land and drains (Manor House Farm, east of Meaux Lane)         Benjamin David James Mackrill West Barn Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE			78 Grey Street	
In the spect of rights contained in a Lease dated 14 July 2015)In the spect of rights contained in a Lease dated 14 July 2015)In the spect of rights contained in a Lease dated 14 July 2015)In the spect of rights contained in a Lease dated 14 July 2015)In the spect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)In the spect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)In the spect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)In the spect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)In the spect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)In the spect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)In the spect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)In the spect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)In the spect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)In the spect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)In the spect of rights of access and use of service media as contained in July 2015)In the spect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)In the spect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)In the spect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)In the spect of rights of acces				
Image:				
LongService media as contained in a Transfer dated 1 July 2015)10-018R9 square metres of public road and access splay (Main Road, A1035)Unknown10-019S853 square metres of agricultural land and rains (Manor House Farm, east of Meaux Lane)Benjamin David James Mackrill Furnace Farm Furnace			(in respect of rights contained in a Lease dated 14 July 2015)	
Meaux Road Routh BEVERLEY North Humberside HU17 9SR (in respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)10-018789 square metres of public road and access splay (Main Road, A1035)Unknown10-01935853 square metres of agricultural land and drains (Manor House Farm, east of Meaux Lane)Benjamin David James Mackrill West Barn Furnace Farm Furnace Farm Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE			T H Caley and Sons Limited	
Routh BEVERLEY North Humberside HUT 95R (in respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)10-018789 square metres of public road and access splay (Main Road, A1035)Unknown10-01935853 square metres of agricultural land and drains (Manor House Farm, east of Meaux Lane)Benjamin David James Mackrill West Barn Furnace Farm Furnace			Church Farm	
BEVERLEY       North Humberside       HU17 95R       in respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)         10-018       789 square metres of public road and access splay (Main Road, A1035)       Unknown         10-019       5853 square metres of agricultural land and drains (Manor House Farm, east of Meaux       Benjamin David James Mackrill         10-019       Lane)       Stass square metres of agricultural land and drains (Manor House Farm, east of Meaux       Benjamin David James Mackrill         10-019       Tuney       Stass square metres of agricultural land and drains (Manor House Farm, east of Meaux       Benjamin David James Mackrill         10-019       Tuney       Stass square metres of agricultural land and drains (Manor House Farm, east of Meaux       Benjamin David James Mackrill         10-019       Tuney       Stass square metres of agricultural land and drains (Manor House Farm, east of Meaux       Benjamin David James Mackrill         10-019       Tuney       Tuney       Stass square metres of agricultural land and drains (Manor House Farm, east of Meaux       Benjamin David James Mackrill         10-019       Tuney       Tuney       Benjamin David James Mackrill       Stass square House Farm, east of Meaux         10-019       Tuney       Tuney       Tuney       Tuney       Stass square House Farm, east of Meaux         10-019       Tuney       Tuney<			Meaux Road	
Image: bit is a bit is				
HU17 9SR       (in respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)         10-018       789 square metres of public road and access splay (Main Road, A1035)       Unknown         10-019       35853 square metres of agricultural land and drains (Manor House Farm, east of Meaux Lane)       Benjamin David James Mackrill         Vest Barn       Furnace Farm       Furnace Farm         Furnace Mathematica       TUNBRIDGE WELLS         TN3 8LE       Furnace Farm				
Image: constraint of a constra				
Image: Constraint of the constraint				
10-019       35853 square metres of agricultural land and drains (Manor House Farm, east of Meaux Lane)       Benjamin David James Mackrill         Vest Barn       Furnace Farm         Furnace Farm       Furnace Lane         Lamberhurst       TUNBRIDGE WELLS         TN3 8LE       TN3 8LE			(in respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)	
Lane) West Barn Furnace Farm Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE	10-018	789 square metres of public road and access splay (Main Road, A1035)	Unknown	
Lane) West Barn Furnace Farm Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE				
Furnace Farm Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE	10-019			
Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE		Lane)		
Lamberhurst TUNBRIDGE WELLS TN3 8LE				
TUNBRIDGE WELLS TN3 8LE				
TN3 8LE				
(in respect of rights to dramage as contained in a conveyance dated 12 June 1986)				
			(in respect or rights to dramage as contained in a conveyance dated 12 june 1980)	



BOOK OF REFERENCE - PART 2 Control of East Right of Versistine           Number on Land Plans         Description of Land         Potential claims under 310 Computory Purchase Act 1995, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008           10-019         VM Sinder # Son Linited Routh         Manor Farm Routh         Beerefay East Vorkhine           10-019         VM Sinder # Son Linited Nanor Farm         Norther Proverptid (Versishing) PLC Loyds Court         Norther Proverptid (Versishing) PLC Loyds Court           10-010         Norther Proverptid (Versishing) PLC Loyds Court         Norther Proverptid (Versishing) PLC Loyds Court         Norther Proverptid (Versishing) PLC Loyds Court           10-010         Immediate the son Linited Immediate the son Linited Norther Proverptid (Versishing) PLC Loyds Court         Norther Proverptid (Versishing) PLC Lo		Dogger Bank South Offshore Wind Farm Development Consent Order		
Number on Land Pins:         Description of Land         Potential dains under S10 Computory Purchase Act 1986, Pur 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008           0.019 cont d         MN Sinker & Son Limited Manor Firm Routh         MN Sinker & Son Limited Manor Firm Routh           0.010         Integration         Potential dains under S10 Computory Purchase Act 1986 Pur 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008           0.010         MN Sinker & Son Limited Manor Firm Routh         Monte Firm Routh           0.011         Potential Act 2014         Morthere Purchase           0.012         Potential Calins under S10 Computory Purchase Act 2014           0.012         Northere         Potential Calins under S10 Computory Purchase Purchase           0.012         Potential Calins under S10 Computory Purchase Act 2014         Northere           0.012         Potential Calins under S10 Computory Purchase Act 2014         Northere           10.012         Potential Calins under S10 Computory Purchase Act 2012         Northere           11.001         119 square metres of agricultural land (Manor House Farm, east of Meaux Lane)         Relignation Durut Lange Machrill Watt Sam         Part Lange Computer Caling Lange Lange Unknown (In respect of rights to drainage as contained in a Lease dated 12 June 1986)         Nt Sinker & Son Limited Manor Farm           11.001         119 square metres of agricultural land (Manor House Farm, east of Meaux Lan			BOOK OF REFERENCE - PART 2	
Land Plans       Other planning Act 2008         100-19       MN Sindler & Son Limited         100-19       MN Sindler & Son Limited         South       Routh         Beverlay       Routhe         HUT 2508       Routhe         (in respect of rights contained in a Lease of a wind turbine site and solar site dated 13 March 2014)         Northem Powerpid (Yorkhire) PLC       Uords Court         Uords Court       T36 Gerg Street         Newsztei Upon Tyne       NE3 E64         Newsztei Upon Tyne       NE3 Bendroff Road         LONDON       Unrown         (in respect of rights contained in a Lease dated 29 March 2012)         11-001       19 square metres of agricultural land (Manor House Farm, east of Meaux Lane)       Benjamin David James Mackrill         Vest Barn       Furnace Lane       Furnace Lane         Lamberthurst       Furnace Lane       Furnace Lane			County of East Riding of Yorkshire	
Land Plans       Other planning Act 2008         100-19       MN Sindler & Son Limited         100-19       MN Sindler & Son Limited         South       Routh         Beverlay       Routhe         HUT 2508       Routhe         (in respect of rights contained in a Lease of a wind turbine site and solar site dated 13 March 2014)         Northem Powerpid (Yorkhire) PLC       Uords Court         Uords Court       T36 Gerg Street         Newsztei Upon Tyne       NE3 E64         Newsztei Upon Tyne       NE3 Bendroff Road         LONDON       Unrown         (in respect of rights contained in a Lease dated 29 March 2012)         11-001       19 square metres of agricultural land (Manor House Farm, east of Meaux Lane)       Benjamin David James Mackrill         Vest Barn       Furnace Lane       Furnace Lane         Lamberthurst       Furnace Lane       Furnace Lane	Number on	Description of Lond	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
cont'd       Manor Fam         Routh       Beverley         East Vorkhire       HU73 S68         (In respect of rights contained in a Lease of a wind turbine site and solar site dated 13 March 2014)       Northem Powerpid (Vorkhire) PLC         Uoyds Court       Northem Powerpid (Vorkhire) PLC       Northem Powerpid (Vorkhire) PLC         Uoyds Court       78 Gersy Street       Newxastie Upon Type         Net Edat       (In respect of rights contained in a Lease dated 29 March 2012)       Northem Net Edat         Not Nether       Net Staf       (In respect of rights contained in a Lease dated 29 March 2012)       Not Nether         11:001       19 square metres of agricultural land (Manor House Fam, east of Meaux Lane)       Benjamin David James Marchill       Respect of rights contained in a Lease dated 29 March 2012)         11:001       19 square metres of agricultural land (Manor House Fam, east of Meaux Lane)       Benjamin David James Marchill       Runow Farm Lamee Lane         In Note       Ket Barn Lamee Lane       Lamberthurst TUNBRIDE VELLS TURBELEAR       TUNBRIDE VELLS TURBELEAR         IN No Net Barn       Kinsher & Son Limited Manor Farm Lamee Lane       Non Farm Lamee Lane       Non Farm Lamee Lane         In No Stafe & Son Limited Manor Farm Lamee Lane       Non Farm Lamee Lane       Non Farm Lamee Lane       Non Farm Lamee Lane         IN Sola       Karu Vorkhire       <	Land Plans	Description of Land		
Image: Section of the section of th	10-019		HN Sinkler & Son Limited	
Image: Provide the second s	cont'd		Manor Farm	
11-001       119 square metres of agricultural land (Manor House Farm, east of Meaus Lane)       Benjamin David James Mackrill         11-001       119 square metres of agricultural land (Manor House Farm, east of Meaus Lane)       Benjamin David James Mackrill         11-001       119 square metres of agricultural land (Manor House Farm, east of Meaus Lane)       Benjamin David James Mackrill         11-001       119 square metres of agricultural land (Manor House Farm, east of Meaus Lane)       Measure Lane         Numer Farm       Formace Farm         Functor Farm       Formace Farm         For Farm       Formace Farm			Routh	
HUI7 958.       HUI7 958.         Image: Contrained in a Lease of a wind turbine site and solar site dated 13 March 2014)       (in respect of rights contained in a Lease of a wind turbine site and solar site dated 13 March 2014)         Image: Contrained in a Lease of a wind turbine site and solar site dated 13 March 2014)       (in respect of rights contained in a Lease of a wind turbine site and solar site dated 13 March 2014)         Image: Contrained in a Lease dated 29 March 2012)       Robin John Mackrill         Image: Contrained in a Lease dated 29 March 2012)       Robin John Mackrill         Image: Contrained in a Lease dated 29 March 2012)       Robin John Mackrill         Image: Contrained in a Lease dated 29 March 2012)       Robin John Mackrill         Image: Contrained in a Lease dated 29 March 2012)       Robin John Mackrill         Image: Contrained in a Lease dated 29 March 2012)       Robin John Mackrill         Image: Contrained in a Lease dated 29 March 2012)       Robin John Mackrill         Image: Contrained in a Lease dated 29 March 2012)       Robin John Mackrill         Image: Contrained in a Lease dated 29 March 2012)       Robin John Mackrill         Image: Contrained in a Lease dated 29 March 2012)       Robin John Mackrill         Image: Contrained in a Lease dated 29 March 2012)       Robin John Mackrill         Image: Contrained in a Lease dated 29 March 2012)       Robin John Mackrill         Image: Contrained in a Conveyance dated			Beverley	
Image: Second				
11-001       119 square metres of agricultural land (Manor House Farm, east of Meaux Lane)       Benjamin David James Mackrill Unknown (In respect of rights contained in a lease dated 29 March 2012)         11-001       119 square metres of agricultural land (Manor House Farm, east of Meaux Lane)       Benjamin David James Mackrill West Barn Furnace Larne Lamberhurst Lamberhurst Lamberhurst Lamberhurst Lamberhurst Lamberhurst Lamberhurst Manor Farm Routh Benjamin David James Mackrill West Barn Furnace Larne Lamberhurst			HU17 9SR	
Image: Second			(in respect of rights contained in a Lease of a wind turbine site and solar site dated 13 March 2014)	
75 Grey Street         Newcasate Upon Tyne         NE1 GAF         (in respect of rights contained in a Lease dated 29 March 2012)         Robin John Mackrill         38 Biandford Road         LONDON         W4 102         Unknown         (in respect of rights contained in a Conveyance dated 12 June 1986)         Unknown         (in respect of rights contained in a lease dated 29 March 2012)         11-001       119 square metres of agricultural land (Manor House Farm, east of Meaux Lane)         Benjamin David James Mackrill         West Barn         Furnase Farm         Routh         Beverley         East torkshire         HN Sinkler & Son Limited         Manor Farm     <				
Newcastle Upon Tyne       Newcastle Upon Tyne         NEI GAF       (in respect of rights contained in a Lease dated 29 March 2012)         Robin John Mackrill       38 Biandford Road         L0NDON       W4 1DX         (in respect of rights contained in a Conveyance dated 12 June 1986)       Unknown         Unknown       W4 1DX         (in respect of rights contained in a lease dated 29 March 2012)       Unknown         11-001       119 square metres of agricultural land (Manor House Farm, east of Meaux Lane)       Benjamin David James Mackrill         Vest Barn       Furnace Farm       Furnace Farm         (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)       HN Sinkler & Son Limited         HW Sinkler & Son Limited       Manor Farm       Beorethey         East Ortkihrine       HW Sinkler & Son Limited         Manor Farm       Boriarm       Boriarm         Boriarm       Boriarm       Boriarm         Boriard       HW Sinkler & Son Limited       Manor Farm         Manor Farm       Boriarm       Boriarm         Boriarm       Boriarm       Boriarm         Boriard       Boriarm       Boriarm         Boriard       Boriarm       Boriarm         Boriard       Boriarm       Boriarm <th></th> <th></th> <th></th>				
NII 6AF       Robin Mackrill         Babandford Road       LONDON         VM 1DX       Unknown         (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)       Unknown         (in respect of rights contained in a lease dated 29 March 2012)       Unknown         11-001       119 square metres of agricultural land (Manor House Farm, east of Meaux Lane)       West Bain David James Mackrill         West Bain David James Mackrill       West Bain David James Mackrill       West Bain David James Mackrill         INI-001       IN Sinkler & Son Limited       In a Conveyance dated 12 June 1986)       HN Sinkler & Son Limited         Robin David James Mackrill       West Bain David James Mackrill       West Bain David James Mackrill       West Bain David James Mackrill         Wind David James Mackrill       West Bain David James Mackrill       West Bain David James Mackrill       West Bain David James Mackrill         Wind David James Mackrill       West Bain David James Mackrill       West Bain David James Mackrill       West Bain David James Mackrill         Wind David James Mackrill       West Bain David James Mackrill       West Bain David James Mackrill       West Bain David James Mackrill         Wind David James Mackrill       West Bain David James Mackrill       West Bain David James Mackrill       West Bain David James Mackrill         Wind David James Mackrill       West				
Image: severe of rights contained in a Lease dated 29 March 2012)         Robin John Mackrill         38 Blandford Road         UNNOWN         Unknown         (in respect of rights contained in a Conveyance dated 12 June 1986)         Unknown         (in respect of rights contained in a lease dated 29 March 2012)         11-001       119 square metres of agricultural land (Manor House Farm, east of Meaux Lane)         Benjamin David James Mackrill         West Barn         Furnace Lane         Lamberhurst         TINNBIDGE WELLS         TN NBIDGE WELLS         TIN SUBGE WELLS         TIN SUBGE WELLS         TIN SUBGE WELLS         TIN SUBGE WELLS         TANS BLE         Youth         Berverlay         East Yorkshire         HUI '75R				
Robin John Mackrill       38 Blandford Road         UNDON       W4 1DX         (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)         Unknown       (in respect of rights contained in a lease dated 29 March 2012)         11-001       119 square metres of agricultural land (Manor House Farm, east of Meaux Lane)       Benjamin David James Mackrill         West Barn       Furnace Farm         Furnace Farm       Furnace Farm         Construction       Construction         TN3 BLE       (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)         HN Sinkler & Son Limited       Manor Farm         Manor Farm       Routh         Routing Structure       Son Limited         Manor Farm       Routh         Routing Structure       Son Limited         Manor Farm       Routh         Routing       Son Limited         Manor Farm       Routh         Routh       Son Limited         Manor Farm       Routh         Routh       Son Limited         Manor Farm       HUT 35 R				
Big			(in respect of rights contained in a Lease dated 29 March 2012)	
LONDON W4 1DX (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986) Unknown (in respect of rights contained in a lease dated 29 March 2012) 11-001 119 square metres of agricultural land (Manor House Farm, east of Meaux Lane) Benjamin David James Mackrill West Barn Furnace Lane Lamberhurst TUNBRIDGE WELLS TNS 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986) HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 95R			Robin John Mackrill	
W4 1DX       (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)         Unknown       (in respect of rights contained in a lease dated 29 March 2012)         11-001       119 square metres of agricultural land (Manor House Farm, east of Meaux Lane)       Benjamin David James Mackrill         West Barn       Furnace Earm         Furnace Lane       Lamberhurst         TUNBRIDGE WELLS       TN3 &LE         (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)       HN Sinkler & Son Limited         Manor Farm       Routh       Beverley         East Yorkshire       HUX7 SSR			38 Blandford Road	
Image:				
11-001       119 square metres of agricultural land (Manor House Farm, east of Meaux Lane)       Benjamin David James Mackrill         West Barn       Furnace Farm         Furnace Farm       Furnace Farm         Furnace Intervention       Furnace Farm         Furnace Farm       Furnace Farm         Furnace Cane       Lamberhurst         TUNBRIOGE WELLS       TN3 8LE         (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)         HN Sinkler & Son Limited         Manor Farm         Routh         Beveley         East Yorkshire         HU17 9SR				
Image: Instant Contrained In a lease dated 29 March 2012)         Image: Instant Contrained In a lease dated 29 March 2012)         Image: Instant Contrained Instant Content Contrained Instant Contrained Instant Con			(in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)	
11-001       119 square metres of agricultural land (Manor House Farm, east of Meaux Lane)       Benjamin David James Mackrill         Vest Barn       Furnace Farm         Furnace Farm       Furnace Farm         Furnace Lane       Lamberhurst         TUNSRIDGE WELLS       TN3 8LE         (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)       HN Sinkler & Son Limited         Manor Farm       Routh         Beverley       East Yorkshire         HU17 9SR       HU17 9SR			Unknown	
West Barn Furnace Farm Furnace Lame Lamberhurst TUNBRIDGE WELLS TN3 BLE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986) HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR			(in respect of rights contained in a lease dated 29 March 2012)	
Furnace Farm Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986) HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR	11-001	119 square metres of agricultural land (Manor House Farm, east of Meaux Lane)	Benjamin David James Mackrill	
Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986) HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR			West Barn	
Lamberhurst TUNBRIDGE WELLS TN3 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986) HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR			Furnace Farm	
TUNBRIDGE WELLS         TN3 8LE         (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)         HN Sinkler & Son Limited         Manor Farm         Routh         Beverley         East Yorkshire         HU17 9SR				
TN3 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986) HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR				
(in respect of rights to drainage as contained in a Conveyance dated 12 June 1986) HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR				
HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR				
Manor Farm Routh Beverley East Yorkshire HU17 9SR			(in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)	
Routh Beverley East Yorkshire HU17 9SR				
Beverley East Yorkshire HU17 9SR				
East Yorkshire HU17 9SR				
HU17 9SR				
(in respect of rights contained in a Lease of a wind turbine site and solar site dated 13 March 2014)				
			(In respect of rights contained in a Lease of a wind turbine site and solar site dated 13 March 2014)	



Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2	
	County of East Riding of Yorkshire	
Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
	Northern Powergrid (Yorkshire) PLC	
	Lloyds Court	
	78 Grey Street	
	Newcastle Upon Tyne	
	NE1 6AF	
	(in respect of rights contained in a Lease dated 29 March 2012)	
	Robin John Mackrill	
	38 Blandford Road	
	LONDON	
	W4 1DX	
	(in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)	
	Unknown	
	(in respect of rights contained in a lease dated 29 March 2012)	
548 square metres of public road and verges (Meaux Lane)	Unknown	
246 square metres of agricultural land (Manor House Farm, west of Meaux Lane)	Benjamin David James Mackrill	
	West Barn	
	Furnace Farm Furnace Lane	
	Lamberhurst	
	TUNBRIDGE WELLS	
	TN3 8LE	
	(in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)	
	HN Sinkler & Son Limited	
	Manor Farm	
	Routh	
	Beverley	
	East Yorkshire	
	HU17 9SR	
	(in respect of a Lease of a wind turbine site and solar site dated 13 March 2014)	
	Description of Land	



	Dogger Bank	South Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on		Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)
Land Plans	Description of Land	of the Planning Act 2008
11-003		Robin John Mackrill
cont'd		38 Blandford Road
		LONDON
		W4 1DX
		(in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)
11-004	775 square metres of public road and verges (Meaux Lane)	Unknown
11-005	136 square metres of agricultural land (Manor House Farm, east of Meaux Lane)	Benjamin David James Mackrill
11-005	150 Square metres of agricultural land (Marior House Farm, east of Meaux Lane)	West Barn
		Furnace Farm
		Furnace Lane
		Lamberhurst
		TUNBRIDGE WELLS
		TN3 8LE
		(in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)
		HN Sinkler & Son Limited
		Manor Farm
		Routh
		Beverley
		East Yorkshire
		HU17 9SR
		(in respect of rights contained in a Lease of a wind turbine site and solar site dated 13 March 2014)
		Northern Powergrid (Yorkshire) PLC
		Lloyds Court
		78 Grey Street
		Newcastle Upon Tyne
		NE1 6AF
		(in respect of rights contained in a Lease dated 29 March 2012)
		Robin John Mackrill
		38 Blandford Road
		LONDON
		W4 1DX
		(in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)



	Dogger Bank Sou	th Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-006	321 square metres of public road and verges (Meaux Lane)	Unknown
11-007	160 square metres of agricultural land (Manor House Farm, west of Meaux Lane)	Benjamin David James Mackrill West Barn Furnace Farm Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986) HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR (in respect of a Lease of a wind turbine site and solar site dated 13 March 2014) Robin John Mackrill 38 Blandford Road LONDON W4 1DX (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)
	73946 square metres of agricultural land, access track and drain (Manor House Farm, east of A1035, Routh and west of Meaux Lane)	Benjamin David James Mackrill West Barn Furnace Farm Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans	Description of Land	of the Planning Act 2008	
11-008		HN Sinkler & Son Limited	
cont'd		Manor Farm	
		Routh	
		Beverley	
		East Yorkshire	
		HU17 9SR	
		(in respect of a Lease of a wind turbine site and solar site dated 13 March 2014)	
		Robin John Mackrill	
		38 Blandford Road	
		LONDON	
		W41DX	
		(in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)	
11-009	24147 square metres of agricultural land (east of A1035, Routh)	Benjamin David James Mackrill	
		West Barn	
		Furnace Farm	
		Furnace Lane	
		Lamberhurst	
		TUNBRIDGE WELLS TN3 8LE	
		(in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)	
		(in respect of rights to drainage as contained in a conveyance dated 12 June 1980)	
		HN Sinkler & Son Limited	
		Manor Farm	
		Routh	
		Beverley	
		East Yorkshire	
		HU17 9SR	
		(in respect of a Lease of a wind turbine site and solar site dated 13 March 2014)	
		Dable Jake Mashell	
		Robin John Mackrill	
		38 Blandford Road LONDON	
		W4 1DX	
		(in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)	
		(in respect or rights to drainage as contained in a conveyance dated 12 June 1900)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Lond	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans	Description of Land	of the Planning Act 2008	
11-010	1415 square metres of public road, verges, footway and access splay (A1035, Routh)	Howard Noel Sinkler	
		Manor House	
		Meaux Road	
		Routh	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 9SR	
		(in respect of rights to maintain hedges as contained in a Conveyance dated 6 June 1974)	
		lan Harold Sinkler	
		Manor House	
		Meaux Road	
		Routh	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 9SR	
		(in respect of rights to maintain hedges as contained in a Conveyance dated 6 June 1974)	
		The Executor of the Estate of the Late Harold Sinkler	
		Manor House	
		Meaux Road	
		Routh	
		BEVERLEY	
		HU17 9SR	
		(in respect of rights to maintain hedges as contained in a Conveyance dated 6 June 1974)	
11-011	1056 square metres of public road, verges, footway and access splay (A1035, Routh)	Unknown	
11-013	520 square metres of agricultural land and access track (Hall Farm and Field House Farm,	Doggerbank Offshore Wind Farm Project 1 Projco Limited	
	south of Tickton Bypass, A1035)	No.1 Forbury Place	
		43 Forbury Road	
		Reading	
		RG1 3JH	
		(in respect of rights and easements contained in a Lease dated 21 February 2020. Also in respect of rights granted as contained in a Deed dated 12 October 2023.)	



	Dogger Ban	k South Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on	Description of Lond	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)
Land Plans	Description of Land	of the Planning Act 2008
11-013		Doggerbank Offshore Wind Farm Project 2 Projco Limited
cont'd		No.1 Forbury Place
		43 Forbury Road
		Reading
		RG1 3JH
		(in respect of rights of access to conduct works as contained in a Deed of Grant dates 6 April 1984. Also in respect of rights contained in a
		Lease dated 21 February 2020)
		East Riding Of Yorkshire Council
		County Hall
		Cross Street
		BEVERLEY
		East Riding of Yorkshire
		HU17 9BA
		(in respect of rights contained in an Option Agreement dated 24 October 2017)
		Hall Farm Wind Farm Limited
		C/O Res Limited
		Beaufort Court
		Egg Farm Lane
		Kings Langley
		Hertfordshire
		WD4 8LR
		(in respect of rights contained in a Lease dated 24 November 2011)
		Northern Powergrid (Yorkshire) PLC
		Lloyds Court
		78 Grey Street
		Newcastle Upon Tyne
		NE1 6AF
		(in respect of rights contained in a Lease)
11-015	607 square metres of public road, verges and access splay (A1035, Routh)	Unknown



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2	
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans	Description of Land	of the Planning Act 2008	
11-017	4338 square metres of agricultural land, access track, hardstanding and access splay (Hall	Doggerbank Offshore Wind Farm Project 1 Projco Limited	
	Farm and Field House Farm, north of Tickton Bypass, A1035)	No.1 Forbury Place	
		43 Forbury Road	
		Reading	
		RG1 3JH	
		(in respect of rights and easements contained in a Lease dated 21 February 2020. Also in respect of rights granted as contained in a Deed dated 12 October 2023.)	
		Doggerbank Offshore Wind Farm Project 2 Projco Limited	
		No.1 Forbury Place	
		43 Forbury Road	
		Reading	
		RG1 3JH	
		(in respect of rights of access to conduct works as contained in a Deed of Grant dates 6 April 1984. Also in respect of rights contained in a	
		Lease dated 21 February 2020)	
		East Riding Of Yorkshire Council	
		County Hall	
		Cross Street BEVERLEY	
		East Riding of Yorkshire	
		HU17 9BA	
		(in respect of rights contained in an Option Agreement dated 24 October 2017)	
		Hall Farm Wind Farm Limited	
		C/O Res Limited	
		Beaufort Court	
		Egg Farm Lane	
		Kings Langley Hertfordshire	
		WD4 8LR	
		(in respect of rights contained in a Lease dated 24 November 2011)	
		Northern Powergrid (Yorkshire) PLC	
		Lloyds Court	
		78 Grey Street	
		Newcastle Upon Tyne	
		NE1 6AF	
		(in respect of rights contained in a Lease dated 14 September 2018)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2	
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans		of the Planning Act 2008	
11-018	175779 square metres of agricultural land, access tracks, hedgerows and drains (Hall Farm	Doggerbank Offshore Wind Farm Project 1 Projco Limited	
	and Field House Farm, east of Eske Lane and west of A1035, Routh)	No.1 Forbury Place	
		43 Forbury Road	
		Reading	
		RG1 3JH	
		(in respect of rights and easements contained in a Lease dated 21 February 2020. Also in respect of rights granted as contained in a Deed	
		dated 12 October 2023.)	
		Doggerbank Offshore Wind Farm Project 2 Projco Limited	
		No.1 Forbury Place	
		43 Forbury Road	
		Reading	
		RG1 3JH	
		(in respect of rights of access to conduct works as contained in a Deed of Grant dates 6 April 1984. Also in respect of rights contained in a	
		Lease dated 21 February 2020)	
		East Riding Of Yorkshire Council	
		County Hall	
		Cross Street	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 9BA	
		(in respect of rights contained in an Option Agreement dated 24 October 2017)	
		Hall Farm Wind Farm Limited	
		C/O Res Limited	
		Beaufort Court	
		Egg Farm Lane	
		Kings Langley	
		Hertfordshire	
		WD4 8LR	
		(in respect of rights contained in a Lease dated 24 November 2011)	
		Northern Powergrid (Yorkshire) PLC	
		Lloyds Court	
		78 Grey Street	
		Newcastle Upon Tyne	
		NE1 6AF	
		(in respect of rights contained in a Lease dated 14 September 2018)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans		of the Planning Act 2008	
11-018		Yorkshire Water Services Limited	
cont'd		Western House	
		Western Way	
		Halifax Road	
		Bradford	
		West Yorkshire	
		BD6 2SZ	
		(in respect of rights to maintain a watermain as contained in a Deed dated 6 April 1984)	
	20685 square metres of agricultural land and drain (Hall Farm and Field House Farm, east	Doggerbank Offshore Wind Farm Project 1 Projco Limited	
	of Eske Lane)	No.1 Forbury Place	
		43 Forbury Road	
		Reading	
		RG1 3JH	
		(in respect of rights and easements contained in a Lease dated 21 February 2020. Also in respect of rights granted as contained in a Deed	
		dated 12 October 2023.)	
		Doggerbank Offshore Wind Farm Project 2 Projco Limited	
		No.1 Forbury Place	
		43 Forbury Road	
		Reading	
		RG1 3JH	
		(in respect of rights of access to conduct works as contained in a Deed of Grant dates 6 April 1984. Also in respect of rights contained in a	
		Lease dated 21 February 2020)	
		East Riding Of Yorkshire Council	
		County Hall	
		Cross Street	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 9BA	
		(in respect of rights contained in an Option Agreement dated 24 October 2017)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans	Description of Land	of the Planning Act 2008	
12-001		Hall Farm Wind Farm Limited	
cont'd		C/O Res Limited	
		Beaufort Court	
		Egg Farm Lane	
		Kings Langley	
		Hertfordshire	
		WD4 8LR	
		(in respect of rights contained in a Lease dated 24 November 2011)	
		Northern Powergrid (Yorkshire) PLC	
		Lloyds Court	
		78 Grey Street	
		Newcastle Upon Tyne	
		NE1 6AF	
		(in respect of rights contained in a Lease dated 14 September 2018)	
12-002	127 square metres of agricultural land and drain (east of Eske Lane)	Unknown	
12-003	959 square metres of public road and verge (Eske Lane)	Unknown	
12-004	85 square metres of agricultural land and drain (east of Eske Lane)	Unknown	
12-005	564 square metres of public road and verge (Eske Lane)	Unknown	
12-006	5985 square metres of public road and verges (Eske Lane and Tickton Bypass, A1035)	Unknown	
12-007	25 square metres of agricultural land and copse (north of Tickton Bypass, A1035 and west	John Scott	
	of Eske Lane)	Tickton Farm	
		56 Main Street	
		Tickton	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 9RZ	
		(in respect of rights to inspect, clean, maintain, repair and renew the buildings, walls and fences contained in a Transfer dated 19	
		December 2017)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
12-009	241 square metres of public road and verges (Tickton Bypass, A1035)	Unknown (in respect of rights contained in a Conveyance dated 19 September 1972)	
12-011	96989 square metres of agricultural land, hedgerows and drains (west of Eske Lane)	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights and rights of access)	
12-013	101 square metres of river bank (east of the River Hull, Beverley)	Unknown	
	770 square metres of grassland, river bank, public bridleway (Leconfield Bridleway No.27) and public footpath (Leconfield Footpath No.33) (west of the River Hull, Beverley)	Unknown	
	2740 square metres of grassland, river bank, public bridleway (Leconfield Bridleway No.27) and public footpath (Leconfield Footpath No.33) (west of the River Hull, Beverley)	Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of rights contained in a Transfer scheme dated 1 September 1989)	
	71530 square metres of agricultural land and copse (Carr Farm, east of Carr Road and north of Hull Bridge Road, A1035)	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of rights of access)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-004		Ministry of Defence	
cont'd		Main Building	
		Horse Guards Avenue	
		Whitehall	
		LONDON	
		SW1A 2HB	
		(in respect of right of access and the free passage of water and electricity contained in a Conveyance dated 25 March 1974)	
		Phillip Brumfield	
		Dunkenhill Farm	
		Highgate	
		Cherry Burton	
		BEVERLEY	
		HU17 7RT	
		(as The Executor of the Estate of The Late David Roy Duncan Brumfield)	
		(in respect of rights contained in a Transfer dated 26 February 1993)	
		The Executor Of The Estate Of The Late David Roy Duncan Brumfield	
		The Beeches	
		Seven Corners Lane	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 7AJ	
		(in respect of rights contained in a Transfer dated 26 February 1993)	
13-005	243 square metres of drain (west of the River Hull, Beverley)	Beverley & North Holderness Internal Drainage Board	
		Derwent House	
		Crockey Hill	
		York	
		YO19 4SR	
		(in respect of riparian rights)	
		Unknown	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of East Riding of Yorkshire		
Number on	Description of land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans	Description of Land	of the Planning Act 2008	
13-006	185 square metres of agricultural land and dike (South Bullock Dike, Carr Farm, east of Carr	Beverley & North Holderness Internal Drainage Board	
	Road and north of Hull Bridge Road, A1035)	Derwent House	
		Crockey Hill	
		York	
		YO19 4SR	
		(in respect of riparian rights and rights of access)	
		Ministry of Defence	
		Main Building	
		Horse Guards Avenue	
		Whitehall	
		LONDON	
		SW1A 2HB	
		(in respect of right of access and the free passage of water and electricity contained in a Conveyance dated 25 March 1974)	
		Phillip Brumfield	
		Dunkenhill Farm	
		Highgate	
		Cherry Burton	
		BEVERLEY	
		HU17 7RT	
		(as The Executor of the Estate of The Late David Roy Duncan Brumfield)	
		(in respect of rights contained in a Transfer dated 26 February 1993)	
		The Executor Of The Estate Of The Late David Roy Duncan Brumfield	
		The Beeches	
		Seven Corners Lane	
		BEVERLEY	
		East Riding of Yorkshire	
		(in respect of rights contained in a Transfer dated 26 February 1993)	
13-007	4825 square metres of agricultural land and dike (South Bullock Dike, east of Carr Road)	Beverley & North Holderness Internal Drainage Board	
		Derwent House	
		Crockey Hill	
		York	
		YO19 4SR	
		(in respect of riparian rights)	
		l	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Lond	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans	Description of Land	of the Planning Act 2008	
13-007		Unknown	
cont'd		(in respect of rights to service media reserved by a Transfer dated 22 June 1989)	
13-008		Devender: O Manthe Haldamann Indonesia Devend	
	4158 square metres of agricultural land, culvert and drain (Molescroft Grange Farm, east of		
	Carr Road)	Derwent House	
		Crockey Hill York	
		YO19 4SR	
		(in respect of riparian rights and rights of access)	
		Michael Jonathan Watson Hall	
		Manor Farm	
		Manor House Lane	
		Walkington	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 8SU	
		(in respect of restrictions contained in a Transfer dated 12 November 2012)	
		S R J Developments (Yorkshire) Limited	
		Belgrave House	
		15 Belgrave Crescent	
		SCARBOROUGH	
		North Yorkshire	
		Y011 1UB	
		(in respect of rights of access and others listed in a Transfer dated 14 July 2000)	
		Yorkshire Water Services Limited	
		Western House	
		Western Way	
		Halifax Road	
		Bradford	
		West Yorkshire	
		BD6 2SZ	
		(in respect of rights and easements relating to water mains, passage of water, entry and covenants listed in a Deed dated 26 September	
		1995)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2			
	County of East Riding of Yorkshire			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
13-009	74687 square metres of agricultural land and drain (Molescroft Grange Farm, north of	Beverley & North Holderness Internal Drainage Board		
	Grange Way, A1035)	Derwent House		
		Crockey Hill		
		York		
		YO19 4SR		
		(in respect of rights of access)		
		Michael Jonathan Watson Hall		
		Manor Farm		
		Manor House Lane		
		Walkington		
		BEVERLEY		
		East Riding of Yorkshire		
		HU17 8SU		
		(in respect of restrictions contained in a Transfer dated 12 November 2012)		
		S R J Developments (Yorkshire) Limited		
		Belgrave House		
		15 Belgrave Crescent		
		SCARBOROUGH		
		North Yorkshire		
		Y0111UB		
		(in respect of rights of access and others listed in a Transfer dated 14 July 2000)		
		Yorkshire Water Services Limited		
		Western House		
		Western Way		
		Halifax Road		
		Bradford		
		West Yorkshire		
		BD6 2SZ		
		(in respect of rights and easements relating to water mains, passage of water, entry and covenants listed in a Deed dated 26 September		
		1995)		



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans		of the Planning Act 2008	
13-010	47466 square metres of agricultural land, hedgerows, drains and access tracks (Molescroft	Beverley & North Holderness Internal Drainage Board	
	Grange Farm, east of Grange Way, A1035)	Derwent House	
		Crockey Hill	
		York	
		YO19 4SR	
		(in respect of riparian rights and rights of access)	
		Michael Jonathan Watson Hall	
		Manor Farm	
		Manor House Lane	
		Walkington	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 8SU	
		(in respect of restrictions contained in a Transfer dated 12 November 2012)	
		Network Rail Infrastructure Limited	
		Waterloo General Office	
		LONDON	
		SE1 8SW	
		(in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a	
		Deed of Release dated 27 February 2019)	
		S R J Developments (Yorkshire) Limited	
		Belgrave House	
		15 Belgrave Crescent	
		SCARBOROUGH	
		North Yorkshire	
		YO11 1UB	
		(in respect of rights of access and others listed in a Transfer dated 14 July 2000)	



	Dogger Bank Sc	outh Offshore Wind Farm Development Consent Order	
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans	Description of Land	of the Planning Act 2008	
13-010		Yorkshire Water Services Limited	
cont'd		Western House	
		Western Way	
		Halifax Road	
		Bradford	
		West Yorkshire	
		BD6 2SZ	
		(in respect of rights and easements relating to water mains, passage of water, entry and covenants listed in a Deed dated 26 September	
		1995)	
13-011	2687 square metres of agricultural land, drains and access track (east of Grange Way,	Beverley & North Holderness Internal Drainage Board	
	A1035)	Derwent House	
		Crockey Hill	
		York	
		YO19 4SR	
		(in respect of riparian rights and rights of access)	
		Unknown	
		(in respect of rights to service media reserved by a Transfer dated 22 June 1989)	
13-013	395 square metres of public road (Hull Bridge Road, A1035)	Unknown	
13-014	1604 square metres of railway track and verges (east of Carr Road)	Unknown	
13-015	11442 square metres of agricultural land and pond (north of Carr Road)	Network Rail Infrastructure Limited	
		Waterloo General Office	
		LONDON SE1 8SW	
		(in respect of rights to access a level crossing contained in a Deed of Release dated 12 September 1995)	
		Phillip Brumfield	
		Dunkenhill Farm	
		Highgate	
		Cherry Burton	
		BEVERLEY	
		HU17 7RT	
		(as The Executor of the Estate of the Late Barbara Brumfield and The Late David Roy Duncan Brumfield)	
		(in respect of rights contained in a Transfer dated 22 June 1989)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans		of the Planning Act 2008	
13-015		S R J Developments (Yorkshire) Limited	
cont'd		Belgrave House	
		15 Belgrave Crescent	
		SCARBOROUGH	
		North Yorkshire	
		Y011 1UB	
		(in respect of rights, agreements and covenants listed in a Transfer dated 14 July 2000)	
		The Executor Of The Estate Of The Late David Roy Duncan Brumfield	
		The Beeches	
		Seven Corners Lane	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 7AJ	
		(in respect of rights contained in a Transfer dated 22 June 1989)	
13-016	8219 square metres of agricultural land, access track and public footpath (Molescroft	Catherine Suzanne Mace	
	Footpath No.5) (Low Parks Farm, north of Carr Road)	The Chestnuts	
		Carr Road	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 7JZ	
		(in respect of rights of access as contained in an Indenture dated 20 September 1920)	
		Thomas Anthony Richard Mace	
		The Chestnuts	
		Carr Road	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 7JZ	
		(in respect of rights of access as contained in an Indenture dated 20 September 1920)	
13-017	59747 square metres of agricultural land and drains (north of Carr Road)	Network Rail Infrastructure Limited	
		Waterloo General Office	
		LONDON	
		SE1 8SW	
		(in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a	
		Deed of Release dated 27 February 2019)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-017 cont'd		S R J Developments (Yorkshire) Limited Belgrave House 15 Belgrave Crescent SCARBOROUGH North Yorkshire YO11 1UB (in respect of rights of access and others listed in a Transfer dated 14 July 2000)	
	21440 square metres of agricultural land (north of Carr Road, Molescroft and north of Ings Road)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a Deed of Release dated 27 February 2019) S R J Developments (Yorkshire) Limited Belgrave House 15 Belgrave Crescent SCARBOROUGH North Yorkshire YO11 1UB (in respect of rights of access and others listed in a Transfer dated 14 July 2000)	
14-002	13 square metres of public road and verge (Carr Road)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rights to access a level crossing contained in a Deed of Release dated 12 September 1995) Phillip Brumfield Dunkenhill Farm Highgate Cherry Burton BEVERLEY HU17 7RT (as The Executor of the Estate of the Late Barbara Brumfield and The Late David Roy Duncan Brumfield) (in respect of rights contained in a Transfer dated 22 June 1989)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans	Description of Land	of the Planning Act 2008	
14-002		The Executor Of The Estate Of The Late David Roy Duncan Brumfield	
cont'd		The Beeches	
		Seven Corners Lane	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 7AJ	
		(in respect of rights contained in a Transfer dated 22 June 1989)	
		Unknown	
		(in respect of rights contained in a Conveyance dated 11 September 1975)	
14-003	2309 square metres of public road and verges (Ings Road)	Unknown	
14-004	1421 square metres of public roads and verges (Ings Road and Grange Way, A1035)	Richmond Properties (UK) Limited	
		Field Head	
		Driffield Road	
		Molescroft	
		Beverley	
		East Yorkshire	
		HU17 7LU	
		(in respect of rights of access to development sites as contained in a Transfer dated 31 December 1993)	
		Shirethorn Limited	
		Unit 8B	
		Marina Court	
		Castle Street	
		HULL	
		East Riding of Yorkshire	
		HU1 1TJ	
		(in respect of rights of access to development sites as contained in a Transfer dated 31 December 1993)	
		Tamara Clara Watson Hall	
		Tamara Clare Watson Hall Manor Farm	
		Manor House Lane	
		Walkington	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 8SU	
		(in respect of rights contained in a Transfer dated 22 June 1989)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
14-008	26576 square metres of agricultural land (east of Driffield Road, A164)	Unknown (in respect of rights to service media reserved by a Transfer dated 22 June 1989)	
14-009	5 square metres of agricultural land (east of Driffield Road, A164)	Unknown (in respect of rights to service media reserved by a Transfer dated 22 June 1989)	
14-010	42 square metres of public road and verges (Driffield Road, A164)	Unknown	
14-011	1998 square metres of public road and verges (Driffield Road, A164)	Unknown	
14-012	3 square metres of agricultural land (west of Driffield Road, A164)	Unknown (in respect of rights to service media reserved by a Transfer dated 22 June 1989)	
14-013	34952 square metres of agricultural land (west of Driffield Road, A164)	Unknown (in respect of rights to service media reserved by a Transfer dated 22 June 1989)	
14-014	63766 square metres of agricultural land and hedgerow (Constitution Hill Farm, north of Malton Road, A1035)	Unknown (in respect of rights relating to a sewer listed in a Conveyance dated 9 November 1920)	
15-001	22062 square metres of agricultural land and hedgerow (Constitution Hill Farm, north of Malton Road, A1035)	Unknown (in respect of rights relating to a sewer listed in a Conveyance dated 9 November 1920)	
15-002	973 square metres of public road, access splay and verges (Malton Road, A1035)	Unknown	
15-004	1845 square metres of public road and verges (Malton Road, A1035)	Unknown	
15-005	15652 square metres of agricultural land (Molescroft Grange Farm, south of Malton Road, A1035)	Michael Jonathan Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (in respect of restrictions contained in a Transfer dated 12 November 2012)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
15-005 cont'd 15-006	30513 square metres of agricultural land (Molescroft Grange Farm, south of Malton Road, A1035)	S R J Developments (Yorkshire) Limited Belgrave House 15 Belgrave Crescent SCARBOROUGH North Yorkshire YO11 1UB (in respect of rights of access and others listed in a Transfer dated 14 July 2000) Michael Jonathan Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (in respect of restrictions contained in a Transfer dated 12 November 2012) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a Deed of Release dated 27 February 2019) S R J Developments (Yorkshire) Limited Belgrave House	
		15 Belgrave Crescent SCARBOROUGH North Yorkshire Y011 1UB (in respect of rights of access and others listed in a Transfer dated 14 July 2000)	
15-007	166 square metres of drain (south of Malton Road, A1035)	Unknown	



	Dogger Bank South Offshore Wind Farm Development Consent Order			
	BOOK OF REFERENCE - PART 2			
	County of East Riding of Yorkshire			
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)		
Land Plans	Description of Land	of the Planning Act 2008		
15-008	15547 square metres of agricultural land and hedgerow (south of Malton Road, A1035)	The Agricultural Mortgage Corporation PLC		
		Keens House		
		Anton Mill Road		
		ANDOVER		
		Hampshire		
		SP10 2NQ		
		(in respect of rights reserved by a Deed dated 4 March 1988)		
16-001	48573 square metres of agricultural land, access track and hedgerows (Mount Pleasant	Lindum Developments Limited		
	Farm, north of York Road)	Lindum Business Park		
		Station Road		
		North Hykeham		
		Lincoln		
		LN6 3QX		
		(in respect of rights of access and covenants listed in a Transfer dated 2 March 2021)		
		Unknown		
		(in respect of rights of way, light and drainage listed in a Conveyance dated 11 September 1953)		
16-002	1418 square metres of public road, verges, scrubland and access splay (York Road)	Lindum Developments Limited		
		Lindum Business Park		
		Station Road		
		North Hykeham		
		Lincoln		
		LN6 3QX		
		(in respect of rights of access and covenants listed in a Transfer dated 2 March 2021)		
		Unknown		
		(in respect of rights of way, light and drainage listed in a Conveyance dated 11 September 1953)		
16-003	624 square metres of scrubland (Mount Pleasant Farm, north of York Road)	Lindum Developments Limited		
		Lindum Business Park		
		Station Road		
		North Hykeham		
		Lincoln		
		(in respect of rights of access and covenants listed in a Transfer dated 2 March 2021)		
		Unknown		
		(in respect of rights of way, light and drainage listed in a Conveyance dated 11 September 1953)		



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-004	91 square metres of public road, verge and access splay (York Road, Beverley) (excluding all interests of the Crown)	Unknown	
16-005	27 square metres of public road and verges (York Road, Beverley) (excluding all interests of the Crown)	Unknown	
	133 square metres of public road and verges (York Road, Beverley) (excluding all interests of the Crown)	Unknown (in respect of restrictive covenants imposed before 11 February 2009)	
	2983 square metres of public road and verges (York Road, Beverley) (excluding all interests of the Crown)	Unknown (in respect of restrictive covenants imposed before 11 February 2009)	
16-008	32635 square metres of agricultural land and hedgerow (Mount Pleasant Farm, south of York Road)	Unknown (in respect of rights of way, light and drainage listed in a Conveyance dated 11 September 1953)	
17-001	832 square metres of public road and verges (Newbald Road)	Unknown	
17-004	1387 square metres of public road and verge (Broadgate, B1230)	Unknown	
17-005	1344 square metres of public road, access splays and verges (Broadgate, B1230)	Unknown	
17-006	541 square metres of public road and verge (Broadgate, B1230)	Unknown	
17-007	430 square metres of grassland (south of Broadgate, B1230)	Bryant Homes Northern Limited Gate House Turnpike Road HIGH WYCOMBE Buckinghamshire HP12 3NR (in respect of the rights of access to sewage reserved by the Conveyance dated 28 September 1967)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans	Description of Land	of the Planning Act 2008	
17-007		The Lord Mayor Aldermen and Citizens of the City and County of Kingston Upon Hull	
cont'd		Kingston upon Hull City Council	
		The Guildhall	
		Alfred Gelder Street	
		HULL	
		East Riding of Yorkshire	
		HU1 2AA	
		(in respect of the right of access granted by a Deed of Grant dated 21 November 1972)	
17-008	512 square metres of car park and hardstanding (south of Broadgate, B1230)	Bryant Homes Northern Limited	
		Gate House	
		Turnpike Road	
		HIGH WYCOMBE	
		Buckinghamshire	
		HP12 3NR	
		(in respect of reserved rights of way and others including restrictive covenants as contained in a Transfer dated 7 May 2010)	
17-009	1657 square metres of grassland and access track (south of Broadgate, B1230)	Bryant Homes Northern Limited	
		Gate House	
		Turnpike Road	
		HIGH WYCOMBE	
		Buckinghamshire	
		HP12 3NR	
		(in respect of the rights of access to sewage reserved by the Conveyance dated 28 September 1967)	
		The Lord Mayor Aldermen and Citizens of the City and County of Kingston Upon Hull	
		Kingston upon Hull City Council	
		The Guildhall	
		Alfred Gelder Street	
		HULL	
		East Riding of Yorkshire	
		HU1 2AA	
		(in respect of the right of access granted by a Deed of Grant dated 21 November 1972)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
17-009 cont'd		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of rights of access)	
	20505 square metres of grassland, copse and building (Broadgates Hospital, south of Broadgate, B1230) (excluding all interests of the Crown)	The Lord Mayor Aldermen and Citizens of the City and County of Kingston Upon Hull Kingston upon Hull City Council The Guildhall Alfred Gelder Street HULL East Riding of Yorkshire HU1 2AA (in respect of rights to construct and maintain a 14 inch water pipe listed in a Deed dated 09 June 1972) Unknown (in respect of rights relating to supply of water listed in a Conveyance dated 14 April 1983)	
17-011	10135 square metres of agricultural land (Vinegar Hill Farm, north of A1079, Beverley)	Beazer Homes Doncaster Limited Persimmon House Fulford YORK North Yorkshire YO19 4FE (in respect of the rights of access reserved by the Conveyance dated 24 January 1962) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of the Agreement relating to the placing, maintaining, repairing, renewing and inspecting of overhead power lines dated 2 July 1931 and 9 July 1931)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
17-011 cont'd		Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights relating to the placing, maintaining, repairing, renewing, inspecting, retaining and removal of poles and overhead and underground lines as contained in Deeds dated 2 August 1957 and 8 August 1961 and 15 March 1962)	
		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of restrictive covenants and the rights granted by a Deed dated 19 March 1968. Also in respect of the rights of access to maintain a water main granted by a Deed dated 19 April 1995)	
18-002	379 square metres of agricultural land (north of A1079, Beverley)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)	
	1483 square metres of public road and verges (A1079, Beverley) (excluding all interests of the Crown)	Unknown	
18-009	6202 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
BOOK OF REFERENCE - PART 2			
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
18-010	103456 square metres of agricultural land, access track, scrubland, hedgerows, drains, pylon and public footpath (Walkington Footpath No.4) (The Risby Estate, west of A1079, Beverley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)	
18-010a	48523 square metres of agricultural land (The Risby Estate, south of A1079, Beverley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
18-011	27093 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)	
18-012	10430 square metres of agricultural land, hedgerow, drain and public footpath (Walkington Footpath No.4) (The Risby Estate, west of A1079, Beverley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	



	Dogger Bank South Offshore Wind Farm Development Consent Order			
	BOOK OF REFERENCE - PART 2			
	County of East Riding of Yorkshire			
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)		
Land Plans		of the Planning Act 2008		
18-012 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)		
	27196 square metres of agricultural land, access tracks, verges, hedgerow and public footpath (Walkington Footpath No.9) (The Risby Estate, south of A1079, Beverley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)		
	131236 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)		
18-014a	8619 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)		



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
18-014a cont'd		Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)	
		Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-015	58902 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	Inees Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43-7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) National Gas Transmission PLC National Gar Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-015a	25825 square metres of agricultural land, scrubland and drain (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	



BOOK OF PERFERCE. PART 2 County of East Riding of Yorkshire           Number on Land Plans         Description of Land         Petertial claims under \$10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act of the Planning Act 2008           18-015         National Gas Transmission PLC National Graft Bouse Warwick Technology Park Gallows Hill WARWICK CV34 60A         National Graft Bouse Warwick Technology Park Gallows Hill WARWICK           18-016         // Part Square metres of agricultural land and drain (The Risby Estate, west of A164, Bentey)         Increspect of rights isted in a Deed dated 12 August 1953. Also in respect of rights to lay and entry listed in a Deed dated 17 August 1993)           18-016         // Part square metres of agricultural land and drain (The Risby Estate, west of A164, Bentey)         Inces Manufacturing (Hull) Limited Hawkisease Chapel Lane Chapel Chapel State Lane Chapel Lane Chapel Lane Chapel Lane Chapel Lane Chapel Lane Chapel Lane Chapel Lane Chapel Chapel State Lane Chapel Chapel State Lane Chapel Chapel State Lane Chapel Lane Chapel Chape	Dogger Bank South Offshore Wind Farm Development Consent Order		
Number on Land Plans         Petential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Ac of the Planning Act 2008           18-015a         National Gas Transmission PLC National Gas Transmission PLC Cont'd         National Gas Transmission PLC National Gas Park Gaslows Hill WRRWICK CV34 GGA (In respect of right of way listed in a Grant of Easement dated 3 April 2006.) Unknown (In respect of right I bade dated 1 December 1954)           18-016         7497 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley) Inex Manufacturing (Hull) Limited Hawkslease Chapel Lane LYND-UNKT Gas Successor of Arthur Thomas Filmer Wilson Filmer) (In respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 200. Unknown (In respect of right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Decessor of Charles Henry Woodmansey)			
Land Plans       of the Planning Act 2008         18-015a       National Grid House         cont'd       Warwick Technology Park         Gallows Hill       WARWICK         CV34 6DA       (in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.)         Unknown       (in respect of rights and easements listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay: and entry listed in a Deed dated 17 August 1953. Also in respect of rights to lay: and entry listed in a Deed dated 17 August 1993)         18-016       7497 square metres of agricultural land and drain (The Risby Estate, west of A164, Bertley)       Incess Manufacturing (Hull) Limited         Hampshire       SO43 7FG       (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 200.         Unknown       (as successor of Arthur Thomas Filmer Wilson Filmer)       (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 200.         Unknown       (as successor of Arthur Thomas Filmer Wilson Filmer)       (in respect of rights in a Deed dated 12 August 1953. Also in respect of rights to lay and and end will we as contained in a Deed and Lease dated 22 May 200.         Unknown       (as successor of Arthur Thomas Filmer Wilson Filmer)       (in respect of rights to lay end will be a Deed dated 12 August 1953. Also in respect of rights to lay a and end vill we are of will be a Deed dated 12 August 1953. Also in respect of rights to lay a and end vill be a Deed dated 12 August 1953. Also in resp			
cont'd       National Grid House         Gallows Hill       Gallows Hill         Gallows Hill       Gallows Hill         VC34 6DA       Cin respect of rights and easements listed in a Grant of Easement dated 3 April 2006.)         Unknown       Cin respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay.         18-016       AP7 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)       Incos Manufacturing (Hull) Limited         HawkiGeae       Chapel Lane       VNDPUNST         Hampshire       S04 37 FG       Cin respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2000.         Unknown       Cin respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2000.         Unknown       Cin respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2000.         Unknown       Cis successor of Arthur Thomas Filmer Wilson Filmer)         (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2000.         Unknown       Cis successor of Charles Henry Wilson Filmer)         (in respect of rights in a Deed dated 1 December 1954)       Unknown         Cis successor of Charles Henry Woodmansey)       Unknown	tion Act 1973 or 152 (3)		
18-015       7497 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)       Incessent of rights integards to a pipeline and value as contained in a Deed and Lease dated 22 May 200:         18-016       7497 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)       Incessent of rights integards to a pipeline and value as contained in a Deed and Lease dated 22 May 200:         18-016       (In respect of rights in the Risby Estate, west of A164, Bentley)       Incessent of rights listed in a Deed dated 17 August 1993)         18-016       (In respect of rights in the Risby Estate, west of A164, Bentley)       Incessent of rights listed in a Deed dated 17 August 1993)         18-016       (In respect of rights in the Risby Estate, west of A164, Bentley)       Incessent of rights listed in a Deed dated 17 August 1993)         18-016       (In respect of rights in regards to a pipeline and value as contained in a Deed and Lease dated 22 May 200:         In respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Deed dated 1 December 1954)         Unknown       (as successor of Arthur Thomas Filmer Wilson Filmer)         (In respect of a right of way listed in a Deced dated 12 August 1953. Also in respect of rights to lay a and entry listed in a December 1954)         Unknown       (as successor of Arthur Thomas Filmer Wilson Filmer)         (In respect of a right of way listed in a December 1954)       Unknown         (In respect of a right o			
Gallows Hill WARWICK CV34 6DA (in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.) Unknown (in respect of rights and easements listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay: and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993) 18-016 7497 Square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley) Inces Manufacturing (Hull) Limited Hawkslease Chapel Lane UNNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2001 Unknown (as successor of Arthur Thomas Filmer Wilson Filmer) (in conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Deed dated 1 December 1954) Unknown (as successor of Charles Henry Woodmansey)			
WARWICK       CV34 6DA       (in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.)         Unknown       (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay: and entry listed in a Deed dated 1 December 1954)         18-016       7497 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)       Incos Manufacturing (Hull) Limited         Hawkiease       Chapel Lane       LYNDHURST         Hampshire       SO33 7FG       (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2000.         Unknown       (as successor of Arthur Thomas Filmer Wilson Filmer)       (in caspect of rights to lay a and entry listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Deed dated 1 December 1954)			
CV34 6DA       (in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.)         Unknown       (in respect of a right of way listed in a Grant of Easement dated 12 August 1953. Also in respect of rights to lay: and entry listed in a Deed dated 1 December 1954)         18-016       7497 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)         HawkSlease       Ineos Manufacturing (Hull) Limited         HawShile       VNNHUKST         HawShile       SO43 7FG         (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 200:         Unknown       (as successor of Arthur Thomas Filmer Wilson Filmer)         (in respect of rights in a Grave of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Deed dated 1 December 1954)			
Image: Second			
18-016       7497 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)       Incomm (in respect of rights listed in a Deed dated 17 August 1993)         18-016       7497 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)       Incos Manufacturing (Hull) Limited Hawkslease Chapel Lane UYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002         Unknown       Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Deed dated 1 December 1954)			
(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay, and entry listed in a Deed dated 1 December 1954)         Unknown       (in respect of rights listed in a Deed dated 17 August 1993)         18-016       7497 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)       Ineos Manufacturing (Hull) Limited         Hawkslease       Chapel Lane       LVND/URST         UNknown       (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002         Unknown       (as successor of Arthur Thomas Filmer Wilson Filmer)         (in respect of a right of way listed in a Deed dated 1 December 1954)       Unknown         (as successor of Charles Henry Woodmansey)       Unknown			
and entry listed in a Deed dated 1 December 1954)         Unknown (in respect of rights listed in a Deed dated 17 August 1993)         18-016       7497 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)         Ineos Manufacturing (Hull) Limited         Hawkslease         Chapel Lane         LYNDHURST         Hampshire         SQ43 7FG         (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)         Unknown         (as successor of Arthur Thomas Filmer Wilson Filmer)         (in respect of right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Deed dated 1 December 1954)         Unknown         (as successor of Charles Henry Woodmansey)			
18-016       7497 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)       Ineos Manufacturing (Hull) Limited         Hawkslease       Chapel Lane         LYNDHURST       Hampshire         SO43 7FG       (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002         Unknown       (as successor of Arthur Thomas Filmer Wilson Filmer)         (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Deed dated 1 December 1954)         Unknown       (as successor of Charles Henry Woodmansey)	to lay a pipe to carry and supply water		
Image: Instant       Instant       Image: Im			
Image: Instant       Instant       Image: Im			
18-016       7497 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)       Ineos Manufacturing (Hull) Limited         18-016       7497 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)       Ineos Manufacturing (Hull) Limited         18-016       7497 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)       Ineos Manufacturing (Hull) Limited         18-016       14wkslease       Chapel Lane       LYNDHURST         Hampshire       SO43 7FG       (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002         Unknown       (as successor of Arthur Thomas Filmer Wilson Filmer)       (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Deed dated 1 December 1954)         Unknown       (as successor of Charles Henry Woodmansey)			
Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002 Unknown (as successor of Arthur Thomas Filmer Wilson Filmer) (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Deed dated 1 December 1954) Unknown (as successor of Charles Henry Woodmansey)			
Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2003 Unknown (as successor of Arthur Thomas Filmer Wilson Filmer) (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Deed dated 1 December 1954) Unknown (as successor of Charles Henry Woodmansey)			
LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002 Unknown (as successor of Arthur Thomas Filmer Wilson Filmer) (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Deed dated 1 December 1954) Unknown (as successor of Charles Henry Woodmansey)			
Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002 Unknown (as successor of Arthur Thomas Filmer Wilson Filmer) (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Deed dated 1 December 1954) Unknown (as successor of Charles Henry Woodmansey)			
SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002 Unknown (as successor of Arthur Thomas Filmer Wilson Filmer) (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Deed dated 1 December 1954) Unknown (as successor of Charles Henry Woodmansey)			
(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2003 Unknown (as successor of Arthur Thomas Filmer Wilson Filmer) (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Deed dated 1 December 1954) Unknown (as successor of Charles Henry Woodmansey)			
Unknown (as successor of Arthur Thomas Filmer Wilson Filmer) (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Deed dated 1 December 1954) Unknown (as successor of Charles Henry Woodmansey)	ay 2002)		
(as successor of Arthur Thomas Filmer Wilson Filmer) (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Deed dated 1 December 1954) Unknown (as successor of Charles Henry Woodmansey)			
(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a and entry listed in a Deed dated 1 December 1954) Unknown (as successor of Charles Henry Woodmansey)			
and entry listed in a Deed dated 1 December 1954) Unknown (as successor of Charles Henry Woodmansey)			
Unknown (as successor of Charles Henry Woodmansey)	o lay a pipe to carry and supply water		
(as successor of Charles Henry Woodmansey)			
(in respect of rights listed in a Deed dated 17 August 1993)			
18-017 13025 square metres of agricultural land (The Risby Estate, west of A164, Bentley) Ineos Manufacturing (Hull) Limited			
Hawkslease			
Chapel Lane			
LYNDHURST			
Hampshire			
SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2003	av 2002)		
(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 200.	ay 2002j		



Number on Land Plans 18-017 cont'd	Description of Land	BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)
Land Plans 18-017		
Land Plans 18-017	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)
		of the Planning Act 2008
cont'd		National Gas Transmission PLC
		National Grid House
		Warwick Technology Park
		Gallows Hill
		WARWICK
		CV34 6DA
		(in respect of rights & easements listed in a Grant of Easement dated 3 April 2006.)
		Unknown
		(as successor of Arthur Thomas Filmer Wilson Filmer)
		(in respect of a right of way listed in a conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water
		and entry listed in a Deed dated 1 December 1954)
		Unknown
		(as successor of Charles Henry Woodmansey)
		(in respect of rights listed in a Deed dated 17 August 1993)
18-018 4 squ	quare metres of agricultural land (The Risby Estate, west of A164, Bentley)	Unknown
		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)
		Unknown
		(in respect of provisions contained in a Conveyance dated 3 September 1976)
		Unknown
		(in respect of rights listed in a Deed dated 17 August 1993)
18-019 417 s	7 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)	Unknown
		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)
		Unknown
		(in respect of provisions contained in a Conveyance dated 3 September 1976)
		Unknown
		(in respect of rights listed in a Deed dated 17 August 1993)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
18-020	328 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of rights & easements listed in a Grant of Easement dated 3 April 2006.) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of provisions contained in a Conveyance dated 3 September 1976) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-021	112 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of provisions contained in a Conveyance dated 3 September 1976) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-021a	87 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of provisions contained in a Conveyance dated 3 September 1976) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
18-022	37869 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-022a	408 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley	<ul> <li>National Gas Transmission PLC</li> <li>National Grid House</li> <li>Warwick Technology Park</li> <li>Gallows Hill</li> <li>WARWICK</li> <li>CV34 6DA</li> <li>(in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.)</li> <li>Unknown</li> <li>(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)</li> <li>Unknown</li> <li>(in respect of rights listed in a Deed dated 17 August 1993)</li> </ul>	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans		of the Planning Act 2008	
18-023	44973 square metres of agricultural land and drain (The Risby Estate, west of A164,	Ineos Manufacturing (Hull) Limited	
	Bentley)	Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
		National Gas Transmission PLC	
		National Grid House	
		Warwick Technology Park	
		Gallows Hill	
		WARWICK	
		CV34 6DA	
		(in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.)	
		Unknown	
		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water	
		and entry listed in a Deed dated 1 December 1954)	
		Unknown	
		(in respect of rights listed in a Deed dated 17 August 1993)	
18-024	13813 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited	
		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
		National Gas Transmission PLC	
		National Grid House	
		Warwick Technology Park	
		Gallows Hill	
		WARWICK	
		CV34 6DA	
		(in respect of rights & easements listed in a Grant of Easement dated 3 April 2006.)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
18-024 cont'd		Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)	
		Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-025	26351 square metres of agricultural land and tree (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-025a	5052 square metres of agricultural land and tree (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans		of the Planning Act 2008	
	23332 square metres of agricultural land, scrubland and copse (The Risby Estate, west of	Ineos Manufacturing (Hull) Limited	
F	A164, Bentley and south of A1079, Beverley)	Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
		National Grid Electricity Transmission PLC	
		1-3 Strand	
		LONDON	
		WC2N 5EH	
		(in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a	
		Deed dated 28 July 1969)	
	3795 square metres of agricultural land, hedgerows, scrubland and drain (The Risby	Ineos Manufacturing (Hull) Limited	
E	Estate, west of A164, Bentley and south of A1079, Beverley)	Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
		National Grid Electricity Transmission PLC	
		1-3 Strand	
		LONDON	
		WC2N 5EH	
		(in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a	
		Deed dated 28 July 1969)	
18-028 3	3970 square metres of woodland (Bentley Moor Wood, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited	
		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans	Description of Land	of the Planning Act 2008	
18-029	270 square metres of drain (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited	
		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
		National Grid Electricity Transmission PLC	
		1-3 Strand	
		LONDON	
		WC2N 5EH	
		(in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a	
		Deed dated 28 July 1969)	
40.000			
18-030	18861 square metres of woodland (Bentley Moor Wood, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited	
		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire SO43 7FG	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
		Unknown	
		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water	
		and entry listed in a Deed dated 1 December 1954)	
		Unknown	
		(in respect of rights listed in a Deed dated 17 August 1993)	
18-031	133 square metres of woodland and drain (Bentley Moor Wood, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited	
		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	



	Dogger Bank Sou	th Offshore Wind Farm Development Consent Order
BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
18-031 cont'd		Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)
18-032	17850 square metres of agricultural land, hedgerows and drain (The Risby Estate, west of A164, Bentley)	
<del>18-033</del>	<del>12 square metres of agricultural land (The Risby Estate, west of A164, Bentley)</del>	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans		of the Planning Act 2008	
<del>18-034</del>	14 square metres of agricultural land and hedgerow (The Risby Estate, west of A164,	Incos Manufacturing (Hull) Limited	
	Bentley)	Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		<del>\$043 7FG</del>	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
		Unknown	
		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water	
		and entry listed in a Deed dated 1 December 1954)	
		Unknown	
		(in respect of rights listed in a Deed dated 17 August 1993)	
18-035	222 square metres of drain and scrubland (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited	
		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
		Unknown	
		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water	
		and entry listed in a Deed dated 1 December 1954)	
		Unknown	
		(in respect of rights listed in a Deed dated 17 August 1993)	
<del>18-036</del>	8 square metres of drain and scrubland (The Risby Estate, west of A164, Bentley)	Incos Manufacturing (Hull) Limited	
		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		<del>\$043 7FG</del>	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
		ļ	



	Dogger Banl	k South Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)
Land Plans		of the Planning Act 2008
<del>18-036</del>		<del>Unknown</del>
<del>cont'd</del>		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply wate and entry listed in a Deed dated 1 December 1954)
		Unknown
		(in respect of rights listed in a Deed dated 17 August 1993)
<del>18-037</del>	62 square metres of scrubland (The Risby Estate, west of A164, Bentley)	Incos Manufacturing (Hull) Limited
		Hawkslease Change Lange
		Chapel Lane LYNDHURST
		Hampshire
		<del>Nampsme</del> <del>SO43-7FG</del>
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)
		······································
		Unknown
		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply wate
		and entry listed in a Deed dated 1 December 1954)
		Unknown
		(in respect of rights listed in a Deed dated 17 August 1993)
18-040	2427 square metres of public road, verges and access splay (A164, Bentley)	Unknown
18-041	49 square metres of public road and verge (A164, Bentley)	Unknown
10.042	ACC assures making of any subural land (The Disky Fatala, ages of A1CA, Danilay)	National Crid Electricity Teacomission DLC
18-042	166 square metres of agricultural land (The Risby Estate, east of A164, Bentley)	National Grid Electricity Transmission PLC 1-3 Strand
		LONDON
		WC2N 5EH
		(in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a
		Deed dated 28 July 1969)
		Unknown
		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)
		and chiry holda in a beed daled i becember 1994)
		Unknown
		(in respect of rights listed in a Deed dated 17 August 1993)



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
18-043	65453 square metres of agricultural land, access tracks, verges, hedgerows, copse, drain and public bridleway (Rowley Bridleway No.13) (The Risby Estate, west of A1079, Beverley)	National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-044	9168 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
BOOK OF REFERENCE - PART 2			
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans		of the Planning Act 2008	
18-045	2756 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown	
		(in respect of rights listed in a Deed dated 17 August 1993)	
	27 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of provisions contained in a Conveyance dated 3 September 1976) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-047	10223 square metres of agricultural land, access tracks, verges, copse and public bridleway (Rowley Bridleway No.13) (The Risby Estate, south of A1079, Beverley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans		of the Planning Act 2008	
18-047		National Grid Electricity Transmission PLC	
cont'd		1-3 Strand	
		LONDON	
		WC2N 5EH	
		(in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a	
		Deed dated 28 July 1969)	
		Unknown	
		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water	
		and entry listed in a Deed dated 1 December 1954)	
		Unknown	
		(in respect of rights listed in a Deed dated 17 August 1993)	
18-048	289 square metres of access track, verge and public bridleway (Rowley Bridleway No.13)		
18-048	(Burn Park Farm, south of A1079, Beverley)	Albanwise Limited c/o Mills & Reeve LLP	
	(Buill Park Failli, South of A1079, Beveney)	Botanic House	
		100 Hills Road	
		CAMBRIDGE	
		Cambridgeshire	
		CB2 1PH	
		(in respect of a right of access to service contained in a Conveyance dated 22 May 1978. Also in respect of rights of access to maintain	
		service media as contained in a Transfer dated 13 January 2021)	
		Doggerbank Offshore Wind Farm Project 1 Projco Limited	
		No.1 Forbury Place 43 Forbury Road	
		Reading	
		RG1 3JH	
		(in respect of rights contained in a Deed of Grant dated 3 March 2020)	
		Doggerbank Offshore Wind Farm Project 2 Projco Limited	
		No.1 Forbury Place	
		43 Forbury Road	
		Reading	
		RG1 3JH	
		(in respect of restrictive covenants contained in a Deed dated 3 March 2020)	



	Dogger Bank So	outh Offshore Wind Farm Development Consent Order	
	BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans		of the Planning Act 2008	
18-048 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to maintaining, repairing, renewing, inspecting or removing electric lines contained in a Deed dated 7 March 1960 and 28 July 1969)	
18-049	2533 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-050	67 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) The Executor Of The Estate Of The Late Andrew White Manor Farm Newland GOOLE DN14 7XD (in respect of rights to use service media and other rights and easements listed in a Lease dated 8 November 2013) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)	



	Dogger Bank S	South Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
18-050 cont'd		Unknown (in respect of provisions contained in a Conveyance dated 3 September 1976) Unknown
18-051	21 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	(in respect of rights listed in a Deed dated 17 August 1993) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH
		<ul> <li>(in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)</li> <li>The Executor Of The Estate Of The Late Andrew White Manor Farm</li> <li>Newland</li> <li>GOOLE</li> <li>DN14 7XD</li> <li>(in respect of rights to use service media and other rights and easements listed in a Lease dated 8 November 2013)</li> <li>Unknown</li> <li>(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)</li> <li>Unknown</li> <li>(in respect of provisions contained in a Conveyance dated 3 September 1976)</li> <li>Unknown</li> <li>(in respect of rights listed in a Deed dated 17 August 1993)</li> </ul>
	26833 square metres of agricultural land, verge and drain (The Risby Estate, west of A10 Beverley)	<ul> <li>Joanne Dransfield</li> <li>Jillywood Farm</li> <li>Bentley</li> <li>BEVERLEY</li> <li>East Riding of Yorkshire</li> <li>HU17 8PP</li> <li>(in respect of rights to pass along a track, lay telephone cables, electricity poles and erect poles to carry said cables; to use a water pipe, drainage and entry listed in a Conveyance dated 7 January 1994)</li> </ul>



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
18-052 cont'd		National Grid Electricity Transmission PLC         1-3 Strand         LONDON         WC2N 5EH         (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a         Deed dated 28 July 1969)         Paul Dransfield         Jillywood Farm         Bentley         BEVERLEY         East Riding of Yorkshire         HU17 8PP         (in respect of rights to pass along a track, lay telephone cables, electricity poles and erect poles to carry said cables; to use a water pipe, drainage and entry listed in a Conveyance dated 7 January 1994)         Unknown         (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)         Unknown         (in respect of rights listed in a Deed dated 17 August 1993)	
18-053	303 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
18-054	22194 square metres of agricultural land, copse, drain and public bridleway (Rowley	Albanwise Limited	
	Bridleway No.13) (Burn Park Farm, south of A1079, Beverley)	c/o Mills & Reeve LLP	
		Botanic House	
		100 Hills Road	
		CAMBRIDGE	
		Cambridgeshire	
		CB2 1PH	
		(in respect of a right of access to service contained in a Conveyance dated 22 May 1978. Also in respect of rights of access to maintain	
		service media as contained in a Transfer dated 13 January 2021)	
		BP Chemicals Limited	
		Chertsey Road	
		SUNBURY ON THAMES	
		Middlesex	
		TW16 7BP	
		(in respect of rights to construct a pipeline contained in Agreements dated 22 May 1998 and 17 June 1998. Also in respect of rights to use	
		pipeline contained in a Deed dated 14 November 2000. Also in respect of rights to construct and maintain a pipeline contained in a Lease	
		of Easements dated 16 May 2002)	
		Doggerbank Offshore Wind Farm Project 1 Projco Limited	
		No.1 Forbury Place	
		43 Forbury Road	
		Reading	
		RG1 3JH	
		(in respect of rights contained in a Deed of Grant dated 3 March 2020)	
		Doggerbank Offshore Wind Farm Project 2 Projco Limited	
		No.1 Forbury Place	
		43 Forbury Road	
		Reading	
		RG1 3JH	
		(in respect of restrictive covenants contained in a Deed dated 3 March 2020)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Country of Fact Britism of Marking		
Number on Land Plans	Description of Land	County of East Riding of Yorkshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
18-054 cont'd		Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access contained in a Lease dated 16 May 2002) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right to maintaining, repairing, renewing, inspecting or removing electric lines contained in a Deed dated 7 March 1960 and 28 July 1969)	
	11580 square metres of agricultural land, verges, drain and public bridleway (Rowley Bridleway No.13) (Burn Park Farm, south of A1079, Beverley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (in respect of a right of access to service contained in a Conveyance dated 22 May 1978. Also in respect of rights of access to maintain service media as contained in a Transfer dated 13 January 2021) BP Chemicals Limited Chertsey Road SUNBURY ON THAMES Middlesex TW16 7BP (in respect of rights to construct a pipeline contained in Agreements dated 22 May 1998 and 17 June 1998. Also in respect of rights to use a pipeline contained in a Deed dated 11 November 2000. Also in respect of rights to construct and maintain a pipeline contained in a Lease of Easements dated 16 May 2002)	



		th Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire
Number		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
18-055		Doggerbank Offshore Wind Farm Project 1 Projco Limited
cont'd		No.1 Forbury Place
		43 Forbury Road
		Reading
		RG1 3JH
		(in respect of rights contained in a Deed of Grant dated 3 March 2020)
		Doggerbank Offshore Wind Farm Project 2 Projco Limited
		No.1 Forbury Place
		43 Forbury Road
		Reading
		RG1 3JH
		(in respect of restrictive covenants contained in a Deed dated 3 March 2020)
		Ineos Manufacturing (Hull) Limited
		Hawkslease
		Chapel Lane
		LYNDHURST
		Hampshire
		SO43 7FG
		(in respect of rights contained in a Lease dated 16 May 2002)
		National Grid Electricity Transmission PLC
		1-3 Strand
		LONDON
		WC2N 5EH
		(in respect of right to maintaining, repairing, renewing, inspecting or removing electric lines contained in a Deed dated 7 March 1960 and
		28 July 1969)
		Unknown
	the Crown)	
19-003	160 square metres of drain (south of A1079, Beverley)	Unknown



	Dogger Bank Sou	th Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
19-004	11423 square metres of agricultural land, verges and public bridleway (Woodmansey Bridleway No.30) (Model Farm, south of A1079, Beverley)	Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant and Transfer dated 15 April 2019. Also rights contained in an Option Agreement dated 4 February 2020 and rights contained in a Lease dated 1 March 2022)
19-005	8558 square metres of agricultural land, access track, hardstanding, scrubland, verges and public bridleway (Woodmansey Bridleway No.6) (Poplar Farm, south of A1079, Beverley)	Albert John Evans Wanlass Farm Park Lane COTTINGHAM East Riding of Yorkshire HU16 5SB (in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010) Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access relating to laying and maintaining a pipeline contained in a Deed dated 11 September 2011. Also in respect of rights and covenants listed in a Transfer dated 11 September 2020 Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of easements and covenants relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September 2011)



Description of Land	BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)
Description of Land	
Description of Land	Potential claims under S10 Compulsory Durchase Act 1965, Part 1 Land Componentian Act 1972 or 152 (2)
	of the Planning Act 2008
	Wind Energy One Limited
	30 Finsbury Square
	LONDON
	EC2A 1AG
	(in respect of rights of access granted by a Lease dated 16 January 2017)
12 square metres of agricultural land (Poplar Farm, south of A1079, Beverley)	Albert John Evans
	Wanlass Farm
	Park Lane
	COTTINGHAM
	East Riding of Yorkshire
	HU16 5SB
	(in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010)
	Ineos Manufacturing (Hull) Limited
	Hawkslease
	Chapel Lane
	LYNDHURST
	Hampshire
	SO43 7FG
	(in respect of rights of access relating to laying and maintaining a pipeline contained in a Deed dated 11 September 2011. Also in respect of
	rights and covenants listed in a Transfer dated 11 September 2002)
	Northern Gas Networks Limited
	1100 Century Way
	Thorpe Park
	LEEDS
	West Yorkshire
	LS15 8TU
	(in respect of easements and covenants relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September
	2011)
12	2 square metres of agricultural land (Poplar Farm, south of A1079, Beverley)



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
19-007	13773 square metres of agricultural land, drains and access splay (Burn Park Farm, south of	Albanwise Limited	
	A1079, Beverley)	c/o Mills & Reeve LLP	
		Botanic House	
		100 Hills Road	
		CAMBRIDGE	
		Cambridgeshire	
		СВ2 1РН	
		(in respect of a right of access to service contained in a Conveyance dated 22 May 1978. Also in respect of rights of access to maintain	
		service media as contained in a Transfer dated 13 January 2021)	
		BP Chemicals Limited	
		Chertsey Road	
		SUNBURY ON THAMES	
		Middlesex	
		TW16 7BP	
		(in respect of rights to construct a pipeline contained in Agreements dated 22 May 1998 and 17 June 1998. Also in respect of rights to use a	
		pipeline contained in a Deed dated 11 November 2000. Also in respect of rights to construct and maintain a pipeline contained in a Lease of Easements dated 16 May 2002)	
		Doggerbank Offshore Wind Farm Project 1 Projco Limited	
		No.1 Forbury Place	
		43 Forbury Road	
		Reading	
		RG1 3JH	
		(in respect of rights contained in a Deed of Grant dated 3 March 2020)	
		Doggerbank Offshore Wind Farm Project 2 Projco Limited	
		No.1 Forbury Place	
		43 Forbury Road	
		Reading	
		RG1 3JH	
		(in respect of restrictive covenants contained in a Deed dated 3 March 2020)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
19-007 cont'd		Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access contained in a Lease dated 16 May 2002) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right to maintaining, repairing, renewing, inspecting or removing electric lines contained in Deeds dated 7 March 1960 and 28 July 1969)	
20-001	26 square metres of agricultural land (Poplar Farm, south of A1079, Beverley)	Albert John Evans Wanlass Farm Park Lane COTTINGHAM East Riding of Yorkshire HU16 5SB (in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010) Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access relating to laying and maintaining a pipeline contained in a Deed dated 11 September 2011. Also in respect of rights and covenants listed in a Transfer dated 11 September 2002)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2	
		County of East Riding of Yorkshire	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans		of the Planning Act 2008	
20-001 cont'd		Northern Gas Networks Limited 1100 Century Way	
		Thorpe Park	
		LEEDS Wast Variation	
		West Yorkshire LS15 8TU	
		(in respect of easements and covenants relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September	
		2011)	
		Unknown	
		(in respect of rights listed in a Transfer dated 11 September 2002)	
	104 square metres of agricultural land, access track and public bridleway (Woodmansey	Albert John Evans	
	Bridleway No.6) (Poplar Farm, south of A1079, Beverley)	Wanlass Farm	
		Park Lane COTTINGHAM	
		East Riding of Yorkshire	
		HU16 5SB	
		(in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010)	
		Northern Gas Networks Limited	
		1100 Century Way	
		Thorpe Park	
		LEEDS	
		West Yorkshire	
		LS15 8TU	
		(in respect of easements and covenants relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September 2011)	
		Unknown	
		(in respect of rights listed in a Transfer dated 11 September 2002)	
		Wind Energy One Limited	
		30 Finsbury Square	
		EC2A 1AG (in respect of rights of access granted by a Lease dated 16 January 2017)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
20-003	3618 square metres of agricultural land, access tracks, verges, hedgerow, drain and public	Albert John Evans	
	bridleway (Woodmansey Bridleway No.6) (Poplar Farm, south of A1079, Beverley)	Wanlass Farm	
		Park Lane	
		COTTINGHAM	
		East Riding of Yorkshire	
		(in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010)	
		Ineos Manufacturing (Hull) Limited	
		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of rights of access relating to laying and maintaining a pipeline contained in a Deed dated 11 September 2011. Also in respect o	
		rights and covenants listed in a Transfer dated 11 September 2002)	
		Northern Gas Networks Limited	
		1100 Century Way	
		Thorpe Park	
		LEEDS	
		West Yorkshire	
		LS15 8TU	
		(in respect of easements and covenants relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September	
		2011)	
		Wind Energy One Limited	
		30 Finsbury Square	
		LONDON	
		EC2A 1AG	
		(in respect of rights of access granted by a Lease dated 16 January 2017)	



	Dogger Bank South Offshore Wind Farm Development Consent Order	
		BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
20-004	118007 square metres of agricultural land, access tracks, verges, hedgerows, electricity pylon, pond, drains, copse, scrubland, building and public bridleway (Woodmansey Bridleway No.6) (Poplar Farm, south of A1079, Beverley)	Albert John Evans Wanlass Farm Park Lane COTTINGHAM East Riding of Yorkshire HU16 558 (in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010) Incos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access relating to laying and maintaining a pipeline contained in a Deed dated 11 September 2011. Also in respect of rights and covenants listed in a Transfer dated 11 September 2002) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Vorkshire LS15 8TU (in respect of easements and covenants relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September 2011) Wind Energy One Limited 30 Finsbury Square LONDON EC2A 1AG (in respect of rights of access granted by a Lease dated 16 January 2017)



	Dogger Bank	South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2	
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
20-005	238 square metres of agricultural land, access track, verges and public bridleway (Woodmansey Bridleway No.6) (Poplar Farm, south of A1079, Beverley)	Albert John Evans Wanlass Farm Park Lane COTTINGHAM East Riding of Yorkshire HU16 5SB (in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010) Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access relating to laying and maintaining a pipeline contained in a Deed dated 11 September 2011. Also in respect of rights and covenants listed in a Transfer dated 11 September 2002) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of easements and covenants relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September 2011)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire		
Number on Land Plans	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
N/A	Andrew Cruickshank Constitution Hill Farm House Malton Road BEVERLEY HU17 7QY Laura Maxine Hayward Constitution Hill Farm House Malton Road BEVERLEY HU17 7QY		
N/A	Heather Mary Hayward Foremans House Cold Harbour Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8JF Paul Wallace Hayward Foremans House Cold Harbour Road Bishop Burton BEVERLEY East Riding of Yorkshire HU17 8JF		
N/A	Dawn Bache Lake Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP		



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of East Riding of Yorkshire		
Number on Land Plans Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)			
	of the Planning Act 2008		
N/A	Joanne Dransfield		
	Jillywood Farm		
	Bentley		
	BEVERLEY		
	East Riding of Yorkshire		
	HU17 8PP		
	Paul Dransfield		
	Jillywood Farm		
	Bentley		
	BEVERLEY		
	East Riding of Yorkshire		
	HU17 8PP		
N/A	Dianne Marion Burnett		
	2 High Eske Farm Cottages		
	Eske Lane		
	Tickton		
	BEVERLEY		
	East Riding of Yorkshire		
	HU17 9SG		
	Geraldine Burnett		
	1 Eske Lane		
	Tickton		
	BEVERLEY		
	East Riding of Yorkshire		
	HU17 9SG		
	(Trading as L B. Burnett & Sons)		
	Michael Burnett		
	2 High Eske Farm Cottages		
	Eske Lane		
	Tickton		
	BEVERLEY		
	East Riding of Yorkshire		
	HU17 9SG		



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire		
Number on Land Plans	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
N/A	Andrew Stewart Buchan 346 Hathersage Road HULL East Riding of Yorkshire HU8 0EY Nicola Jayne Smith 346 Hathersage Road HULL East Riding of Yorkshire HU8 0EY		
N/A	Naomi Orcades Maguire Cowden Holiday Park Main Road Cowden HULL East Riding of Yorkshire HU11 4UD Philip Lawrence Henry Maguire Cowden Holiday Park Main Road Cowden HULL East Riding of Yorkshire HU11 4UD		
N/A	Andrew Paul Usher Southdene Hornsea Road Skipsea DRIFFIELD YO25 8ST		



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of East Riding of Yorkshire		
Number of Lond Dise	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)		
of the Planning Act 2008			
N/A cont'd	Jeanette Usher		
	Southdene		
	Hornsea Road		
	Skipsea		
	DRIFFIELD		
	YO25 8ST		
N/A	John Francis Tunnicliffe		
N/A	Rose Cottage		
	Bentley		
	BEVERLEY		
	East Riding of Yorkshire		
	HU17 8PP		
	HOI7 8PP		
	Ruth Jennifer Tunnicliffe		
	Rose Cottage		
	Bentley		
	BEVERLEY		
	East Riding of Yorkshire		
	HU17 8PP		
N/A	Elisabeth Rebecca Lilley		
N/A	Mouse Hill		
	Bentley		
	BEVERLEY		
	HU17 8PP		
	Graham Charles Lilley		
	Mouse Hill		
	Bentley		
	BEVERLEY		
	HU17 8PP		
N/A	Susan Janette McLoughlin		
	99 Megson Way		
	Walkington		
	BEVERLEY		
	HU17 8YA		



Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of East Riding of Yorkshire		
Number on Land Plans	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Jennifer Myers 71 Megson Way Walkington BEVERLEY HU17 8YA Jonathan Waller Myers 71 Megson Way Walkington BEVERLEY	
	HU17 8YA	
N/A	Anita Burnley 89 Megson Way Walkington BEVERLEY HU17 8YA Neil Antony Burnley 89 Megson Way Walkington BEVERLEY HU17 8YA	
N/A	James Andrew Fairburn 83 Megson Way Walkington BEVERLEY HU17 8YA Laura Jane Gulley 83 Megson Way Walkington BEVERLEY HU17 8YA	



Dogger Bank South Offshore Wind Farm Development Consent Order			
	BOOK OF REFERENCE - PART 2		
	County of East Riding of Yorkshire		
Number on Land Plans	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)		
	of the Planning Act 2008		
N/A	David Anthony Robinson		
	87 Megson Way		
	Walkington		
	BEVERLEY		
	HU17 8YA		
	Philipa Jane Robinson		
	87 Megson Way		
	Walkington		
	BEVERLEY		
	HU17 8YA		
N/A	Clark Ernest Anthony Dean		
	91 Megson Way		
	Walkington		
	BEVERLEY		
	HU17 8YA		
	Rachel Marie Dean		
	91 Megson Way		
	Walkington		
	BEVERLEY		
	HU17 8YA		
N/A	David Christopher Harris		
	76 Megson Way		
	Walkington		
	BEVERLEY		
	HU17 8YA		
N/A	Georgina Laura Clarkson		
	85 Megson Way		
	Walkington		
	BEVERLEY		
	HU17 8YA		



Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
	County of East Riding of Yorkshire	
Number on Land Plans	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd	William Frederick Clarkson 85 Megson Way Walkington BEVERLEY HU17 8YA	
N/A	Richard Douglas Watson 93 Megson Way Walkington BEVERLEY HU17 8YA Suzanne Watson 93 Megson Way Walkington BEVERLEY HU17 8YA	
N/A	Brendan Jeremy Crosby 95 Megson Way Walkington BEVERLEY HU17 8YA	
N/A	Irene Turner 103 Megson Way Walkington BEVERLEY HU17 8YA	
N/A	Christopher Jewell 101 Megson Way Walkington BEVERLEY HU17 8YA	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of East Riding of Yorkshire		
Number of Level Disease	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)		
Number on Land Plans	of the Planning Act 2008		
N/A cont'd	Kim Jewell		
	101 Megson Way		
	Walkington		
	BEVERLEY		
	HU17 8YA		
N/A	John Billany		
,	74 Megson Way		
	Walkington		
	BEVERLEY		
	HU17 8YA		
	Samantha Billany		
	74 Megson Way		
	Walkington		
	BEVERLEY		
	HU17 8YA		
N/A	Audrey Dawson		
	Bentley Lodge		
	Victoria Road		
	BEVERLEY		
	East Riding of Yorkshire		
	НU17 8РЈ		
	Stephen Michael Knaggs		
	Bentley Lodge		
	Victoria Road		
	BEVERLEY		
	East Riding of Yorkshire		
	HU17 8PJ		
N/A	Adelle Elizabeth Brewitt		
	St. Peters House		
1	Bentley		
1	BEVERLEY		
1	East Riding of Yorkshire		
	HU17 8PP		



	Dogger Bank South Offshore Wind Farm Development Consent Order	
BOOK OF REFERENCE - PART 2		
County of East Riding of Yorkshire		
Number on Land Plans	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
	of the Planning Act 2008	
N/A cont'd	Sam Brewitt	
	St. Peters House	
	Bentley	
	BEVERLEY	
	East Riding of Yorkshire	
	HU17 8PP	
N/A	Andrew Harland	
·	Threeways	
	Hornsea Road	
	Skipsea	
	DRIFFIELD	
	East Riding of Yorkshire	
	YO25 8ST	
	Jacqueline Ann Harland	
	Threeways	
	Hornsea Road	
	Skipsea	
	DRIFFIELD	
	East Riding of Yorkshire	
	YO25 8ST	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-001	2797 square metres of public road and verges (North Turnpike)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Unknown	
01-004	2586 square metres of agricultural land and access track (east of North Turnpike)	Unknown (in respect of rights, covenants and restrictions as contained in a Transfer dated 3 February 1977)	
01-006	3107 square metres of agricultural land and access track (east of North Turnpike)	Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 29 April 2019) Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 29 April 2019) Unknown (in respect of rights of access to maintain and repair service media dated 3 February 1977)	
01-007	1903 square metres of agricultural land and access track (east of North Turnpike)	Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 29 April 2019)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-007		Doggerbank Offshore Wind Farm Project 2 Projec Limited	
cont'd		No.1 Forbury Place 43 Forbury Road Reading RG1 3JH	
		(in respect of rights contained in a Deed of Grant dated 29 April 2019)	
		Unknown (in respect of rights of access to maintain and repair service media dated 3 February 1977)	
	20 square metres of shore landward of the Mean High Water at Ulrome (east of North Turnpike)	Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 28 May 2019)	
	92 square metres of shore landward of the Mean High Water at Ulrome (east of North Turnpike)	Clifford Noel Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD (in respect of rights contained in a Lease dated 5 August 2005) Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 28 May 2019)	



01-000         Number on Land Plass         Number on Pescription of Land         Pescription of Land         Pescription of Land         Pescription of Land         Pescription Pescription Pescription of Land         Pescription Pescripticon Pescription Pescripticon Pescription Pescription Pescr		Dogger Bank South Offshore Wind Farm Development Consent Order		
Number on Land Plans         Names of all tose entitled to enjoy essements or other private rights or waited with its proposed ability to waite and suppended or interfered with under Regulation 71(1)(c) of the infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2000           01-003 control         Graham Clifford Warkup         Marciff           Beverley Road Beeford         DRFIFELD         East Riding of Yorkshire YO25 RAD           In respect of rights contained in a Lease dated 5 August 2005)         Marciff           Marciff         Beverley Road           Beeford         DRFIFELD           East Riding of Yorkshire YO25 RAD         In respect of rights contained in a Lease dated 5 August 2005)           Marciff         Beverley Road Beeford         DRFIFELD           DRFIELD         East Riding of Yorkshire YO25 RAD         In respect of rights contained in a Lease dated 5 August 2005)           Marciff         Beverley Road         Beeford           DRFIELD         East Riding of Yorkshire YO25 RAD         In respect of rights contained in a Lease dated 5 August 2005)           Nail Anthony Warkup         Marciff         Beverley Road         Beeford           DRFIELD         East Riding of Yorkshire         YO25 RAD         In respect of rights contained in a Lease dated 5 August 2005)           Nail Anthony Warkup         Marciff         Beverley Road         Beeford      <		BOOK OF REFERENCE - PART 3		
Number of Land Plans         Official difference which it is proposed shall be extinguished, suspended or interfered with under Regulation value which it is proposed shall be extinguished.           0.009         cont difference which it is proposed shall be extinguished.           cont difference which it is proposed shall be extinguished.         Suspended or interference which it is proposed shall be extinguished.           cont difference which it is proposed shall be extinguished.         Graham Clifford Warkup         Control interference which it is proposed shall be extinguished.           seveley Road         Beeverence Road         Beeverence Road         Beeverence Road           Beeverence Road         In respect of rights contained in a Lease dated 5 August 2005)         Arrin Geoffrey Warkup           Marrin Geoffrey Warkup         Marrin Geoffrey Warkup         Marrin Geoffrey Warkup           Marrin Geoffrey Warkup         Marrin Geoffrey Warkup         Marrin Geoffrey Warkup           Marrin Geoffrey Warkup         Marrin Geoffrey Warkup         Marrin Geoffrey Warkup           Marrin Geoffrey Warkup         Marrin Geoffrey Warkup         Marrin Geoffrey Warkup           Marrin Geoffrey Warkup         Marrin Geoffrey Warkup         Marrin Geoffrey Warkup           Marrin Geoffrey Warkup         Marrin Geoffrey Warkup         Marrin Geoffrey Warkup           Marrin Geoffrey Geodf         Beeverence Foord Fights of access)         Del Aufonov Warkup			County of East Riding of Yorkshire	
01.09 cont'd       Graham Clifford Warkup         Marcliff       Beverley Road         Bedraf       DIFFICD         East Riding of Yorkshire       YO25 8AD         (in respect of rights contained in a Lesse dated 5 August 2005)       Marcliff         Beverley Road       Beeford         DIFFICD       East Riding of Yorkshire         YO25 8AD       (in respect of rights contained in a Lesse dated 5 August 2005)         Marcliff       Beverley Road         Beeford       DIFFICD         DIFFICD       East Riding of Yorkshire         YO25 8AD       (in respect of rights contained in a Lesse dated 5 August 2005)         Neil Anthony Warkup       Marcliff         Beverley Road       Beeford         VO25 8AD       (in respect of rights contained in a Lesse dated 5 August 2005)         Neil Anthony Warkup       Marcliff         Beverley Road       Beeford         DIFFICD       East Riding of Yorkshire         YO25 8AD       (in respect of rights contained in a Lesse dated 5 August 2005)         Unknown       (in respect of rights contained in a Lesse dated 5 August 2005)         Unknown       (in respect of rights contained in a Lesse dated 5 August 2005)         Unknown       (in respect of rights contained in a Lesse dated 5 August 2005)      <		Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
cont'd     Marcliff       Beverley Road     Beerlord       DFIFFLD     East Riding of Yorkshire       VO2S RAD     (In respect of rights contained in a Lease dated 5 August 2005)       Marcliff     Beerlord       DRIFFLD     East Riding of Yorkshire       VO2S RAD     (In respect of rights contained in a Lease dated 5 August 2005)       Marcliff     Beerlord       DRIFFLD     East Riding of Yorkshire       VO2S RAD     (In respect of rights contained in a Lease dated 5 August 2005)       Neil Anthony Warkup     Marcliff       Beerlord     DRIFFLD       East Riding of Yorkshire     VO2S RAD       (In respect of rights contained in a Lease dated 5 August 2005)     Neil Anthony Warkup       Marcliff     Beerlord       DRIFFLD     East Riding of Yorkshire       VO2S RAD     (In respect of rights contained in a Lease dated 5 August 2005)       (In respect of rights contained in a Lease dated 5 August 2005)     Unespect of rights of access)       DI-010     Ba Square metres of foreshore seaward of the Mean High Water lying to the east of     Degerbank Offshore Vind Farm Project 1 Projco Limited       No.1 Forbury Place     As Forbury Road     Reading       Reading     Reading     Reading				
Beverley Road       Beverley Road         Beerlord       DRIFFIELD         East Niding of Yorkshire       Y225 8AD         (In respect of rights contained in a Lease dated 5 August 2005)       Martin Geoffrey Warkup         Marcliff       Beverley Road         Beerlord       DRIFFIELD         East Riding of Yorkshire       Y225 8AD         (In respect of rights contained in a Lease dated 5 August 2005)       Martin Geoffrey Warkup         Marcliff       Beverley Road         Beerlord       DRIFFIELD         East Riding of Yorkshire       Y025 8AD         (In respect of rights contained in a Lease dated 5 August 2005)       Neil Anthony Warkup         Marcliff       Beverley Road         Beeford       DRIFFIELD         East Riding of Yorkshire       Y025 8AD         (In respect of rights contained in a Lease dated 5 August 2005)       Neil Anthony Warkup         Marcliff       Beverley Road         Beeford       DRIFFIELD         East Riding of Yorkshire       Y025 8AD         (In respect of rights contained in a Lease dated 5 August 2005)       Un respect of rights of access)         Unknown       In respect of rights of access)       Unknown         Unknown       In respect of rights of access)       No.1 Forbury Place <td></td> <td></td> <td></td>				
01-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of       Deast Riding of Vorkshire         01-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of       Dogerhank Offshre Wrole Caressian         01-010       30 square metres of foreshore seaward of the Crown)       Dogerhank Offshre Wrole Place All State St	cont'd			
01:1FIE.D       East Riding of Yorkshire         Y025 & RAD       (In respect of rights contained in a Lease dated 5 August 2005)         Martin Geffrey Warkup       Marcine fifty Warkup         Marcine Geford       DBIFFIE.D         East Riding of Yorkshire       Y025 & RAD         V025 & RAD       DBIFFIE.D         East Riding of Yorkshire       Y025 & RAD         V025 & RAD       DBIFFIE.D         East Riding of Yorkshire       Y025 & RAD         (In respect of rights contained in a Lease dated 5 August 2005)       Neil Anthony Warkup         Marcilif       Beverley Road         Beverley Road       Beverley Ro				
10101       30 square metres of foreshore seaward of the Mean High Water lying to the east of        Visite for Company       Visite for Company         101020       30 square metres of foreshore seaward of the Mean High Water lying to the east of        Digmet for Company       Digmet for Company         101020       30 square metres of foreshore seaward of the Mean High Water lying to the east of        Digmet for Company       Digmet for Company         101020       30 square metres of foreshore seaward of the Mean High Water lying to the east of        Digmet for Company       Digmet for Company         101020       30 square metres of foreshore seaward of the Mean High Water lying to the east of        Digmet for Company       Digmet for Company         101020       30 square metres of foreshore seaward of the Mean High Water lying to the east of        Digmet for Shire        Visite for Company         101020       30 square metres of foreshore seaward of the Mean High Water lying to the east of        Digmet for Company       Digmet for Company         101020       30 square metres of foreshore seaward of the Crown       Dispect for Rights of access)       Dispect for Rights of access)         101020       1000000000000000000000000000000000000				
1       30 square metres of foreshore seaward of the Mean High Water lying to the east of       V025 8AD         1       10 square metres of foreshore seaward of the Mean High Water lying to the east of       Point Provide Water Provide Water lying to the east of         1       10 square metres of foreshore seaward of the Mean High Water lying to the east of       Point Provide Water lying to the east of         1       10 square metres of foreshore seaward of the Mean High Water lying to the east of       Point Provide Water Provide Water lying to the east of         1       10 square metres of foreshore seaward of the Mean High Water lying to the east of       Point Provide Water Provide Water lying to the east of         1       10 square metres of foreshore seaward of the Mean High Water lying to the east of       Point Provide Variant Provide Vari				
01-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of       Water of rights on tained in a Lease dated 5 August 2005)         01-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of       Water of rights on tained in a Lease dated 5 August 2005)         01-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of       No.1 Forbury Place         01-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of       No.1 Forbury Place         01-010       30 square metres of the Crown)       No.1 Forbury Place         01-010       30 square metres of the Crown)       Regional Group Place         01-010       30 square metres of the Crown)       Regional Group Place         01-010       30 square metres of the Crown)       Regional Group Place         01-010       30 square metres of the Crown)       Regional Group Place         01-010       30 square metres of the Crown)       Regional Group Place         01-010       Regional Group Place       Regional Group Place         01-010       Source metres of the Crown)       Regional Group Place         01-010       Regional Group Place       Regional Group Place         01-010       Regional Group Place       Regional Group Place				
01-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)       Water lying to the east of Not Strike				
D1-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of       Unknown         D1-010       30 square metres of foreshore seaward of the Crown)       Date metres of the Crown)         D1-010       30 square metres of the Crown)       Date metres of the Crown)				
D1-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of       Unknown         D1-010       30 square metres of foreshore seaward of the Crown)       Date metres of the Crown)         D1-010       30 square metres of the Crown)       Date metres of the Crown)			Martin Geoffrey Warkup	
01-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)       Beeford Dogger Data Project 1 Project 1 Project Limited Project 1 Proj				
01-010       30 square metres of foreshore seaward of the Mean High Water Iying to the east of Ulrome (excluding all interests of the Crown)       Degerbank Offshore Wind Farm Project 1 Projco Limited No. 1 Forbury Place 43 Forbury Road Reading RG3 JH			Beverley Road	
01-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of       Degerbank Offshore Wind Farm Project 1 Projco Limited         01-010       30 square metres of foreshore seaward of the Crown)       Degerbank Offshore Wind Farm Project 1 Projco Limited         01-010       30 square metres of foreshore seaward of the Crown)       Degerbank Offshore Wind Farm Project 1 Projco Limited         01-010       30 square metres of foreshore seaward of the Crown)       Reat of Ages Ages Ages Ages Ages Ages Ages Ages				
V025 BAD       (in respect of rights contained in a Lease dated 5 August 2005)         Neil Anthony Warkup       Neil Anthony Warkup         Marcliff       Beverley Road         Beeford       DRIFFIELD         East Riding of Yorkshire       Y025 8AD         (in respect of rights contained in a Lease dated 5 August 2005)       Neil Anthony Warkup         Marcliff       Beverley Road         Beeford       DRIFFIELD         Control (in respect of rights contained in a Lease dated 5 August 2005)       Neil Anthony Warkup         (in respect of rights contained in a Lease dated 5 August 2005)       Unknown         (in respect of rights contained in a Lease dated 5 August 2005)       Unknown         (in respect of rights contained in a Lease dated 5 August 2005)       Unknown         (in respect of rights of access)       Doggerbank Offshore Wind Farm Project 1 Projco Limited         No.1 Forbury Place       A 5 orbury Road         Reading       Reiding         Roft 31H       Neil Anthony Place				
Image: Control of the seaward of the Mean High Water lying to the east of Urome (excluding all interests of the Crown)       Image: Control of the Mean High Water lying to the east of Ref 13 JH				
Neil Anthony Warkup         Neil Anthony Warkup         Marcliff         Beverley Road         Beeford         DIFFIELD         East Riding of Yorkshire         Y025 8AD         (in respect of rights contained in a Lease dated 5 August 2005)         Unknown         (in respect of rights of access)         01-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of         Ulrome (excluding all interests of the Crown)       Doggerbank Offshore Wind Farm Project 1 Projco Limited         No. 1 Forbury Place       43 Forbury Road         Reading       RG1 3)H				
Marcliff       Beverley Road         Beeford       DRIFFIELD         Cast Riding of Yorkshire       VO25 8AD         (in respect of rights contained in a Lease dated 5 August 2005)       Unknown         (in respect of rights of access)       Unknown         V1-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of       Doggerbank Offshore Wind Farm Project 1 Projco Limited         No.1 Forbury Place       43 Forbury Road       Reading         Reading       Reading       Reading         RG1 3JH       State Stat			(in respect of rights contained in a Lease dated 5 August 2005)	
Beverley Road       Beeford         DRIFFIELD       East Riding of Yorkshire         YO25 8AD       (in respect of rights contained in a Lease dated 5 August 2005)         Unknown       (in respect of rights of access)         01-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of       Doggerbank Offshore Wind Farm Project 1 Projco Limited         No.1 Forbury Place       A3 Forbury Road       Reading         Red 31 JH       RG1 3JH				
Beeford       DRIFFIELD         East Riding of Yorkshire       YO25 8AD         YO25 8AD       (in respect of rights contained in a Lease dated 5 August 2005)         Unknown       (in respect of rights of access)         01-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of       Doggerbank Offshore Wind Farm Project 1 Projco Limited         No.1 Forbury Place       43 Forbury Road       Reading         Reading       RG1 3JH				
DRIFFIELD       East Riding of Yorkshire         YQ25 8AD       (in respect of rights contained in a Lease dated 5 August 2005)         Unknown       (in respect of rights of access)         01-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)       Doggerbank Offshore Wind Farm Project 1 Projco Limited         No.1 Forbury Place       A3 Forbury Road         Reading       Reading         RG1 3JH       Reading				
Base Riding of Yorkshire       YO25 8AD       (in respect of rights contained in a Lease dated 5 August 2005)         Unknown       Unknown       (in respect of rights of access)         01-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of       Doggerbank Offshore Wind Farm Project 1 Projco Limited         No.1 Forbury Place       43 Forbury Road       Reading         Reading       Reading         Ro1 3JH       Keating				
Y025 8AD (in respect of rights contained in a Lease dated 5 August 2005)         Unknown (in respect of rights of access)         01-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)       Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH				
Image: Control of the seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)       Doggerbank Offshore Wind Farm Project 1 Projco Limited         No.1 Forbury Place       A3 Forbury Road         Reading       Reading         Rof JH       Rof JH				
01-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)       Doggerbank Offshore Wind Farm Project 1 Projco Limited         No.1 Forbury Place       43 Forbury Road         Reading       Reading         RG1 3JH       RG1 3JH				
01-010       30 square metres of foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)       Doggerbank Offshore Wind Farm Project 1 Projco Limited         0.1 Forbury Place       A3 Forbury Road         Reading       Reading         RG1 3JH       RG1 3JH				
01-010     30 square metres of foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)     Doggerbank Offshore Wind Farm Project 1 Projco Limited       No.1 Forbury Place       43 Forbury Road       Reading       RG1 3JH			Unknown	
Ulrome (excluding all interests of the Crown) No.1 Forbury Place 43 Forbury Road Reading RG1 3JH			(in respect of rights of access)	
Ulrome (excluding all interests of the Crown) No.1 Forbury Place 43 Forbury Road Reading RG1 3JH	01.010	20 causes motions of foreshore segurard of the Mean High Water luiss to the cost of	Degraphank Offshara Wind Form Project 1 Project Limited	
43 Forbury Road Reading RG1 3JH	01-010			
Reading RG1 3JH				
RG1 3JH				
			· ····································	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3		
Number on Land Plans	Description of Land	County of East Riding of Yorkshire           Names of all those entitled to enjoy easements or other private rights over land           (including private rights of navigation over water) which it is proposed shall be extinguished,           suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning           (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-010 cont'd		Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 29 April 2019) Unknown (in respect of rights of access)	
01-011	356 square metres of foreshore seaward of the Mean High Water lying to the north east of Ulrome (excluding all interests of the Crown)	Clifford Noel Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO2S 8AD (in respect of rights contained in a Lease dated 5 August 2005) Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 28 May 2019) Graham Clifford Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO2S 8AD (in respect of rights contained in a Lease dated 5 August 2005)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-011		Martin Geoffrey Warkup	
cont'd		Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD (in respect of rights contained in a Lease dated 5 August 2005) Neil Anthony Warkup Marcliff Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AD (in respect of rights contained in a Lease dated 5 August 2005)	
02-005	3948 square metres of shore landward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	Unknown	
02-013	599 square metres of shore landward of the Mean High Water lying to the east of Skipsea (north of Hornsea Road)	Unknown	
02-014	8 square metres of shore landward of the Mean High Water lying to the east of Skipsea (north of Hornsea Road)	Unknown	
02-015	2144 square metres of shore landward of the Mean High Water lying to the east of Skipsea (north of Hornsea Road)	Unknown (in respect of right of way contained in a Conveyance dated 4 June 1963)	
02-016	185977 square metres of agricultural land and drains (east of Cleeton Lane and north of Hornsea Road, B1242)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-016 cont'd		Unknown (in respect of rights of way contained in a Conveyance dated 4 June 1963)	
02-017	6728 square metres of agricultural land and drains (east of Cleeton Lane and north of Hornsea Road, B1242)	Unknown (in respect of rights of way contained in a Conveyance dated 4 June 1963)	
02-019	319 square metres of agricultural land and drains (east of Cleeton Lane)	Unknown	
02-020	85758 square metres of agricultural land and drains (north of Hornsea Road, B1242)	Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus)	
02-024	25135 square metres of agricultural land (north of Hornsea Road, B1242)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	
02-025	3873 square metres of agricultural land (north of Hornsea Road, B1242)	Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus)	
02-026	3 square metres of public road and verges (Hornsea Road, B1242)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	



	Dogger Bank South Offshore Wind Farm Development Consent Order			
	BOOK OF REFERENCE - PART 3			
	County of East Riding of Yorkshire			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
02-027	1607 square metres of public road and verges (Hornsea Road, B1242)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)		
02-028	20 square metres of public road and verges (Hornsea Road, B1242)	Unknown Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)		
02-029	1 square metres of verges (south of Hornsea Road, B1242)	Unknown		
02-030	632 square metres of verges (south of Hornsea Road, B1242)	Unknown		



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-031	10884 square metres of agricultural land and verges (east of Hornsea Road, B1242)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)	
02-032	21808 square metres of agricultural land and verges (east of Hornsea Road, B1242)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	
02-033	2484 square metres of public road, verges and hedgerow (Hornsea Road, B1242)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus) Unknown Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-034	9128 square metres of agricultural land and verges (west of Hornsea Road, B1242)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus)	
02-036	881 square metres of public road, access track and verges (Hornsea Road, B1242)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground sewerage and water apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-037	570 square metres of agricultural land and access track (west of Hornsea Road, B1242)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of rights reserved by a Transfer dated 15 February 2012)	
02-038	72661 square metres of agricultural land and public footpath (Skipsea Footpath No.6) (west of Hornsea Road, B1242)	East Riding Of Yorkshire Council County Hall Cross Street BEVERLEY East Riding of Yorkshire HU17 9BA (in respect of rights reserved by a Transfer dated 15 February 2012)	
03-001	276 square metres of watercourse (Skipsea Drain) (east of Bewholme Lane)	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights) Unknown	
03-002	26356 square metres of agricultural land (east of Bewholme Lane)	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of rights of access)	
03-003	90 square metres of hedgerow (east of Bewholme Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	



	Dogg	er Bank South Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-004	417 square metres of public road and verges (Bewholme Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown
03-005	52 square metres of agricultural land (west of Bewholme Lane)	Unknown (in respect of restrictive covenants as may have been imposed before 11 May 1791) Unknown (in respect of rights of way, water support, drainage, light and other easements granted by a Conveyance dated 20 April 1988)
03-006	730 square metres of public road and verges (Bewholme Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown
03-007	140 square metres of agricultural land (east of Bewholme Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)
03-008	590 square metres of public road and verges (Bewholme Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-009	127 square metres of agricultural land (west of Bewholme Lane)	Unknown (in respect of restrictive covenants as may have been imposed before 11 May 1971) Unknown (in respect of rights of way, water support, drainage, light and other easements granted by a Conveyance dated 20 April 1988)	
03-010	9561 square metres of agricultural land (west of Bewholme Lane)	Unknown (in respect of restrictive covenants as may have been imposed before 11 May 1971) Unknown (in respect of rights of way, water support, drainage, light and other easements granted by a Conveyance dated 20 April 1988)	
	33116 square metres of agricultural land and drain (Low Bonwick Farm, south of Dunnington Lane)	Carr Hill Farm Limited Office F12 Beverley Enterprise Centre Beck View Road BEVERLEY East Riding of Yorkshire HU17 OJT (in respect of rights of access to maintain, inspect and repair service media contained in a Transfer dated 30 November 2017)	
03-013	26372 square metres of agricultural land (Manor Farm, south of Dunnington Lane)	Manor Farm Energy Limited Medina House 2 Station Avenue BRIDLINGTON East Riding of Yorkshire YO16 4LZ (in respect of rights of access and maintaining, inspecting and repairing conducting media contained in a Lease dated 5 September 2014)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-001	76 square metres of access splay and hedgerow (Manor Farm, east of Skipsea Lane)	Manor Farm Energy Limited Medina House 2 Station Avenue BRIDLINGTON East Riding of Yorkshire YO16 4LZ (in respect of rights of access and maintaining, inspecting and repairing conducting media contained in a Lease dated 5 September 2014)	
04-002	523 square metres of public road and verges (Skipsea Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
04-003	120 square metres of agricultural land (Manor Farm, west of Skipsea Lane)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in Deeds dated 11 May 1972, 18 February 1981 and 26 March 1991)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-004	865 square metres of public road and verges (Skipsea Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
04-005	78 square metres of hedgerow (east of Skipsea Lane)	Manor Farm Energy Limited Medina House 2 Station Avenue BRIDLINGTON East Riding of Yorkshire YO16 4LZ (in respect of rights of access and maintaining, inspecting and repairing media contained in a Lease dated 5 September 2014)	
04-006	561 square metres of public road and verges (Skipsea Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	



	Dogger Bank South Offshore Wind Farm Development Consent Order	
		BOOK OF REFERENCE - PART 3
Number on Land Plans	Description of Land	County of East Riding of Yorkshire Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-006 cont'd		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
04-007	58 square metres of agricultural land (Manor Farm, west of Skipsea Lane)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in Deeds dated 11 May 1972, 18 February 1981 and 26 March 1991)
04-008	42776 square metres of agricultural land (Manor Farm, west of Skipsea Lane)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in Deeds dated 11 May 1972, 18 February 1981 and 26 March 1991) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)



	Dogger Bank	South Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-009	490 square metres of watercourse (Dunnington Sewer, east of Dunnington Lane)	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights) Unknown
04-010	42868 square metres of agricultural land and drain (east of Dunnington Lane)	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of rights of access) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in a Deed of Grant dated 25 June 1971) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of underground gas apparatus)



	Dogger Bank South Offshore Wind Farm Development Consent Order	
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-011	2103 square metres of agricultural land (east of Dunnington Lane)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, construct, use, maintain, inspect, alter, renew, replace and remove a main or pipe contained in a Deed of Grant dated 25 June 1971)
04-012	79 square metres of public road and verge (Dunnington Lane)	Unknown
04-013	858 square metres of public road and verge (Dunnington Lane)	Unknown
04-014	125 square metres of verge (west of Dunnington Lane)	Unknown
04-015	2432 square metres of public road and verge (Dunnington Lane)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Unknown
04-016	305 square metres of verge (west of Dunnington Lane)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Unknown



	Dogger Bank South Offshore Wind Farm Development Consent Order	
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
	3499 square metres of agricultural land (west of Dunnington Lane)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)
	18022 square metres of public road and verges (Dunnington Lane and Beverley Road, A165)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Unknown Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
04-021	21402 square metres of agricultural land (west of Dunnington Lane)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)



	Dogger Bank Sou	th Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-024	27263 square metres of agricultural land and drain (Moor Grange Farm, south of Dunnington Lane)	Gregory Peter Atkin Moor Grange Beverley Road Beeford Driffield East Riding of Yorkshire YO25 8AE (in respect of rights of drainage and access to the retained land contained in a Transfer dated 14 September 2010) Lucy Sheila Elizabeth Atkin Moor Grange Cottage Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AE (in respect of rights of drainage and access to the retained land contained in a Transfer dated 14 September 2010)
05-001	45117 square metres of agricultural land and hedgerow (Moor Grange Farm, east of Moor Grange Access Road, west of Billings Lane)	Alexander Douglas Robinson Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG (in respect of rights to maintain a ditch contained in a Conveyance dated 1 August 1984) Gregory Peter Atkin Moor Grange Beverley Road Beeford Driffield East Riding of Yorkshire YO25 8AE (in respect of rights of way contained in a Transfer dated 10 May 1988 and in respect of rights of drainage and access to the retained land contained in a Transfer dated 14 September 2010)



	Dogger Bank Sou	th Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-001		Lucy Sheila Elizabeth Atkin
cont'd		Moor Grange Cottage Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AE (in respect of rights of way contained in a Transfer dated 10 May 1988 and in respect of rights of drainage and access to the retained land contained in a Transfer dated 14 September 2010) R & CM Rivis Limited Beachcomber Lodge 56 South Marine Drive BRIDLINGTON YO15 3JN (in respect of rights of way and paying a proportion of costs to repair, maintain, clean, dredging and renewing drains contained in a Transfer dated 14 September 2010)
	6271 square metres of agricultural land and hedgerow (Moor Grange Farm, east of Moor Grange Access Road, west of Billings Lane)	Alexander Douglas Robinson Happy Land Farm Dringhoe DRIFFIELD East Riding of Yorkshire YO25 8AG (in respect of rights to maintain a ditch contained in a Conveyance dated 1 August 1984) Gregory Peter Atkin Moor Grange Beverley Road Beeford Driffield East Riding of Yorkshire YO25 8AE (in respect of rights of way contained in a Transfer dated 10 May 1988 and in respect of rights of drainage and access to the retained land contained in a Transfer dated 14 September 2010)



	Dogger Bank South Offshore Wind Farm Development Consent Order	
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-002 cont'd		Lucy Sheila Elizabeth Atkin Moor Grange Cottage Beverley Road Beeford DRIFFIELD East Riding of Yorkshire YO25 8AE (in respect of rights of way contained in a Transfer dated 10 May 1988 and in respect of rights of drainage and access to the retained land contained in a Transfer dated 14 September 2010) R & CM Rivis Limited Beachcomber Lodge 56 South Marine Drive BRIDLINGTON YO15 3JN (in respect of rights of way and paying a proportion of costs to repair, maintain, clean, dredging and renewing drains contained in a Transfer dated 14 September 2010)
05-003	6575 square metres of agricultural land and drain (west of Billings Lane)	Unknown
05-004	69585 square metres of agricultural land and hedgerow (east of New Road) (excluding all interests of the Crown)	Julia Alexa Clover Todd Nunkeeling Priory Nunkeeling DRIFFIELD East Riding of Yorkshire YO25 8EH (in respect of rights to maintain the dwellinghouse, drains and septic tank stated in the Conveyance dated 14 January 1966) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)



Dogger Bank South Offshore Wind Farm D		ith Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-004		Northern Powergrid (Yorkshire) PLC
cont'd		Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of wayleave consents dated 28 September 1967, 13 May 1969 and 28 November 1983)
		Nunkeeling Renewables Limited Hall Garth Manor Farm Nunkeeling Driffield East Yorkshire YO25 8EH (in respect of rights contained in a Lease dated 1 May 2012 and 3 December 2012)
		Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of restrictive covenants and rights listed in a Deed dated 30 May 1949) Unknown
		(in respect of rights granted and terms listed in a Conveyance dated 18 March 1987)
	29550 square metres of agricultural land, access track and hedgerow (Moor House Farm, Moor House Access Road)	Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)



		th Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3 County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
	10450 square metres of agricultural land and watercourse (Nunkeeling Drain, west of Main Road)	
06-003	519 square metres of public road and verge (Main Road)	Unknown
		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of abandoned underground water apparatus)
	891 square metres of public road and verge (Main Road)	Unknown Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
06-005	30 square metres of verge (east of Main Road)	Unknown



	Dogger	Bank South Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
06-005 cont'd		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)
06-006	136 square metres of agricultural land and verge (east of Main Road)	Foster Farms Limited Willow Croft Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 SPR (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1 February 1980 and 17 July 1973) Laurazena Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of easements as contained in a Transfer dated 4 May 2023) Roy Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of easements as contained in a Transfer dated 4 May 2023)



06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       February 1980 and 17 July 1973)         06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       For the same timed wink time and the same timed wink time and the same		Dogger Bank So	outh Offshore Wind Farm Development Consent Order
Number on Land Plans         Names of all those entitled to enjoy essentents or other private rights over land (including private rights or availation over wately while the extragulation, suspended or interfered with under Regulation 71()(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009           06-007         SSO74 square metres of agricultural land (north of Catfoss Road, east of Main Road)         Foster Farms Limited (Applications: Prescribed Forms and Procedures) Regulations 2009           06-007         SSO74 square metres of agricultural land (north of Catfoss Road, east of Main Road)         Foster Farm Riston Road Catvick BVEVRLY         Foster Farm Riston Road Catvick BVEVRLY           06-008         Catvick BVEVRLY         East Riding of Yorkhire HU17 5PR (In respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1 February 1980 and 17 July 1973)           Laurazena Thompson Haverhan Farm Hempholme DEFIFICIO Bast Riding of Yorkhire YO25 SNB (In respect of easements as contained in a Transfer dated 4 May 2023)           06-008         2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)         Foster Farm Riston Road Catvick BVEVRLY           06-008         2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)         Foster Farm Riston Road Catvick BVEVRLY           06-008         2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)         Foster Farm Riston Road Catvick BVEVRLY           06-008         2047 square metres of agric			BOOK OF REFERENCE - PART 3
Number on Land Plans         Description of Land         (including private rights of navigation over water) which it is proposed shall be extinguished, suspendend or interfered with onder Regulations TU(Lo) of the Infrastructure Planning           06-007         \$5074 square metres of agricultural land (north of Catfoss Read, east of Main Read)         Foster Farms Limited         (Applications: Prescribed Forms and Procedures) Regulations 2009           06-007         \$5074 square metres of agricultural land (north of Catfoss Read, east of Main Read)         Foster Farms Limited         Foster Farms Limited           0F-007         \$5074 square metres of agricultural land (north of Catfoss Read, east of Main Read)         Foster Farms Limited         Foster Farms Limited           0F-007         \$6074 square metres of agricultural land (north of Catfoss Read, east of Main Read)         Foster Farms Limited         Foster Farms Limited           0F-007         \$6074 square metres of agricultural land (north of Catfoss Read, east of Main Read)         Foster Farms Limited         Foster Farms Limited           0F-008         \$2074 square metres of agricultural land (north of Catfoss Read, east of Main Read)         Foster Farms Limited         Foster Farms Limited           0F-008         \$2074 square metres of agricultural land (north of Catfoss Read, east of Main Read)         Foster Farm Stimited         Foster Farm Stimited           0F-008         \$2074 square metres of agricultural land (north of Catfoss Read, east of Main Read)         Foster Farm Stimited<			County of East Riding of Yorkshire
Land Plans     Description of LMO     suspended or interfered with under Regulation 7(1)(c) of the infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009       06-007     \$5074 square metres of agricultural land (north of Catfoss Road, east of Main Road)     Foster Farms Limited Willow Core Farm Biston Road     Willow Core Farm Biston Road       0F-007     \$5074 square metres of agricultural land (north of Catfoss Road, east of Main Road)     Vieto Cart Farm Biston Road     Vieto Cart Farm Biston Road       0F-008     \$074 square metres of agricultural land (north of Catfoss Road, east of Main Road)     Vieto Cart Farm Biston Road     Vieto Lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1       0F-008     \$0847 square metres of agricultural land (north of Catfoss Road, east of Main Road)     Rev Thompson Haverham Farm Hempholine       0F-008     \$2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)     Rev Thompson Haverham Farm Hempholine       0F-008     \$2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)     Rev Thompson Haverham Farm Hempholine       0F-008     \$2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)     Rev Thompson Haverham Farm Hempholine       0F-008     \$2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)     Rev Thompson Haverham Farm Yillow Croft Farm Roton Road Catwick       0F-008     \$2047 square metres of agricultural land (north of Catfoss Road, east of Main Road) <th></th> <th></th> <th>Names of all those entitled to enjoy easements or other private rights over land</th>			Names of all those entitled to enjoy easements or other private rights over land
Land Plans         suggende or interfered with under Regulation /11(c) of the Intrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009           06-007         55074 square metres of agricultural land (north of Catfoss Road, east of Main Road)         Fort Farms Limited Willow Croft Farm Riston Road Critick           BVVR.LCY         East Riding of Vorshire HU17 59R         Fort Farm Right of Vorshire HU17 59R           Language and the state of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1 February 1980 and 12 July 1973)           Laurazera Thompson Historiking of Vorshire V225 8N8         Roy Thompson Historiking of Vorshire V225 8N8           06-008         2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)         Fost Farms Limited Willow Croft Farm Hemptoine DIHFIELD           06-008         2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)         Fost Farms Limited Willow Croft Farm Niston Road Critick           06-008         2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)         Fost Farms Limited Willow Croft Farm Niston Road Critick           0F-008         2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)         Fost Farm Niston Road Critick           0F-008         2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)         Fost Farm Limited Willow Croft Farm           Nore report of agricultural land (north of C	Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,
06-007       55074 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited         Wilkow Croft Farm       Biston Road       Cattrock         BV:VRLEY       East Riding of Yorkshire         HU12 SPR       (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1         Laurazena Thompson       Havefam Farm         Heget of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1         Quirzena Thompson       Havefam Farm         Heget of the right to fay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1         Quirzena Thompson       Havefam Farm         Heget of reasements as contained in a Transfer dated 4 May 2023)         Roy Thompson       Havefam Farm         Heget of reasements as contained in a Transfer dated 4 May 2023)         Quirties Constrained       Foster Farms Limited         VIDE Const Farm       Riston Road         Catrock       Exerct of easements as contained in a Transfer dated 4 May 2023)         Color Farm       Riston Road         Catrock       Exerct of easements as contained in a Transfer dated 4 May 2023)         Color Farm       Riston Road         Catrock       East Miding of Yorkshire	Land Plans	Description of Land	suspended or interfered with under Regulation $7(1)(c)$ of the Infrastructure Planning
Willow Cord Farm         Ristion Road         Catwick         BVFREY         East Riding of Yorkshire         HU12 SPR         Ruin respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1         February 1980 and 17 July 1973)         Laurazena Thompson         Heurpholme         DBF/FIELD         East Riding of Yorkshire         Y025 SNB         (in respect of easements as contained in a Transfer dated 4 May 2023)         Rey Thompson         Haverhain Farm         Heupholme         DIFFIELD         East Riding of Yorkshire         Y025 SNB         (in respect of easements as contained in a Transfer dated 4 May 2023)         Rey Thompson         Haverhain Farm         Heupholme         DIFFIELD         East Riding of Yorkshire         Y025 SNB         (in respect of easements as contained in a Transfer dated 4 May 2023)         Rey Thompson         Haverhain Farm         Heupholme         DIFFIELD         East Riding of Yorkshire         Y025 SNB         (in respect of easements as contained in a Transfer dated 4 May 2023)			(Applications: Prescribed Forms and Procedures) Regulations 2009
Biton Road         Crivick         BVERLEY         East Roing of Vorkshire         Hu17 SPR         (in respect of regiments)         Lauracena Thompson         Hiverham Farm         Hempholme         OB-008         2047 square metres of agricultural land (north of Catfoss Read, east of Main Road)         Foster Farms Limited         Willow Cord Farm         Rition Road         Catwick         Biton Road         Biton Road         Catwick	06-007	55074 square metres of agricultural land (north of Catfoss Road, east of Main Road)	Foster Farms Limited
OE-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited         06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited         016-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited			Willow Croft Farm
06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited         06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited			
East Riding of Yorkshire         HU17 SPR         (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1         February 1980 and 17 July 1973)         Laurazena Thompson         Haverham Farm         Hempholme         DRFIFILD         East Riding of Yorkshire         YO25 SN8         (in respect of easements as contained in a Transfer dated 4 May 2023)         Roy Thompson         Haverham Farm         Hempholme         DRIFFIELD         East Riding of Yorkshire         YO25 SN8         (in respect of easements as contained in a Transfer dated 4 May 2023)         D6-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)         Foster Farms Limited       Willow Cord Farm         Nilston Road       Catwick         BVEVELEY       East Riding of Yorkshire         HU12 YSR       (in respect of the right to lay and maintain water and gas pipes, drains, severs and electricity cables stated in Conveyances dated 1			Catwick
06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Koster Fams Limited         06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Koster Fams Limited         06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Koster Fams Limited         06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Koster Fams Limited         010       Kiston Road       Catwick BEVENLEY       East Riding of Yorkshire         011       Kiston Road       Catwick BEVENLEY       East Riding of Yorkshire			BEVERLEY
06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited         06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited         06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited         VI: Star Riding of Yorkshire       YO25 SN8         (in respect of easements as contained in a Transfer dated 4 May 2023)			East Riding of Yorkshire
06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       February 1980 and 17 July 1973)         06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Following Yorkshire         016-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Following Yorkshire         016-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Following Yorkshire         016-017       East Riding of Yorkshire       YOLS SNB         016-018       East Riding of Yorkshire       YOLS SNB         016-019       Following Yorkshire       YOLS SNB         017       Following Yorkshire       YOLS Yorkshire         018       Following Yorkshire       YOLS Yorkshire         019       Following Yorkshire       YOLS Yorkshire         019       Yoll Yorkshire       Yoll Yorkshir			HU17 5PR
06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Kain Road         06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Villow Croft Farm Riston Road         06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Villow Croft Farm Riston Road			(in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1
Non-one       Participation         Non-one       Part			February 1980 and 17 July 1973)
Non-one       Participation         Non-one       Part			
06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited         V05-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited         Willow Croft Farm       Riston Road       Catwick         BEVERLEY       East Riding of Yorkshire         In respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1			
DRIFFIELD         East Riding of Yorkshire         YO25 8NB         (in respect of easements as contained in a Transfer dated 4 May 2023)         Roy Thompson         Haverham Farm         Hempholme         DRIFFIELD         East Riding of Yorkshire         YO25 8NB         (in respect of easements as contained in a Transfer dated 4 May 2023)         O6-008         2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)         Foster Farms Limited         Willow Croft Farm         Riston Road         Catvick         BEVERLEY         East Riding of Yorkshire         HU17 Spran         Riston Road         Catvick         BEVERLEY         East Riding of Yorkshire         HU17 Spran         Riston Road         Catvick         BEVERLEY         East Riding of Yorkshire         HU17 Spran         Riton Road         Catvick         BEVERLEY         East Riding of Yorkshire         HU17 Spran         Riton Road         (In respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances d			
East Riding of Yorkshire       Y025 8NB       (in respect of easements as contained in a Transfer dated 4 May 2023)         Roy Thompson       Haverham Farm       Hempholine         DRIFFIELD       East Riding of Yorkshire         East Riding of Yorkshire       Y025 8NB         (in respect of easements as contained in a Transfer dated 4 May 2023)         06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited         Willow Croft Farm       Riston Road       Catwick         BEVERLEY       East Riding of Yorkshire         East Riding of Yorkshire       HU17 SPR         (in respect of the right to lay and maintain water and gas pipes, drains, severs and electricity cables stated in Conveyances dated 1			
YO25 8NB       (in respect of easements as contained in a Transfer dated 4 May 2023)         Roy Thompson       Haverham Farm         Hemphoime       DRIFFIELD         East Riding of Yorkshire       YO25 8NB         (in respect of easements as contained in a Transfer dated 4 May 2023)         06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited         Willow Croft Farm       Riston Road       Catwick         BEVERLEY       East Riding of Yorkshire         HU17 SPR       (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1			
Image: Constraint of the constraint			-
06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited         Willow Croft Farm       Riston Road         Catwick       BEVERLEY         East Riding of Yorkshire       Hul17 SPR         (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1			
Haverham Farm         Hempholme         DRIFFIELD         East Riding of Yorkshire         YO25 8NB         (in respect of easements as contained in a Transfer dated 4 May 2023)         06-008         2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)         Foster Farms Limited         Willow Croft Farm         Riston Road         Catwick         BEVERLEY         East Riding of Yorkshire         HU17 5PR         (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1			(in respect of easements as contained in a Transfer dated 4 May 2023)
Haverham Farm         Hempholme         DRIFFIELD         East Riding of Yorkshire         YO25 8NB         (in respect of easements as contained in a Transfer dated 4 May 2023)         06-008         2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)         Foster Farms Limited         Willow Croft Farm         Riston Road         Catwick         BEVERLEY         East Riding of Yorkshire         HU17 5PR         (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1			Day Thompson
Hempholme         DRIFFIELD       East Riding of Yorkshire         YO25 8NB       (in respect of easements as contained in a Transfer dated 4 May 2023)         06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited         Willow Croft Farm       Riston Road       Catwick         BEVERLEY       East Riding of Yorkshire         HUIT 5PR       (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1			
DRIFFIELD         East Riding of Yorkshire         YO25 8NB         (in respect of easements as contained in a Transfer dated 4 May 2023)         06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)         Foster Farms Limited         Willow Croft Farm         Riston Road         Catwick         BEVERLEY         East Riding of Yorkshire         HU17 5PR         (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1			
East Riding of Yorkshire       YO25 8NB       YO25 8NB         (in respect of easements as contained in a Transfer dated 4 May 2023)       Solar Farms Limited         06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited         Willow Croft Farm       Riston Road       Catwick         BEVERLEY       East Riding of Yorkshire         HU17 5PR       (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1			
VO25 8NB       (in respect of easements as contained in a Transfer dated 4 May 2023)         06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited         Willow Croft Farm       Riston Road       Catwick         BEVERLEY       East Riding of Yorkshire         HU17 5PR       (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1			
06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited         Willow Croft Farm       Riston Road         Catwick       BEVERLEY         East Riding of Yorkshire       HU17 SPR         (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1			•
06-008       2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)       Foster Farms Limited         Willow Croft Farm       Riston Road         Catwick       BEVERLEY         East Riding of Yorkshire       HU17 SPR         (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1			
Willow Croft Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1			(in respect of easements as contained in a fransfer dated 4 May 2025)
Willow Croft Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1	06-008	2047 square metres of agricultural land (north of Catfoss Road, east of Main Road)	Foster Farms Limited
Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1			Willow Croft Farm
BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1			Riston Road
East Riding of Yorkshire HU17 5PR (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1			Catwick
HU17 5PR (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1			BEVERLEY
HU17 5PR (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1	1		East Riding of Yorkshire
(in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1	1		
			(in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1
			February 1980 and 17 July 1973)



	Dogger Bank South Offshore Wind Farm Development Consent Order	
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
06.000		(Applications: Prescribed Forms and Procedures) Regulations 2009
06-008 cont'd		Laurazena Thompson Haverham Farm
cont u		Hempholme
		DRIFFIELD
		East Riding of Yorkshire
		YO25 8NB
		(in respect of easements as contained in a Transfer dated 4 May 2023)
		Roy Thompson
		Haverham Farm
		Hempholme
		DRIFFIELD
		East Riding of Yorkshire YO25 8NB
		(in respect of easements as contained in a Transfer dated 4 May 2023)
06-009	16 square metres of hedgerow and drain (north of Catfoss Road)	Centrica PLC
		Millstream
		Maidenhead Road
		WINDSOR
		Berkshire SL4 5GD
		(in respect of the rights granted by the Deeds dated 27 June 1972 and 10 April 1975)
		Nunkeeling Renewables Limited
		Hall Garth Manor Farm
		Nunkeeling
		Driffield
		East Yorkshire
		YO25 8EH
		(in respect of the rights reserved by the Transfer dated 9 October 2013)
06-010	1004 square metres of public road and verge (Catfoss Road)	Openreach Limited
06-010	1004 square metres of public road and verge (Catfoss Road)	6 Gracechurch Street
06-010	1004 square metres of public road and verge (Catfoss Road)	6 Gracechurch Street LONDON
06-010	1004 square metres of public road and verge (Catfoss Road)	6 Gracechurch Street LONDON EC3V 0AT
06-010	1004 square metres of public road and verge (Catfoss Road)	6 Gracechurch Street LONDON



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-010		Unknown	
cont'd			
		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
06-011	4 square metres of verge (south of Catfoss Road)	Unknown	
06-012	1012 square metres of public road and verge (Catfoss Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	



	Dogger Bank S	South Offshore Wind Farm Development Consent Order	
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-013	43 square metres of verge (north of Catfoss Road)	Foster Farms Limited	
		Willow Croft Farm Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 SPR (in respect of the right to lay and maintain water and gas pipes, drains, sewers and electricity cables stated in Conveyances dated 1 February 1980 and 17 July 1973) Laurazena Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of easements as contained in a Transfer dated 4 May 2023) Roy Thompson Haverham Farm Hempholme DRIFFIELD East Riding of Yorkshire YO25 8NB (in respect of easements as contained in a Transfer dated 4 May 2023)	
06-014	9777 square metres of public road and verge (Catfoss Road and Catfoss Lane)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-014 cont'd		Unknown	
		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
06-016	8429 square metres of agricultural land and access track (south of Catfoss Road)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the right to lay, maintain, inspect and repair and in respect to easement stated in the Deed dated 27 June 1972)	
06-017	31268 square metres of agricultural land and drain (south of Catfoss Road)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the right to lay, maintain, inspect and repair and in respect to easement stated in the Deed dated 27 June 1972)	
	30325 square metres of agricultural land and drain (north of Harsell Lane, excluding all interests of the Crown)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights contained in Deeds dated 19 April 1972 and 6 February 1973)	



	Dogger Bank Sou	uth Offshore Wind Farm Development Consent Order	
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
06-018		(Applications: Prescribed Forms and Procedures) Regulations 2009 Northern Powergrid (Yorkshire) PLC	
cont'd		Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)	
		Peter James Boynton	
		Hind House Catfoss Lane Sigglesthorne	
		HULL	
		East Riding of Yorkshire	
		HU11 5QN	
		(as The Executor of the Estate of the Late Patricia Anne Boynton)	
		(in respect of rights reserved by a Conveyance dated 24 April 1981)	
		Secretary Of State For Defence	
		Main Building	
		Horse Guards Avenue	
		Whitehall LONDON	
		LONDON SW1A 2HB	
		(in respect of covenants and grants contained in a Deed dated 11 November 1948)	
		Unknown	
		(in respect to rights reserved by a Conveyance dated 1 October 1965)	
06-019	136 square metres of verge (north of Harsell Lane, excluding all interests of the Crown)	Centrica PLC Millstream	
		Maidenhead Road	
		WINDSOR	
		Berkshire	
		SL4 5GD	
		(in respect of rights contained in Deeds dated 19 April 1972 and 6 February 1973)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-019		Colin Stuart Hazell	
cont'd		Astral House Catfoss HULL East Riding of Yorkshire HU11 5SP (in respect of rights of access as contained in a Transfer dated 16 December 2010) Peter James Boynton Hind House Catfoss Lane	
		Sigglesthorne HULL East Riding of Yorkshire HU11 5QN (as The Executor of the Estate of the Late Patricia Anne Boynton) (in respect of rights reserved by a Conveyance dated 24 April 1981)	
		Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of covenants and grants contained in a Deed dated 11 November 1948)	
		Unknown (in respect to rights reserved by a Conveyance dated 1 October 1965)	
	374 square metres of public road and verge (Harsell Lane, excluding all interests of the Crown)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights contained in Deeds dated 19 April 1972 and 6 February 1973)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
06-020		Colin Stuart Hazell	
cont'd		Astral House	
		Catfoss HULL	
		East Riding of Yorkshire	
		HU11 5SP	
		(in respect of rights listed in a Transfer dated 16 December 2010)	
		(	
		Peter James Boynton	
		Hind House	
		Catfoss Lane	
		Sigglesthorne	
		HULL	
		East Riding of Yorkshire	
		HU11 5QN	
		(as The Executor of the Estate of the Late Patricia Anne Boynton)	
		(in respect of rights reserved by a Conveyance dated 24 April 1981)	
		Secretary Of State For Defence	
		Main Building	
		Horse Guards Avenue	
		Whitehall	
		LONDON	
		SW1A 2HB	
		(in respect of covenants and grants contained in a Deed dated 11 November 1948)	
		Unknown	
		(in respect to rights reserved by a Conveyance dated 1 October 1965)	
06-021	472 square metres of public road (Harsell Lane, exlucing all interests of the Crown)	Centrica PLC	
		Millstream	
		Maidenhead Road	
		WINDSOR	
		Berkshire	
		SL4 5GD	
		(in respect of rights contained in Deeds dated 19 April 1972 and 6 February 1973)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
06-021		(Applications: Prescribed Forms and Procedures) Regulations 2009 Colin Stuart Hazell	
cont'd		Astral House Catfoss HULL East Riding of Yorkshire	
		HU11 5SP (in respect of rights listed in a Transfer dated 16 December 2010)	
		Peter James Boynton Hind House Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HU11 5QN	
		(as The Executor of the Estate of the Late Patricia Anne Boynton) (in respect of rights reserved by a Conveyance dated 24 April 1981) Secretary Of State For Defence	
		Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of covenants and grants contained in a Deed dated 11 November 1948)	
		Unknown (in respect to rights reserved by a Conveyance dated 1 October 1965)	
06-022	116 square metres of verge (north of Harsell Lane, excluding all interests of the Crown)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights contained in Deeds dated 19 April 1972 and 6 February 1973)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-022		Colin Stuart Hazell	
cont'd		Astral House Catfoss HULL East Riding of Yorkshire HU11 5SP (in respect of rights of access as contained in a Transfer dated 16 December 2010)	
		Peter James Boynton Hind House Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HU11 5QN (as The Executor of the Estate of the Late Patricia Anne Boynton) (in respect of rights reserved by a Conveyance dated 24 April 1981) Secretary Of State For Defence	
		Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of covenants and grants contained in a Deed dated 11 November 1948) Unknown (in respect to rights reserved by a Conveyance dated 1 October 1965)	
06-023	177 square metres of public road (Harsell Lane, exlucing all interests of the Crown)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, maintain and renew a pipe contained in a Deed dated 19 April 1972 and in respect of rights to lay, maintain and renew a pipe contained in a Deed dated 6 February 1973)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-023 cont'd		Colin Stuart Hazell Astral House Catfoss HULL East Riding of Yorkshire HU11 5SP (in respect of rights to maintain, repair and renew a fence contained in a Transfer dated 16 December 2010) Peter James Boynton Hind House Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HULL East Riding of Yorkshire HULL Saft Riding of Yorkshire HULL (a Sthe Executor of the Estate of the Late Patricia Anne Boynton) (in respect of rights of way contained in a Conveyance dated 24 April 1981) Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights to maintain hedges and trees contained in a Deed dated 11 November 1948) Unknown (in respect to rights of way contained in a Conveyance dated 1 October 1965)	
06-024	9 square metres of agricultural land (south of Harsell Lane)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, maintain and renew a pipe contained in a Deed dated 19 April 1972 and in respect of rights to lay, maintain and renew a pipe contained in a Deed dated 6 February 1973)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
06-024		(Applications: Prescribed Forms and Procedures) Regulations 2009 Colin Stuart Hazell	
cont'd		Astral House Catfoss HULL East Riding of Yorkshire HU11 5SP (in respect of rights to maintain, repair and renew a fence contained in a Transfer dated 16 December 2010) Peter James Boynton Hind House Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HULL East Riding of Yorkshire HULL East Riding of Yorkshire HUL1 SQN (as The Executor of the Estate of the Late Patricia Anne Boynton) (in respect of rights of way contained in a Conveyance dated 24 April 1981) Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights to maintain hedges and trees contained in a Deed dated 11 November 1948) Unknown (in respect to rights of way contained in a Conveyance dated 1 October 1965)	
06-025	64907 square metres of agricultural land, drain and hedgerow (south of Harsell Lane, east of Catwick Road, excluding all interests of the Crown)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights to lay, maintain and renew a pipe contained in a Deed dated 19 April 1972 and in respect of rights to lay, maintain and renew a pipe contained in a Deed dated 6 February 1973)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-025 cont'd		Colin Stuart Hazell Astral House Catfoss HULL East Riding of Yorkshire HU11 5SP (in respect of rights to maintain, repair and renew a fence contained in a Transfer dated 16 December 2010) Peter James Boynton Hind House Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HULL East Riding of Yorkshire HULL East Riding of Yorkshire HU11 SQN (as The Executor of the Estate of the Late Patricia Anne Boynton) (in respect of rights of way contained in a Conveyance dated 24 April 1981) Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights to maintain hedges and trees contained in a Deed dated 11 November 1948)	
07-001	20924 square metres of agricultural land (east of Catwick Road)	(in respect to rights of way contained in a Conveyance dated 1 October 1965) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect to the rights to lay and maintain pipes and in respect to a right to easements as contained in a Deed dated 19 April 1972)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-002	6926 square metres of agricultural land (east of Catwick Road)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect to the right to easement and in respect to the right to lay, maintain, inspect and repair a gas pipeline contained in a Deed dated 19 April 1972)	
07-003	4081 square metres of agricultural land, copse and drain (east of Catwick Road)	David Hobson Manor Lodge Catfoss Lane Sigglesthorne HULL HU11 5QN (in respect of rights of wayleave easements, rights of way and drainage contained in a transfer dated 1 August 1990)	
07-004	21226 square metres of agricultural land, drain and public footpath (Seaton Footpath No.10) (east of Catwick Road)	Andrew Stewart Buchan 346 Hathersage Road HULL East Riding of Yorkshire HU8 0EY (in respect of rights of light or air contained in a Conveyance dated 6 November 1990) Nicola Jayne Smith 346 Hathersage Road HULL East Riding of Yorkshire HU8 0EY (in respect of rights of light and air contained in a Conveyance dated 6 November 1990)	
07-005	62273 square metres of agricultural land, access track, copse and watercourse (Catfoss Drain) (Catfoss Hall Farm Access Road, east of Catwick Road)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order			
	BOOK OF REFERENCE - PART 3			
	County of East Riding of Yorkshire			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
07-005		Openreach Limited		
cont'd		6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)		
		Timothy Duncan Rose Catfoss Hall Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HU11 5QN (in respect of rights of way contained in a Conveyances dated 26 January 1999 and 29 October 1970)		
08-001	1135 square metres of agricultural land and access splay (east of Catwick Road)	Timothy Duncan Rose Catfoss Hall Catfoss Lane Sigglesthorne HULL East Riding of Yorkshire HU11 5QN (in respect of rights of way contained in a Conveyances dated 26 January 1999 and 29 October 1970)		
08-002	6119 square metres of public road, access splays, drains and verges (Catwick Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus) Unknown		



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-002 cont'd		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
08-003	230 square metres of bridge, verge and copse (Washdike Bridge, east of Catwick Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
08-004	515 square metres of public road and verges (West Road, A1035)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground sewerage and water apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-005	1912 square metres of public road and verges (West Road, A1035)	Northern Powergrid (Yorkshire) PLC	
00-003		Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown Yorkshire Water Services Limited Western House	
00.000		Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground sewerage and water apparatus)	
08-006	966 square metres of public road and verges (West Road, A1035)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-006		Unknown	
cont'd			
		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground sewerage and water apparatus)	
08-007	17149 square metres of agricultural land (south of West Road, A1035, east of Catwick	Northern Gas Networks Limited	
	Heads)	1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus) Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford	
08-008	11017 square metres of agricultural land and drain (south of West Road, A1035, east of	West Yorkshire BD6 2SZ (in respect of underground water apparatus) Northern Gas Networks Limited	
	Catwick Heads)	1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus)	



	Dogger Bank Sou	th Offshore Wind Farm Development Consent Order	
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-008 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus)	
	14359 square metres of agricultural land, access splay, pond and drain (south of West Road, A1035 and east of Catwick Heads)	Unknown (in respect of rights of easements and covenants contained in a Conveyance dated 19 January 1987)	
08-010	218 square metres of verge (south of West Road, A1035, east of Catwick Heads)	Unknown (in respect of rights of easements and covenants contained in a Conveyance dated 19 January 1987)	
08-011	583 square metres of public road, access splay and verge (Catwick Heads)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
08-012	4675 square metres of agricultural land, access splay and verge (west of Catwick Heads)	Unknown (in respect of rights of light, air, footpaths and carriageways contained in a Conveyance dated 6 April 1965)	
08-013	1637 square metres of agricultural land (east of Catwick Heads)	Unknown	
08-014	1919 square metres of public road and verge (Catwick Heads)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
08-015	19001 square metres of agricultural land, access track, drain and public footpath (Catwick Footpath No.8) (west of Catwick Heads)	Unknown (in respect of rights of light, air, footpaths and carriageways contained in a Conveyance dated 6 April 1965)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-016	20212 square metres of agricultural land and pond (west of Catwick Heads)	Unknown (in respect of right of way as contained in a Conveyance dated 10 May 1989)	
08-017	30792 square metres of agrilcutural land (Cobble Hall Farm, north of Rise Lane)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)	
08-026	46823 square metres of agricultural land, access track and drain (Field House Farm, south of Rise Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown (in respect of rights contained in a Transfer dated 2 September 2009)	
09-001	15487 square metres of agricultural land (east of Riston Road)	Amy Clark-Blakey Glenholme Leven Road Catwick BEVERLEY HU17 SPT (in respect of rights contained in a Transfer dated 2 March 2008) Gemma Louise Elizabeth Clark-Blakey Glenholme Leven Road Catwick BEVERLEY HU17 SPT (in respect of rights contained in a Transfer dated 2 March 2008)	



	Dogger	Bank South Offshore Wind Farm Development Consent Order	
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
09-001		(Applications: Prescribed Forms and Procedures) Regulations 2009 John Christopher Bird	
cont'd		The Old Hall Lakeview Riston Road Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of rights contained in a Transfer dated 30 July 2009 ) Mark Desmond Bird Old Hall Farm Riston Road	
		Catwick BEVERLEY East Riding of Yorkshire HU17 5PR (in respect of rights contained in a Transfer dated 30 July 2009 )	
		Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of rights contained in a Deed of Grant of Easement dated 5 September 1967. Also in respect of rights and restrictive covenants contained in a Deed of Grant of Easement dated 3 October 1966) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF	
		(in respect of rights contained in a Lease dated 14 October 2021)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
09-001		Paul Adrian Bird	
cont'd		Old Hall Farm	
		Riston Road	
		Catwick BEVERLEY	
		East Riding of Yorkshire HU17 5PR	
		(in respect of rights contained in a Transfer dated 30 July 2009 )	
		(in respect of rights contained in a transfer dated so July 2009)	
09-002	5341 square metres of agricultural land (east of Riston Road)	Unknown	
09-005	89 square metres of public road and verges (Riston Road)	Unknown	
09-006	1059 square metres of public road and verges (Riston Road)	Unknown	
09-007	117 square metres of agricultural land (west of Riston Road)	Holderness Hunt (Holdings) Limited	
		Citadel House	
		58 High Street	
		HULL	
		East Riding of Yorkshire	
		HU1 1QE	
		(in respect of rights of access)	
		Unknown	
		(in respect of right of access contained in a Conveyance dated 6 April 1918)	
09-008	15705 square metres of agricultural land and drain (west of Riston Road)	Holderness Hunt (Holdings) Limited	
		Citadel House	
		58 High Street	
		HULL	
		East Riding of Yorkshire	
		HU1 1QE	
		(in respect of rights of access)	
		Unknown	
		(in respect of right of access contained in a Conveyance dated 6 April 1918)	
II			



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation $7(1)(c)$ of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
	19717 square metres of agricultural land and drain (Willow Croft Farm, west of Riston	John Christopher Bird	
	Road)	The Old Hall Lakeview	
		Riston Road	
		Catwick	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 SPR	
		(in respect of rights granted by a Conveyance dated 27 November 1925)	
		Mark Desmond Bird	
		Old Hall Farm	
		Riston Road	
		Catwick	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 5PR	
		(in respect of rights granted by a Conveyance dated 27 November 1925)	
		Paul Adrian Bird	
		Old Hall Farm	
		Riston Road	
		Catwick	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 5PR	
		(in respect of rights granted by a Conveyance dated 27 November 1925)	
09-010	70690 square metres of agricultural land, hedgerow and drain (Prospect Farm, east of	Unknown	
	Whitecross Road, A165 and west of Catwick Lane)	(in respect of rights listed in a conveyance dated 19 January 1987)	
		Unknown	
	Road, A165 and west of Catwick Lane)	(in respect of rights as contained in a Conveyance dated 19 January 1987)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-012	6000 square metres of agricultural land and drain (Riston Grange Farm, east of Whitecross Road, A165)		
		Unknown (in respect of right of entry and access to services as contained in a Conveyance dated 4 October 1977)	
09-013	39182 square metres of agricultural land and drain (Riston Grange Farm, east of Whitecross Road, A165)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of a gas main and rights listed in a Deed dated 3 October 1966) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Unknown (in respect of right of entry and access to services as contained in a Conveyance dated 4 October 1977)	
10-001	2704 square metres of agricultural land (Riston Grange Farm, east of Whitecross Road, A165)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of a gas main and rights listed in a Deed dated 3 October 1966) Unknown (in respect of right of entry and access to services as contained in a Conveyance dated 4 October 1977)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-002	2119 square metres of public road and verges (Whitecross Road, A165)	KCOM Group Limited 37 Carr Lane	
		Hull East Yorkshire HU1 3RE	
		(in respect of underground telecommunication apparatus) Openreach Limited	
		6 Gracechurch Street LONDON	
		EC3V 0AT (in respect of underground telecommunication apparatus)	
		Unknown	
		Yorkshire Water Services Limited Western House	
		Western Way Halifax Road	
		Bradford West Yorkshire	
		BD6 2SZ (in respect of underground water apparatus)	
10-003	3692 square metres of public road and verges (Whitecross Road, A165)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire	
		HU1 3RE (in respect of underground telecommunication apparatus)	
		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT	
		(in respect of underground telecommunication apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-003 cont'd		Unknown	
	21177 square metres of agricultural land (Riston Grange Farm, west of Whitecross Road, A165)	Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 252 (in respect of underground water apparatus) Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 45R (in respect of rights of access) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect of a gas main and rights listed in a Deed dated 3 October 1966) Openreach Limited 6 Gracechurch Street LONDON EGJY 0AT (in respect of underground telecommunication apparatus)	
		(in respect of right of entry and access to services as contained in a Conveyance dated 4 October 1977)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-005	44238 square metres of agricultural land and drain (Arnold and Riston Drain) (Riston	Beverley & North Holderness Internal Drainage Board	
	Grange Farm, west of Whitecross Road, A165)	Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights and rights of access) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of a gas main and rights listed in a Deed dated 3 October 1966) Northern Powergrid (Yorkshire) PLC Lloyds Court	
10-006	27667 square metres of agricultural land and drain (Arnold and Riston Drain) (Riston Grange Farm, west of Whitecross Road, A165)	78 Grey Street         Newcastle Upon Tyne         NE1 6AF         (in respect of overhead electricity apparatus)         Unknown         (in respect of right of entry and access to services as contained in a Conveyance dated 4 October 1977)         Beverley & North Holderness Internal Drainage Board         Derwent House         Crockey Hill         Yorls         YO19 4SR         (in respect of rights of access)	



	Dogger Bank So	uth Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-007	18189 square metres of agricultural land and public footpath (Riston Footpath No.2) (nort	h Walter Stuart Leonard Kirkwood
	of Carr Lane)	Criftins Farm Catwick Lane Long Riston HULL East Riding of Yorkshire HU11 5JR (as The Executor of the Estate of the Late Walter Trevor Kirkwood) (in respect of a right of way as contained in a Conveyance dated 19 July 1967)
10-008	2 square metres of verge (north of Carr Lane)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect of rights listed in a Deed of Grant dated 5 September 1967) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of the rights of access to maintain drainage as contained in a Conveyance dated 4 February 1957) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights contained in a Lease dated 14 July 2015)



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-008		T H Caley and Sons Limited	
cont'd		Church Farm	
		Meaux Road	
		Routh	
		BEVERLEY	
		North Humberside	
		HU17 9SR /in recent of rights of access and use of convise modia as contained in a Transfor dated 1 July 2015)	
		(in respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)	
10-009	2273 square metres of agricultural land, verges and drain (north of Carr Lane)	Environment Agency	
		Horizon House	
		Deanery Road	
		BRISTOL	
		Avon	
		BS1 5AH	
		(in respect of rights to maintain)	
		Richard Guy Caley	
		North Grange	
		Meaux	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 9SS	
		(in respect of rights of drainage)	
		Thomas Stephen Caley	
		Church Farm	
		Meaux Road	
		Routh	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 9SR	
		(in respect of rights of drainage)	
		Unknown	
		·	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-010	16062 square metres of agricultural land (Church Farm, south of Main Road, A1035)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights of access to enter and conduct works to maintain drainage as contained in a Conveyance dated 28 March 1956) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus)	
10-014	64311 square metres of agricultural land, access tracks and drains (Church Farm, south of Main Road, A1035 and east of Meaux Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights of access to enter and conduct works to maintain drainage as contained in a Conveyance dated 28 March 1956)	
10-015	15746 square metres of agricultural land and access tracks (Church Farm, south of Main Road, A1035 and east of Meaux Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights of access to enter and conduct works to maintain drainage as contained in a Conveyance dated 28 March 1956)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
10-016	18019 square metres of agricultural land, access track and drains (south of Main Road,	Centrica PLC	
	A1035)	Millstream	
		Maidenhead Road	
		WINDSOR .	
		Berkshire	
		SL4 5GD	
		(in respect of rights listed in a Deed of Grant dated 5 September 1967)	
		Environment Agency	
		Horizon House	
		Deanery Road	
		BRISTOL	
		Avon	
		BS1 5AH	
		(in respect of the rights of access to maintain drainage as contained in a Conveyance dated 4 February 1957)	
		Northern Powergrid (Yorkshire) PLC	
		Lloyds Court	
		78 Grey Street	
		Newcastle Upon Tyne	
		NE1 6AF (in respect of overhead and underground electricity apparatus)	
		(in respect of overnead and underground electricity apparatus)	
		Northern Powergrid (Yorkshire) PLC	
		Lloyds Court	
		78 Grey Street	
		Newcastle Upon Tyne	
		NE1 6AF	
		(in respect of rights contained in a Lease dated 14 July 2015)	
		T H Caley and Sons Limited	
		Church Farm	
		Meaux Road	
		Routh	
		BEVERLEY	
		North Humberside	
		HU17 9SR (in respect of rights of access and use of service modia as contained in a Transfer dated 1 July 2015)	
		(in respect of rights of access and use of service media as contained in a Transfer dated 1 July 2015)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-017	971 square metres of public road, verges and access splay (Main Road, A1035)	KCOM Group Limited         37 Carr Lane         Hull         East Yorkshire         HU1 3RE         (in respect of underground telecommunication apparatus)         Northern Powergrid (Yorkshire) PLC         Lloyds Court         78 Grey Street         Newcastle Upon Tyne         NE1 6AF         (in respect of overhead and underground electricity apparatus)         Openreach Limited         6 Gracechurch Street         LONDON         EC3V 0AT         (in respect of underground telecommunication apparatus)         Yorkshire Water Services Limited         Western House         Western House         West Yorkshire         Bradford         West Yorkshire         BD 625Z         (in respect of underground water apparatus)	
10-018	789 square metres of public road and access splay (Main Road, A1035)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne	
		NE1 6AF (in respect of overhead and underground electricity apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-018		Openreach Limited	
cont'd		6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus)	
		Unknown	
10-019	35853 square metres of agricultural land and drains (Manor House Farm, east of Meaux	Benjamin David James Mackrill	
	Lane)	West Barn	
		Furnace Farm	
		Furnace Lane	
		Lamberhurst	
		TUNBRIDGE WELLS	
		TN3 8LE	
		(in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)	
		HN Sinkler & Son Limited	
		Manor Farm	
		Routh	
		Beverley	
		East Yorkshire	
		HU17 9SR	
		(in respect of rights contained in a Lease of a wind turbine site and solar site dated 13 March 2014)	
		Northern Powergrid (Yorkshire) PLC Lloyds Court	
		78 Grey Street	
		Newcastle Upon Tyne	
		NE1 6AF	
		(in respect of overhead electricity apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
10-019		Northern Powergrid (Yorkshire) PLC	
cont'd		Lloyds Court	
		78 Grey Street	
		Newcastle Upon Tyne	
		NE1 6AF	
		(in respect of rights contained in a Lease dated 29 March 2012)	
		Robin John Mackrill	
		38 Blandford Road	
		LONDON	
		W4 1DX	
		(in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)	
		Unknown	
		(in respect of rights contained in a lease dated 29 March 2012)	
11-001	119 square metres of agricultural land (Manor House Farm, east of Meaux Lane)	Benjamin David James Mackrill	
		West Barn	
		Furnace Farm	
		Furnace Lane	
		Lamberhurst	
		TUNBRIDGE WELLS	
		TN3 8LE	
		(in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)	
		HN Sinkler & Son Limited	
		(in respect of rights contained in a Lease of a wind turbine site and solar site dated 13 March 2014)	
		HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-001 cont'd		Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights contained in a Lease dated 29 March 2012) Robin John Mackrill 38 Blandford Road LONDON W4 1DX (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986) Unknown (in respect of rights contained in a lease dated 29 March 2012)	
11-002	548 square metres of public road and verges (Meaux Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Described Forme and Descriptions: Descriptions 2000)	
11-003	246 square metres of agricultural land (Manor House Farm, west of Meaux Lane)	(Applications: Prescribed Forms and Procedures) Regulations 2009 Benjamin David James Mackrill	
11-003	246 square metres of agricultural land (Manor House Farm, west of Meaux Lane)	West Barn	
		Furnace Farm	
		Furnace Lane	
		Lamberhurst	
		TUNBRIDGE WELLS	
		TN3 8LE	
		(in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)	
		HN Sinkler & Son Limited	
		Manor Farm	
		Routh	
		Beverley	
		East Yorkshire	
		HU17 9SR	
		(in respect of a Lease of a wind turbine site and solar site dated 13 March 2014)	
		Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V 0AT	
		(in respect of underground telecommunication apparatus)	
		Robin John Mackrill	
		38 Blandford Road	
		LONDON	
		W4 1DX	
		(in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)	
11-004	775 square metres of public road and verges (Meaux Lane)	Openreach Limited	
		6 Gracechurch Street	
		EC3V 0AT (in respect of underground telecommunication apparatus)	
		(in respect of underground telecommunication apparatus)	
		Unknown	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-004 cont'd	116 cause motions of agricultural land (Mapor House Form, part of Mapur Jana)	Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
11-005	136 square metres of agricultural land (Manor House Farm, east of Meaux Lane)	Benjamin David James Mackrill West Barn Furnace Farm Furnace Farm Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986) HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR (in respect of rights contained in a Lease of a wind turbine site and solar site dated 13 March 2014) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights contained in a Lease dated 29 March 2012) Robin John Mackrill 38 Blandford Road LONDON W4 1DX (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
Number on Land Plans	Description of Land	County of East Riding of Yorkshire Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-006	321 square metres of public road and verges (Meaux Lane)	Openreach Limited         6 Gracechurch Street         LONDON         EC3V 0AT         (in respect of underground telecommunication apparatus)         Unknown         Yorkshire Water Services Limited         Western House         Western Way         Halifax Road         Bradford         West Yorkshire         BD6 2SZ         (in respect of underground water apparatus)	
11-007	160 square metres of agricultural land (Manor House Farm, west of Meaux Lane)	Benjamin David James Mackrill West Barn Furnace Farm Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986) HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 9SR (in respect of a Lease of a wind turbine site and solar site dated 13 March 2014)	



	Dogger Bank Sou	th Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number or Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-007 cont'd		Robin John Mackrill 38 Blandford Road LONDON W4 1DX (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)
11-008	73946 square metres of agricultural land, access track and drain (Manor House Farm, east of A1035, Routh and west of Meaux Lane)	Benjamin David James Mackrill West Barn Furnace Farm Furnace Lane Lamberhurst TUNBRIDGE WELLS TN3 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986) HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU12 9SR (in respect of a Lease of a wind turbine site and solar site dated 13 March 2014) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Robin John Mackrill 38 Blandford Road LONDON W4 1DX (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-008 cont'd		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
11-009	24147 square metres of agricultural land (east of A1035, Routh)	Benjamin David James Mackrill West Barn Furnace Farm Furnace Lane Lamberhurst TUNRIDGE WELLS TN3 8LE (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986) HN Sinkler & Son Limited Manor Farm Routh Beverley East Yorkshire HU17 95R (in respect of a Lease of a wind turbine site and solar site dated 13 March 2014) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NEI 6AF (in respect of overhead electricity apparatus) Robin John Mackrill 38 Blandford Road LONDON W4 1DX (in respect of rights to drainage as contained in a Conveyance dated 12 June 1986)	



	Dogger Bank South Offshore Wind Farm Development Consent Order			
	BOOK OF REFERENCE - PART 3			
	County of East Riding of Yorkshire			
		Names of all those entitled to enjoy easements or other private rights over land		
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,		
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning		
		(Applications: Prescribed Forms and Procedures) Regulations 2009		
11-010	1415 square metres of public road, verges, footway and access splay (A1035, Routh)	Howard Noel Sinkler		
		Manor House		
		Meaux Road		
		Routh		
		BEVERLEY		
		East Riding of Yorkshire		
		HU17 9SR		
		(in respect of rights to maintain hedges as contained in a Conveyance dated 6 June 1974)		
		lan Harold Sinkler		
		Manor House		
		Meaux Road		
		Routh		
		BEVERLEY		
		East Riding of Yorkshire		
		HU17 9SR		
		(in respect of rights to maintain hedges as contained in a Conveyance dated 6 June 1974)		
		KCOM Group Limited		
		37 Carr Lane		
		Hull		
		East Yorkshire		
		HU1 3RE		
		(in respect of underground telecommunication apparatus)		
		Openreach Limited		
		6 Gracechurch Street		
		LONDON		
		EC3V 0AT		
		(in respect of underground telecommunication apparatus)		
		The Executor of the Estate of the Late Harold Sinkler		
		Manor House		
		Meaux Road		
		Routh		
		BEVERLEY		
		HU17 9SR		
		(in respect of rights to maintain hedges as contained in a Conveyance dated 6 June 1974)		



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3			
	County of East Riding of Yorkshire			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
11-010 cont'd		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)		
11-011	1056 square metres of public road, verges, footway and access splay (A1035, Routh)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown		
	520 square metres of agricultural land and access track (Hall Farm and Field House Farm, south of Tickton Bypass, A1035)	Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights and easements contained in a Lease dated 21 February 2020. Also in respect of rights granted as contained in a Deed dated 12 October 2023.)		



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-013		Doggerbank Offshore Wind Farm Project 2 Projco Limited	
cont'd		No.1 Forbury Place	
1		43 Forbury Road	
1		Reading	
1		RG1 3JH	
		(in respect of rights of access to conduct works as contained in a Deed of Grant dates 6 April 1984. Also in respect of rights contained in a	
		Lease dated 21 February 2020)	
		East Riding Of Yorkshire Council	
		County Hall	
		Cross Street	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 9BA	
		(in respect of rights contained in an Option Agreement dated 24 October 2017)	
		Hall Farm Wind Farm Limited	
		C/O Res Limited	
		Beaufort Court	
		Egg Farm Lane	
		Kings Langley	
		Hertfordshire	
		WD4 8LR	
		(in respect of rights contained in a Lease dated 24 November 2011)	
		Northern Powergrid (Yorkshire) PLC	
		Lloyds Court	
		78 Grey Street	
		Newcastle Upon Tyne	
		NE1 6AF	
		(in respect of rights contained in a Lease)	
		Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V 0AT	
		(in respect of overhead telecommunication apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-014	77 square metres of public road, verge and access splay (A1035, Routh)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
11-015	607 square metres of public road, verges and access splay (A1035, Routh)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-015 cont'd		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
11-016	1393 square metres of public road, verges, footway and access splay (A1035, Routh)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	
	4338 square metres of agricultural land, access track, hardstanding and access splay (Hall Farm and Field House Farm, north of Tickton Bypass, A1035)	Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights and easements contained in a Lease dated 21 February 2020. Also in respect of rights granted as contained in a Deed dated 12 October 2023.) Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights of access to conduct works as contained in a Deed of Grant dates 6 April 1984. Also in respect of rights contained in a Lease dated 21 February 2020)	



	Dogger E	Bank South Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
11-017		East Riding Of Yorkshire Council
cont'd		County Hall
		Cross Street
		BEVERLEY
		East Riding of Yorkshire
		HU17 9BA
		(in respect of rights contained in an Option Agreement dated 24 October 2017)
		Hall Farm Wind Farm Limited
		C/O Res Limited
		Beaufort Court
		Egg Farm Lane
		Kings Langley
		Hertfordshire
		WD4 8LR
		(in respect of rights contained in a Lease dated 24 November 2011)
		KCOM Group Limited
		37 Carr Lane
		Hull
		East Yorkshire
		HU1 3RE
		(in respect of underground telecommunication apparatus)
		Northern Powergrid (Yorkshire) PLC
		Lloyds Court
		78 Grey Street
		Newcastle Upon Tyne
		NE1 6AF
		(in respect of overhead electricity apparatus)
		Northern Powergrid (Yorkshire) PLC
		Lloyds Court
		78 Grey Street
		Newcastle Upon Tyne
		NE1 6AF
		(in respect of rights contained in a Lease dated 14 September 2018)



	Dogger Bank Sou	th Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3			
		County of East Riding of Yorkshire		
		Names of all those entitled to enjoy easements or other private rights over land		
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,		
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning		
		(Applications: Prescribed Forms and Procedures) Regulations 2009		
	175779 square metres of agricultural land, access tracks, hedgerows and drains (Hall Farm	Doggerbank Offshore Wind Farm Project 1 Projco Limited		
	and Field House Farm, east of Eske Lane and west of A1035, Routh)	No.1 Forbury Place		
		43 Forbury Road		
		Reading		
		RG1 3JH		
		(in respect of rights and easements contained in a Lease dated 21 February 2020. Also in respect of rights granted as contained in a Deed dated 12 October 2023.)		
		Doggerbank Offshore Wind Farm Project 2 Projco Limited		
		No.1 Forbury Place		
		43 Forbury Road		
		Reading		
		RG1 3JH		
		(in respect of rights of access to conduct works as contained in a Deed of Grant dates 6 April 1984. Also in respect of rights contained in a		
		Lease dated 21 February 2020)		
		East Riding Of Yorkshire Council		
		County Hall		
		Cross Street		
		BEVERLEY		
		East Riding of Yorkshire		
		НU17 9ВА		
		(in respect of rights contained in an Option Agreement dated 24 October 2017)		
		Hall Farm Wind Farm Limited		
		C/O Res Limited		
		Beaufort Court		
		Egg Farm Lane		
		Kings Langley		
		Hertfordshire		
		WD4 8LR		
		(in respect of rights contained in a Lease dated 24 November 2011)		



Dogger Bank South Offshore Wind Farm Development Consent Order			
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-018 cont'd		Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NEI 6AF (in respect of overhead electricity apparatus) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NEI 6AF (in respect of rights contained in a Lease dated 14 September 2018) Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 25Z (in respect of rights to maintain a watermain as contained in a Deed dated 6 April 1984) Yorkshire Water Services Limited Western Way Halifax Road Bradford Western House Western Way Halifax Road Bradford Western Way Halifax Road Bradford Western Way Halifax Road Bradford Western Way Halifax Road Bradford West Yorkshire BD6 25Z (in respect of underground water apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order			
		BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire		
		Names of all those entitled to enjoy easements or other private rights over land		
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,		
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning		
		(Applications: Prescribed Forms and Procedures) Regulations 2009		
12-001	20685 square metres of agricultural land and drain (Hall Farm and Field House Farm, east	Doggerbank Offshore Wind Farm Project 1 Projco Limited		
1	of Eske Lane)	No.1 Forbury Place		
		43 Forbury Road		
		Reading		
		RG1 3JH		
		(in respect of rights and easements contained in a Lease dated 21 February 2020. Also in respect of rights granted as contained in a Deed dated 12 October 2023.)		
		Doggerbank Offshore Wind Farm Project 2 Projco Limited		
		No.1 Forbury Place		
		43 Forbury Road		
		Reading		
		RG1 3JH		
		(in respect of rights of access to conduct works as contained in a Deed of Grant dates 6 April 1984. Also in respect of rights contained in a		
		Lease dated 21 February 2020)		
		East Riding Of Yorkshire Council		
		County Hall		
		Cross Street		
		BEVERLEY		
		East Riding of Yorkshire		
		HU17 9BA		
		(in respect of rights contained in an Option Agreement dated 24 October 2017)		
		Hall Farm Wind Farm Limited		
		C/O Res Limited		
		Beaufort Court		
		Egg Farm Lane		
		Kings Langley		
		Hertfordshire		
		WD4 8LR		
		(in respect of rights contained in a Lease dated 24 November 2011)		



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-001 cont'd		Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights contained in a Lease dated 14 September 2018)	
12-002	127 square metres of agricultural land and drain (east of Eske Lane)	Unknown	
12-003	959 square metres of public road and verge (Eske Lane)	Unknown	
12-004	85 square metres of agricultural land and drain (east of Eske Lane)	Unknown	
12-005	564 square metres of public road and verge (Eske Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
12-006	5985 square metres of public road and verges (Eske Lane and Tickton Bypass, A1035)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
12-006		Northern Powergrid (Yorkshire) PLC	
cont'd		Lloyds Court	
		78 Grey Street	
		Newcastle Upon Tyne	
		NE1 6AF	
		(in respect of underground and overhead electricity apparatus)	
		Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V 0AT	
		(in respect of underground telecommunication apparatus)	
		Unknown	
12-007	25 square metres of agricultural land and copse (north of Tickton Bypass, A1035 and west	John Scott	
	of Eske Lane)	Tickton Farm	
		56 Main Street	
		Tickton	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 9RZ	
		(in respect of rights to inspect, clean, maintain, repair and renew the buildings, walls and fences contained in a Transfer dated 19	
		December 2017)	
12-008	378 square metres of public road and verges (Eske Lane and Tickton Bypass, A1035)	KCOM Group Limited	
		37 Carr Lane	
		Hull	
		East Yorkshire	
		HU1 3RE	
		(in respect of underground telecommunication apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
12-008		Northern Powergrid (Yorkshire) PLC	
cont'd		Lloyds Court	
		78 Grey Street	
		Newcastle Upon Tyne	
		NE1 6AF	
		(in respect of underground electricity apparatus)	
12-009	241 square metres of public road and verges (Tickton Bypass, A1035)	KCOM Group Limited	
		37 Carr Lane	
		Hull	
		East Yorkshire	
		HU1 3RE	
		(in respect of underground telecommunication apparatus)	
		Northern Powergrid (Yorkshire) PLC	
		Lloyds Court	
		78 Grey Street	
		Newcastle Upon Tyne	
		NE1 6AF	
		(in respect of overhead and underground electricity apparatus)	
		Unknown	
		(in respect of rights contained in a Conveyance dated 19 September 1972)	
12-010	2920 square metres of agricultural land (west of Eske Lane)	Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V OAT	
		(in respect of underground telecommunication apparatus)	
		(,	
12-011	96989 square metres of agricultural land, hedgerows and drains (west of Eske Lane)	Beverley & North Holderness Internal Drainage Board	
		Derwent House	
		Crockey Hill	
		York	
		YO19 4SR	
		(in respect of riparian rights and rights of access)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3		
		BOOK OF REFERENCE - PART 3 County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-011 cont'd		Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	
12-013	101 square metres of river bank (east of the River Hull, Beverley)	Unknown	
	770 square metres of grassland, river bank, public bridleway (Leconfield Bridleway No.27) and public footpath (Leconfield Footpath No.33) (west of the River Hull, Beverley)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus) Unknown	
	2740 square metres of grassland, river bank, public bridleway (Leconfield Bridleway No.27) and public footpath (Leconfield Footpath No.33) (west of the River Hull, Beverley)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-002 cont'd		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of rights contained in a Transfer scheme dated 1 September 1989)	
13-004	71530 square metres of agricultural land and copse (Carr Farm, east of Carr Road and north of Hull Bridge Road, A1035)	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of rights of access) Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access and the free passage of water and electricity contained in a Conveyance dated 25 March 1974) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-004		Phillip Brumfield	
cont'd		Dunkenhill Farm Highgate Cherry Burton BEVERLEY HU17 7RT (as The Executor of the Estate of The Late David Roy Duncan Brumfield) (in respect of rights contained in a Transfer dated 26 February 1993) The Executor Of The Estate Of The Late David Roy Duncan Brumfield The Beeches Seven Corners Lane BEVERLEY East Riding of Yorkshire HU17 7AJ	
13-005	243 square metres of drain (west of the River Hull, Beverley)	(in respect of rights contained in a Transfer dated 26 February 1993) Beverley & North Holderness Internal Drainage Board	
		Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights) Unknown	
	185 square metres of agricultural land and dike (South Bullock Dike, Carr Farm, east of Carr Road and north of Hull Bridge Road, A1035)	r Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights and rights of access)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-006 cont'd		Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access and the free passage of water and electricity contained in a Conveyance dated 25 March 1974) Phillip Brumfield Dunkenhill Farm Highgate Cherry Burton BEVERLEY HU17 7RT (as The Executor of the Estate of The Late David Roy Duncan Brumfield) (in respect of rights contained in a Transfer dated 26 February 1993) The Executor Of The Estate Of The Late David Roy Duncan Brumfield The Beeches Seven Corners Lane BEVERLEY East Riding of Yorkshire HU17 7AJ	
13-007	4825 square metres of agricultural land and dike (South Bullock Dike, east of Carr Road)	(in respect of rights contained in a Transfer dated 26 February 1993) Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights) Unknown (in respect of rights to service media reserved by a Transfer dated 22 June 1989)	



	Dogger Bank Sout	th Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
13-008	4158 square metres of agricultural land, culvert and drain (Molescroft Grange Farm, east of	
	Carr Road)	Derwent House
		Crockey Hill
		York
		YO19 4SR
		(in respect of riparian rights and rights of access)
		Michael Jonathan Watson Hall
		Manor Farm
		Manor House Lane
		Walkington
		BEVERLEY
		East Riding of Yorkshire
		HU17 8SU
		(in respect of restrictions contained in a Transfer dated 12 November 2012)
		S R J Developments (Yorkshire) Limited
		Belgrave House
		15 Belgrave Crescent
		SCARBOROUGH
		North Yorkshire
		Y011 1UB
		(in respect of rights of access and others listed in a Transfer dated 14 July 2000)
		Yorkshire Water Services Limited
		Western House
		Western Way
		Halifax Road
		Bradford
		West Yorkshire
		BD6 2SZ
		(in respect of rights and easements relating to water mains, passage of water, entry and covenants listed in a Deed dated 26 September
		1995)



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
13-009	74687 square metres of agricultural land and drain (Molescroft Grange Farm, north of	Beverley & North Holderness Internal Drainage Board	
	Grange Way, A1035)	Derwent House	
		Crockey Hill	
		York	
		YO19 4SR	
		(in respect of rights of access)	
		Michael Jonathan Watson Hall	
		Manor Farm	
		Manor House Lane	
		Walkington	
		BEVERLEY	
		East Riding of Yorkshire	
		HU17 8SU	
		(in respect of restrictions contained in a Transfer dated 12 November 2012)	
		Northern Powergrid (Yorkshire) PLC	
		Lloyds Court	
		78 Grey Street	
		Newcastle Upon Tyne	
		NE1 6AF	
		(in respect of overhead electricity apparatus)	
		S R J Developments (Yorkshire) Limited	
		Belgrave House	
		15 Belgrave Crescent	
		SCARBOROUGH	
		North Yorkshire	
		YO11 1UB	
		(in respect of rights of access and others listed in a Transfer dated 14 July 2000)	



	Dogger Bank South Offshore Wind Farm Development Consent Order	
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
13-009		Yorkshire Water Services Limited
cont'd		Western House
		Western Way
		Halifax Road
		Bradford
		West Yorkshire
		BD6 2SZ
		(in respect of rights and easements relating to water mains, passage of water, entry and covenants listed in a Deed dated 26 September
		1995)
	47466 square metres of agricultural land, hedgerows, drains and access tracks (Molescroft	
	Grange Farm, east of Grange Way, A1035)	Derwent House
		Crockey Hill
		York
		YO19 4SR
		(in respect of riparian rights and rights of access)
		Michael Jonathan Watson Hall
		Manor Farm
		Manor House Lane
		Walkington
		BEVERLEY
		East Riding of Yorkshire
		HU17 8SU
		(in respect of restrictions contained in a Transfer dated 12 November 2012)
		Network Rail Infrastructure Limited
		Waterloo General Office
		LONDON
		SE1 8SW
		(in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a
		Deed of Release dated 27 February 2019)



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-010		Northern Powergrid (Yorkshire) PLC	
cont'd		Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF	
		(in respect of overhead electricity apparatus)	
		S R J Developments (Yorkshire) Limited Belgrave House 15 Belgrave Crescent SCARBOROUGH North Yorkshire YO11 1UB (in respect of rights of access and others listed in a Transfer dated 14 July 2000) Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of rights and easements relating to water mains, passage of water, entry and covenants listed in a Deed dated 26 September 1995)	
	2687 square metres of agricultural land, drains and access track (east of Grange Way, A1035)	Beverley & North Holderness Internal Drainage Board Derwent House Crockey Hill York YO19 4SR (in respect of riparian rights and rights of access) Unknown (in respect of rights to service media reserved by a Transfer dated 22 June 1989)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-012	752 square metres of public road and verge (Hull Bridge Road, A1035)	(Applications: Prescribed Forms and Procedures) Regulations 2009 KCOM Group Limited	
15 012		37 Carr Lane	
		Hull	
		East Yorkshire	
		HU1 3RE	
		(in respect of underground telecommunication apparatus)	
		Northern Gas Networks Limited	
		1100 Century Way	
		Thorpe Park	
		LEEDS	
		West Yorkshire	
		LS15 8TU	
		(in respect of underground gas apparatus)	
		Northern Powergrid (Yorkshire) PLC	
		Lloyds Court	
		78 Grey Street	
		Newcastle Upon Tyne	
		NE1 6AF	
		(in respect of underground electricity apparatus)	
		Yorkshire Water Services Limited	
		Western House	
		Western Way	
		Halifax Road	
		Bradford	
		West Yorkshire	
		BD6 2SZ	
		(in respect of underground water apparatus)	
13-013	395 square metres of public road (Hull Bridge Road, A1035)	Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V OAT	
		(in respect of underground telecommunication apparatus)	
i			



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-013 cont'd		Unknown	
13-014	1604 square metres of railway track and verges (east of Carr Road)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) Unknown	
13-015	11442 square metres of agricultural land and pond (north of Carr Road)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rights to access a level crossing contained in a Deed of Release dated 12 September 1995) Phillip Brumfield Dunkenhill Farm Highgate Cherry Burton BEVERLEY HU17 7RT (as The Executor of the Estate of the Late Barbara Brumfield and The Late David Roy Duncan Brumfield) (in respect of rights contained in a Transfer dated 22 June 1989) S R J Developments (Yorkshire) Limited Belgrave House 15 Belgrave Crescent SCARBOROUGH North Yorkshire YO11 1UB (in respect of rights, agreements and covenants listed in a Transfer dated 14 July 2000)	



Subscription         Subscription<		Dogger Bank South Offshore Wind Farm Development Consent Order		
Number on Land Pins         Description of Land         Number on Induffing private regists over land (including private regists over worker) which its proposed ball between worker which its proposed ball between social between worker which its proposed ball between worker) which its proposed ball between worker which its proposed ball between worker which its proposed ball between social between worker which its proposed ball between social between worker which its proposed ball between social between so		BOOK OF REFERENCE - PART 3		
Number of Land Plans         Description of Land         Including private rights of nalgebiations: Proceedings Integration (1) the Interacture Internation (Applications: Proceeding Interference)           13-055         cont'd         The Execution (1) the Extee Of the		County of East Riding of Yorkshire		
13-015 cont'd       The Executor Of the Exit and Full Late David Roy Duncan Brumfield         13-016       Ber Verker Ber Riding of Yorkshire         13-016       8219 square metres of agricultural land, access track and public footpath (Molescroft Footpath No.5) (Low Parks Farm, north of Carr Road)       Catherine Suzame Make The Chestinuts Carr Road BEVERLEY         13-016       8219 square metres of agricultural land, access track and public footpath (Molescroft Footpath No.5) (Low Parks Farm, north of Carr Road)       Catherine Suzame Make The Chestinuts Carr Road BEVERLEY         13-017       Road Bever Carr Road       Northern Powergid (Vorkalire) HUT 77 Zr (In respect of rights of access as contained in an indenture dated 20 September 1920)         13-017       By Park square metres of agricultural land and drains (north of Carr Road)       Northern Powergid (Vorkalire) HUT 77 Zr (In respect of rights of access as contained in an indenture dated 20 September 1920)         13-017       By Park square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited Waterion General Office LUD 77 Zr (In respect of rights of access as contained in an indenture dated 20 September 1920)         13-017       System Street Str		Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
cont'd     be tenches       seven Corres Jane       13-016       202 square metres of agricultural land, access track and public footpath (Molescroft)       The Chestnuts       Carr Road       BOYENLEY       isat Riding of Yorkshire       HUL7 7A       (in respect of rights contained in an Indenture dated 20 September 1920)       Northern Powergrid (Yorkshire)       HL2       Vertex Vertex       Vertex Vertex <td>13-015</td> <td></td> <td></td>	13-015			
13-016       Seven Correst Jane         82/07/201       State metres of agricultural land, access track and public footpath (Molescroft footpath Ro.5) (Low Parks Farm, north of Carr Road)       Catherine Suzame Mace         13-016       State metres of agricultural land, access track and public footpath (Molescroft footpath Ro.5) (Low Parks Farm, north of Carr Road)       Catherine Suzame Mace         13-016       Northern Powergind (Yortshire)       Catherine Suzame Mace         13-017       Northern Powergind (Yortshire)       Northern Powergind (Yortshire)         113-018       State Suzame Mace       Northern Powergind (Yortshire)         113-017       Northern Powergind (Yortshire)       Northern Powergind (Yortshire)         113-017       State Suzame Mace       Network Rail Infrastructure Limited         Waterio General Office       Northern Powergind office       Northern Powergind (Yortshire)         113-017       State Suzame Mace       Network Rail Infrastructure Limited				
Image: Set Riding of Yorkshire         13-016       8219 square metres of agricultural land, access track and public footpath (Molescroft Footpath Road)       Catherine Suzanne Mace The Chestnuls Catherine Suzanne Mace The Chestnuls Catherine Suzanne Mace The Chestnuls Catherine Road BEVERLEY         Image: Road BEVERLEY       Catherine Suzanne Mace The Chestnuls Catherine Road (Yorkshire HUT 772)         Image: Road BEVERLEY       Catherine Suzanne Mace The Chestnuls Catherine Road (Yorkshire HUT 772)         Image: Road BEVERLEY       Catherine Suzanne Mace The Chestnuls Catherine Road (Yorkshire) PLC Lloyds Court Road)         Northern Powergid (Yorkshire) PLC Lloyds Court The Road Catherine Powergid (Yorkshire) PLC Lloyds Court The Road Electricity apparatus)         Thomas Anthrony Roberd Mace The Chestnuts Care Road (PUT) PLOC Road (Yorkshire PUT)         13-017       \$747 square metres of agricultural land and drains (north of Carr Road)         Network Rail Infrastructure Limited Waterioo General Office LONDON SET ISSW (In respect of rights released relating to a level crossing listed in a longer 1055, Also in respect of rights released relating to a level crossing listed in a longer 1055, Also in respect of rights released relating to a level crossing listed in a longer 1055, Also in respect of rights released relating to a level crossing listed in a longer 1055, Also in respect of rights released relating to a level crossing listed in a longer 1055, Also in respect of rights released relating to a level crossing listed in a longer 1055, Also in respect of rights release			Seven Corners Lane	
13-016       HUT 7A/ (In respect of rights contained in a Transfer dated 22 June 1989)         13-016       82219 square metres of agricultural land, access track and public footpath (Molescroft Footpath No.5.) (Low Panis Farm, north of Carr Road)       Catherine Suzanne Mace The Chestnutus Carr Road         13-016       Footpath No.5.) (Low Panis Farm, north of Carr Road)       Catherine Suzanne Mace The Chestnutus Carr Road         13-017       Footpath No.5.) (Low Panis Farm, north of Carr Road)       Northere Deworged (Vorkshire HUT 77Z) (In respect of rights of access as contained in an Indenture dated 20 September 1920)         Northere Deworged (Vorkshire) PLC Uoyds Court 78 Grey Street Newcasle Upon Type Net GAF (In respect of rights of access as contained in an Indenture dated 20 September 1920)         13-017       Sp747 square metres of agricultural land and drains (north of Carr Road)         Network Rail Infrastructure Limited Waterboo General Offlice UNDON St 1 85W       Network Rail Infrastructure Limited Waterboo General Offlice UNDON			BEVERLEY	
13-016       (in respect of rights contained in a Transfer dated 22 June 1989)         13-016       2219 square metres of agricultural land, access track and public footpath (Molescroft Footpath No.5) (Low Parks Farm, north of Carr Road)       Catherine Suzanne Mace The Chestnuss Carr Road BEVERLY         13-016       Read Gevent Carr Road Carr Road)       EVERLY         14       No.5.1 (Low Parks Farm, north of Carr Road)       EVERLY         15       Read Gevent Carr Road Carr Road       EVERLY         16       Northern Powergrid (Vorkshire)       HU17 71Z         (In respect of rights of access as contained in an Indenture dated 20 September 1920)       Northern Powergrid (Vorkshire) PLC         Northern Powergrid (Vorkshire) PLC       Northern Powergrid (Vorkshire) PLC         NUT 702       In respect of rights of access as contained in an Indenture dated 20 September 1920)         13-017       SirV47 square metres of agricultural land and drains (north of Carr Road)       Network Rain Grad         Nut 702       In respect of rights of access as contained in an Indenture dated 20 September 1920)			East Riding of Yorkshire	
13-016       8219 square metres of agricultural land, access track and public footpath (Molescroft Footpath No.5) (Low Parks Farm, north of Carr Road)       Catherine Suzane Mace The Chestnuls Carr Road BEVRELY East Riding of Yorkhire HU17 712 (In respect of rights of access as contained in an indenture dated 20 September 1920)         Northern Powergrid (Yorkshire) PLC Uoyds Court 78 Grey Street Newcast Upon Type NE1 GAF (In respect of rights of access as contained in an indenture dated 20 September 1920)       Northern Powergrid (Yorkshire) PLC Uoyds Court 78 Grey Street Newcast Upon Type NE1 GAF (In respect of overhead electricity apparatus)         Thomas Anthony Richard Mace The Chestnuls Carr Road       Thomas Anthony Richard Mace The Chestnuls Carr Road BEVRELY East Riding of Yorkshire HU17 712 (In respect of rights of access as contained in an indenture dated 20 September 1920)         13-017       \$9747 square metres of agricultural land and drains (north of Carr Road)       Network Rule Infrastructure Limited Waterloo General Office LONDON SE 1 83W (In respect of rights released relating to a level crossing listed in a first released to a level crossing listed in a first released relating to a level crossing listed in a first released by a Deed dated 12 September 1925, Also in respect of rights released relating to a level crossing listed in a first released by a Deed dated 12 September 1925, Also in respect of rights of access as contained at 20 September 1925, Also in respect of rights released relating to a level crossing listed in a first released by a Deed dated 12 September 1925, Also in respect of rights released relating to a level crossing listed in a first released by a Deed dated 12 September 1925, Also in respect of rights released relating to a level crossing listed in a first released by a Deed dated 12 September 1925, Also in			HU17 7AJ	
Footpath No.5) (Low Parks Farm, north of Carr Road)       The Chestruus       Carr Road         BVFRILY       East Riding of Yorkshire       HU17 712         (In respect of rights of access as contained in an Indenture dated 20 September 1920)       Northern Powergrid (Yorkshire) PLC         Log Strep Street       Newstate Upon Tyre       Newstate Upon Tyre         New cast Log Do Tyre       New cast Log Northead electricity apparatus)       Thomas Anthony Richard Mace         The Chestruus       East Riding of Yorkshire       HU17 712         (In respect of rights of access as contained in an Indenture dated 20 September 1920)       Northern Powergrid (Yorkshire) PLC         Log Strep Street       Newstate Upon Tyre       Net Cast         New cast Autony Richard Mace       The Chestruus       Thomas Anthony Richard Mace         The Chestruus       Carr Road       BEVERILY       East Riding of Yorkshire         13-017       \$9747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited       Waterloo General Office         LONDON       S1 S500       S1 Stop Stop Stop Stop Stop Stop Stop Stop			(in respect of rights contained in a Transfer dated 22 June 1989)	
13-017       \$9747 square metres of agricultural land and drains (north of Carr Road)       Reverage         13-017       \$9747 square metres of agricultural land and drains (north of Carr Road)       Nether of rights released by a Deed dated 12 September 1920)				
BeVRLEY       BeVRLEY         BevR Riding of Yorkshire       HU17 7/2         (in respect of rights of access as contained in an Indenture dated 20 September 1920)         Northern Powergid (Yorkshire) PLC         Uoyds Court         78 Grey Street         Newcastle Upon Tyne         NEI GAF         (in respect of overhead electricity apparatus)         Thomas Anthony Richard Mace         The Chestnuts         Carr Road         BEVERLEY         BevERLEY         (in respect of rights of access as contained in an Indenture dated 20 September 1920)		Footpath No.5) (Low Parks Farm, north of Carr Road)		
13-017       S9747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited         13-017       S9747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited         13-017       S9747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited         13-017       S1400 NO       Northor Rowers of the select of rights released leased by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a				
HU17 7JZ       (in respect of rights of access as contained in an Indenture dated 20 September 1920)         Northern Powergrid (Yorkshire) PLC       Loyds Court         Loyds Court       78 Grey Street         Newcastle Upon Tyne       Nexe Street         Newcastle Upon Tyne       Net GAF         (in respect of overhead electricity apparatus)       Thomas Anthony Richard Mace         The Chestnuts       Carr Road         BEVERLEY       East Riding of Yorkshire         East Riding of Yorkshire       HU17 7JZ         (in respect of rights of access as contained in an Indenture dated 20 September 1920)       Nothern Ower and Development of the Chestnuts         13-017       \$9747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited         Naterioo General Office       LONDON       Site SWV         (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a				
13-017       S9747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited         13-017       S9747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited         13-017       S9747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited         13-017       S9747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited			-	
13-017       \$9747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited         VAME       Network Rail Infrastructure Limited         VA       VA         VA       Network Rail Infrastructure Limited         VA       VA <td></td> <td></td> <td></td>				
Lloyds Court       78 Grey Street         Newcastle Upon Tyne       Newcastle Upon Tyne         Net 16 AF       (in respect of overhead electricity apparatus)         Thomas Anthony Richard Mace       The Chestnuts         Carr Road       BEVERLEY         East Riding of Yorkshire       HU17 7JZ         (in respect of rights of access as contained in an Indenture dated 20 September 1920)         13-017       59747 square metres of agricultural land and drains (north of Carr Road)         Network Rail Infrastructure Limited         Waterloo General Office         LONDON         SE1 85W         (in respect of rights released relating to a level crossing listed in a			(in respect of rights of access as contained in an Indenture dated 20 September 1920)	
13-017       \$9747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited         Network Rail Infrastructure Limited       Waterloo General Office         UNDDON       \$514 85W				
Newcastle Upon Tyne       NE1 6AF         (in respect of overhead electricity apparatus)       Thomas Anthony Richard Mace         The Chestnuts       The Chestnuts         Carr Road       BEVERLEY         East Riding of Yorkshire       HU17 7JZ         (in respect of rights of access as contained in an Indenture dated 20 September 1920)         13-017       \$9747 square metres of agricultural land and drains (north of Carr Road)         Network Rail Infrastructure Limited         Waterloo General Office         LONDON         SE1 8SW         (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a				
NE1 6AF       (in respect of overhead electricity apparatus)         Thomas Anthony Richard Mace       The Chestnuts         Carr Road       BEVERLEY         East Riding of Yorkshire       HU17 7JZ         (in respect of rights of access as contained in an Indenture dated 20 September 1920)         13-017       \$9747 square metres of agricultural land and drains (north of Carr Road)         Network Rail Infrastructure Limited         Waterloo General Office         LONDON         SEI 8SW         (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a				
Image:				
13-017       59747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited         Waterloo General Office       LONDON         SE1 85W       (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a				
13-017       59747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited         Waterloo General Office       LONDON         SEI 85W       (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a			(in respect of overhead electricity apparatus)	
Carr Road         BEVERLEY         East Riding of Yorkshire         HU17 7JZ         (in respect of rights of access as contained in an Indenture dated 20 September 1920)         13-017       59747 square metres of agricultural land and drains (north of Carr Road)         Network Rail Infrastructure Limited         Waterloo General Office         LONDON         SE1 8SW         (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a			Thomas Anthony Richard Mace	
BEVERLEY       East Riding of Yorkshire         HU17 7JZ       (in respect of rights of access as contained in an Indenture dated 20 September 1920)         13-017       59747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited         Waterloo General Office       LONDON         SE1 8SW       (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a			The Chestnuts	
East Riding of Yorkshire         HU17 7JZ         (in respect of rights of access as contained in an Indenture dated 20 September 1920)         13-017       59747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited         Waterloo General Office       LONDON         SE1 8SW       in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a				
HU17 7JZ (in respect of rights of access as contained in an Indenture dated 20 September 1920)         13-017       59747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited         Waterloo General Office LONDON SE1 8SW (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a				
Image: spect of rights of access as contained in an Indenture dated 20 September 1920)         13-017       59747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited         Waterloo General Office       LONDON         SE1 8SW       (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a				
13-017       59747 square metres of agricultural land and drains (north of Carr Road)       Network Rail Infrastructure Limited         Waterloo General Office       LONDON         SE1 8SW       (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a				
Waterloo General Office LONDON SE1 8SW (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a			(in respect of rights of access as contained in an indenture dated 20 September 1920)	
LONDON SE1 8SW (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a	13-017	59747 square metres of agricultural land and drains (north of Carr Road)		
SE1 8SW (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a				
(in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a				



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-017		Openreach Limited	
cont'd		G Gracechurch Street LONDON EC3V 0AT	
		(in respect of underground telecommunication apparatus)	
		S R J Developments (Yorkshire) Limited Belgrave House 15 Belgrave Crescent	
		SCARBOROUGH North Yorkshire YO11 1UB (in respect of rights of access and others listed in a Transfer dated 14 July 2000)	
	21440 square metres of agricultural land (north of Carr Road, Molescroft and north of Ings Road)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a Deed of Release dated 27 February 2019)	
		S R J Developments (Yorkshire) Limited Belgrave House 15 Belgrave Crescent SCARBOROUGH North Yorkshire YO11 1UB (in respect of rights of access and others listed in a Transfer dated 14 July 2000)	
14-002	13 square metres of public road and verge (Carr Road)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rights to access a level crossing contained in a Deed of Release dated 12 September 1995)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-002		Phillip Brumfield	
cont'd		Dunkenhill Farm Highgate Cherry Burton BEVERLEY HU17 7RT (as The Executor of the Estate of the Late Barbara Brumfield and The Late David Roy Duncan Brumfield) (in respect of rights contained in a Transfer dated 22 June 1989) The Executor Of The Estate Of The Late David Roy Duncan Brumfield The Beeches Seven Corners Lane BEVERLEY East Riding of Yorkshire HU17 7AJ (in respect of rights contained in a Transfer dated 22 June 1989) Unknown (in respect of rights contained in a Conveyance dated 11 September 1975)	
14-003	2309 square metres of public road and verges (Ings Road)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus) Unknown	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-004	1421 square metres of public roads and verges (Ings Road and Grange Way, A1035)		
14-004	1421 square metres of public roads and verges (Ings Road and Grange Way, A1035)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Richmond Properties (UK) Limited Field Head Driffield Road Molescroft Beverley East Yorkshire HU17 7LU (in respect of rights of access to development sites as contained in a Transfer dated 31 December 1993) Shirethorn Limited Unit 88 Marina Court Castle Street HULL East Riding of Yorkshire HULL East Riding of Yorkshire HULL Manor Farm Manor House Lane	
		Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (in respect of rights contained in a Transfer dated 22 June 1989)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-008	26576 square metres of agricultural land (east of Driffield Road, A164)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of underground electricity apparatus) Unknown (in respect of rights to service media reserved by a Transfer dated 22 June 1989)	
14-009	5 square metres of agricultural land (east of Driffield Road, A164)	Unknown (in respect of rights to service media reserved by a Transfer dated 22 June 1989)	
14-010	42 square metres of public road and verges (Driffield Road, A164)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
14-010		Yorkshire Water Services Limited	
cont'd		Western House	
		Western Way	
		Halifax Road	
		Bradford	
		West Yorkshire	
I I		BD6 2SZ	
		(in respect of underground water apparatus)	
14-011	1998 square metres of public road and verges (Driffield Road, A164)	KCOM Group Limited	
		37 Carr Lane	
		Hull	
		East Yorkshire	
I I		HU1 3RE	
		(in respect of underground telecommunication apparatus)	
		Northern Gas Networks Limited	
I I		1100 Century Way	
I I		Thorpe Park	
		LEEDS	
		West Yorkshire	
		LS15 8TU	
		(in respect of underground gas apparatus)	
		Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V OAT	
		(in respect of underground telecommunication apparatus)	
		Unknown	
íI		1	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-011		Yorkshire Water Services Limited	
cont'd		Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
14-012	3 square metres of agricultural land (west of Driffield Road, A164)	Unknown (in respect of rights to service media reserved by a Transfer dated 22 June 1989)	
14-013	34952 square metres of agricultural land (west of Driffield Road, A164)	Unknown (in respect of rights to service media reserved by a Transfer dated 22 June 1989)	
14-014	63766 square metres of agricultural land and hedgerow (Constitution Hill Farm, north of Malton Road, A1035)	Unknown (in respect of rights relating to a sewer listed in a Conveyance dated 9 November 1920)	
15-001	22062 square metres of agricultural land and hedgerow (Constitution Hill Farm, north of Malton Road, A1035)	Unknown (in respect of rights relating to a sewer listed in a Conveyance dated 9 November 1920)	
15-002	973 square metres of public road, access splay and verges (Malton Road, A1035)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus) Unknown	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-002 cont'd		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
15-004	1845 square metres of public road and verges (Malton Road, A1035)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (In respect of underground telecommunication apparatus) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (In respect of underground gas apparatus) Unknown Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (In respect of underground water apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-005	15652 square metres of agricultural land (Molescroft Grange Farm, south of Malton Road,	Michael Jonathan Watson Hall	
	A1035)	Manor Farm Manor House Lane	
		Walkington BEVERLEY East Riding of Yorkshire	
		HU17 8SU (in respect of restrictions contained in a Transfer dated 12 November 2012)	
		S R J Developments (Yorkshire) Limited Belgrave House 15 Belgrave Crescent SCARBOROUGH North Yorkshire YO11 1UB	
		(in respect of rights of access and others listed in a Transfer dated 14 July 2000)	
	30513 square metres of agricultural land (Molescroft Grange Farm, south of Malton Road, A1035)	Michael Jonathan Watson Hall Manor Farm Manor House Lane Walkington BEVERLEY East Riding of Yorkshire HU17 8SU (in respect of restrictions contained in a Transfer dated 12 November 2012) Network Rail Infrastructure Limited	
		Waterloo General Office LONDON SE1 8SW (in respect of rights released by a Deed dated 12 September 1995. Also in respect of rights released relating to a level crossing listed in a Deed of Release dated 27 February 2019)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3		
Number on Land Plans	Description of Land	County of East Riding of Yorkshire Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
15-006 cont'd		(Applications: Prescribed Forms and Procedures) Regulations 2009         Northern Powergrid (Yorkshire) PLC         Lloyds Court         78 Grey Street         Newcastle Upon Tyne         NE1 6AF         (in respect of overhead electricity apparatus)         S R J Developments (Yorkshire) Limited         Belgrave House         15 Belgrave Crescent         SCARBOROUGH         North Yorkshire         YO11 1UB         (in respect of rights of access and others listed in a Transfer dated 14 July 2000)	
15-007	166 square metres of drain (south of Malton Road, A1035)	Unknown	
15-008	15547 square metres of agricultural land and hedgerow (south of Malton Road, A1035)	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (in respect of rights reserved by a Deed dated 4 March 1988)	
	48573 square metres of agricultural land, access track and hedgerows (Mount Pleasant Farm, north of York Road)	Lindum Developments Limited Lindum Business Park Station Road North Hykeham Lincoln LN6 3QX (in respect of rights of access and covenants listed in a Transfer dated 2 March 2021)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-001 cont'd		Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Unknown (in respect of rights of way, light and drainage listed in a Conveyance dated 11 September 1953)	
16-002	1418 square metres of public road, verges, scrubland and access splay (York Road)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Lindum Developments Limited Lindum Business Park Station Road North Hykeham Lincoln LN6 3QX (in respect of rights of access and covenants listed in a Transfer dated 2 March 2021) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans 16-002	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 Unknown	
cont'd		(in respect of rights of way, light and drainage listed in a Conveyance dated 11 September 1953)	
16-003	624 square metres of scrubland (Mount Pleasant Farm, north of York Road)	Lindum Developments Limited Lindum Business Park Station Road North Hykeham Lincoln LN6 3QX (in respect of rights of access and covenants listed in a Transfer dated 2 March 2021) Unknown (in respect of rights of way, light and drainage listed in a Conveyance dated 11 September 1953)	
16-004	91 square metres of public road, verge and access splay (York Road, Beverley) (excluding all interests of the Crown)	Unknown	
16-005	27 square metres of public road and verges (York Road, Beverley) (excluding all interests of the Crown)	Unknown	
16-006	133 square metres of public road and verges (York Road, Beverley) (excluding all interests of the Crown)	Unknown (in respect of restrictive covenants imposed before 11 February 2009)	
16-007	2983 square metres of public road and verges (York Road, Beverley) (excluding all interests of the Crown)	Unknown (in respect of restrictive covenants imposed before 11 February 2009)	
16-008	32635 square metres of agricultural land and hedgerow (Mount Pleasant Farm, south of York Road)	Unknown (in respect of rights of way, light and drainage listed in a Conveyance dated 11 September 1953)	
17-001	832 square metres of public road and verges (Newbald Road)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HUI 3RE (in respect of underground telecommunication apparatus) Unknown	



	Dogger Bank South Offshore Wind Farm Development Consent Order			
	BOOK OF REFERENCE - PART 3			
	County of East Riding of Yorkshire			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
17-004	1387 square metres of public road and verge (Broadgate, B1230)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Unknown Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground sewerage and water apparatus)		
17-005	1344 square metres of public road, access splays and verges (Broadgate, B1230)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus) Unknown		



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-005 cont'd		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground sewerage and water apparatus)	
17-006	541 square metres of public road and verge (Broadgate, B1230)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of underground gas apparatus) Unknown Yorkshire Water Services Limited Western House Western May Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground sewerage and water apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
17-007	430 square metres of grassland (south of Broadgate, B1230)	Bryant Homes Northern Limited	
		Gate House	
		Turnpike Road	
		нідн шусомве	
		Buckinghamshire	
		HP12 3NR	
		(in respect of the rights of access to sewage reserved by the Conveyance dated 28 September 1967)	
		KCOM Group Limited	
		37 Carr Lane	
		Hull	
		East Yorkshire	
		HU1 3RE	
		(in respect of underground telecommunication apparatus)	
		The Lord Mayor Aldermen and Citizens of the City and County of Kingston Upon Hull	
		Kingston upon Hull City Council	
		The Guildhall	
		Alfred Gelder Street	
		HULL	
		East Riding of Yorkshire	
		HU1 2AA	
		(in respect of the right of access granted by a Deed of Grant dated 21 November 1972)	
		Yorkshire Water Services Limited	
		Western House	
		Western Way	
		Halifax Road	
		Bradford	
		West Yorkshire	
		BD6 2SZ	
		(in respect of underground water apparatus)	



		BOOK OF REFERENCE - PART 3	
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on		Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
17-008 51	12 square metres of car park and hardstanding (south of Broadgate, B1230)	Bryant Homes Northern Limited	
		Gate House	
		Turnpike Road	
		НIGH WYCOMBE	
		Buckinghamshire	
		HP12 3NR	
		(in respect of reserved rights of way and others including restrictive covenants as contained in a Transfer dated 7 May 2010)	
17-009 16	657 square metres of grassland and access track (south of Broadgate, B1230)	Bryant Homes Northern Limited	
		Gate House	
		Turnpike Road	
		HIGH WYCOMBE	
		Buckinghamshire	
		HP12 3NR	
		(in respect of the rights of access to sewage reserved by the Conveyance dated 28 September 1967)	
		KCOM Group Limited	
		37 Carr Lane	
		Hull	
		East Yorkshire	
		HU1 3RE	
		(in respect of underground telecommunication apparatus)	
		Northern Gas Networks Limited	
		1100 Century Way	
		Thorpe Park	
		LEEDS	
		West Yorkshire	
1		LS15 8TU	
		(in respect of underground gas apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-009		The Lord Mayor Aldermen and Citizens of the City and County of Kingston Upon Hull	
cont'd		Kingston upon Hull City Council The Guildhall Alfred Gelder Street HULL East Riding of Yorkshire HU1 2AA (in respect of the right of access granted by a Deed of Grant dated 21 November 1972)	
		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of rights of access) Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
17-010	20505 square metres of grassland, copse and building (Broadgates Hospital, south of Broadgate, B1230) (excluding all interests of the Crown)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)	



	Dogger Bank So	uth Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-010		The Lord Mayor Aldermen and Citizens of the City and County of Kingston Upon Hull
cont'd		Kingston upon Hull City Council The Guildhall Alfred Gelder Street HULL East Riding of Yorkshire HU1 2AA (in respect of rights to construct and maintain a 14 inch water pipe listed in a Deed dated 09 June 1972) Unknown (in respect of rights relating to supply of water listed in a Conveyance dated 14 April 1983) Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground sewerage and water apparatus)
17-011	10135 square metres of agricultural land (Vinegar Hill Farm, north of A1079, Beverley)	Beazer Homes Doncaster Limited Persimmon House Fulford YORK North Yorkshire YO19 4FE (in respect of the rights of access reserved by the Conveyance dated 24 January 1962) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of the Agreement relating to the placing, maintaining, repairing, renewing and inspecting of overhead power lines dated 2 July 1931 and 9 July 1931)



	Dogger	Bank South Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-011		Northern Powergrid (Yorkshire) PLC
cont'd		Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of rights relating to the placing, maintaining, repairing, renewing, inspecting, retaining and removal of poles and overhead and
		underground lines as contained in Deeds dated 2 August 1957 and 8 August 1961 and 15 March 1962) Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of restrictive covenants and the rights granted by a Deed dated 19 March 1968. Also in respect of the rights of access to maintain a water main granted by a Deed dated 19 April 1995) Yorkshire Water Services Limited Western House Western House Western Way Halifax Road Bradford Western Way Halifax Road Bradford West Yorkshire BD6 2SZ
		(in respect of underground water apparatus)



	Dogger Bank South Offshore Wind Farm Development Consent Order		
BC		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-001	32638 square metres of agricultural land (north of A1079, Beverley)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)	
18-002	379 square metres of agricultural land (north of A1079, Beverley)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)	
18-003	1483 square metres of public road and verges (A1079, Beverley) (excluding all interests of the Crown)	Unknown	
18-009	6202 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of overhead electricity apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)	
18-010	103456 square metres of agricultural land, access track, scrubland, hedgerows, drains, pylon and public footpath (Walkington Footpath No.4) (The Risby Estate, west of A1079, Beverley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	



	Dogger Bank Sou	th Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation $7(1)(c)$ of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
18-010		National Grid Electricity Transmission PLC
cont'd		1-3 Strand
		LONDON
		WC2N 5EH
		(in respect of overhead electricity apparatus)
		National Grid Electricity Transmission PLC
		1-3 Strand
		LONDON
		WC2N 5EH
		(in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a
		Deed dated 28 July 1969)
		Northern Powergrid (Yorkshire) PLC
		Lloyds Court
		78 Grey Street
		Newcastle Upon Tyne
		NE1 6AF
		(in respect of overhead electricity apparatus)
		Yorkshire Water Services Limited
		Western House
		Western Way
		Halifax Road
		Bradford
		West Yorkshire
		BD6 2SZ
		(in respect of underground water and sewerage apparatus)
18-010a	48523 square metres of agricultural land (The Risby Estate, south of A1079, Beverley)	Ineos Manufacturing (Hull) Limited
		Hawkslease
		Chapel Lane
		LYNDHURST
		Hampshire
		SO43 7FG
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)
		l de la constante de



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
18-010a		Yorkshire Water Services Limited	
cont'd		Western House	
		Western Way	
		Halifax Road	
		Bradford	
		West Yorkshire	
		BD6 25Z	
		(in respect of underground water and sewerage apparatus)	
18-011	27093 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	National Grid Electricity Transmission PLC	
		1-3 Strand	
		LONDON	
		WC2N 5EH	
		(in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a	
		Deed dated 28 July 1969)	
40.040			
18-012	10430 square metres of agricultural land, hedgerow, drain and public footpath (Walkingtor	n Ineos Manufacturing (Hull) Limited	
18-012	Footpath No.4) (The Risby Estate, west of A1079, Beverley)	Hawkslease	
18-012		Hawkslease Chapel Lane	
18-012		Hawkslease Chapel Lane LYNDHURST	
18-012		Hawkslease Chapel Lane LYNDHURST Hampshire	
18-012		Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG	
18-012		Hawkslease Chapel Lane LYNDHURST Hampshire	
18-012		Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG	
18-012		Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
18-012		Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) National Grid Electricity Transmission PLC	
18-012		Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) National Grid Electricity Transmission PLC 1-3 Strand	
18-012		Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a	
18-012		Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	
18-012		Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) Northern Powergrid (Yorkshire) PLC	
18-012		Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) Northern Powergrid (Yorkshire) PLC Lloyds Court	
18-012		Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street	
18-012		Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne	
18-012		Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne Net1 6AF	
18-012		Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-012 cont'd		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water and sewerage apparatus)	
	27196 square metres of agricultural land, access tracks, verges, hedgerow and public footpath (Walkington Footpath No.9) (The Risby Estate, south of A1079, Beverley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
	131236 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-014a	8619 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited	
		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
		Unknown	
		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water	
		and entry listed in a Deed dated 1 December 1954)	
		Unknown	
		(in respect of rights listed in a Deed dated 17 August 1993)	
18-015	58902 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited	
		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		<del>SO43 7FG</del>	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
		Ineos Manufacturing (Hull) Limited	
		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire SO43-7FG	
		(in respect of underground gas pipeline apparatus)	
		KCOM Group Limited	
		<del>37 Carr Lane</del>	
		Hull Fact Marketing	
		East Yorkshire HUI 3RE	
		(in respect of underground telecommunication apparatus)	
		In respect of underground teleconfinantiation apparatasy	
		•	



Dogger Bank South Offshore Wind Farm Development Consent Order			
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-015		National Gas Transmission PLC	
cont'd		National Grid House	
		Warwick Technology Park	
		Gallows Hill	
		WARWICK	
		CV34 6DA	
		(in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.)	
		National Gas Transmission PLC	
		National Grid House	
		Warwick Technology Park	
		Gallows Hill	
		WARWICK	
		CV34 6DA	
		(in respect of underground gas apparatus)	
		Unknown	
		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)	
		Unknown	
		(in respect of rights listed in a Deed dated 17 August 1993)	
		Yorkshire Water Services Limited	
		Western House	
		Western Way	
		Halifax Road	
		Bradford	
		West Yorkshire	
		BD6 2SZ	
		(in respect of underground water apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
18-015a	25825 square metres of agricultural land, scrubland and drain (The Risby Estate, west of	Ineos Manufacturing (Hull) Limited	
	A164, Bentley)	Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
		Ineos Manufacturing (Hull) Limited	
		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of underground gas pipeline apparatus)	
		KCOM Group Limited	
		37 Carr Lane	
		Hull	
		East Yorkshire	
		HU1 3RE	
		(in respect of underground telecommunication apparatus)	
		National Gas Transmission PLC	
		National Grid House	
		Warwick Technology Park	
		Gallows Hill	
		WARWICK	
		CV34 6DA	
		(in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.)	
		National Gas Transmission PLC	
		National Grid House	
		Warwick Technology Park	
		Gallows Hill	
		WARWICK	
		CV34 6DA	
		(in respect of underground gas apparatus)	
		In respect of and Broand Bas abhalaras)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-015a cont'd		Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993) Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
18-016	7497 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Unknown (as successor of Arthur Thomas Filmer Wilson Filmer) (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (as successor of Charles Henry Woodmansey) (in respect of rights listed in a Deed dated 17 August 1993)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
County of East Riding of Yorkshire			
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
18-017	13025 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited	
		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
		National Gas Transmission PLC	
		National Grid House	
		Warwick Technology Park	
		Gallows Hill	
		WARWICK	
		CV34 6DA	
		(in respect of rights & easements listed in a Grant of Easement dated 3 April 2006.)	
		National Gas Transmission PLC	
		National Grid House	
		Warwick Technology Park	
		Gallows Hill	
		WARWICK	
		CV34 6DA	
		(in respect of underground gas apparatus)	
		Unknown	
		(as successor of Arthur Thomas Filmer Wilson Filmer)	
		(in respect of a right of way listed in a conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water	
		and entry listed in a Deed dated 1 December 1954)	
		Unknown	
		(as successor of Charles Henry Woodmansey)	
		(in respect of rights listed in a Deed dated 17 August 1993)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-017 cont'd		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
18-018	4 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of provisions contained in a Conveyance dated 3 September 1976) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-019	417 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)	Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of provisions contained in a Conveyance dated 3 September 1976) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-020	328 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of rights & easements listed in a Grant of Easement dated 3 April 2006.)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-020 cont'd		National Gas Transmission PLC         National Grid House         Warwick Technology Park         Gallows Hill         WARWICK         CV34 6DA         (in respect of underground gas apparatus)         Unknown         (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)         Unknown         (in respect of provisions contained in a Conveyance dated 3 September 1976)         Unknown         (in respect of rights listed in a Deed dated 17 August 1993)	
18-021	112 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of provisions contained in a Conveyance dated 3 September 1976) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-021 cont'd		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
18-021a	87 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of provisions contained in a Conveyance dated 3 September 1976) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-022	37869 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation $7(1)(c)$ of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
18-022		Ineos Manufacturing (Hull) Limited	
cont'd		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of underground gas pipeline apparatus)	
		National Gas Transmission PLC	
		National Grid House	
		Warwick Technology Park	
		Gallows Hill	
		WARWICK	
		CV34 6DA	
		(in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.)	
		National Gas Transmission PLC	
		National Grid House	
		Warwick Technology Park	
		Gallows Hill	
		WARWICK	
		CV34 6DA	
		(in respect of underground gas apparatus)	
		Unknown	
		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water	
		and entry listed in a Deed dated 1 December 1954)	
		Unknown	
		(in respect of rights listed in a Deed dated 17 August 1993)	
18-022a	408 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)		
		National Grid House	
		Warwick Technology Park	
		Gallows Hill	
		WARWICK	
		CV34 6DA	
		(in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-022a cont'd		Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-023	44973 square metres of agricultural land and drain (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus) National Gas Transmission PLC National Gai Thouse Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of underground gas apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
	County of East Riding of Yorkshire		
Number or Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-023 cont'd		Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-024	13813 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDPIURST Hampshire SQ43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SQ43 7FG (in respect of underground gas pipeline apparatus) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of rights & easements listed in a Grant of Easement dated 3 April 2006.) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK CV34 6DA (in respect of underground gas apparatus)	



	FERENCE - PART 3 Riding of Yorkshire Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished,
Number on Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished,
Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,
	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
	a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water ed in a Deed dated 1 December 1954)
	rights listed in a Deed dated 17 August 1993)
Bentley) Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of Ineos Manufa Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of SO43 7FG (in respect of KCOM Group 37 Carr Lane Hull East Yorkshire HUI 3RE (in respect of and entry list Unknown	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-025		Yorkshire Water Services Limited	
cont'd		Western House	
		Western Way	
		Halifax Road	
		Bradford	
		West Yorkshire	
		BD6 2SZ	
		(in respect of underground water apparatus)	
18-025a	5052 square metres of agricultural land and tree (The Risby Estate, west of A164, Bentley)		
		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
		Ineos Manufacturing (Hull) Limited	
		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of underground gas pipeline apparatus)	
		Unknown	
		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water	
		and entry listed in a Deed dated 1 December 1954)	
		Unknown	
		(in respect of rights listed in a Deed dated 17 August 1993)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
BOOK OF REFERENCE - PART 3			
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-025a cont'd		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
<del>18 026</del>	<del>2 square metres of drain (south of A1079, Beverley)</del>	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NEL 6AF (in respect of overhead electricity apparatus)	
18-027	23322 square metres of agricultural land, scrubland and copse (The Risby Estate, west of A164, Bentley and south of A1079, Beverley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N SEH (in respect of overhead electricity apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N SEH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
10.027		(Applications: Prescribed Forms and Procedures) Regulations 2009	
18-027 cont'd		Northern Powergrid (Yorkshire) PLC Lloyds Court	
cont u		<del>rioyas court</del> <del>78 Grey Street</del>	
		Newcastle Upon Tyne	
		NEI 6AF	
		<del>(in respect of overhead electricity apparatus)</del>	
		Yorkshire Water Services Limited	
		Western House	
		<del>Western Way</del>	
		Halifax Road	
		Bradford	
		West Yorkshire	
		BD6 2SZ	
		<del>(in respect of underground water apparatus)</del>	
18-027a	3795 square metres of agricultural land, hedgerows, scrubland and drain (The Risby	Ineos Manufacturing (Hull) Limited	
	Estate, west of A164, Bentley and south of A1079, Beverley)	Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
		National Grid Electricity Transmission PLC	
		1-3 Strand	
		LONDON	
		WC2N 5EH	
		(in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a	
		Deed dated 28 July 1969)	
		Northern Powergrid (Yorkshire) PLC	
		Lloyds Court	
		78 Grey Street	
		Newcastle Upon Tyne	
		NE1 6AF	
		(in respect of overhead electricity apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-028	3970 square metres of woodland (Bentley Moor Wood, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)	
18-029	270 square metres of drain (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)	
18-030	18861 square metres of woodland (Bentley Moor Wood, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-030 cont'd		Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-031	133 square metres of woodland and drain (Bentley Moor Wood, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-032	17850 square metres of agricultural land, hedgerows and drain (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)	



	Dogger Bank So	uth Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-032		Unknown
cont'd		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)
		Unknown (in respect of rights listed in a Deed dated 17 August 1993)
<del>18-033</del>	12 square metres of agricultural land (The Risby Estate, west of A164, Bentley)	Incos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43-7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water- and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)
<del>18-03</del> 4	14 square metres of agricultural land and hedgerow (The Risby Estate, west of A164, Bentley)	Incos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43-7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-035	222 square metres of drain and scrubland (The Risby Estate, west of A164, Bentley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
<del>18-036</del>	8 square metres of drain and scrubland (The Risby Estate, west of A164, Bentley)	Incos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43-7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
<del>18-037</del>	<del>62 square metres of scrubland (The Risby Estate, west of A164, Bentley)</del>	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43-7FG (in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	



	Dogger Bank Sou	ith Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
<del>18-037</del> <del>cont'd</del>		Unknown ( <del>in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)</del>
18-038	12918 square metres of agricultural land, hedgerows and copse (west of A164, Bentley)	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)
18-039	386 square metres of public road and verge (A164, Bentley)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus)
18-040	2427 square metres of public road, verges and access splay (A164, Bentley)	KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Unknown



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
Number on Land Plans	Description of Land	County of East Riding of Yorkshire           Names of all those entitled to enjoy easements or other private rights over land           (including private rights of navigation over water) which it is proposed shall be extinguished,           suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning           (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-040 cont'd		Yorkshire Water Services Limited Western House Western Way Halifax Road Bradford West Yorkshire BD6 2SZ (in respect of underground water apparatus)	
18-041	49 square metres of public road and verge (A164, Bentley)	Unknown	
18-042	166 square metres of agricultural land (The Risby Estate, east of A164, Bentley)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
	65453 square metres of agricultural land, access tracks, verges, hedgerows, copse, drain and public bridleway (Rowley Bridleway No.13) (The Risby Estate, west of A1079, Beverley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-043		National Gas Transmission PLC	
cont'd		National Grid House	
		Warwick Technology Park	
		Gallows Hill	
		WARWICK	
		CV34 6DA	
		(in respect of rights and easements listed in a Grant of Easement dated 3 April 2006.)	
		National Gas Transmission PLC	
		National Grid House	
		Warwick Technology Park	
		Gallows Hill	
1		WARWICK	
		CV34 6DA	
		(in respect of underground gas apparatus)	
		National Grid Electricity Transmission PLC	
		1-3 Strand	
		LONDON	
		WC2N 5EH	
		(in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)	
		Northern Powergrid (Yorkshire) PLC	
		Lloyds Court	
		78 Grey Street	
		Newcastle Upon Tyne	
		NE1 6AF	
		(in respect of overhead electricity apparatus)	
		Unknown	
		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)	
		Unknown	
		(in respect of rights listed in a Deed dated 17 August 1993)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-044	9168 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-045	2756 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
18-046	27 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	National Grid Electricity Transmission PLC	
		1-3 Strand	
		LONDON	
		WC2N 5EH	
		(in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a	
		Deed dated 28 July 1969)	
		Underson	
		Unknown (is represent of a right of your listed in a Company dated 12 August 1952, Alex is present of rights to love size to company and supply unter	
		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)	
		and entry listed in a beed dated 1 betenber 1954)	
		Unknown	
		(in respect of provisions contained in a Conveyance dated 3 September 1976)	
		· ····································	
		Unknown	
		(in respect of rights listed in a Deed dated 17 August 1993)	
18-047	10223 square metres of agricultural land, access tracks, verges, copse and public bridleway	Ineos Manufacturing (Hull) Limited	
	(Rowley Bridleway No.13) (The Risby Estate, south of A1079, Beverley)	Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of rights in regards to a pipeline and valve as contained in a Deed and Lease dated 22 May 2002)	
		Ineos Manufacturing (Hull) Limited	
		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire	
		SO43 7FG	
		(in respect of underground gas pipeline apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
BC		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-047		KCOM Group Limited	
cont'd		37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) National Grid Electricity Transmission PLC	
		1-3 Strand	
		LONDON WC2N 5EH	
		(in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)	
		Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)	
		Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown	
		(in respect of rights listed in a Deed dated 17 August 1993)	
18-048	289 square metres of access track, verge and public bridleway (Rowley Bridleway No.13) (Burn Park Farm, south of A1079, Beverley)	Albanwise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (in respect of a right of access to service contained in a Conveyance dated 22 May 1978. Also in respect of rights of access to maintain service media as contained in a Transfer dated 13 January 2021)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-048		Doggerbank Offshore Wind Farm Project 1 Projco Limited	
cont'd		No.1 Forbury Place 43 Forbury Road Reading	
		RG1 3JH (in respect of rights contained in a Deed of Grant dated 3 March 2020)	
		Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road	
		Reading RG1 3JH (in respect of restrictive covenants contained in a Deed dated 3 March 2020)	
		KCOM Group Limited	
		37 Carr Lane Hull East Yorkshire HU1 3RE	
		(in respect of proposed underground telecommunication apparatus)	
		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	
		(in respect of rights of access to maintaining, repairing, renewing, inspecting or removing electric lines contained in a Deed dated 7 March 1960 and 28 July 1969)	
18-049	2533 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG	
		(in respect of underground gas pipeline apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-049 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-050	67 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	National Grid Electricity Transmission PLC         1-3 Strand         LONDON         WC2N SEH         (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)         The Executor Of The Estate Of The Late Andrew White         Manor Farm         Newland         GOLE         DN14 7XD         (in respect of rights to use service media and other rights and easements listed in a Lease dated 8 November 2013)         Unknown         (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954)         Unknown         (in respect of provisions contained in a Conveyance dated 3 September 1976)         Unknown         (in respect of provisions contained in a Conveyance dated 3 September 1976)         Unknown         (in respect of provisions contained in a Conveyance dated 3 September 1976)         Unknown         (in respect of rights listed in a Deed dated 17 August 1993)	



	Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3		
	County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-051	21 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N SEH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969) The Executor Of The Estate Of The Late Andrew White Manor Farm Newland GOOLE DN14 7XD (in respect of rights to use service media and other rights and easements listed in a Lease dated 8 November 2013) Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of provisions contained in a Conveyance dated 3 September 1976) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
	26833 square metres of agricultural land, verge and drain (The Risby Estate, west of A1079, Beverley)	Joanne Dransfield Jillywood Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP (in respect of rights to pass along a track, lay telephone cables, electricity poles and erect poles to carry said cables; to use a water pipe, drainage and entry listed in a Conveyance dated 7 January 1994)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-052		National Grid Electricity Transmission PLC	
cont'd		1-3 Strand LONDON WC2N 5EH (in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)	
		Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)	
		Paul Dransfield Jillywood Farm Bentley BEVERLEY East Riding of Yorkshire HU17 8PP (in respect of rights to pass along a track, lay telephone cables, electricity poles and erect poles to carry said cables; to use a water pipe, drainage and entry listed in a Conveyance dated 7 January 1994)	
		Unknown (in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water and entry listed in a Deed dated 1 December 1954) Unknown (in respect of rights listed in a Deed dated 17 August 1993)	
18-053	303 square metres of agricultural land (The Risby Estate, west of A1079, Beverley)	Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3 County of East Riding of Yorkshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
10.050		(Applications: Prescribed Forms and Procedures) Regulations 2009	
18-053		National Grid Electricity Transmission PLC	
cont'd		1-3 Strand	
		WC2N 5EH	
		(in respect of rights of access to erect, retain, use, maintain, repair, renew, inspect and remove electric lines and covenants listed by a Deed dated 28 July 1969)	
		Unknown	
		(in respect of a right of way listed in a Conveyance dated 12 August 1953. Also in respect of rights to lay a pipe to carry and supply water	
		and entry listed in a Deed dated 1 December 1954)	
		and entry listed in a Deed dated 1 December 1994)	
		Unknown	
		(in respect of rights listed in a Deed dated 17 August 1993)	
18-054	22194 square metres of agricultural land, copse, drain and public bridleway (Rowley	Albanwise Limited	
	Bridleway No.13) (Burn Park Farm, south of A1079, Beverley)	c/o Mills & Reeve LLP	
		Botanic House	
		100 Hills Road	
		CAMBRIDGE	
		Cambridgeshire	
		CB2 1PH	
		(in respect of a right of access to service contained in a Conveyance dated 22 May 1978. Also in respect of rights of access to maintain	
		service media as contained in a Transfer dated 13 January 2021)	
		BP Chemicals Limited	
		Chertsey Road	
		SUNBURY ON THAMES	
		Middlesex	
		TW16 7BP	
		(in respect of rights to construct a pipeline contained in Agreements dated 22 May 1998 and 17 June 1998. Also in respect of rights to use a	
		pipeline contained in a Deed dated 14 November 2000. Also in respect of rights to construct and maintain a pipeline contained in a Lease	
		of Easements dated 16 May 2002)	



	Do	ogger Bank South Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-054		Doggerbank Offshore Wind Farm Project 1 Projec Limited
cont'd		No.1 Forbury Place
1		43 Forbury Road
		Reading
		RG1 3JH
		(in respect of rights contained in a Deed of Grant dated 3 March 2020)
		Doggerbank Offshore Wind Farm Project 2 Projco Limited
		No.1 Forbury Place
		43 Forbury Road
		Reading
		RG1 3JH
		(in respect of restrictive covenants contained in a Deed dated 3 March 2020)
		Ineos Manufacturing (Hull) Limited
		Hawkslease
		Chapel Lane
		LYNDHURST Hampshire
		SO43 7FG
		(in respect of rights of access contained in a Lease dated 16 May 2002)
		National Grid Electricity Transmission PLC
		1-3 Strand
		LONDON
		WC2N 5EH
		(in respect of overhead electricity apparatus)
		National Grid Electricity Transmission PLC
		1-3 Strand
		LONDON
		WC2N 5EH
		(in respect of right to maintaining, repairing, renewing, inspecting or removing electric lines contained in a Deed dated 7 March 1960 and 28 July 1969)



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-054 cont'd	11500 martin da si standarda da si su da da la da si su da si da si su da si da si su da si da si si su da si s	Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus)	
18-055	11580 square metres of agricultural land, verges, drain and public bridleway (Rowley Bridleway No.13) (Burn Park Farm, south of A1079, Beverley)	Albamvise Limited c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (in respect of a right of access to service contained in a Conveyance dated 22 May 1978. Also in respect of rights of access to maintain service media as contained in a Transfer dated 13 January 2021) BP Chemicals Limited Chertsey Road SUNBURY ON THAMES Middlesex TW16 7BP (in respect of rights to construct a pipeline contained in Agreements dated 22 May 1998 and 17 June 1998. Also in respect of rights to use a pipeline contained in a Deed dated 11 November 2000. Also in respect of rights to construct and maintain a pipeline contained in a Lease of Easements dated 16 May 2002) Doggerbank Offshore Wind Farm Project 1 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 3 March 2020)	





	Dogger Bank South Offshore Wind Farm Development Consent Order		
BOOK OF		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-055		Doggerbank Offshore Wind Farm Project 2 Projec Limited	
cont'd		No.1 Forbury Place 43 Forbury Road Reading RG1 3JH	
		(in respect of restrictive covenants contained in a Deed dated 3 March 2020)	
		Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights contained in a Lease dated 16 May 2002) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right to maintaining, repairing, renewing, inspecting or removing electric lines contained in a Deed dated 7 March 1960 and 28 July 1969) Northern Powergrid (Yorkshire) PLC Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF	
19-002	5820 square metres of public road and verges (A1079, Beverley) (excluding all interests of	(in respect of overhead electricity apparatus) Northern Powergrid (Yorkshire) PLC	
	the Crown)	Lloyds Court 78 Grey Street Newcastle Upon Tyne NE1 6AF (in respect of overhead electricity apparatus) Unknown	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-003	160 square metres of drain (south of A1079, Beverley)	Unknown	
19-004	11423 square metres of agricultural land, verges and public bridleway (Woodmansey Bridleway No.30) (Model Farm, south of A1079, Beverley)	Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant and Transfer dated 15 April 2019. Also rights contained in an Option Agreement dated 4 February 2020 and rights contained in a Lease dated 1 March 2022)	
19-005	8558 square metres of agricultural land, access track, hardstanding, scrubland, verges and public bridleway (Woodmansey Bridleway No.6) (Poplar Farm, south of A1079, Beverley)	Albert John Evans Wanlass Farm Park Lane COTTINGHAM East Riding of Yorkshire HU16 5SB (in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010) Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access relating to laying and maintaining a pipeline contained in a Deed dated 11 September 2011. Also in respect of rights and covenants listed in a Transfer dated 11 September 2002) Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-005 cont'd		Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of easements and covenants relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September 2011) Wind Energy One Limited 30 Finsbury Square LONDON EC2A 1AG (in respect of rights of access granted by a Lease dated 16 January 2017)	
19-006	812 square metres of agricultural land (Poplar Farm, south of A1079, Beverley)	Albert John Evans Wanlass Farm Park Lane COTTINGHAM East Riding of Yorkshire HU16 5SB (in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010) Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access relating to laying and maintaining a pipeline contained in a Deed dated 11 September 2011. Also in respect of rights and covenants listed in a Transfer dated 11 September 2002)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-006 cont'd		Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of underground gas pipeline apparatus) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU	
		(in respect of easements and covenants relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September 2011)	
	13773 square metres of agricultural land, drains and access splay (Burn Park Farm, south of A1079, Beverley)	c/o Mills & Reeve LLP Botanic House 100 Hills Road CAMBRIDGE Cambridgeshire CB2 1PH (in respect of a right of access to service contained in a Conveyance dated 22 May 1978. Also in respect of rights of access to maintain service media as contained in a Transfer dated 13 January 2021) BP Chemicals Limited Chertsey Road SUNBURY ON THAMES	
		Middlesex TW16 7BP (in respect of rights to construct a pipeline contained in Agreements dated 22 May 1998 and 17 June 1998. Also in respect of rights to use a pipeline contained in a Deed dated 11 November 2000. Also in respect of rights to construct and maintain a pipeline contained in a Lease of Easements dated 16 May 2002)	



	Dogger Bank South Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of East Riding of Yorkshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-007		Doggerbank Offshore Wind Farm Project 1 Projec Limited	
cont'd		No.1 Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of rights contained in a Deed of Grant dated 3 March 2020) Doggerbank Offshore Wind Farm Project 2 Projco Limited No.1 Forbury Place 43 Forbury Place	
		43 Forbury Road	
		Reading RG1 3JH	
		(in respect of restrictive covenants contained in a Deed dated 3 March 2020)	
		Ineos Manufacturing (Hull) Limited	
		Hawkslease	
		Chapel Lane	
		LYNDHURST	
		Hampshire SO43 7FG	
		(in respect of rights of access contained in a Lease dated 16 May 2002)	
		National Grid Electricity Transmission PLC	
		1-3 Strand	
		LONDON	
		WC2N 5EH	
		(in respect of overhead electricity apparatus)	
		National Grid Electricity Transmission PLC	
		1-3 Strand	
		LONDON WC2N 5EH	
		(in respect of right to maintaining, repairing, renewing, inspecting or removing electric lines contained in Deeds dated 7 March 1960 and	
		28 July 1969)	



	Dogger Bank Sou	uth Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 3			
		County of East Riding of Yorkshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
20-001	26 square metres of agricultural land (Poplar Farm, south of A1079, Beverley)	Albert John Evans Wanlass Farm Park Lane COTTINGHAM East Riding of Yorkshire HU16 5SB (in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010) Incos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access relating to laying and maintaining a pipeline contained in a Deed dated 11 September 2011. Also in respect of rights and covenants listed in a Transfer dated 11 September 2002) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of easements and covenants relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September 2011) Unknown (in respect of rights listed in a Transfer dated 11 September 2002)		
20-002	104 square metres of agricultural land, access track and public bridleway (Woodmansey Bridleway No.6) (Poplar Farm, south of A1079, Beverley)	Albert John Evans Wanlass Farm Park Lane COTTINGHAM East Riding of Yorkshire HU16 5SB (in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010)		



	Dogger Bank Sou	th Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
20-002		Northern Gas Networks Limited
cont'd		1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of easements and covenants relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September 2011) Unknown (in respect of rights listed in a Transfer dated 11 September 2002) Wind Energy One Limited 30 Finsbury Square LONDON
		EC2A 1AG (in respect of rights of access granted by a Lease dated 16 January 2017)
20-003	3618 square metres of agricultural land, access tracks, verges, hedgerow, drain and public bridleway (Woodmansey Bridleway No.6) (Poplar Farm, south of A1079, Beverley)	Albert John Evans Wanlass Farm Park Lane COTTINGHAM East Riding of Yorkshire HU16 5SB (in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010)
		Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access relating to laying and maintaining a pipeline contained in a Deed dated 11 September 2011. Also in respect of rights and covenants listed in a Transfer dated 11 September 2002)



	Dogger Bank So	uth Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
20-003		Northern Gas Networks Limited
cont'd		1100 Century Way
		Thorpe Park
		LEEDS
		West Yorkshire
		LS15 8TU
		(in respect of easements and covenants relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September
		2011)
		Wind Energy One Limited
		30 Finsbury Square
		LONDON
		EC2A 1AG
		(in respect of rights of access granted by a Lease dated 16 January 2017)
20-004	118007 square metres of agricultural land, access tracks, verges, hedgerows, electricity	Albert John Evans
	pylon, pond, drains, copse, scrubland, building and public bridleway (Woodmansey	Wanlass Farm
	Bridleway No.6) (Poplar Farm, south of A1079, Beverley)	Park Lane
		COTTINGHAM
		East Riding of Yorkshire
		HU16 5SB
		(in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010)
		Ineos Manufacturing (Hull) Limited
		Hawkslease
		Chapel Lane
		LYNDHURST
		Hampshire
		SO43 7FG
		(in respect of rights of access relating to laying and maintaining a pipeline contained in a Deed dated 11 September 2011. Also in respect of
		rights and covenants listed in a Transfer dated 11 September 2002)



	Dogger Bank S	South Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
20-004		KCOM Group Limited
cont'd		37 Carr Lane
		Hull
		East Yorkshire
		HU1 3RE
		(in respect of underground telecommunication apparatus)
		National Grid Electricity Transmission PLC
		1-3 Strand
		LONDON
		WC2N 5EH
		(in respect of overhead electricity apparatus)
		Northern Gas Networks Limited
		1100 Century Way
		Thorpe Park
		LEEDS
		West Yorkshire
		LS15 8TU
		(in respect of easements and covenants relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September 2011)
		Wind Energy One Limited
		30 Finsbury Square
		LONDON
		EC2A 1AG
		(in respect of rights of access granted by a Lease dated 16 January 2017)
20-005	238 square metres of agricultural land, access track, verges and public bridleway	Albert John Evans
	(Woodmansey Bridleway No.6) (Poplar Farm, south of A1079, Beverley)	Wanlass Farm
		Park Lane
		COTTINGHAM
		East Riding of Yorkshire
		HU16 5SB
		(in respect of rights to passage of services or supplies through service media listed in a Transfer dated 10 February 2010)
L		



		Dogger Bank South Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of East Riding of Yorkshire
Number on       Description of Land       (including private rights of navigation over water) which it is proposed shall be suspended or interfered with under Regulation 7(1)(c) of the Infrastructure		Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
20-005 cont'd		Ineos Manufacturing (Hull) Limited Hawkslease Chapel Lane LYNDHURST Hampshire SO43 7FG (in respect of rights of access relating to laying and maintaining a pipeline contained in a Deed dated 11 September 2011. Also in respect of rights and covenants listed in a Transfer dated 11 September 2002) KCOM Group Limited 37 Carr Lane Hull East Yorkshire HU1 3RE (in respect of underground telecommunication apparatus) Northern Gas Networks Limited 1100 Century Way Thorpe Park LEEDS West Yorkshire LS15 8TU (in respect of easements and covenants relating to the laying and maintenance of a pipeline listed in a Deed of Grant dated 21 September 2011)



		Dogger Bank South Offshore Wind Farm BOOK OF REFERENCE	- PART 4
Lan	nd which is proposed to be used for the pur	poses of the order for which application is being made under Regulatio County of East Riding o	n 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 f Yorkshire
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made
01-010	Temporary Possession	30 square metres of foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
01-011	Temporary Possession	356 square metres of foreshore seaward of the Mean High Water lying to the north east of Ulrome (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
01-012	Temporary Possession	8013 square metres of foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
01-013	Temporary Possession	1772 square metres of foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
01-014	Temporary Possession	1144 square metres of foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
01-015	Temporary Possession	2258 square metres of foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
02-001	Acquisition of Rights	80265 square metres of foreshore and sea landward and seaward of the Mean Low Water lying to the east of Skipsea (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
02-002	Acquisition of Rights	103 square metres of foreshore seaward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH



Land	d which is proposed to be used for the pur		- PART 4 n 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009
Number on Land Plans	Extent of acquisition or use	County of East Riding o Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made
02-003	Acquisition of Rights	30458 square metres of foreshore seaward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
02-004	Acquisition of Rights	122 square metres of foreshore seaward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
02-005	Acquisition of Rights	3948 square metres of shore landward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)
02-007	Acquisition of Rights	343 square metres of foreshore seaward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
02-008	Acquisition of Rights	67 square metres of foreshore seaward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
02-010	Acquisition of Rights	1994 square metres of foreshore seaward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
02-011	Acquisition of Rights	71 square metres of foreshore seaward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH



Land	d which is proposed to be used for the pur	Dogger Bank South Offshore Wind Farm BOOK OF REFERENCE poses of the order for which application is being made under Regulatio County of East Riding c	- PART 4 in 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made
02-012	Acquisition of Rights	157 square metres of foreshore seaward of the Mean High Water at Skipsea (north of Cliff Road, excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
02-013	Acquisition of Rights	599 square metres of shore landward of the Mean High Water lying to the east of Skipsea (north of Hornsea Road)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)
02-014	Acquisition of Rights	8 square metres of shore landward of the Mean High Water lying to the east of Skipsea (north of Hornsea Road)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)
05-004	Acquisition of Rights	69585 square metres of agricultural land and hedgerow (east of New Road) (excluding all interests of the Crown)	Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of restrictive covenants and rights listed in a Deed dated 30 May 1949)
06-018	Acquisition of Rights	30325 square metres of agricultural land and drain (north of Harsell Lane, excluding all interests of the Crown)	Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of covenants and grants contained in a Deed dated 11 November 1948)



La	nd which is proposed to be used for the purp	Dogger Bank South Offshore Wind Farm BOOK OF REFERENC oses of the order for which application is being made under Regulati County of East Riding	E - PART 4 on 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made
06-019	Temporary Possession	136 square metres of verge (north of Harsell Lane, excluding all interests of the Crown)	Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of covenants and grants contained in a Deed dated 11 November 1948)
06-020	Temporary Possession	374 square metres of public road and verge (Harsell Lane, excluding all interests of the Crown)	Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of covenants and grants contained in a Deed dated 11 November 1948)
06-021	Acquisition of Rights	472 square metres of public road (Harsell Lane, exlucing all interests of the Crown)	Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of covenants and grants contained in a Deed dated 11 November 1948)
06-022	Temporary Possession	116 square metres of verge (north of Harsell Lane, excluding all interests of the Crown)	Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of covenants and grants contained in a Deed dated 11 November 1948)



	nd which is proposed to be used for the pur	Dogger Bank South Offshore Wind Farm BOOK OF REFERENC	E - PART 4		
Ldi	Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of East Riding of Yorkshire				
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made		
06-023	Temporary Possession	177 square metres of public road (Harsell Lane, exlucing all interests of the Crown)	Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights to maintain hedges and trees contained in a Deed dated 11 November 1948)		
06-024	Temporary Possession	9 square metres of agricultural land (south of Harsell Lane)	Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights to maintain hedges and trees contained in a Deed dated 11 November 1948)		
06-025	Acquisition of Rights	64907 square metres of agricultural land, drain and hedgerow (south of Harsell Lane, east of Catwick Road, excluding all interests of the Crown)	Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights to maintain hedges and trees contained in a Deed dated 11 November 1948)		
12-014	Acquisition of Rights	2356 square metres of foreshore and bed (River Hull, Beverley) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH		
13-004	Acquisition of Rights	71530 square metres of agricultural land and copse (Carr Farm, east of Carr Road and north of Hull Bridge Road, A1035)	Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access and the free passage of water and electricity contained in a Conveyance dated 25 March 1974)		



Lai	nd which is proposed to be used for the pur	Dogger Bank South Offshore Wind Farm BOOK OF REFERENCE poses of the order for which application is being made under Regulatio	
Number on Land Plans	Extent of acquisition or use	County of East Riding o Description of land	f Yorkshire The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made
13-006	Temporary Possession	185 square metres of agricultural land and dike (South Bullock Dike, Carr Farm, east of Carr Road and north of Hull Bridge Road, A1035)	Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access and the free passage of water and electricity contained in a Conveyance dated 25 March 1974)
16-004	Acquisition of Rights	91 square metres of public road, verge and access splay (York Road, Beverley) (excluding all interests of the Crown)	Secretary Of State For Transport Great Minster House 33 Horseferry Road Marsham LONDON SW1P 4DR (in respect of subsoil beneath public highway)
16-005	Temporary Possession	27 square metres of public road and verges (York Road, Beverley) (excluding all interests of the Crown)	Secretary Of State For Transport Great Minster House 33 Horseferry Road Marsham LONDON SW1P 4DR (in respect of subsoil beneath public highway)
16-006	Temporary Possession	133 square metres of public road and verges (York Road, Beverley) (excluding all interests of the Crown)	Secretary Of State For Transport Great Minster House 33 Horseferry Road Marsham LONDON SW1P 4DR
16-007	Acquisition of Rights	2983 square metres of public road and verges (York Road, Beverley) (excluding all interests of the Crown)	Secretary Of State For Transport Great Minster House 33 Horseferry Road Marsham LONDON SW1P 4DR



		Dogger Bank South Offshore Wind Farm	
Lar	nd which is proposed to be used for the pur	BOOK OF REFERENCE poses of the order for which application is being made under Regulatic	E - PART 4 on 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009
		County of East Riding of	
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made
18-003	Acquisition of Rights	1483 square metres of public road and verges (A1079, Beverley) (excluding all interests of the Crown)	Secretary Of State For Transport Great Minster House 33 Horseferry Road Marsham LONDON SW1P 4DR (in respect of subsoil beneath public highway)
18-004	Acquisition of Rights	1535 square metres of public road and verges (A1079, Beverley) (excluding all interests of the Crown)	Secretary Of State For Transport Great Minster House 33 Horseferry Road Marsham LONDON SW1P 4DR
18-005	Temporary Possession	2307 square metres of public road (A1079, Beverley) (excluding al interests of the Crown)	I Secretary Of State For Transport Great Minster House 33 Horseferry Road Marsham LONDON SW1P 4DR
18-006	Freehold Acquisition	1836 square metres of public road and verge (A1079, Beverley) (excluding all interests of the Crown)	Secretary Of State For Transport Great Minster House 33 Horseferry Road Marsham LONDON SW1P 4DR



Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 5 County of East Riding of Yorkshire				
Number on Land Plans	Extent of acquisition or use	Description of land	Category of Land	
01-010	Temporary Possession	30 square metres of foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)	Open Space	
01-011	Temporary Possession	356 square metres of foreshore seaward of the Mean High Water lying to the north east of Ulrome (excluding all interests of the Crown)	Open Space	
01-012	Temporary Possession	8013 square metres of foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)	Open Space	
01-013	Temporary Possession	1772 square metres of foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)	Open Space	
01-014	Temporary Possession	1144 square metres of foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)	Open Space	
01-015	Temporary Possession	2258 square metres of foreshore seaward of the Mean High Water lying to the east of Ulrome (excluding all interests of the Crown)	Open Space	
02-001	Acquisition of Rights	80265 square metres of foreshore and sea landward and seaward of the Mean Low Water lying to the east of Skipsea (excluding all interests of the Crown)	Open Space	
02-002	Acquisition of Rights	103 square metres of foreshore seaward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	Open Space	
02-003	Acquisition of Rights	30458 square metres of foreshore seaward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	Open Space	
02-004	Acquisition of Rights	122 square metres of foreshore seaward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	Open Space	
02-005	Acquisition of Rights	3948 square metres of shore landward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	Open Space	
02-006	Acquisition of Rights	108 square metres of shore landward of the Mean High Water lying to the east of Skipsea (north of Hornsea Road)	Open Space	
02-007	Acquisition of Rights	343 square metres of foreshore seaward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	Open Space	
02-008	Acquisition of Rights	67 square metres of foreshore seaward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	Open Space	



Dogger Bank South Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 5 County of East Riding of Yorkshire					
Number on Land Plans	Extent of acquisition or use	Description of land	Category of Land		
02-009	Acquisition of Rights	7 square metres of shore landward of the Mean High Water at Skipsea (north of Hornsea Road)	Open Space		
02-010	Acquisition of Rights	1994 square metres of foreshore seaward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	Open Space		
02-011	Acquisition of Rights	71 square metres of foreshore seaward of the Mean High Water lying to the east of Skipsea (excluding all interests of the Crown)	Open Space		
02-012	Acquisition of Rights	157 square metres of foreshore seaward of the Mean High Water at Skipsea (north of Cliff Road, excluding all interests of the Crown)	Open Space		

RWE Renewables UK Dogger Bank South (West) Limited

RWE Renewables UK Dogger Bank South (East) Limited

Windmill Hill Business Park Whitehill Way Swindon Wiltshire, SN5 6PB